A study of constrained Navier-Stokes equations and related problems

GAURAV DHARIWAL

DOCTOR OF PHILOSOPHY

UNIVERSITY OF YORK MATHEMATICS

JULY 2017

ABSTRACT

In the solution. In this thesis we address these questions corresponding to two models governing the dynamics of incompressible fluids, both being the modification of classical Navier-Stokes equations: constrained Navier-Stokes equations and tamed Navier-Stokes equations.

The former being Navier-Stokes equations with a constraint on the L^2 norm of the solution considered on a two-dimensional domain with periodic boundary conditions. We prove existence of the unique global-in-time solution in deterministic setting and establish existence of a pathwise unique strong solution under the impact of a stochastic forcing.

The tamed Navier-Stokes equations were introduced by Röckner and Zhang [75], to study the properties of solutions of the 3D Navier-Stokes equations. We use three new ideas to prove the existence of a strong solution and existence of invariant measures: approximating equation on an infinite dimensional space in contrast to classical Faedo-Galerkin approximation; tightness criterion related to the Dubinsky's compactness theorem introduced recently by Brzeźniak and Motyl [23]; and lastly proving the existence of invariant measures based on continuity and compactness in the weak topologies [62].

TABLE OF CONTENTS

	$\mathbf{P}\mathbf{a}_{i}$									
Ał	Abstract 3									
Та	Table of Contents5									
Ac	Acknowledgements 11									
Author's Declaration 13										
1	Intr	oduction	15							
	1.1	Stochastic and deterministic constrained partial differential equations	15							
	1.2	Thesis layout	18							
2	Pre	liminaries	21							
2.1 Hilbert space and orthogonal projection		Hilbert space and orthogonal projection	21							
	2.2	Linear operators	23							
		2.2.1 Closed operators	24							
		2.2.2 Adjoint operators	25							
		2.2.3 Compact operators	25							
	2.3	Semigroups	26							
	2.4	Deterministic compactness criterion	27							
	2.5	Random variables	28							
	2.6	Miscellaneous preliminaries	31							
		2.6.1 Kuratowski-Zorn Lemma	32							
	2.7	Stochastic processes and martingale	33							
	2.8	Wiener process and the martingale representation theorem	36							
	2.9	Tightness and Skorohod Theorem	38							
3	Con	strained Navier-Stokes equations	41							
	3.1 Functional setting									
		3.1.1 Functional setting for \mathbb{R}^2	42							
		3.1.2 Functional setting for a periodic domain	43							

	3.2	Convective term					
	3.3	NSEs	and CNSE	. 45			
4	Det	erministic CNSE on a 2D torus 4					
	4.1	Introduction					
	4.2	Local solution : Existence and Uniqueness					
		4.2.1	Construction of a globally Lipschitz map	51			
		4.2.2	Definition of a solution	. 56			
		4.2.3	Local existence	. 57			
		4.2.4	The local solution stays on the manifold ${\mathscr M}$ \ldots \ldots \ldots \ldots \ldots	. 59			
	4.3	Globa	l solution: Existence and Uniqueness	. 60			
	4.4	Convergence to the Euler equation					
	4.5	CNSE	in the fractional Sobolev spaces	. 66			
		4.5.1	Local solution : Existence and Uniqueness	. 68			
		4.5.2	Maximal solution	. 71			
		4.5.3	Global solution: Existence and Uniqueness	. 73			
	4.6	Lower	bound on the regularity of the initial data	. 76			
5 Stochastic constrained Navier-Stokes equations							
	5.1	Introd	luction	. 79			
	5.2	Stocha	astic constrained Navier-Stokes equations	. 82			
	5.3	Assun	nptions, definitions and results	. 83			
	5.4	Comp	actness	. 85			
		5.4.1	Tightness	. 87			
		5.4.2	The Skorohod Theorem	. 88			
	5.5	Faedo-Galerkin approximation and existence of a martingale solution		. 89			
		5.5.1	Tightness of the laws of approximating solutions	. 95			
		5.5.2	Proof of Theorem 5.3.3	. 97			
	5.6	Pathwise uniqueness and strong solution					
	5.7	The co	ontinuous dependence of solutions on the initial data	. 111			
		5.7.1	Tightness criterion and the Jakubowski-Skorohod Theorem	. 112			
		5.7.2	The continuous dependence	. 114			
	5.8	Seque	ntially weak Feller property	. 122			
6	Sto	chastic	${f c}$ tamed Navier-Stokes equations on ${\Bbb R}^3$	125			
	6.1	Introduction					
	6.2	Functional setting					
		6.2.1	Notations	. 127			
		6.2.2	Some operators	. 128			

Bi	Bibliography								
D	Con	verger	ace of P_n	197					
С	Kur	xi Theorem	193						
B	Som	ie resu	lts in the support of Section 4.4	189					
A	Ortł	nogona	lity of bilinear map to the Stokes operator	187					
		7.2.3	Stochastic hyperbolic constrained Navier-Stokes equations	185					
		7.2.2	Slightly compressible approximation of CNSE	184					
		7.2.1	Stochastic 2D viscous shallow water equations	183					
	7.2 Possible future research directions		le future research directions	183					
		7.1.5	Stochastic tamed Navier-Stokes equations: Invariant measures	183					
		7.1.4	Existence of invariant measures for SCNSE	183					
		7.1.3	SCNSE on \mathbb{R}^2	182					
		7.1.2	Lower bound on the regularity of the initial data	182					
		7.1.1	CNSE on a general bounded domain	181					
	7.1	Open]	problems	181					
7	Оре	Open problems and future directions 18							
		6.6.2	Sequentially weak Feller property	177					
		6.6.1	Boundedness in probability	175					
	6.6	Invari	ant measures	172					
		6.5.4	Uniqueness and strong solutions	171					
		6.5.3	Proof of Theorem 6.5.4	156					
		6.5.2	Tightness of measures	153					
		6.5.1	A'priori estimates	146					
	6.5	6.5 Existence of solution							
	6.4 Truncated SPDE								
		6.3.3	Martingale and strong solution	138					
		6.3.2	The Skorohod Theorem	137					
		6.3.1	Tightness	136					
	6.3	actness	134						
		6.2.3	Assumptions	130					

To my parents Suneel and Vijayshree Dhariwal

ACKNOWLEDGEMENTS

Prof. Zdzisław Brzeźniak has been pivotal to this thesis. He is a very enthusiastic, caring supervisor and an exceptional teacher. I have not only learnt mathematics from him but also, how to be a good teacher and researcher. I thank him for this and for countless values that were instilled in me during my stay at the University of York.

I also thank Dr. Alexei Daletskii and Dr. Simon Eveson, who went the extra mile to make sure that I am on track with my thesis and career planning. Their guidance and encouragement has been valuable and I am indebted to them for it. The collaboration with Dr. Mauro Mariani was one of the most fruitful experience of my academic life. His critical, precise and timely feedback helped me develop into a better academic writer - I greatly appreciate all his effort and thank him for that. I sincerely thank Dr. Utpal Manna, for introducing me to the spellbinding world of SPDEs and for recommending me to Prof. Brzeźniak when I was first looking for a PhD supervisor. To the Department of Mathematics, University of York, for the administrative and financial help this dissertation wouldn't have been possible without your unswerving support.

A special note of thanks to my colleagues and friends. Tayfun, Nimit and Aakash - this thesis would have not seen its end without your fruitful feedbacks. To Demi, Mirjam, Manos, Francis, Kayleigh and Sam, for a productive and refreshing office environment. To Pranav, Ani, Varma and Nilesh, for their endless support, despite being thousands of miles away. To all my friends (an endless list) from York and other parts of the world, for a healthy dose of the real world when I got lost in the abstract world of SPDEs. To my family for always supporting my research interests and for their unsurpassed love and endless patience, thank you all!

Lastly, but most importantly I would like to acknowledge the University of York and the city of York, for making me feel a part of it and providing me with the most conducive environment to pursue my post-graduate studies.

AUTHOR'S DECLARATION

declare that the work in this thesis was carried out in accordance with the requirements of the University's regulations for Research Degree Programmes and I am the sole author. This work was carried out under the supervision of Prof. Zdzisław Brzeźniak, and has not previously been presented for an award at this, or any other, University. All sources are acknowledged and have been listed in the Bibliography.

List of Publications

- (1) Z. Brzeźniak, G. Dhariwal and M. Mariani, 2D Constrained Navier-Stokes equations, Preprint arXiv:1606.08360v2 (Submitted, 2016).
- (2) Z. Brzeźniak and G. Dhariwal, Stochastic Constrained Navier-Stokes Equations on T², Preprint arxiv:1701.01385v2 (2017).
- (3) Z. Brzeźniak, G. Dhariwal, J. Hussain and M. Mariani *Stochastic and deterministic constrained partial differential equations*, (Submitted, 2017).
- (4) Z. Brzeźniak and G. Dhariwal, Stochastic tamed Navier-Stokes equations on \mathbb{R}^3 : existence, uniqueness of solutions and existence of invariant measures, (In preparation, 2017).



INTRODUCTION

he time dependent partial differential equations, commonly known as evolution equations play a crucial role in modelling various natural processes mathematically, which are used to study the behaviour of physical entities like wave function of a particle, temperature profile of a system, stocks in a financial market and velocity of a fluid. Well-known examples are Schrödinger equation from quantum mechanics, reaction diffusion equations modelling biological processes and heat flow, Black-Scholes equation from finance and Navier-Stokes equations from fluid mechanics.

At instances these physical processes are subject to external forcing, which mostly is random in nature. Thus, one has to modify the mathematical models accordingly to incorporate this randomness, which in turn gives rise to stochastic partial differential equations (SPDE), providing us with more robust model to study these natural processes.

Though (S)PDE serve the purpose of analysing these physical entities well, they pose quite basic mathematical questions, like global in time existence and uniqueness of the solution, existence of invariant measures. This thesis deals with such questions for constrained Navier-Stokes equations, stochastic constrained Navier-Stokes equations and stochastic tamed Navier-Stokes equations.

1.1 Stochastic and deterministic constrained partial differential equations

In the theory of partial differential equations one often studies equations with constraints on the values of the unknown function. Here primary examples are geometric heat and wave equations where it is required that the solution is a manifold-valued function. Such models have been

extensively studied, one could mention Eells-Sampson [41], Struwe and Shatah [83–85] for the deterministic problems; Funaki [44], Carroll [32] and Brzeźniak *et al* [11, 27] for the stochastic problems. If the target manifold is a sphere, one can study a generalisation of the heat flow map, called the Landau-Lifshitz-Gilbert Equations [2, 4, 19, 22]. Recently, different kind of constraints, the nonlocal ones, were investigated by Rybka [81], Caffarelli-Lin [30] and Cagliotti *et al.* [31]. For instance, one imposes the constraint that the L^p norm of the solution remains constant.

It is well understood that how to construct a stochastic or deterministic equation on hypersurfaces of an Euclidean space (or even general Hilbert space) from a given equation on an ambient space, provided the latter is given in terms of smooth functions. To be precise let us describe this procedure.

Suppose that $(H, \langle \cdot, \cdot \rangle)$ is a Hilbert space and $\mathcal{M} = \varphi^{-1}(\{1\}) \subset H$ is a hypersurface for some non-degenerate smooth function $\varphi \colon H \to [0, \infty)$. Each element $\omega \in \mathcal{M}$ has a tangent space $T_{\omega}\mathcal{M}$ which can be identified with the closed subspace of H (of co-dimension 1) given by $\ker d_{\omega}\varphi = \{x \in$ $H \colon (d_{\omega}\varphi)(x) = 0\}$, where $d_{\omega}\varphi \in \mathcal{L}(H,\mathbb{R})$ is the Fréchet derivative of φ at ω . By the Riesz Lemma there exists a unique element in H, denoted by $(D\varphi)(\omega)$, such that $(d_{\omega}\varphi)(x) = \langle (D\varphi)(\omega), x \rangle, x \in H$. Since $(D\varphi)(\omega) \neq 0$, the orthogonal projection $\pi_{\omega} \colon H \to T_{\omega}\mathcal{M}$ is given (with $|\cdot|$ being the norm on H) by the formula

(1.1.1)
$$\pi_{\omega}(x) = x - \langle x, \vec{n}(\omega) \rangle \vec{n}(\omega), \quad x \in \mathbf{H}$$

where

$$\vec{n}(\omega) = \frac{\mathbf{D}\varphi(\omega)}{|\mathbf{D}\varphi(\omega)|}, \quad \omega \in \mathcal{M}.$$

Given a vector field $f: H \to H$ we can consider the "tangent projection" \hat{f} of the restriction of f to \mathcal{M} (which is a "tangent" vector field on \mathcal{M}) defined by

(1.1.2)
$$\hat{f}(\omega) := \pi_{\omega}(f(\omega)) \in T_{\omega}\mathcal{M}, \quad \omega \in \mathcal{M}.$$

The associated ODE

(1.1.3)
$$\frac{\mathrm{d}x(t)}{\mathrm{d}t} = f(x(t)), \quad t \ge 0$$

takes the following well-known form on ${\mathcal M}$

(1.1.4)
$$\frac{\mathrm{d}x(t)}{\mathrm{d}t} = \hat{f}(x(t)), \quad t \ge 0$$

Note that \hat{f} has a smooth extension to an open neighbourhood of \mathcal{M} and the ODE (1.1.4) is locally well-posed on that neighbourhood. One can then show that given $x_0 \in \mathcal{M}$, the local solution stays on \mathcal{M} , by either using local diffeomorphism of some neighbourhood of x_0 in \mathcal{M} (i.e. de facto the Hilbert manifold structure of \mathcal{M}) or by showing that $\varphi(x(t)) = 0$ for t in the domain of the solution. If one knows that \mathcal{M} is a compact set (that requires H to be finite dimensional) then we can easily deduce that each solution starting at $x_0 \in \mathcal{M}$ is a global one, i.e. defined on $[0,\infty)$. However, if \mathcal{M} is not compact the solutions may not be global-in-time.

Similar argument can be made for stochastic differential equations, with one small but important difference. Suppose f_0, f_1, \dots, f_N is a finite collection of vector fields on H and $W = (W(t)), t \ge 0$ is an \mathbb{R}^N -valued Wiener process, we write $W(t) = (W_j(t))$ defined on some filtered probability space $(\Omega, \mathscr{F}, \mathbb{F}, \mathbb{P}), \mathbb{F} = (\mathscr{F}_t)_{t\ge 0}$, satisfying the so-called usual conditions [63].

In the whole ambient space we can study stochastic differential equations either in the Itô or Stratonovich form, the latter requiring more regularity assumptions on the vector fields f_1, \dots, f_N :

(1.1.5)
$$dx = f_0(x)dt + \sum_{i=1}^N f_i(x)dW_i,$$

or

(1.1.6)
$$dx = f_0(x)dt + \sum_{i=1}^N f_i(x) \circ dW_i$$
$$= f_0(x)dt + \left[\frac{1}{2}\sum_{i=1}^N f_i'(x)f_i(x)\right]dt + \sum_{i=1}^N f_i(x)dW_i,$$

where $f'_i(x) = d_x f_i$, $x \in H$. On the other hand, it turns out that the correct form of these equations on \mathcal{M} is the Stratonovich one. This fact is related to the Wong-Zakai type theorems, see [10] or the rough paths theory proposed recently by Terry Lyons [59]. With the same notation as before one can consider an equation

(1.1.7)
$$dx = \hat{f}_0(x) dt + \sum_{j=1}^N \hat{f}_j(x) \circ dW_j$$
$$= \hat{f}_0(x) dt + \sum_{j=1}^N \hat{f}_j(x) dW_j + \frac{1}{2} \sum_{j=1}^N \hat{f}_j'(x) \hat{f}_j(x) dt.$$

The issues of local and global solutions to the above problem can be solved through a similar approach as the one used to answer the analogous questions in the deterministic case, see for instance [15] and references therein.

However, when the vector fields are not smooth or not everywhere defined or both, the situation changes. For instance, let us consider an unbounded, self-adjoint and non-negative operator A on a Hilbert space H. The domain of A, denoted by D(A), is a Hilbert space endowed with the "graph norm" :

$$|x|_{D(A)}^2 = |x|^2 + |Ax|^2, \qquad x \in D(A).$$

Such an operator A induces a (only densely defined) vector field $f_0(x) = -Ax$. Theory of corresponding deterministic and stochastic problems related to equations (1.1.5) and (1.1.6) is now well developed and understood, see e.g. a monograph [38] by Da Prato and Zabczyk. However, this is not the case for equations (1.1.7) with a vector field \hat{f}_0 defined by

(1.1.8)
$$\hat{f}_0(x) = f_0(x) - \langle f_0(x), \vec{n}(x) \rangle \vec{n}(x), \quad x \in \mathcal{M} \cap \mathbf{D}(\mathbf{A}),$$

in the view of (1.1.1) and (1.1.2). In [48], Hussain studied reaction diffusion equations under a non-local constraint, in both deterministic and stochastic cases. He established the existence of "unique" global "solution" for these equations.

The aim of this thesis is to present a detailed study of such questions in the case of Navier-Stokes equations. We address the questions of existence and uniqueness for two dimensional Navier-Stokes equations with a constraint on the L^2 norm of the solution. Another related problem that we have addressed in this thesis is that of three dimensional tamed Navier-Stokes equations, which can be used to study the properties of the solution (if exist) of three dimensional Navier-Stokes equations. We provide a new approach to prove the existence of a pathwise unique strong solution for the tamed Navier-Stokes equations under the impact of stochastic forcing.

1.2 Thesis layout

Chapter 2 includes all the necessary preliminaries required by the reader to understand the thesis. The majority of this thesis focuses on constrained Navier-Stokes equations. Thus, in Chapter 3 after introducing certain functional spaces and operators we deduce constrained Navier-Stokes equations (CNSE) from Navier-Stokes equations under the constraint of constant energy.

The questions of existence and uniqueness of a global-in-time solution of the deterministic CNSE are considered in Chapter 4. We start by giving the motivation behind studying such a system and stating the main results of the chapter. We prove the global existence of the solutions using the Banach Fixed Point Theorem and a non-explosion principle, i.e. proving the enstrophy (gradient norm) of the solution remains bounded. The periodic boundary conditions play a crucial role in obtaining the boundedness of enstrophy. We show that if the solution starts from the manifold \mathcal{M} , then it stays on \mathcal{M} . Furthermore we prove the existence of local solutions to CNSE with Dirichlet boundary conditions (as well as invariance of \mathcal{M}). In the vanishing viscosity limit, we show that the CNSE converges to the Bardos solution (see [5]) of the Euler equation. We extend our analysis to fractional Sobolev spaces, where using the Banach Fixed Point Theorem, Kuratowski-Zorn Lemma and maximal solutions we prove the global existence of solutions to CNSE. We end the chapter with an informal discussion about the lower bound on the regularity of the initial data corresponding to the well-posedness of CNSE.

The stochastic generalisation of CNSE is studied in Chapter 5. We consider the noise of gradient type in the Stratonovich form. The structure of the noise is such that it is "tangent" to the manifold \mathcal{M} . Here we take more classical approach of Faedo-Galerkin approximation to prove the existence of local solutions. We start by showing that each approximating equation has a global solution, which satisfy suitable a'priori estimates. Then, using Aldous condition along a'priori estimates, we prove that the laws of the solutions of these approximating equations are tight on a suitably chosen topological space \mathcal{Z}_T . By applications of the Jakubowski-Skorohod

and martingale representation theorems we deduce the existence of martingale solutions. We also prove the so-called maximum regularity of solutions and their pathwise uniqueness, which helps us to establish the existence of a strong solution to stochastic constrained Navier-Stokes equations (SCNSE) by invoking Yamada-Watanabe type theorem. We end the chapter by showing that the solution of SCNSE depends continuously on the initial data.

Moving away from constrained equations, in Chapter 6 we shift our focus to tamed Navier-Stokes equations, which were introduced by Röckner and Zhang [75]. In this chapter we study stochastic tamed Navier-Stokes equations on \mathbb{R}^3 and reprove the results from [76] using a different approach and in doing so we generalise the L^4 estimate of the solution from \mathbb{T}^3 to the whole Euclidean space. We use three new ideas to prove the existence and uniqueness of global solutions and the existence of invariant measures on the whole Euclidean space. Firstly, in contrast to classical Faedo-Galerkin approximation where one uses finite dimensional spaces, we study the truncated equations on infinite dimensional spaces. We prove the existence of global solutions to these truncated equations satisfying suitable a'priori estimates. Secondly, we use a tightness criterion related to Dubinsky's compactness criterion introduced recently by Brzeźniak and Motyl [23]. Finally, we end the chapter by proving the existence of invariant measures using the method based on continuity and compactness in the weak topologies [62].

Chapter 7, the final chapter of the thesis summarises some of the open problems arising from this thesis and also enlists other related problems that will be part of my future research.



PRELIMINARIES

his chapter has been included in the thesis so as to make it self-contained. We hope to provide all the necessary mathematical concepts that the reader might require to understand this thesis. The content of this chapter has been taken from various textbooks which have been aptly listed in the bibliography.

2.1 Hilbert space and orthogonal projection

Let *H* be a Hilbert space with the norm $|\cdot|_H$ induced by the inner product $\langle \cdot, \cdot \rangle_H$.

Definition 2.1.1. Let $x, y \in H$, then we say x is orthogonal to y if $\langle x, y \rangle_H = 0$. In general, if V is a subspace of H, then so is the set

$$V^{\perp} = \{ x \in H : \langle x, y \rangle_H = 0 \ \forall y \in V \} .$$

Lemma 2.1.2. [50, Theorem 21.4] Let V be a closed subspace of H. Then every $x \in H$ has a unique decomposition

$$x = y + z$$
 with $y \in V, z \in V^{\perp}$.

Theorem 2.1.3 (Projection Theorem). Let V be a closed subspace of H. Then for every $x \in H$, there exists a unique element $\hat{x} \in V$ (same as y from the previous lemma) such that

$$|x - \hat{x}|_H = \inf_{\mathbf{v} \in V} |\mathbf{v} - x|_H.$$

Moreover

i) $\hat{x} = x$ *iff* $x \in V$.

ii) $x - \hat{x} \in V^{\perp}$ and

$$|x|_{H}^{2} = |\hat{x}|_{H}^{2} + |x - \hat{x}|_{H}^{2}.$$

Corollary 2.1.4. [50, Corollary 21.5] For every closed subspace V of H there exists a unique linear map

$$\pi: H \ni x \mapsto \hat{x} \in V,$$

with

$$\|\pi\| := \sup_{x \in H, x \neq 0} \frac{|\pi x|_H}{|x|_H} = 1,$$

$$\pi^2 = \pi \text{ and } \ker \pi = V^{\perp}.$$

Remark 2.1.5. The existence of the element \hat{x} in Corollary 2.1.4 is guaranteed by Theorem 2.1.3. The map π described in Corollary 2.1.4 is called the orthogonal projection of H onto V.

Definition 2.1.6. The sequence $(x_n)_{n \in \mathbb{N}} \subset H$ is said to converge (strongly) to $x \in H$, symbolically $x_n \to x$, if

$$\forall \varepsilon > 0 \exists N \in \mathbb{N} \colon \forall n \ge N \ |x_n - x|_H < \varepsilon.$$

Definition 2.1.7. A sequence $(x_n)_{n \in \mathbb{N}} \subset H$ is called weakly convergent to $x \in H$, symbolically $x_n \rightharpoonup x$, if

$$\langle x_n, y \rangle_H \to \langle x, y \rangle_H$$
, for all $y \in H$.

One can easily see that a sequence converging in usual sense also converges weakly, since,

$$|\langle x_n, y \rangle - \langle x, y \rangle| \le |x_n - x|_H |y|_H.$$

Theorem 2.1.8. [50, Theorem 21.8] Every bounded sequence $(x_n)_{n \in \mathbb{N}} \subset H$ has a weakly convergent subsequence.

Lemma 2.1.9. [50, Lemma 21.11] Every weakly convergent sequence $(x_n)_{n \in \mathbb{N}} \subset H$, is bounded.

Remark 2.1.10. H is not metrizable w.r.t weak convergence. But a closed unit ball in a separable Hilbert space H w.r.t weak convergence is metrizable.

Definition 2.1.11. *H* is called a separable Hilbert space if it contains a countable dense subset.

Definition 2.1.12. An orthonormal basis of a Hilbert space *H* is a sequence $\{e_j\}_{j \in \mathbb{N}} \subset H$, such that linear span of $\{e_j\}$ is dense in *H* and

$$\langle e_j, e_k \rangle_H = \delta_{jk}, \qquad j,k \in \mathbb{N},$$
$$|e_j|_H = 1, \qquad \forall j \in \mathbb{N}$$

where δ_{jk} is the Kronecker delta.

Theorem 2.1.13. Every separable Hilbert space H admits an orthonormal basis.

2.2 Linear operators

Let *X* and *Y* be normed spaces with $|\cdot|_X$ and $|\cdot|_Y$ norms respectively.

Definition 2.2.1. A linear operator from *X* to *Y* is a map $L: X \to Y$ such that for $\alpha, \beta \in \mathbb{R}$

$$L(\alpha x + \beta y) = \alpha L x + \beta L y, \quad x, y \in X.$$

Definition 2.2.2. If $L: X \to Y$ is a linear operator, the kernel of *L* is defined as the pre-image of the null vector in *Y* i.e.

$$\ker L := \{x \in X : Lx = 0\}.$$

The range of *L* is the set of all images i.e.

$$L(X) := \{Lx : x \in X\}.$$

Definition 2.2.3. A linear operator $L: X \to Y$ is called bounded if there exists a C > 0 such that

$$|Lx|_Y \le C|x|_X, \quad x \in X.$$

Theorem 2.2.4. [50, Theorem 7.18] A linear operator $L: X \to Y$ is bounded if and only if it is continuous i.e. if $x_n \to x$, then $Lx_n \to Lx$.

The set of all bounded linear operators from X into Y is denoted by $\mathscr{L}(X,Y)$. If X = Y then we will write $\mathscr{L}(X)$.

Theorem 2.2.5. [50, Theorem 7.21] Let Y be a Banach space and X a normed space. Then, $\mathcal{L}(X,Y)$, with the norm

$$\|L\|_{\mathscr{L}(X,Y)} := \sup_{x \in X, |x|_X = 1} |Lx|_Y, \quad L \in \mathscr{L}(X,Y),$$

is a Banach space.

Lemma 2.2.6. Let Z be a real normed space with the norm $|\cdot|_Z$. Assume that $L_1: X \to Y$ and $L_2: Y \to Z$ are bounded linear operators. Then the composition $L_2 \circ L_1$ is a bounded linear operator from X into Z and

$$(2.2.2) |L_2 \circ L_1|_{\mathscr{L}(X,Z)} \le |L_2|_{\mathscr{L}(Y,Z)}|L_1|_{\mathscr{L}(X,Y)}.$$

Definition 2.2.7. A sequence of operators $(L_n)_{n \in N} \subset \mathscr{L}(X, Y)$ is said to:

a) converge in operator norm to $L \in \mathscr{L}(X, Y)$, if

$$||L_n - L||_{\mathscr{L}(X,Y)} \to 0 \text{ as } n \to \infty.$$

b) strongly converge to $L \in \mathcal{L}(X, Y)$, iff $L_n x$ converges to Lx strongly in Y for each x in X.

Definition 2.2.8. A linear operator $L: X \to \mathbb{R}$ is called a linear functional. The space of all bounded linear functionals $f: X \to \mathbb{R}$ is called the dual space of X and is denoted by X^* or X', i.e. $X^* = \mathscr{L}(X, \mathbb{R})$.

Remark 2.2.9. By Theorem 2.2.5, X^* is a Banach space with the norm

(2.2.3) $|L|_{\mathscr{L}(X,\mathbb{R})} = \sup_{x \in X : |x|_X = 1} |Lx|, \quad L \in X^*.$

Theorem 2.2.10 (Riesz Representation Theorem). [50, Theorem 21.6] Let L be a bounded linear functional on the Hilbert space H. Then there exists a uniquely determined $y \in H$ with

$$Lx = \langle x, y \rangle_H, \quad \forall x \in H.$$

Moreover

$$|L|_{\mathscr{L}(H,R)} = |y|_H$$

Definition 2.2.11. If there exits an injective continuous linear map L from X into Y, then X is said to be embeddable in Y. Such a map L is called the embedding.

Definition 2.2.12. Assume that $X \subseteq Y$. Then X is called continuously embedded in Y if the inclusion map (identity function) $i: X \to Y$ is continuous, i.e. there exists a constant c > 0 such that

$$|u|_X \le c|u|_Y \quad u \in X.$$

In this case we denote this embedding symbolically by $X \hookrightarrow Y$ and the map *i* is called the embedding operator.

Definition 2.2.13. Let X^* be dual and X^{**} be double dual of X respectively. Then we have a canonical map $x \to \hat{x}$ defined by:

$$\widehat{x}(f) = f(x) \quad f \in X^*,$$

gives an isometric linear isomorphism (embedding) from X into X^{**} . The space X is called reflexive if this map is also surjective.

2.2.1 Closed operators

Definition 2.2.14. $L: D(L) \to Y$ is a linear operator, $D(L) \subset X$, then L is called a closed linear operator if its graph

$$G(L) = \{(x, y) : x \in D(L), y = Lx\}$$

is closed in the normed space $X \times Y$.

Theorem 2.2.15. [50] Let L be a linear operator in the Banach space $(X, |\cdot|_X)$. Define norm on D(L) by

$$|x|_{\mathcal{D}(L)} = |x|_X + |Lx|_X, \quad x \in \mathcal{D}(L).$$

Then $(D(L), |\cdot|_{D(L)})$ is a Banach space iff L is closed.

2.2.2 Adjoint operators

Let *H* be a Hilbert space with the norm $|\cdot|_H$. Let $L: D(L) \to H$ be a densely defined operator, with $D(L) \subset H$. Let us denote by $D(L^*)$ the set,

$$D(L^*) = \{y \in H : D(L) \ni x \to \langle Lx, y \rangle_H \in \mathbb{R} \text{ is } H\text{-continuous} \}.$$

Note that if *L* is bounded then $D(L^*) = H$.

Hence by Riesz Representation Theorem 2.2.10, for every $y \in D(L^*)$ there exists a unique (uniqueness is guaranteed by the denseness of L) $z \in H$ such that

$$\langle Lx, y \rangle_H = \langle x, z \rangle_H, \quad \forall x \in \mathrm{D}(L).$$

For $y \in D(L^*)$, put $L^* y = z$.

Definition 2.2.16. For a densely defined linear operator $L: D(L) \rightarrow H$, with $D(L) \subset H$, its adjoint is an operator $L^*: D(L^*) \rightarrow H$ that satisfies the identity

$$\langle Lx, y \rangle_H = \langle x, L^*y \rangle_H, \quad x \in \mathcal{D}(L), y \in \mathcal{D}(L^*).$$

Definition 2.2.17. A densely defined operator $L: D(L) \to H$, with $D(L) \subset H$, is called self-adjoint iff $D(L) = D(L^*)$ and

$$\langle Lx, y \rangle_H = \langle x, Ly \rangle_H, \quad x, y \in \mathrm{D}(L).$$

2.2.3 Compact operators

Definition 2.2.18. Let *X* and *Y* be normed spaces. A linear operator $L: X \to Y$ is called compact if for each bounded sequence $(x_n)_{n \in \mathbb{N}} \subset X$, the sequence $(Lx_n)_{n \in \mathbb{N}}$ has a convergent subsequence in *Y*.

Definition 2.2.19. The embedding $X \hookrightarrow Y$ is compact iff the identity map $i: X \to Y$ is compact.

Lemma 2.2.20. [96, Theorem 8.3] Let X and Y be Banach spaces, then $\mathcal{L}C(X,Y)$, set of all compact operators from X into Y is a closed (and hence complete) subspace of $\mathcal{L}(X,Y)$ with operator norm.

Definition 2.2.21. Let *X* and *Y* be Hilbert spaces. A linear operator $L: X \to Y$ is called Hilbert-Schmidt if for every complete orthonormal basis $(e_n)_{n \in \mathbb{N}} \subset X$

(2.2.4)
$$\|L\|_{HS} := \sum_{j=1}^{\infty} |Le_j|_Y^2 < \infty.$$

The space of all Hilbert-Schmidt operators $L: X \to Y$ will be denoted by $\mathcal{T}_2(X,Y)$ and the operator norm will be denoted by $\|\cdot\|_2$ or $\|\cdot\|_{\mathcal{T}_2(X,Y)}$ which is equal to $\|\cdot\|_{HS}$ norm defined in (2.2.4).

Theorem 2.2.22. [96, Theorem 8.7] Hilbert-Schmidt operators are compact.

2.3 Semigroups

For this section we will assume that X is a Banach space with the norm $|\cdot|_X$. The content of this section is based on [73].

Definition 2.3.1. A function $S : [0,\infty) \ni t \mapsto S(t) \in \mathscr{L}(X)$, which is usually denoted by $S = \{S(t)\}_{t \ge 0}$, is called a semigroup of linear bounded operators on X if

- i) S(0) = I, where I is the identity operator on X,
- ii) for all $t, s \ge 0$,

$$S(t+s) = S(t)S(s),$$

where S(t)S(s) denotes the composition of the operators S(t) and S(s).

Definition 2.3.2. Let *S* be a semigroup on *X*. If

(2.3.1)
$$\lim_{t \to 0} |S(t) - I|_{\mathscr{L}(X)} = 0,$$

then S is called a uniformly continuous semigroup.

Definition 2.3.3. A semigroup *S* on *X* is called a C_0 -semigroup (or strongly continuous semigroup) iff for each $x \in X$,

(2.3.2)
$$\lim_{t \to 0} |S(t)x - x|_X = 0.$$

Theorem 2.3.4. If S be a C₀-semigroup on X, then there exist constants $M \ge 1$ and $\beta \ge 0$ such that

$$|S(t)|_{\mathscr{L}(X)} \le M e^{\beta t}, \quad t \ge 0$$

Corollary 2.3.5. If S is a C_0 -semigroup on X, then for each $x \in X$, the function

$$S(\cdot)(x): [0,\infty) \ni t \mapsto S(t)(x) \in X$$

is continuous on $[0,\infty)$.

Definition 2.3.6. Let *S* be a semigroup on *X*. A linear operator *L* defined by

(2.3.4)
$$D(L) = \left\{ x \in X : \lim_{t \to 0} \frac{S(t)x - x}{t} \text{ exists} \right\}$$

and

(2.3.5)
$$Lx = \lim_{t \to 0} \frac{S(t)x - x}{t} = \frac{dS(t)x}{dt}\Big|_{t=0}, \quad x \in D(L)$$

is called the infinitesimal generator of the semigroup S.

Theorem 2.3.7. A linear operator L is the infinitesimal generator of a uniformly continuous semigroup iff L is a bounded linear operator.

Theorem 2.3.8. Let S be a C_0 -semigroup on X with the infinitesimal generator L. Then for all $t \ge 0$,

i) for each $x \in X$,

(2.3.6)
$$\lim_{h \to 0} \frac{1}{h} \int_{t}^{t+h} S(s) x \, ds = S(t) x \, ,$$

ii) for each $x \in X$, $\int_0^t S(s)x \, ds \in D(L)$ and

(2.3.7)
$$L\left(\int_0^t S(s)x\,ds\right) = S(t)x - x$$

iii) for each $x \in D(L)$, $S(t)x \in D(L)$ and

(2.3.8)
$$\frac{d}{dt}S(t)x = LS(t)x = S(t)Lx,$$

iv) for each $x \in D(L)$,

(2.3.9)
$$S(t)x - S(s)x = \int_s^t S(\tau)Lx \, d\tau = \int_s^t LS(\tau)x \, d\tau$$

Corollary 2.3.9. If *L* is the infinitesimal generator of a C_0 semigroup *S* on *X*, then D(L) is dense in *X* and *L* is a closed linear operator.

Definition 2.3.10. A C_0 -semigroup S on X is called uniformly bounded if there exists a constant $M \ge 1$ such that for each $t \ge 0$,

$$(2.3.10) |S(t)|_{\mathcal{L}(X)} \le M.$$

Definition 2.3.11. A C_0 -semigroup *S* on *X* is called a contraction semigroup if for each $t \ge 0$,

$$(2.3.11) |S(t)|_{\mathscr{L}(X)} \le 1.$$

2.4 Deterministic compactness criterion

Let (X, ρ) be a metric space. Consider the set,

 $\mathscr{C}(X) := \{ f : X \to \mathbb{R} \text{ is continuous} \},\$

then $\mathcal{C}(X)$ is a Banach space endowed with the norm

$$\|f\|_{\mathscr{C}(X)} := \sup_{x \in X} |f(x)|.$$

We say that a sequence (f_n) converges to f in $\mathscr{C}(X)$ if

$$\sup_{x\in X} |f_n(x) - f(x)| \to 0,$$

i.e. (f_n) converges uniformly to f in X.

Definition 2.4.1. Let (X, ρ) be a metric space. A family of functions $\Lambda \subset \mathscr{C}(X)$ is called equicontinuous if for every $\varepsilon > 0$ there exists a $\delta > 0$ such that for all $f \in \Lambda$,

$$|f(u) - f(v)| < \varepsilon$$
, for all $u, v \in X$ satisfying $\rho(u, v) < \delta$.

The family Λ is called uniformly bounded if there exists a constant C > 0 such that

 $|f(u)| \le C$, for all $f \in \Lambda$ and for all $u \in X$.

Definition 2.4.2. If (X, ρ) be a metric space, a subset $K \subset X$ is called precompact (or relatively compact) if closure \overline{K} of K is compact in X.

Theorem 2.4.3 (Arzelà-Ascoli Theorem). [50, Theorem 5.20] Let (X, ρ) be a compact metric space. For a family of functions $\Lambda \subset \mathscr{C}(X)$ following conditions are equivalent:

- i) Λ is relatively compact,
- ii) Λ is equicontinuous and uniformly bounded.

Following is an immediate consequence of the Arzelà-Ascoli theorem.

Corollary 2.4.4. [50, Corollary 5.21] Let (X, ρ) be a compact metric space. If a sequence of functions $(f_n) \subset \mathscr{C}(X)$ is equicontinuous and uniformly bounded then it contains a uniformly convergent subsequence.

Now we state the classical compactness criteria due to Dubinsky [94, Theorem IV.4.1] (see also [56]).

Theorem 2.4.5 (Dubinsky Theorem). Let E_0, E and E_1 be reflexive Banach spaces such that $E_0 \hookrightarrow E \hookrightarrow E_1$ and the embedding $E_0 \hookrightarrow E$ is compact. Let $q \in (1, \infty)$ and let K be a bounded subset in $L^q(0,T;E_0)$ consisting of functions equicontinuous in $\mathcal{C}([0,T];E_1)$. Then K is relatively compact in $L^q(0,T;E) \cap \mathcal{C}([0,T];E_1)$.

2.5 Random variables

Throughout this section, we assume that X is a separable Banach space with norm $|\cdot|_X$ and Ω is a non-empty set. The contents of this section are based on [38, 79].

Definition 2.5.1. A family \mathscr{F} of subsets of Ω is called a σ -field on Ω if

- i) $\Omega \in \mathscr{F}$,
- ii) if $A \in \mathscr{F}$, then $\Omega \setminus A \in \mathscr{F}$,
- iii) if A_1, A_2, \dots is a sequence of sets in \mathscr{F} , then the countable union $\bigcup_{n \in \mathbb{N}} A_n$ also belongs to \mathscr{F} .

The pair (Ω, \mathscr{F}) is called a measurable space.

Definition 2.5.2. Let (Ω_1, \mathscr{F}) and (Ω_2, \mathscr{G}) be two measurable spaces. A map $\xi : \Omega_1 \to \Omega_2$ is said to be measurable if for every $A \in \mathscr{G}$,

$$\xi^{-1}(A) = \{\omega \in \Omega_1 : \xi(\omega) \in A\} \in \mathscr{F}.$$

Such a measurable map is called a random variable on Ω_1 .

Definition 2.5.3. Let \mathscr{H} be a family of subsets of Ω . The smallest σ -field on Ω containing \mathscr{H} is called the σ -field generated by \mathscr{H} and it is denoted by $\sigma(\mathscr{H})$.

Definition 2.5.4. The smallest σ -field containing all closed (or open) subsets of X is called the Borel σ -field of X and it is denoted by $\mathscr{B}(X)$.

Lemma 2.5.5. [38, Proposition 1.3] Let X^* be the dual space of X. Then $\mathscr{B}(X)$ is the smallest σ -field of X containing all sets of the form

$$\{x \in X : \varphi(x) \le \alpha\}, \quad \varphi \in X^*, \quad \alpha \in \mathbb{R}.$$

Definition 2.5.6. Let (Ω, \mathscr{F}) be a measurable space. A mapping $\xi : \Omega \to X$ is said to be Borel measurable if for each $A \in \mathscr{B}(X)$, $\xi^{-1}(A) \in \mathscr{F}$. Such a Borel measurable map is called an *X*-valued random variable on Ω .

Definition 2.5.7. Let (Ω, \mathscr{F}) be a measurable space and ξ be an *X*-valued random variable on Ω . The smallest σ -field $\sigma(\xi)$ containing all sets $\xi^{-1}(A)$, $A \in \mathscr{B}(X)$, is called the σ -field generated by ξ .

Lemma 2.5.8. [38, Lemma 1.5] Let (Ω, \mathscr{F}) be a measurable space. Assume that ξ and ζ are X-valued random variables on Ω . Then

- i) for $\alpha, \beta \in \mathbb{R}$, $\alpha\xi + \beta\zeta$ is an X-valued random variable on Ω ,
- ii) the mapping $\Omega \ni \omega \mapsto |\xi(\omega)|_X$ is a real-valued random variable on Ω .

Definition 2.5.9. Let (Ω, \mathscr{F}) be a measurable space. A map $\mu : \mathscr{F} \to \mathbb{R}$ is called a non-negative measure if

- i) for all $A \in \mathcal{F}$, $\mu(A) \ge 0$,
- ii) $\mu(\phi) = 0$,
- iii) for all countable collection $\{A_i\}_{i=1}^{\infty}$ of pairwise disjoint sets in \mathscr{F} ,

$$\mu\left(\bigcup_{i=1}^{\infty}A_i\right)=\sum_{i=1}^{\infty}\mu(A_i).$$

The third condition in the definition is called the σ -additivity and the triple $(\Omega, \mathscr{F}, \mu)$ is called a measure space. A non-negative measure \mathbb{P} satisfying $\mathbb{P}(\Omega) = 1$ is called a probability measure and the triple $(\Omega, \mathscr{F}, \mathbb{P})$ is called a probability space.

Theorem 2.5.10 (Lebesgue's Monotone Convergence Theorem). [79, Theorem 1.26] Let $(f_n)_{n \in \mathbb{N}}$ be a sequence of real valued measurable functions on Ω , satisfying

- a) $0 \le f_1 \le f_2 \le \dots \le \infty$
- b) $f_n \rightarrow f$ point-wise as $n \rightarrow \infty$.

Then f is measurable and

$$\int_{\Omega} f_n \, d\mu \to \int_{\Omega} f \, d\mu \ as \ n \to \infty.$$

Theorem 2.5.11 (Fatou's Lemma). [79, Lemma 1.28] If $f_n : \Omega \to [0,\infty]$ is measurable for all $n \in \mathbb{N}$, then

$$\int_{\Omega} \left(\liminf_{n \to \infty} \inf f_n \right) d\mu \leq \liminf_{n \to \infty} \inf \int_{\Omega} f_n d\mu,$$
$$\lim_{n \to \infty} \sup \int_{\Omega} f_n d\mu \leq \int_{\Omega} \left(\limsup_{n \to \infty} \sup f_n \right) d\mu.$$

Theorem 2.5.12. [79, Theorem 1.33] If $f \in L^{1}(\mu)$ i.e. $\int_{E} f d\mu < \infty$, then

$$\left|\int_{\Omega} f \, d\mu\right| \leq \int_{\Omega} |f| \, d\mu$$

Theorem 2.5.13 (Lebesgue's Dominated Convergence Theorem). [79, Theorem 1.34] Suppose $(f_n)_{n \in \mathbb{N}}$ be the sequence of real valued measurable functions on Ω such that $f_n \to f$ point-wise as $n \to \infty$. If there exists a function $g \in L^1(\mu)$ such that $|f_n(\omega)| \leq g(\omega)$ for every $\omega \in \Omega$ then $f \in L^1(\mu)$ and

$$\lim_{n \to \infty} \int_{\Omega} |f_n - f| \, d\mu = 0,$$
$$\lim_{n \to \infty} \int_{\Omega} f_n \, d\mu = \int_{\Omega} f \, d\mu.$$

Theorem 2.5.14 (Vitali Convergence Theorem or Vitali Theorem). [79, Exercise 6.10 (b)] Let $(\Omega, \mathcal{F}, \mu)$ be a measure space. If $\mu(\Omega) < \infty$ and $\{f_n\}_{n \in \mathbb{N}}$ is a sequence of functions on Ω such that

- i) $\{f_n\}$ is uniformly integrable,
- ii) $f_n \rightarrow f$ pointwise a.e. on Ω ,
- *iii*) $|f(x)| < \infty a.e.$

Then $f \in L^1(\mu)$ and

$$\lim_{n\to\infty}\int_{\Omega}|f_n-f|\,d\mu=0$$

Definition 2.5.15. Let $(\Omega, \mathscr{F}, \mathbb{P})$ be a probability space. A set $\overline{\mathscr{F}}$ defined by

$$\overline{\mathscr{F}} = \{A \subset \Omega : \exists B, C \in \mathscr{F}; B \subset A \subset C, \mathbb{P}(B) = \mathbb{P}(C)\}$$

is a σ -field and is called the completion of \mathscr{F} . If $\mathscr{F} = \overline{\mathscr{F}}$, then the probability space $(\Omega, \mathscr{F}, \mathbb{P})$ is said to be complete.

Definition 2.5.16. Let $(\Omega, \mathscr{F}, \mathbb{P})$ be a probability space and ξ be an *X*-valued random variable on Ω . Then a mapping $L(\xi) : \mathscr{B}(X) \to [0, 1]$ defined by

$$L(\xi)(A) = \mathbb{P}\left(\xi^{-1}(A)\right) = \mathbb{P}\left(\{\omega \in \Omega : \xi(\omega) \in A\}\right), \quad A \in \mathscr{B}(X)$$

is called the law (or the distribution) of ξ .

Theorem 2.5.17 (Kuratowski Theorem). [70, Theorem 3.9] Assume that X_1, X_2 are the Polish spaces with their Borel σ -fields denoted respectively by $\mathscr{B}(X_1), \mathscr{B}(X_2)$. If $\varphi: X_1 \to X_2$ is an injective Borel measurable map then for any $E_1 \in \mathscr{B}(X_1), E_2 := \varphi(E_1) \in \mathscr{B}(X_2)$.

2.6 Miscellaneous preliminaries

Lemma 2.6.1 (Gagliardo - Nirenberg inequality). [91] Assume that $r, q \in [1, \infty)$, and $j, m \in \mathbb{Z}$ satisfy $0 \le j < m$. Then for all $\alpha \in \left[\frac{j}{m}, 1\right]$ there exists a constant C > 0 such that

(2.6.1)
$$\left| D^{j} u \right|_{L^{p}(\mathbb{R}^{n})} \leq C \left| D^{m} u \right|_{L^{r}(\mathbb{R}^{n})}^{\alpha} \left| u \right|_{L^{q}(\mathbb{R}^{n})}^{1-\alpha}, \quad u \in C_{0}^{\infty}(\mathbb{R}^{n}),$$

where $\frac{1}{p} = \frac{j}{n} + \alpha \left(\frac{1}{r} - \frac{m}{d}\right) + (1 - \alpha)\frac{1}{q}$. If $m - j - \frac{n}{r}$ is a non-negative integer, then the equality holds only for $\alpha \in \left[\frac{j}{m}, 1\right]$.

The next two results play a pivotal role in this thesis.

Theorem 2.6.2 (Banach Fixed Point Theorem). [50, Theorem 4.7] Let (X,d) be a complete metric space, $K \subset X$ be a closed subset, $f : K \to K$ be a function that satisfies the inequality, for some $0 \le \alpha < 1$,

$$d(f(u), f(v)) \le \alpha d(u, v), \quad \text{for all } u, v \in X,$$

Then f has uniquely determined fixed point in K i.e. there exists a unique $a \in K$ such that f(a) = a.

Lemma 2.6.3. [88, Lemma III.1.2] Let T > 0, V, H be two Hilbert spaces, V^* and H^* be corresponding dual spaces. Assume that $V \hookrightarrow H \equiv H^* \hookrightarrow V^*$, where embeddings are dense too. If a function u belongs to $L^2(0,T;V)$ and its weak derivative u' belongs to $L^2(0,T;V')$, then u is a.e. equal to a continuous function $v : [0,T] \to H$ such that the function $[0,T] \ni t \mapsto |v(t)|_{H}^2 \in \mathbb{R}$ is absolutely continuous and

$$\frac{d}{dt}|\mathbf{v}(t)|_{H}^{2} = 2\left\langle \mathbf{v}', \mathbf{v} \right\rangle_{H}, \quad for \ almost \ all \ t \in [0, T].$$

The following lemma is used repeatedly in this thesis. We will later state a generalisation of it for random variables.

Lemma 2.6.4 (Bellman–Gronwall Inequality or Gronwall Lemma). [86, Section 1.3.6] Suppose $\phi \in L^1[a, b]$ satisfies

$$\phi(t) \leq f(t) + \beta \int_a^t \phi(s) ds, \quad a.e.$$

where $f \in L^1[a,b]$ and β is a positive constant, then

$$\phi(t) \le f(t) + \beta \int_a^t f(s) e^{\beta(t-s)} ds, \quad \text{for a.e. } t \in [a,b].$$

In particular if $f(t) = \alpha$ (constant) then

$$\phi(t) \le \alpha e^{\beta(t-a)}, \quad \text{for a.e. } t \in [a,b].$$

Lemma 2.6.5. [58, Chapter 2] Let $f \in L^{p_0} \cap L^{p_1}$. Then for $\theta \in (0,1)$ there exists a constant c > 0 such that

$$|f|_{L^{p_{\theta}}} \leq c|f|_{L^{p_{0}}}^{1-\theta}|f|_{L^{p_{1}}}^{\theta}, \quad f \in L^{p_{0}} \cap L^{p_{1}},$$

where

$$\frac{1}{p_{\theta}} = \frac{1-\theta}{p_0} + \frac{\theta}{p_1}.$$

Lemma 2.6.6 (Poincaré Inequality). [50, Corollary 20.16] Let $\Omega \subset \mathbb{R}^d$ be open and bounded. Then there exists a constant C, depending only on Ω

$$|u|_{L^2(\Omega)} \le C |\nabla u|_{L^2(\Omega)} \qquad u \in W_0^{1,2}(\Omega).$$

Theorem 2.6.7 (Plancherel Theorem). [79, Theorem 9.13] Let $f \in L^2$ and we denote its Fourier transform by \hat{f} . Then

- *i)* for every $f \in L^2$, $|f|_{L^2} = |\hat{f}|_{L^2}$,
- ii) the mapping $f \rightarrow \hat{f}$ is a Hilbert space isomorphism of L^2 onto L^2 .

2.6.1 Kuratowski-Zorn Lemma

This subsection is based on [60, Chapter 1].

Definition 2.6.8. A relation \leq on Ξ is called **partial order** if \leq is:

- i) reflexive i.e. $x \leq x$ for all $x \in \Xi$,
- ii) antisymmetric i.e. for all $x, y \in \Xi$ if $x \le y$ and $y \le x$, then x = y,
- iii) transitive i.e. for all $x, y, z \in \Xi$ if $x \leq y$ and $y \leq z$, implies $x \leq z$.

In this case (Ξ, \preceq) is called partially ordered set (or poset).

Definition 2.6.9. A chain in a poset (Ξ, \leq) is a subset $B \subseteq \Xi$ such that any two elements in B are comparable.

Definition 2.6.10. Let (Ξ, \preceq) be a poset and $B \subseteq \Xi$ then an element $u \in \Xi$ is called upper bound of *B*, if $x \preceq u$ for all $x \in B$.

Definition 2.6.11. Let (Ξ, \preceq) be a poset. An element $m \in \Xi$ is called maximal element of Ξ , if there is no element $x \in \Xi$ such that $m \preceq x$ and $m \neq x$.

Now we state the main result of this subsection.

Lemma 2.6.12 (Kuratowski-Zorn Lemma). *If every chain in a poset* (Ξ, \preceq) *has an upper bound in* Ξ , *then* Ξ *contains a maximal element.*

2.7 Stochastic processes and martingale

We fix a probability space $(\Omega, \mathscr{F}, \mathbb{P})$ for this section and assume that *H* is a separable Hilbert space with the norm $|\cdot|_H$. The content of this section is based on [28, 38].

Definition 2.7.1. An *H*-valued continuous-time stochastic process $\{\xi_t\}_{t \in \mathbb{T}}$ is a family of *H*-valued random variables indexed by time *t*. Moreover, either $\mathbb{T} := [0, T]$ or $\mathbb{T} := [0, \infty)$.

For each $\omega \in \Omega$, the map

$$\xi(\omega): \mathbb{T} \ni t \to \xi_t(\omega) \in H$$

is called a path (or trajectory) of the process ξ .

We will use the notation ξ instead of $\{\xi(t)\}_{t \in \mathbb{T}}$ for simplicity. And throughout this section we will assume that ξ is an *H*-valued stochastic process on \mathbb{T} unless specified otherwise.

Definition 2.7.2. A process ξ is called continuous if \mathbb{P} -a.s. the trajectories of ξ are continuous on \mathbb{T} , i.e. there exists $\overline{\Omega} \in \mathscr{F}$ with $\mathbb{P}(\overline{\Omega}) = 1$ such that for each $\omega \in \overline{\Omega}$, the mapping $\mathbb{T} \ni t \mapsto \xi(t, \omega) \in H$ is continuous.

Definition 2.7.3. An *H*-valued stochastic process ζ on \mathbb{T} is called a modification (or version) of the process ξ if

$$\mathbb{P}(\{w \in \Omega : \xi(t,w) \neq \zeta(t,w)\}) = 0 \text{ for every } t \in \mathbb{T}.$$

Definition 2.7.4. A process ξ is called measurable if the following mapping

$$\xi: [0,T] \times \Omega \to H$$

is $\mathscr{B}([0,T]) \otimes \mathscr{F}$ -measurable.

Definition 2.7.5. A process ξ is called stochastically continuous at $t_0 \in [0, T]$ if for every $\epsilon, \delta > 0$, there exists $\rho > 0$ such that for each $t \in [t_0 - \rho, t_0 + \rho] \cap [0, T]$,

$$\mathbb{P}(\{\omega \in \Omega : |\xi(t,\omega) - \xi(t_0,\omega)|_H \ge \epsilon\}) \le \delta.$$

If the process ξ is stochastically continuous for each $t_0 \in [0, T]$, then it is said to be stochastically continuous on [0, T].

Lemma 2.7.6. [38, Proposition 3.2] If a process ξ is stochastically continuous on [0, T], then it has a measurable modification on [0, T].

Definition 2.7.7. A family $\{\mathscr{F}_t\}_{t\geq 0}$ of σ -fields such that for all $t \ \mathscr{F}_t \subset \mathscr{F}$, is called a filtration if for any $0 \leq s \leq t < \infty$, $\mathscr{F}_s \subset \mathscr{F}_t$.

From now on, we will assume that $\mathbb{F} = \{\mathscr{F}_t\}_{t \ge 0}$ is a filtration.

Definition 2.7.8. A process ξ is said to be adapted to \mathbb{F} if for each $t \in [0, T]$, $\xi(t)$ is \mathscr{F}_t -measurable.

Definition 2.7.9. A process ξ is called progressively measurable if for each $t \in [0, T]$, the following mapping

$$\xi: [0,t] \times \Omega \ni (s,w) \mapsto \xi(s,w) \in H$$

is $\mathcal{B}([0,t]) \otimes \mathcal{F}_t$ -measurable.

Definition 2.7.10. A subset $P \subseteq [0,\infty) \times \Omega$ is said to be progressively measurable, if the process $\xi_s(\omega) := \mathbb{1}_P(s,\omega)$ is progressively measurable. The σ -field generated by all such subsets P of $[0,\infty) \times \Omega$ is called progressively measurable σ -field.

Remark 2.7.11. If the process ξ is progressively measurable, then it is adapted to \mathbb{F} .

Lemma 2.7.12. Limits of progressively measurable processes are progressively measurable.

Lemma 2.7.13. [38, Proposition 3.5] If a process ξ is stochastically continuous on [0,T] and adapted to \mathbb{F} , then it has a progressively measurable modification.

Definition 2.7.14. A random variable $\tau : \Omega \to [0, \infty]$ i.e. a random time, is a stopping time (w.r.t filtration \mathbb{F}) if for all $t \in \mathbb{T}$,

$$\{\tau \leq t\} := \{\omega \in \Omega : \tau(\omega) \leq t\} \in \mathscr{F}_t.$$

- **Remark 2.7.15.** i) One can easily see that if τ and σ are two stopping times then $\tau \land \sigma$, $\tau \lor \sigma$ and $\tau + \sigma$ are also stopping times.
 - ii) Every stopping time τ is \mathscr{F}_{τ} -measurable, where

$$\mathscr{F}_{\tau} = \{B \in \mathscr{F} : B \cap \{\tau \leq t\} \in \mathscr{F}_t, \text{ for all } t \in \mathbb{T}\}$$

is a σ -field.

iii) A r.v ζ is \mathscr{F}_{τ} -measurable if and only if for all $t \in \mathbb{T}$, $\zeta \mathbb{I}_{\{\tau \leq t\}}$ is \mathscr{F}_{t} -measurable.

Lemma 2.7.16. [74, Proposition 1.1.3] Let ξ be a progressively measurable process, and τ a stopping time. Then $\xi_{\tau} \mathbb{1}_{\{\tau \leq t\}}$ is \mathscr{F}_{τ} -measurable and the stopped process $\xi_{t \wedge \tau}$ is also progressively measurable.

Definition 2.7.17. An *H*-valued process ξ is called an \mathbb{F} -martingale if

- i) ξ is adapted to \mathbb{F} ,
- ii) for each $t \in [0, T]$, $\mathbb{E}(|\xi(t)|_H) < \infty$,
- iii) for each $t, s \in [0, T]$ with $t \ge s$,

$$\mathbb{E}(\xi(t)|\mathscr{F}_s) = \xi(s), \text{ i.e. } \forall A \in \mathscr{F}_s \quad \int_A \xi(t) d\mathbb{P} = \int_A \xi(s) d\mathbb{P}$$

Lemma 2.7.18. [38, Proposition 3.9] Let \mathcal{M}_T^2 be the space of all H-valued, continuous and square integrable martingales ξ on [0,T]. Then \mathcal{M}_T^2 is a Banach space with respect to the following norm

$$|\xi|_{\mathcal{M}^2_T} = \left(\mathbb{E}\sup_{t\in[0,T]} |\xi(t)|^2_H\right)^{\frac{1}{2}}, \quad \xi\in\mathcal{M}^2_T.$$

Theorem 2.7.19 (Burkholder-Davis-Gundy inequality). [74, Theorem 1.1.6] Let $1 \le p < \infty$, then for all H-valued continuous martingales M with $M_0 = 0$ and stopping time τ , there exist positive constants c_p and C_p such that

(2.7.1)
$$\mathbb{E}\left[\langle M(\tau)\rangle^{p/2}\right] \le \mathbb{E}\left(\sup_{0\le t\le \tau} |M_t|\right)^p \le C_p \mathbb{E}\left[\langle M(\tau)\rangle^{p/2}\right],$$

where $\langle M \rangle$ denotes the quadratic variation of M.

We will require following generalisation of the Gronwall Lemma [40, Lemma 3.9]:

Lemma 2.7.20 (Generalised Gronwall Lemma). Let X, Y, I and φ be non-negative processes and Z be a non-negative integrable random variable. Assume that I is non-decreasing and there exist non-negative constants $C, \alpha, \beta, \gamma, \eta$ with the following properties

(2.7.2)
$$\int_0^T \varphi(s) ds \le C a.s., \qquad 2\beta e^C \le 1, \qquad 2\eta e^C \le \alpha,$$

and such that for $0 \le t \le T$,

(2.7.3)
$$X(t) + \alpha Y(t) \le Z + \int_0^t \varphi(r) X(r) dr + I(t), \quad a.s.,$$

(2.7.4)
$$\mathbb{E}(I(t)) \leq \beta \mathbb{E}(X(t)) + \gamma \int_0^t \mathbb{E}(X(s)) \, ds + \eta \mathbb{E}(Y(t)) + \tilde{C},$$

where $\tilde{C} > 0$ is a constant. If $X \in L^{\infty}([0,T] \times \Omega)$, then we have

(2.7.5)
$$\mathbb{E}[X(t) + \alpha Y(t)] \le 2 \exp\left(C + 2t\gamma e^{C}\right) \left(\mathbb{E}(Z) + \tilde{C}\right), \qquad t \in [0, T].$$

2.8 Wiener process and the martingale representation theorem

We assume that *H* is a separable Hilbert space with the norm $|\cdot|_H$ and $(\Omega, \mathscr{F}, \mathbb{P})$ is a probability space. The definitions and results are taken from [28, 38].

Definition 2.8.1. A probability measure μ on $(H, \mathscr{B}(H))$ is called Gaussian if for arbitrary $h \in H$ there exist $m \in \mathbb{R}$, $\sigma \ge 0$, such that

$$\mu\{x \in H : \langle h, x \rangle_H \in A\} = \mathcal{N}(m, \sigma)(A), \quad A \in \mathscr{B}(\mathbb{R}).$$

Definition 2.8.2. An *H*-valued stochastic process *X* on $[0,\infty)$ is said to be Gaussian if, for any $n \in N$ and for arbitrary positive numbers, t_1, t_2, \dots, t_n , the H^n -valued random variable $(X(t_1), X(t_2), \dots, X(t_n))$ is Gaussian.

Definition 2.8.3. A real valued Wiener process (or Brownian motion) is a stochastic process W(t) with values in \mathbb{R} defined for $t \in [0, \infty)$ such that

- i) W(0) = 0 a.s.,
- ii) the sample paths $t \mapsto W(t)$ are a.s. continuous,
- iii) for $0 \le s \le t < \infty$, W(t) W(s) is normally distributed with mean 0 and variance t s,
- iv) for any $0 = t_0 \le t_1 \le \cdots \le t_n < \infty$, the increments

$$W(t_1) - W(t_0), \cdots, W(t_n) - W(t_{n-1})$$

are independent.

Definition 2.8.4. We call $W(t) = (W^1(t), W^2(t), \dots, W^d(t))$, a *d*-dimensional Wiener process if $W^1(t), \dots, W^d(t)$ are independent \mathbb{R} -valued Wiener processes.

Let *U* be a Hilbert space (can be finite dimensional too) with norm $|\cdot|_U$ and $Q \in \mathcal{L}(U)$ be a symmetric non-negative operator. We also assume that $\operatorname{Tr} Q < \infty$. Then there exists a complete orthonormal basis $\{e_k\}$ in *U*, and a bounded sequence of non-negative real numbers λ_k such that

$$Qe_k = \lambda_k e_k, \quad k = 1, 2, \cdots$$

Without loss of generality, we may assume that Q is injective.

Definition 2.8.5. A *U*-valued stochastic process W(t), $t \ge 0$ is called a *Q*-Wiener process if

- i) W(0) = 0 a.s.,
- ii) W has a.s. continuous trajectories,
- iii) W has independent increments,
iv) $\mathscr{L}(W(t) - W(s)) = \mathscr{N}(0, (t-s)Q), \ 0 \le s \le t.$

If a process $W(t), t \in [0,T]$ satisfies (i) - (iii) and (iv) for $t, s \in [0,T]$ then we say that W is a Q-Wiener process on [0,T].

Lemma 2.8.6. [38, Proposition 4.1] Assume that W is a Q-Wiener process on U, with $\text{Tr}Q < \infty$. Then the following statements hold:

i) W is a Gaussian process on U and

(2.8.1)
$$\mathbb{E}(W(t)) = 0, \qquad \operatorname{Cov}(W(t)) = tQ, \quad t \ge 0.$$

ii) For arbitrary t,

(2.8.2)
$$W(t) = \sum_{j=1}^{\infty} \sqrt{\lambda_j} \beta_j(t) e_j \quad a.e$$

where

$$\beta_j(t) = \frac{1}{\lambda_j} \langle W(t), e_j \rangle_U, \quad j = 1, 2, \cdots,$$

are real valued Brownian motions mutually independent on $(\Omega, \mathscr{F}, \mathbb{P})$ and the series (2.8.2) is convergent in $L^2(\Omega, \mathscr{F}, \mathbb{P})$.

Theorem 2.8.7 (Martingale Representation Theorem). [38, Theorem 8.2] Assume that $M \in \mathcal{M}^2_T(H)$ and

$$\langle\langle M \rangle\rangle_t = \int_0^t \left(\varphi(s)Q^{1/2}\right) \left(\varphi(s)Q^{1/2}\right)^* ds, \qquad t \in [0,T],$$

where φ is a predictable $\mathcal{T}_2(U_0, H)$ process; $U_0 = Q^{1/2}U$ is a Hilbert space endowed with the inner product

$$\langle u, \mathbf{v} \rangle_{U_0} = \langle Q^{-1/2} u, Q^{-1/2} \mathbf{v} \rangle_U \quad u, \mathbf{v} \in U_0,$$

and Q a given bounded symmetric non-negative operator in U. Then there exists a probability space $(\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{P}})$, a filtration $\{\tilde{\mathscr{F}}_t\}$ and a Q-Wiener process W, with values in U, defined on $(\Omega \times \tilde{\Omega}, \mathscr{F} \times \tilde{\mathscr{F}}, \mathbb{P} \times \tilde{\mathbb{P}})$ adapted to $\{\mathscr{F}_t \times \tilde{\mathscr{F}}_t\}$, such that

(2.8.3)
$$M(t,\omega,\tilde{\omega}) = \int_0^t \varphi(s,\omega,\tilde{\omega}) dW(s,\omega,\tilde{\omega}), \quad t \in [0,T], (\omega,\tilde{\omega}) \in \Omega \times \tilde{\Omega},$$

where

(2.8.4)
$$M(t,\omega,\tilde{\omega}) = M(t,\omega), \quad and \quad \varphi(t,\omega,\tilde{\omega}) = \varphi(t,\omega), \quad (\omega,\tilde{\omega}) \in \Omega \times \tilde{\Omega}.$$

Next we state a simplified version of an existence theorem for the stochastic differential equation

(2.8.5)
$$X(t) = \xi + \int_0^t \sigma(X(s)) dW(s) + \int_0^t b(X(s)) ds,$$

where the maps

$$\sigma \colon \mathbb{R}^d \to \mathcal{T}_2(H, \mathbb{R}^d), \qquad b \colon \mathbb{R}^d \to \mathbb{R}^d,$$

where *H* is a real separable, possibly infinite dimensional, Hilbert space, are measurable. Suppose that $\mathcal{U} := (\Omega, \mathscr{F}, \mathbb{P}; \mathbb{F})$ is a filtered probability space and $W = (W(t))_{t \ge 0}$ be an *H*-cylindrical Wiener process on \mathcal{U} .

Theorem 2.8.8. [1, Theorem 3.1] Assume that the functions σ and b satisfy the following conditions

(i) For any R > 0 there exists a constant C > 0 such that

$$\|\sigma(x) - \sigma(y)\|_{\mathcal{T}_{2}(\mathbb{R}^{d};H)} + |b(x) - b(y)|_{\mathbb{R}^{d}} \le C|x - y|_{\mathbb{R}^{d}}^{2}, \quad |x|_{\mathbb{R}^{d}}, |y|_{\mathbb{R}^{d}} \le R$$

(ii) There exists a constant $K_1 > 0$ such that

$$\|\sigma(x)\|_{\mathscr{T}_2(\mathbb{R}^d:H)}^2 + 2\langle x, b(x) \rangle_{\mathbb{R}^d} \le K_1(1 + \|x\|_{\mathbb{R}^d}^2), \qquad x \in \mathbb{R}^d.$$

Then, for any \mathbb{R}^d -valued \mathscr{F}_0 -measurable random variable ξ , there exists a unique global solution $X = (X(t))_{t\geq 0}$ to (2.8.5).

2.9 Tightness and Skorohod Theorem

In this section we take E to be a separable Banach space with the norm $|\cdot|_E$ and let $\mathscr{B}(E)$ be its Borel σ -field. The family of probability measures on $(E, \mathscr{B}(E))$ will be denoted by Λ . The set of all bounded and continuous E-valued functions is denoted by $\mathscr{C}_b(E)$. The content of this section is based on [23, 24] and [38, Chapter 2].

Definition 2.9.1. The family Λ of probability measures on $(E, \mathscr{B}(E))$ is said to be tight if for arbitrary $\varepsilon > 0$ there exists a compact set $K_{\varepsilon} \subset E$ such that

$$\mu(K_{\varepsilon}) \ge 1 - \varepsilon$$
, for all $\mu \in \Lambda$.

Definition 2.9.2. A sequence of measures $\{\mu_n\}_{n \in \mathbb{N}}$ on $(E, \mathscr{B}(E))$ is said to be weakly convergent to a measure μ if for every $\varphi \in \mathscr{C}_b(E)$ we have

$$\lim_{n\to\infty}\int_E\varphi(x)\mu_n(dx)=\int_E\varphi(x)\,\mu(dx)\,.$$

Definition 2.9.3. The family Λ is said to be compact (respectively relatively compact), if an arbitrary sequence $\{\mu_n\}_{n\in\mathbb{N}}$ of elements from Λ contains a subsequence $\{\mu_{n_k}\}_{k\in\mathbb{N}}$ weakly convergent to a measure $\mu \in \Lambda$ (respectively to a measure μ on $(E, \mathscr{B}(E))$).

Theorem 2.9.4 (Prokhorov Theorem). [38, Theorem 2.3] The family Λ of probability measures on $(E, \mathscr{B}(E))$ is relatively compact if and only if it is tight.

The following theorem, due to Skorohod, links the concept of weak convergence of probability measures with that of almost sure convergence of random variables.

Theorem 2.9.5 (Skorohod Theorem). [38, Theorem 2.4] For an arbitrary sequence of probability measures $\{\mu_n\}_{n\in\mathbb{N}}$ on $\mathscr{B}(E)$ weakly convergent to a probability measure μ , there exists a probability space $(\Omega, \mathscr{F}, \mathbb{P})$ and random variables X, X_1, \dots , such that $\mathscr{L}(X_m) = \mu_m$, $\mathscr{L}(X) = \mu$ and

$$\lim_{n\to\infty} X_n = X, \quad \mathbb{P}-a.s.$$

We will need the following Jakubowski's generalisation of the Skorohod Theorem, in the form given by Brzeźniak and Ondreját [26, Theorem C.1], see also [49], as we deal with non-metric spaces.

Theorem 2.9.6. Let \mathscr{X} be a topological space such that there exists a sequence $\{f_m\}_{m \in \mathbb{N}}$ of continuous functions $f_m : \mathscr{X} \to \mathbb{R}$ that separates points of \mathscr{X} . Let us denote by \mathscr{S} the σ -algebra generated by the maps $\{f_m\}$. Then

- a) every compact subset of \mathscr{X} is metrizable,
- b) if $(\mu_m)_{m \in \mathbb{N}}$ is a tight sequence of probability measures on $(\mathcal{X}, \mathscr{S})$, then there exists a subsequence $(m_k)_{k \in \mathbb{N}}$, a probability space $(\Omega, \mathscr{F}, \mathbb{P})$ with \mathcal{X} -valued Borel measurable variables ξ_k, ξ such that μ_{m_k} is the law of ξ_k and ξ_k converges to ξ almost surely on Ω .

Let (\mathbb{S}, ρ) be a separable and complete metric space.

Definition 2.9.7. Let $u \in \mathcal{C}([0,T]; \mathbb{S})$. The modulus of continuity of u on [0,T] is defined by

$$m(u,\delta) := \sup_{s,t\in[0,T], |t-s|\leq\delta} \rho(u(t),u(s)), \quad \delta > 0.$$

Let $(\Omega, \mathscr{F}, \mathbb{P})$ be a probability space with filtration $\mathbb{F} := (\mathscr{F}_t)_{t \in [0,T]}$ satisfying the usual conditions, see [63], and let $(X_n)_{n \in \mathbb{N}}$ be a sequence of continuous \mathbb{F} -adapted \mathbb{S} -valued processes.

Definition 2.9.8. We say that the sequence $(X_n)_{n \in \mathbb{N}}$ of S-valued random variables satisfies condition **[T]** iff $\forall \varepsilon > 0, \forall \eta > 0, \exists \delta > 0$:

(2.9.1)
$$\sup_{n\in\mathbb{N}}\mathbb{P}\left\{m(X_n,\delta)>\eta\right\}\leq\varepsilon.$$

Lemma 2.9.9. [24, Lemma 2.4] Assume that $(X_n)_{n \in \mathbb{N}}$ satisfies condition [**T**]. Let \mathbb{P}_n be the law of X_n on $\mathscr{C}([0,T];\mathbb{S})$, $n \in \mathbb{N}$. Then for every $\varepsilon > 0$ there exists a subset $A_{\varepsilon} \subset \mathscr{C}([0,T];\mathbb{S})$ such that

$$\sup_{n\in\mathbb{N}}\mathbb{P}_n(A_{\varepsilon})\geq 1-\varepsilon$$

and

(2.9.2)
$$\lim_{\delta \to 0} \sup_{u \in A_c} m(u, \delta) = 0.$$

Now we recall the Aldous condition [**A**], which is connected with condition [**T**]. This condition allows to investigate the modulus of continuity for the sequence of stochastic processes by means of stopped processes.

Definition 2.9.10. [Aldous condition] A sequence $(X_n)_{n \in \mathbb{N}}$ satisfies condition [A] iff $\forall \varepsilon > 0$, $\forall \eta > 0$, $\exists \delta > 0$ such that for every sequence $(\tau_n)_{n \in \mathbb{N}}$ of \mathbb{F} -stopping times with $\tau_n \leq T$ one has

 $\sup_{n \in \mathbb{N}} \sup_{0 \le \theta \le \delta} \mathbb{P}\left\{ \varrho(X_n(\tau_n + \theta), X_n(\tau_n)) \ge \eta \right\} \le \varepsilon.$

Lemma 2.9.11. [64, Theorem 3.2] Conditions [A] and [T] are equivalent.

C H A P T E R

CONSTRAINED NAVIER-STOKES EQUATIONS

Incompressible Navier-Stokes equations describe the dynamics of an incompressible viscous fluid. These equations were proposed by C. Navier in 1822 on the basis of a suitable molecular model and were later derived by G. Stokes by means of the theory of continua. A solution to these equations predicts the behaviour of the fluid, in particular, describes the evolution of velocity of the fluid as a function of space and time, given the initial and boundary states. Even though Navier-Stokes equations have variety of applications ranging from aerodynamics to biology, such as modelling the flow of blood in the circulatory system; the basic mathematical question of the existence of a unique global-in-time solution to these parabolic PDEs on a bounded domain in \mathbb{R}^3 still remains open. The non-linear convective term poses a lot of problems during the analysis as well as in physical systems by causing physical phenomenons, such as eddy flows and turbulence.

The existence of a unique global-in-time solution to the Navier-Stokes equations on \mathbb{R}^2 has been known for a long time. In her seminal paper [54], Ladyzhenskaya proved an inequality to control the non-linear convective term on a bounded domain in \mathbb{R}^2 , which was later used to prove the global-in-time existence of a unique solution to the Navier-Stokes equations. This trick already fails in the case of a bounded domain in \mathbb{R}^3 . One can prove the existence of a global-in-time weak solutions [47, 55], also known as Leray solutions on a general bounded domain in \mathbb{R}^3 .

In this chapter we introduce the constrained Navier-Stokes equations, which are Navier-Stokes equations with a constraint on the L^2 -energy of the solution. We assume that the L^2 -energy of the solution remains constant and is assumed to be equal to 1. The motivation to study such a constrained problem, is that these equations should be a better approximation to incompressible Euler equations, since for the Euler equations, the energy of (sufficiently smooth) solutions is constant (see [31]).

We end the introduction by giving a brief overview of the chapter: in Section 3.1 we define our functional spaces along with the Stokes operator, for both \mathbb{R}^2 and bounded domain with periodic boundary conditions (i.e. a torus). We introduce the bilinear map corresponding to the non-linear convective term along with some of its important properties in Section 3.2. We conclude the chapter by introducing the constraint and corresponding orthogonal projection map which projects the Hilbert manifold \mathcal{M} onto its tangent space, that along with all the functional setting is used to describe the constrained Navier-Stokes equations (CNSE) in Section 3.3.

3.1 Functional setting

Let \mathcal{O} be either a bounded domain in \mathbb{R}^2 , the full Euclidean space \mathbb{R}^2 or the torus \mathbb{T}^2 . For $p \in [1,\infty]$ and $k \in \mathbb{N}$, the Lebesgue and Sobolev spaces of \mathbb{R}^2 -valued functions will be denoted by $L^p(\mathcal{O}, \mathbb{R}^2)$ and $W^{k,p}(\mathcal{O}, \mathbb{R}^2)$ respectively, and often L^p and $W^{k,p}$ whenever the context is understood. The usual scalar product on L^2 is denoted by $\langle u, v \rangle$ for $u, v \in L^2$. The associated norm is given by $|u|_{L^2}, u \in L^2$. We also write $W^{k,2}(\mathcal{O}, \mathbb{R}^2) := H^k(\mathcal{O})$ and will denote it's norm by $\|\cdot\|_{H^k}$. In particular the scalar product for $H^1(\mathcal{O})$ is given by

$$\langle u, v \rangle_{H^1} = \langle u, v \rangle + \langle \nabla u, \nabla v \rangle, \qquad u, v \in H^1(\mathcal{O}),$$

and thus the norm is

$$\|u\|_{H^1} = \left[|u|_{L^2}^2 + |\nabla u|_{L^2}^2\right]^{1/2}$$

In the following two subsections we will introduce some additional spaces. The structure of the spaces will depend on the choice of \mathcal{O} .

3.1.1 Functional setting for \mathbb{R}^2

We consider the whole space \mathbb{R}^2 . We introduce the following spaces:

(3.1.1)
$$\mathbf{H} = \left\{ u \in L^2(\mathbb{R}^2, \mathbb{R}^2) : \operatorname{div} u = 0 \right\},$$
$$\mathbf{V} = H^1 \cap \mathbf{H}.$$

We endow H with the scalar product and norm of L^2 and denote it by $\langle u, v \rangle_H$, $|u|_H$ respectively for $u, v \in H$. We equip the space V with the scalar product and norm of H^1 and will denote it by $\langle \cdot, \cdot \rangle_V$ and $\|\cdot\|_V$ respectively.

Let $\Pi: L^2 \to H$ be the Leray-Helmholtz projection operator [88] which projects vector fields on to the plane of divergence free vector fields. We denote by $A: D(A) \to H$, the Stokes operator which is defined by

$$D(A) = H \cap H^{2}(\mathbb{R}^{2}),$$
$$Au = -\Pi \Delta u, \quad u \in D(A).$$

It is well known that A is a self adjoint non-negative operator in H [33]. Note that Δ and Π commute with each other. Moreover

$$D((A+I)^{1/2}) = V$$
 and $\langle Au, u \rangle = |\nabla u|_{L^2}^2, u \in D(A).$

From now onwards we will denote E := D(A).

3.1.2 Functional setting for a periodic domain

We denote the bounded domain with periodic boundary conditions by \mathbb{T}^2 which can be identified to a two dimensional torus. Let $\mathscr{C}^{\infty}_{c}(\mathbb{T}^2, \mathbb{R}^2)$ denote the space of all \mathbb{R}^2 -valued functions of class \mathscr{C}^{∞} with compact supports contained in \mathbb{T}^2 . We introduce the following spaces:

$$\mathcal{V} = \left\{ u \in \mathscr{C}^{\infty}_{c}(\mathscr{O}, \mathbb{R}^{2}) : \operatorname{div} u = 0 \right\},$$

$$L^{2}_{0} = \left\{ u \in L^{2}(\mathbb{T}^{2}, \mathbb{R}^{2}) : \int_{\mathbb{T}^{2}} u(x) \, dx = 0 \right\},$$

$$H = \left\{ u \in \mathbb{L}^{2}_{0} : \operatorname{div} u = 0 \right\},$$

$$V = H^{1} \cap H.$$

We endow H with the scalar product and norm of L^2 and denote it by $\langle u, v \rangle_H$, $|u|_H$ respectively for $u, v \in H$. We equip the space V with the scalar product $\langle \nabla u, \nabla v \rangle_{L^2}$ and norm $||u||_V, u, v \in V$. One can show that in the case of \mathbb{T}^2 , V-norm $||\cdot||_V$, and H^1 -norm $||\cdot||_{H^1}$ are equivalent on V.

As before we denote by $A: D(A) \rightarrow H$, the Stokes operator which is defined by

$$D(A) = H \cap H^{2}(\mathbb{T}^{2}),$$
$$Au = -\Pi \Delta u, \quad u \in D(A).$$

It is well known that A is a self adjoint positive operator in H [90]. Moreover

$$D(A^{1/2}) = V$$
 and $\langle Au, u \rangle = ||u||_V^2 = |\nabla u|_{L^2}^2, \quad u \in D(A)$

In the following section we will introduce a tri-linear form corresponding to the non-linear convective term from Navier-Stokes equations, which is well defined for any general domain \mathcal{O} and will state some of its properties.

3.2 Convective term

From now onwards we denote our domain by \mathcal{O} which can be either \mathbb{R}^2 or \mathbb{T}^2 . We introduce a continuous tri-linear form $b: L^p \times W^{1,q} \times L^r \to \mathbb{R}$,

$$b(u,\mathbf{v},w) = \sum_{i,j=1}^{2} \int_{\mathcal{O}} u^{i} \frac{\partial \mathbf{v}^{j}}{\partial x^{i}} w^{j} dx,$$

where $p, q, r \in [1, \infty]$ satisfies

$$\frac{1}{p} + \frac{1}{q} + \frac{1}{r} \le 1.$$

By the Sobolev Embedding Theorem [91] and the Hölder inequality [91], we obtain the following estimates

(3.2.1)
$$|b(u,v,w)| \le |u|_{L^4} ||v||_{V} |w|_{L^4}, \qquad u,w \in L^4, v \in V, \\ \le c ||u||_{V} ||v||_{V} ||w||_{V}, \qquad u,v,w \in V.$$

Hence, we can define a bilinear map $B: V \times V \rightarrow V'$ such that

$$\langle B(u,v),\varphi\rangle = b(u,v,\varphi), \text{ for } u,v,\varphi \in V,$$

where $\langle \cdot, \cdot \rangle$ denotes the duality between *V* and *V*'.

Using the following well-known Ladyzhenskaya's inequality on $\mathcal{O} \subset \mathbb{R}^2$ [88] (this is a special case of Gagliardo - Nirenberg inequality):

$$\|u\|_{L^4} \le 2^{1/4} \|u\|_{L^2}^{1/2} |\nabla u|_{L^2}^{1/2}, \qquad u \in \mathbf{V},$$

and the Hölder inequality, we obtain

$$(3.2.3) |b(u,v,\varphi)| \le \sqrt{2} |u|_{\mathrm{H}}^{\frac{1}{2}} ||u|_{\mathrm{V}}^{\frac{1}{2}} ||v|_{\mathrm{V}}^{\frac{1}{2}} |v|_{\mathrm{D}(\mathrm{A})}^{\frac{1}{2}} |\varphi|_{\mathrm{H}}, u \in \mathrm{V}, v \in \mathrm{D}(\mathrm{A}), \varphi \in \mathrm{H}.$$

Thus b can be uniquely extended to the tri-linear form (denoted by the same letter)

$$b: \mathbf{V} \times \mathbf{D}(\mathbf{A}) \times \mathbf{H} \to \mathbb{R}.$$

We can now also extend the operator B uniquely to a bounded bilinear operator

$$(3.2.4) B: V \times D(A) \to H.$$

The following properties of the tri-linear map b and the bilinear map B are very well established in [88] and Appendix A:

$$b(u, u, u) = 0, \qquad u \in \mathbf{V},$$

$$(3.2.5) \qquad b(u, w, w) = 0, \qquad u \in \mathbf{V}, w \in H^1,$$

$$\langle B(u, u), \mathbf{A}u \rangle_{\mathbf{H}} = 0, \qquad u \in \mathbf{D}(\mathbf{A}).$$

Note that the last identity in (3.2.5) holds only in the two cases that we have considered here, i.e. on the whole Euclidean space \mathbb{R}^2 and torus \mathbb{T}^2 .

3.3 NSEs and CNSE

The 2D Navier-Stokes equations (NSE) governing the dynamics of an incompressible viscous fluid are given as following:

(3.3.1)
$$\begin{cases} \frac{\partial u(x,t)}{\partial t} - v\Delta u(x,t) + (u(x,t)\cdot\nabla)u(x,t) + \nabla p(x,t) = 0, \\ \operatorname{div} u(x,t) = 0, \\ u(x,0) = u_0(x), \end{cases}$$

where $x \in \mathcal{O}$ and $t \in [0, T]$ for every T > 0; $u : \mathcal{O} \to \mathbb{R}^2$ and $p : \mathcal{O} \to \mathbb{R}$ are velocity and pressure of the fluid respectively. v is the viscosity of the fluid.

With all the notations as defined in the Sections 3.1 and 3.2, the Navier-Stokes equations (3.3.1) projected on divergence free vector field using the Leray-Helmholtz projection operator is given by

(3.3.2)
$$\begin{cases} \frac{du}{dt} + vAu + B(u, u) = 0, \\ u(0) = u_0. \end{cases}$$

Let us denote the set of divergence free \mathbb{R}^2 -valued functions with unit L^2 norm, as following

$$\mathcal{M} = \{ u \in \mathbf{H} : |u|_{\mathbf{H}} = 1 \} .$$

Then the tangent space at u is defined as the following closed subspace of H,

$$T_u \mathcal{M} = \{ \mathbf{v} \in \mathbf{H} : \langle \mathbf{v}, u \rangle_{\mathbf{H}} = 0 \}, \qquad u \in \mathcal{M}.$$

A linear map $\pi_u : \mathbf{H} \to T_u \mathcal{M}$ defined by

$$\pi_u(\mathbf{v}) = \mathbf{v} - \langle \mathbf{v}, u \rangle_{\mathrm{H}} u \,,$$

is the orthogonal projection from H onto $T_u \mathcal{M}$.

Remark 3.3.1. It follows from (3.2.5) that

$$B(u,u) \in T_u \mathcal{M}, \quad u \in \mathcal{M} \cap D(A).$$

In particular,

$$\pi_u(B(u,u)) = B(u,u), \quad u \in \mathcal{M} \cap \mathcal{D}(\mathcal{A}).$$

Let

$$F(u) = vAu + B(u, u)$$

and $\hat{F}(u)$ be the projection of F(u) on the tangent space $T_u \mathcal{M}$, then for $u \in D(A)$,

$$\begin{split} \hat{F}(u) &= \pi_u(F(u)) = F(u) - \langle F(u), u \rangle_{\mathrm{H}} u \\ &= v \mathrm{A} u + B(u, u) - \langle v \mathrm{A} u + B(u, u), u \rangle_{\mathrm{H}} u \\ &= v \mathrm{A} u - v \langle \mathrm{A} u, u \rangle_{\mathrm{H}} u + B(u, u) - \langle B(u, u), u \rangle_{\mathrm{H}} u \\ &= v \mathrm{A} u - v |\nabla u|_{L^2}^2 u + B(u, u). \end{split}$$

The last equality follows heuristically from Remark 3.3.1.

Thus by projecting NSEs (3.3.2) on the tangent space $T_u \mathcal{M}$, we obtain our constrained Navier-Stokes equations (CNSE) which is given by

(3.3.3)
$$\begin{cases} \frac{du}{dt} + vAu - v|\nabla u|_{L^2}^2 u + B(u, u) = 0, \\ u(0) = u_0. \end{cases}$$

The majority of this thesis is dedicated to the study of the constrained Navier-Stokes equations (3.3.3) under the impact of external forcing, both deterministic and stochastic. Even though in Chapter 4 the analysis has been carried out in the absence of any deterministic external forcing (i.e. assuming external force is identically zero) one can easily generalise the results obtained there for non-zero deterministic external forcing, under suitable assumptions. We also show that the solution of CNSE (3.3.3) converge to the unique solution (Bardos solution, see [5]) of the Euler equations (formally obtained by putting v = 0 in (3.3.1)) in inviscid limit ($v \searrow 0$) with appropriate assumptions on the initial data.

In Chapter 5 we shift our focus to the stochastic generalisation of (3.3.3) where we assume that the stochastic forcing is tangent to the manifold \mathcal{M} , enabling the solution to stay on the manifold. The analysis is carried out using classical tools from the theory of partial differential equations, like Faedo-Galerkin approximations and compactness.



DETERMINISTIC CONSTRAINED NAVIER-STOKES EQUATIONS ON A 2D TORUS

e study 2D Navier-Stokes equations with a constraint on L^2 energy of the solution. We prove the existence and uniqueness of a global solution for the constrained Navier-Stokes equations on \mathbb{R}^2 and \mathbb{T}^2 , by a fixed point argument. We also show that the solution of constrained Navier-Stokes equations converges to the solution of Euler equations as viscosity v vanishes.

4.1 Introduction

The motivation behind this chapter is threefold. Firstly Caglioti *et.al.* in [31] studied the wellposedness and asymptotic behaviour of two dimensional Navier-Stokes equations in the vorticity form with two constraints: constant energy $E(\omega)$ and moment of inertia $I(\omega)$

$$\frac{\partial \omega}{\partial t} + u \cdot \nabla \omega = v \Delta \omega - v \operatorname{div} \left[\omega \nabla \left(b \psi + a \frac{|x|^2}{2} \right) \right],$$

which can be rewritten as

(4.1.1)
$$\frac{\partial \omega}{\partial t} + u \cdot \nabla \omega = v \operatorname{div} \left[\omega \nabla \left(\log \omega - b \psi - a \frac{|x|^2}{2} \right) \right],$$

where $\omega = \text{Curl}(u)$, $a = a(\omega)$ and $b = b(\omega)$ are the Lagrange multipliers associated to those constraints and

$$E(\omega) = \int_{\mathbb{R}^2} \psi \omega \, dx \,, \qquad I(\omega) = \int_{\mathbb{R}^2} |x|^2 \omega \, dx \,, \quad \psi = -\Delta^{-1} \omega \,.$$

They were able to show the existence of a unique classical global-in-time solution to (4.1.1) for a family of initial data [31, Theorem 5]. They also showed that the solution to (4.1.1) converges, as

time tends to $+\infty$, to the unique solution of an associated microcanonical variational problem [31, Theorem 8].

Secondly, Rybka [81] and Caffarelli & Lin [30] study the linear heat equation with constraints. Rybka studied heat flow on a manifold \mathcal{M} given by

$$\mathcal{M} = \left\{ u \in L^{2}(\Omega) \cap \mathscr{C}(\Omega) : \int_{\Omega} u^{k}(x) \, dx = C_{k}, \, k = 1, \dots, N \right\}$$

where Ω denotes a connected bounded region in \mathbb{R}^2 with smooth boundary. He proved [81, Theorem 2.5] the existence of the unique global solution for the projected heat equation

(4.1.2)
$$\begin{cases} \frac{du}{dt} = \Delta u - \sum_{k=1}^{N} \lambda_k u^{k-1} & \text{in } \Omega \subset \mathbb{R}^2 \\ \frac{\partial u}{\partial n} = 0 \text{ on } \partial \Omega, \qquad u(0,x) = u_0, \end{cases}$$

where $\lambda_k = \lambda_k(u)$ are such that u_t is orthogonal to Span $\{u^{k-1}\}$, for a more regular initial data. He also showed that the solutions to (4.1.2) converge to a steady state as time tends to $+\infty$. On the other hand Caffarelli and Lin initially establish the existence and uniqueness of a global, energy-conserving solution to the heat equation [30, Theorem 1.1]. They were then able to extend these results to more general family of singularly perturbed systems of non-local parabolic equations [30, Theorem 3.1]. Their main result was to prove the strong convergence of the solutions of these perturbed systems to some weak-solutions of the limiting constrained non-local heat flows of maps into a singular space.

Finally, these equations should be a better approximation of the Euler equations (for small viscosity), since for the Euler equations, the energy of (sufficiently smooth) solutions is constant (see [31]).

In this chapter we consider a problem which links the aforementioned works. We consider Navier-Stokes equations as in [31], but subject to the same energy constraint as in [30, 81]. Contrary to [31] we prove global-in-time existence of the solution but only on a torus, namely in the periodic case. Surprisingly our proof of global existence does not hold for a general bounded domain, although the local existence holds. We also prove our result of global existence of the solution for \mathbb{R}^2 . We additionally show that, in vanishing viscosity limit, the solution of the constrained equation (4.1.3) below, converges to the Bardos solution (see [5]) of the Euler equation (formally obtained setting v = 0).

We are interested in the Cauchy problem

(4.1.3)
$$\begin{cases} \frac{du}{dt} = -vAu + v|\nabla u|_{L^2}^2 u - B(u, u), \\ u(0) = u_0, \end{cases}$$

where $u \in H$, and H is a space of divergence free, mean zero vector fields on a torus, see (3.1.2) for the precise definition.

The above problem has a local *maximal* solution for each $u_0 \in V \cap \mathcal{M}$, where V is defined in (3.1.2) and

$$\mathcal{M} = \{ u \in \mathbf{H} : |u|_{\mathbf{H}} = 1 \}$$

Moreover $u(t) \in \mathcal{M}$ for all times *t*. This result is true, both for constrained Navier-Stokes equations on a bounded domain or with periodic boundary conditions (i.e. on a torus). In a more geometrical fashion, equation (4.1.3) can be also written as

$$\frac{du}{dt} = -\nabla_{\mathcal{M}} \mathcal{E}(u) - B(u, u),$$

where $\mathscr{E}(u) = \frac{1}{2} |\nabla u|_{L^2}^2$, $u \in \mathcal{M}$ and $\nabla_{\mathcal{M}} \mathscr{E}(u)$ is the gradient of \mathscr{E} with respect to H-norm projected onto $T_u \mathscr{M}$. The remarkable feature of this is that on a torus $\nabla_{\mathscr{M}} \mathscr{E}(u)$ and B(u, u) are orthogonal in H. This orthogonality holds for the Navier-Stokes without constraint too, i.e. on a torus $\nabla \mathscr{E}(u)$ is orthogonal to B(u, u) in H. The fact that this constraint preserves the orthogonality somehow makes it a natural constraint.

Hence, at least in a heuristic way

$$\begin{split} \frac{d}{dt} \mathscr{E}(u(t)) &= \left\langle \nabla_{\mathscr{M}} \mathscr{E}(u(t)), \frac{du}{dt} \right\rangle_{\mathrm{H}} \\ &= \left\langle \nabla_{\mathscr{M}} \mathscr{E}(u(t)), -\nabla_{\mathscr{M}} \mathscr{E}(u(t)) - B(u, u) \right\rangle_{\mathrm{H}} \\ &= - \left| \nabla_{\mathscr{M}} \mathscr{E}(u(t)) \right|_{\mathrm{H}}^{2}, \end{split}$$

so that $\mathscr{E}(u(t))$ is decreasing and thus the $H^{1,2}$ norm of the solution remains bounded.

Next we state the two main results of this chapter on a torus.

Let us denote

$$X_T = \mathscr{C}([0,T]; \mathbf{V}) \cap L^2(0,T; \mathbf{E}).$$

Theorem 4.1.1. Let v > 0 be fixed. Let $u_0 \in V \cap \mathcal{M}$. Then there exists a global and locally unique solution u of (4.1.3) such that $u \in X_T$ for each T > 0.

The space X_T with more details and the precise definition of the solution of (4.1.3) will be given in Section 4.2. Theorem 4.1.1 will be proved in steps in Sections 4.2 and 4.3.

Theorem 4.1.2. Let $u_0, u_0^v \in V \cap \mathcal{M}$ and u^v be the solution of (4.1.3) (existence and uniqueness of u^v follows from Theorem 4.1.1). Assume that $u_0^v \to u_0$ in V as $v \searrow 0$, and that $\operatorname{Curl}(u_0^v)$, $\operatorname{Curl}(u_0)$ stays uniformly bounded in $L^{\infty}(\mathbb{T}^2)$. Then for each T > 0, u^v converges in $\mathscr{C}([0,T]; L^2(\mathbb{T}^2))$ to the unique solution u of the limiting equation (namely (4.1.3) with v = 0).

We end the introduction with a brief description of the content of the chapter. In Section 4.2, a precise definition of the solution to problem (4.1.3) is given, and local existence and uniqueness are proved, together with some basic properties of the solution. In Section 4.3, global existence is established. After proving Theorem 4.1.2 in Section 4.4, we study CNSE (4.1.3) in fractional Sobolev spaces and establish the existence of a unique solution for much more regular initial data in Section 4.5. We end the chapter by presenting a formal discussion regarding the lower bound on the regularity of the initial data so as to have the existence of a solution to problem (4.1.3).

4.2 Local solution : Existence and Uniqueness

In this section we will establish the existence of a local solution to

(4.2.1)
$$\begin{cases} \frac{du}{dt} + Au - |\nabla u|_{L^2}^2 u + B(u, u) = 0, \\ u(0) = u_0 \in \mathbf{V} \cap \mathcal{M}, \end{cases}$$

by using the Banach fixed point theorem. We obtain certain estimates for non-linear terms of (4.2.1) using results from Chapter 3. After obtaining these estimates we construct a globally Lipschitz map. Some ideas in the Subsection 4.2.1 are based on [29].

In what follows we put E := D(A) and V, H are spaces as defined in Section 3.1.

Lemma 4.2.1. Let $G_1: V \rightarrow H$ be defined by

$$G_1(u) = |\nabla u|_{L^2}^2 u, \quad u \in \mathbf{V}.$$

Then, there exists C > 0 such that for $u_1, u_2 \in V$,

$$(4.2.2) |G_1(u_1) - G_1(u_2)|_{\mathbf{H}} \le C ||u_1 - u_2||_{\mathbf{V}} [||u_1||_{\mathbf{V}} + ||u_2||_{\mathbf{V}}]^2.$$

Proof. Let us consider $u_1, u_2 \in V$, then

$$\begin{split} |G_{1}(u_{1}) - G_{1}(u_{2})|_{\mathrm{H}} &= \left| |\nabla u_{1}|_{L^{2}}^{2} u_{1} - |\nabla u_{2}|_{L^{2}}^{2} u_{2} \right|_{\mathrm{H}} \\ &= \left| |\nabla u_{1}|_{L^{2}}^{2} u_{1} - |\nabla u_{1}|_{L^{2}}^{2} u_{2} + |\nabla u_{1}|_{L^{2}}^{2} u_{2} - |\nabla u_{2}|_{L^{2}}^{2} u_{2} \right|_{\mathrm{H}} \\ &= \left| |\nabla u_{1}|_{L^{2}}^{2} (u_{1} - u_{2}) + (|\nabla u_{1}|_{L^{2}}^{2} - |\nabla u_{2}|_{L^{2}}^{2}) u_{2} \right|_{\mathrm{H}} \\ &\leq |\nabla u_{1}|_{L^{2}}^{2} |u_{1} - u_{2}|_{\mathrm{H}} + \left[|\nabla u_{1}|_{L^{2}} + |\nabla u_{2}|_{L^{2}} \right] \left[|\nabla u_{1}|_{L^{2}} - |\nabla u_{2}|_{L^{2}} \right] |u_{2}|_{\mathrm{H}} \\ &\leq |\nabla u_{1}|_{L^{2}}^{2} |u_{1} - u_{2}|_{\mathrm{H}} + \left[|\nabla u_{1}|_{L^{2}} + |\nabla u_{2}|_{L^{2}} \right] |\nabla (u_{1} - u_{2})|_{L^{2}} |u_{2}|_{\mathrm{H}} \\ &\leq C \left[|\nabla u_{1}|_{L^{2}}^{2} ||u_{1} - u_{2}|_{\mathrm{H}} + \left[|\nabla u_{1}|_{L^{2}} + |\nabla u_{2}|_{L^{2}} \right] |\nabla (u_{1} - u_{2})|_{L^{2}} ||u_{2}|_{\mathrm{H}} \\ &\leq C \left[|\nabla u_{1}|_{L^{2}}^{2} ||u_{1} - u_{2}|_{\mathrm{H}} + \left[|\nabla u_{1}|_{L^{2}} + |\nabla u_{2}|_{L^{2}} \right] |\nabla (u_{1} - u_{2})|_{L^{2}} ||u_{2}|_{\mathrm{H}} \right] \\ &\leq C \left[|u_{1} - u_{2}|_{\mathrm{V}} \left[|\nabla u_{1}|_{L^{2}}^{2} + |\nabla u_{2}|_{L^{2}} ||u_{2}|_{\mathrm{V}} + |\nabla u_{1}|_{L^{2}} ||u_{2}|_{\mathrm{V}} \right], \end{split}$$

where we have repeatedly used the fact that V is continuously embedded in H. Thus, we obtain (4.2.2).

Lemma 4.2.2. Let $G_2: E \to H$ be defined by

$$G_2(u) = B(u, u), \qquad u \in \mathbf{E}.$$

Then, there exists $\tilde{C} > 0$ such that for $u_1, u_2 \in E$,

$$(4.2.3) |G_2(u_1) - G_2(u_2)|_{\mathcal{H}} \le \tilde{C} \left[\|u_1\|_{\mathcal{V}}^{1/2} |u_1|_{\mathcal{E}}^{1/2} \|u_1 - u_2\|_{\mathcal{V}} + \|u_2\|_{\mathcal{V}} \|u_1 - u_2\|_{\mathcal{V}}^{1/2} |u_1 - u_2|_{\mathcal{E}}^{1/2} \right].$$

Proof. Let us take $u_1, u_2 \in E$, then

$$\begin{split} |G_{2}(u_{1}) - G_{2}(u_{2})|_{\mathrm{H}} &= |B(u_{1}, u_{1}) - B(u_{2}, u_{2})|_{\mathrm{H}} \\ &= |B(u_{1}, u_{1}) - B(u_{2}, u_{1}) + B(u_{2}, u_{1}) - B(u_{2}, u_{2})|_{\mathrm{H}} \\ &= |B(u_{1} - u_{2}, u_{1}) + B(u_{2}, u_{1} - u_{2})|_{\mathrm{H}} \\ &= |\Pi[(u_{1} - u_{2}) \cdot \nabla u_{1}] + \Pi[u_{2} \cdot \nabla (u_{1} - u_{2})]|_{\mathrm{H}} \\ &\leq |(u_{1} - u_{2}) \cdot \nabla u_{1}|_{\mathrm{H}} + |u_{2} \cdot \nabla (u_{1} - u_{2})|_{\mathrm{H}} \\ &\leq |u_{1} - u_{2}|_{L^{4}(\mathcal{O})} |\nabla u_{1}|_{L^{4}(\mathcal{O})} + |u_{2}|_{L^{4}(\mathcal{O})} |\nabla (u_{1} - u_{2})|_{L^{4}(\mathcal{O})} \,. \end{split}$$

Now using the Ladyzhenkaya's inequality (3.2.2) and the embedding of V in H, we obtain

$$\begin{split} |G_{2}(u_{1}) - G_{2}(u_{2})|_{\mathcal{H}} &\leq \sqrt{2} |u_{1} - u_{2}|_{\mathcal{H}}^{1/2} |\nabla(u_{1} - u_{2})|_{\mathcal{H}}^{1/2} |\nabla u_{1}|_{\mathcal{H}}^{1/2} |\nabla^{2}u_{1}|_{\mathcal{H}}^{1/2} \\ &+ \sqrt{2} |u_{2}|_{\mathcal{H}}^{1/2} |\nabla u_{2}|_{\mathcal{H}}^{1/2} |\nabla(u_{1} - u_{2})|_{\mathcal{H}}^{1/2} |\nabla^{2}(u_{1} - u_{2})|_{\mathcal{H}}^{1/2} \\ &\leq \sqrt{2} C \Big[\|u_{1} - u_{2}\|_{\mathcal{V}} \|u_{1}\|_{\mathcal{V}}^{1/2} |u_{1}|_{\mathcal{E}}^{1/2} \\ &+ \|u_{2}\|_{\mathcal{V}} \|u_{1} - u_{2}\|_{\mathcal{V}}^{1/2} |u_{1} - u_{2}|_{\mathcal{E}}^{1/2} \Big] \,. \end{split}$$

Thus, we obtain the inequality (4.2.3).

4.2.1 Construction of a globally Lipschitz map

Let $\theta:\mathbb{R}_+\to [0,1]$ be a C_0^∞ non-increasing function such that

$$\min_{x \in \mathbb{R}_+} \theta'(x) \ge -1, \ \theta(x) = 1 \text{ iff } x \in [0,1] \text{ and } \theta(x) = 0 \text{ iff } x \in [3,\infty)$$

and for $n \ge 1$ set $\theta_n(\cdot) = \theta(\frac{\cdot}{n})$. Observe that if $h : \mathbb{R}_+ \to \mathbb{R}_+$ is a non-decreasing function, then for every $x, y \in \mathbb{R}_+$,

$$\theta_n(x)h(x) \le h(3n), \ |\theta_n(x) - \theta_n(y)| \le 3n|x - y|.$$

Let us fix T > 0. We will first construct a solution on [0, T]. For that, set

$$X_T = \mathscr{C}([0,T]; \mathbf{V}) \cap L^2(0,T; \mathbf{E}),$$

with norm

$$|u|_{X_T}^2 = \sup_{t \in [0,T]} ||u(t)||_{\mathbf{V}}^2 + \int_0^T |u(t)|_{\mathbf{E}}^2 dt$$

Let us define $G \colon \mathbf{E} \to \mathbf{H}$ as

(4.2.4)
$$G(u) := G_1(u) - G_2(u) = |\nabla u|_{L^2}^2 u - B(u, u).$$

Lemma 4.2.3. Suppose $G: E \rightarrow H$ is a map defined in (4.2.4). Define a map

$$\Phi_{n,T}: X_T \to L^2(0,T;\mathbf{H})$$

by

(4.2.5)
$$\Phi_{n,T}(u)(t) = \theta_n(|u|_{X_t})G(u(t)), \quad t \in [0,T].$$

Then $\Phi_{n,T}$ is globally Lipschitz and moreover, for any $u_1, u_2 \in X_T$

(4.2.6)
$$|\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2)|_{L^2(0,T;\mathbf{H})} \le K(n,T)|u_1 - u_2|_{X_T} T^{\frac{1}{4}},$$

where

$$K(n,T) = 3n \left(27n^3 T^{1/4} + 9n^2 + 12n T^{1/4} + 2 \right),$$

depends on n and T only.

Proof. Assume that $u_1, u_2 \in X_T$. Set

$$\tau_i = \inf \left\{ t \in [0, T]; |u_i|_{X_t} \ge 3n \right\}, \qquad i = 1, 2.$$

Without loss of generality assume that $\tau_1 \leq \tau_2$. Consider

$$\begin{split} |\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2)|_{L^2(0,T;\mathrm{H})} &= \left[\int_0^T |\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2)|_{\mathrm{H}}^2 dt\right]^{\frac{1}{2}} \\ &= \left[\int_0^T \left|\theta_n(|u_1|_{X_t})G(u_1) - \theta_n(|u_2|_{X_t})G(u_2)\right|_{\mathrm{H}}^2 dt\right]^{\frac{1}{2}}, \end{split}$$

for $i = 1, 2 \theta_n(|u_i|_{X_t}) = 0$ for $t \ge \tau_2$, thus we have

$$\begin{split} |\Phi_{n,T}(u_{1}) - \Phi_{n,T}(u_{2})|_{L^{2}(0,T;\mathrm{H})} &= \left[\int_{0}^{\tau_{2}} \left| \theta_{n}(|u_{1}|_{X_{t}})G(u_{1}) - \theta_{n}(|u_{2}|_{X_{t}})G(u_{2}) \right|_{\mathrm{H}}^{2} dt \right]^{\frac{1}{2}} \\ &= \left[\int_{0}^{\tau_{2}} \left| \theta_{n}(|u_{1}|_{X_{t}})[G_{1}(u_{1}) - G_{2}(u_{1})] - \theta_{n}(|u_{2}|_{X_{t}})[G_{1}(u_{2}) - G_{2}(u_{2})] \right|_{\mathrm{H}}^{2} dt \right]^{\frac{1}{2}} \\ &= \left[\int_{0}^{\tau_{2}} \left| \theta_{n}(|u_{1}|_{X_{t}})G_{1}(u_{1}) - \theta_{n}(|u_{1}|_{X_{t}})G_{1}(u_{2}) + \theta_{n}(|u_{1}|_{X_{t}})G_{1}(u_{2}) - \theta_{n}(|u_{2}|_{X_{t}})G_{1}(u_{2}) \right. \\ &+ \left. \theta_{n}(|u_{1}|_{X_{t}})G_{2}(u_{2}) - \theta_{n}(|u_{1}|_{X_{t}})G_{2}(u_{1}) + \theta_{n}(|u_{2}|_{X_{t}})G_{2}(u_{2}) - \theta_{n}(|u_{1}|_{X_{t}})G_{2}(u_{2}) \right|_{\mathrm{H}}^{2} dt \right]^{\frac{1}{2}} \end{split}$$

Using the Minkowski inequality we get,

$$\begin{split} |\Phi_{n,T}(u_{1}) - \Phi_{n,T}(u_{2})|_{L^{2}(0,T;\mathrm{H})} \\ &\leq \left[\int_{0}^{\tau_{2}} \left|\theta_{n}(|u_{1}|_{X_{t}})[G_{1}(u_{1}) - G_{1}(u_{2})]\right|_{\mathrm{H}}^{2} dt\right]^{\frac{1}{2}} + \left[\int_{0}^{\tau_{2}} \left|\left[\theta_{n}(|u_{1}|_{X_{t}}) - \theta_{n}(|u_{2}|_{X_{t}})\right]G_{1}(u_{2})\right|_{\mathrm{H}}^{2} dt\right]^{\frac{1}{2}} \\ &+ \left[\int_{0}^{\tau_{2}} \left|\theta_{n}(|u_{1}|_{X_{t}})[G_{2}(u_{2}) - G_{2}(u_{1})]\right|_{\mathrm{H}}^{2} dt\right]^{\frac{1}{2}} + \left[\int_{0}^{\tau_{2}} \left|\left[\theta_{n}(|u_{2}|_{X_{t}}) - \theta_{n}(|u_{1}|_{X_{t}})\right]G_{2}(u_{2})\right|_{\mathrm{H}}^{2} dt\right]^{\frac{1}{2}} \end{split}$$

.

 \mathbf{Set}

$$A_{1} = \left[\int_{0}^{\tau_{2}} \left| \left[\theta_{n}(|u_{1}|_{X_{t}}) - \theta_{n}(|u_{2}|_{X_{t}})\right] G_{1}(u_{2}) \right|_{H}^{2} dt \right]^{\frac{1}{2}},$$

$$A_{2} = \left[\int_{0}^{\tau_{2}} \left|\theta_{n}(|u_{1}|_{X_{t}})[G_{1}(u_{1}) - G_{1}(u_{2})]\right|_{H}^{2} dt \right]^{\frac{1}{2}},$$

$$A_{3} = \left[\int_{0}^{\tau_{2}} \left|\left[\theta_{n}(|u_{2}|_{X_{t}}) - \theta_{n}(|u_{1}|_{X_{t}})\right] G_{2}(u_{2}) \right|_{H}^{2} dt \right]^{\frac{1}{2}},$$

$$A_{4} = \left[\int_{0}^{\tau_{2}} \left|\theta_{n}(|u_{1}|_{X_{t}})[G_{2}(u_{2}) - G_{2}(u_{1})]\right|_{H}^{2} dt \right]^{\frac{1}{2}},$$

and hence

(4.2.7)
$$|\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2)|_{L^2(0,T;\mathbf{H})} \le A_1 + A_2 + A_3 + A_4.$$

Since θ_n is a Lipschitz function with Lipschitz constant 3n, we obtain

$$A_{1}^{2} = \int_{0}^{\tau_{2}} \left| \left[\theta_{n}(|u_{1}|_{X_{t}}) - \theta_{n}(|u_{2}|_{X_{t}}) \right] G_{1}(u_{2}) \right|_{\mathrm{H}}^{2} dt \leq 9n^{2} \int_{0}^{\tau_{2}} \left| |u_{1}|_{X_{t}} - |u_{2}|_{X_{t}} \right|_{\mathrm{H}}^{2} |G_{1}(u_{2})|_{\mathrm{H}}^{2} dt.$$

Again, using the Minkowski inequality, we get

(4.2.8)
$$A_{1}^{2} \leq 9n^{2} \int_{0}^{\tau_{2}} |u_{1} - u_{2}|_{X_{t}}^{2} |G_{1}(u_{2})|_{H}^{2} dt$$
$$\leq 9n^{2} |u_{1} - u_{2}|_{X_{T}}^{2} \int_{0}^{\tau_{2}} |G_{1}(u_{2})|_{H}^{2} dt.$$

Now consider $\int_0^{ au_2} |G_1(u_2)|_{\mathrm{H}}^2 dt$. Using (4.2.2) we get

$$\int_0^{\tau_2} |G_1(u_2)|_{\mathrm{H}}^2 \, dt \le C \int_0^{\tau_2} \|u_2(t)\|_{\mathrm{V}}^6 \, dt \le C \left[\sup_{t \in [0,\tau_2]} \|u_2(t)\|_{\mathrm{V}}^2 \right]^3 \tau_2 \, .$$

Since

$$|u_2|_{X_{\tau_2}}^2 = \sup_{t \in [0,\tau_2]} ||u_2(t)||_{\mathbf{V}}^2 + \int_0^{\tau_2} |u_2(t)|_{\mathbf{E}}^2 dt,$$

thus

$$\sup_{t\in[0,\tau_2]} \|u_2(t)\|_{\mathbf{V}}^2 \le |u_2|_{X_{\tau_2}}^2,$$

and using

 $|u_2|_{X_{\tau_2}} \leq 3n,$

we get

$$\int_0^{\tau_2} |G_1(u_2)|_{\mathrm{H}}^2 dt \le C \left[\sup_{t \in [0, \tau_2]} \|u_2(t)\|_{\mathrm{V}}^2 \right]^3 \tau_2 \le C |u_2|_{X_{\tau_2}}^6 \tau_2 \le C (3n)^6 \tau_2 \,.$$

Hence, the inequality (4.2.8) takes the form

$$A_1^2 \le 9n^2 C |u_1 - u_2|_{X_T}^2 (3n)^6 \tau_2,$$

from where we deduce

(4.2.9)
$$A_1 \le (3n)^4 C |u_1 - u_2|_{X_T} \tau_2^{\frac{1}{2}}$$

Similarly, since $\theta_n(|u_1|_{X_t}) = 0$ for $t \ge \tau_1$ and $\tau_1 \le \tau_2$, we have

$$A_{2} = \left[\int_{0}^{\tau_{2}} \left|\theta_{n}(|u_{1}|_{X_{t}})[G_{1}(u_{1}) - G_{1}(u_{2})]\right|_{\mathrm{H}}^{2} dt\right]^{\frac{1}{2}} = \left[\int_{0}^{\tau_{1}} \left|\theta_{n}(|u_{1}|_{X_{t}})[G_{1}(u_{1}) - G_{1}(u_{2})]\right|_{\mathrm{H}}^{2} dt\right]^{\frac{1}{2}}.$$

Since $\theta_n(|u_1|_{X_t}) \le 1$ for $t \in [0, \tau_1)$ and using (4.2.2), we have

$$\begin{split} A_2^2 &\leq \int_0^{\tau_1} |G_1(u_1) - G_1(u_2)|_{\mathcal{H}}^2 \, dt \leq C \int_0^{\tau_1} \|u_1(t) - u_2(t)\|_{\mathcal{V}}^2 [\|u_1(t)\|_{\mathcal{V}} + \|u_2(t)\|_{\mathcal{V}}]^4 \, dt \\ &\leq C \sup_{t \in [0, \tau_1]} \|u_1(t) - u_2(t)\|_{\mathcal{V}}^2 \int_0^{\tau_1} [\|u_1(t)\|_{\mathcal{V}} + \|u_2(t)\|_{\mathcal{V}}]^4 \, dt \\ &\leq C |u_1 - u_2|_{X_T}^2 \sup_{t \in [0, \tau_1]} [\|u_1(t)\|_{\mathcal{V}} + \|u_2(t)\|_{\mathcal{V}}]^4 \int_0^{\tau_1} dt \\ &\leq C |u_1 - u_2|_{X_T}^2 \left[|u_1|_{X_{\tau_1}} + |u_2|_{X_{\tau_1}} \right]^4 \tau_1 \, . \end{split}$$

Since $|u_i|_{X_{\tau_i}} \leq 3n, i = 1, 2$, we get,

$$A_{2}^{2} \leq C|u_{1}-u_{2}|_{X_{T}}^{2} \left[|u_{1}|_{X_{\tau_{1}}}+|u_{2}|_{X_{\tau_{1}}}\right]^{4} \tau_{1} \leq (6n)^{4}C|u_{1}-u_{2}|_{X_{T}}^{2} \tau_{1}.$$

Thus

(4.2.10)
$$A_2 \le (6n)^2 C |u_1 - u_2|_{X_T} \tau_1^{\frac{1}{2}}$$

Now we consider

$$A_{3}^{2} = \int_{0}^{\tau_{2}} \left| \left[\theta_{n}(|u_{2}|_{X_{t}}) - \theta_{n}(|u_{1}|_{X_{t}}) \right] G_{2}(u_{2}) \right|_{\mathrm{H}}^{2} dt$$

Since θ_n is a Lipschitz function with Lipschitz constant 3n, we obtain

$$A_3^2 \leq 9n^2 \int_0^{\tau_2} \left| |u_2|_{X_t} - |u_1|_{X_t} \right|_{\mathrm{H}}^2 \left| G_2(u_2) \right|_{\mathrm{H}}^2 dt.$$

Since

$$||u_2|_{X_t} - |u_1|_{X_t}|_{\mathrm{H}} \leq |u_1 - u_2|_{X_t},$$

we get

(4.2.11)
$$A_{3}^{2} \leq 9n^{2} \int_{0}^{\tau_{2}} \left| u_{1} - u_{2} \right|_{X_{t}}^{2} \left| G_{2}(u_{2}) \right|_{H}^{2} dt$$
$$\leq 9n^{2} \left| u_{1} - u_{2} \right|_{X_{T}}^{2} \int_{0}^{\tau_{2}} \left| G_{2}(u_{2}) \right|_{H}^{2} dt.$$

Now consider $\int_0^{ au_2} \left| G_2(u_2) \right|_{\mathrm{H}}^2 dt$. Using (4.2.3) we get

$$\int_0^{\tau_2} \left| G_2(u_2) \right|_{\mathrm{H}}^2 dt \le \tilde{C} \int_0^{\tau_2} \|u_2(t)\|_{\mathrm{V}}^3 |u_2(t)|_{\mathrm{E}} dt \le \tilde{C} \left[\sup_{t \in [0, \tau_2]} \|u_2(t)\|_{\mathrm{V}}^2 \right]^{\frac{3}{2}} \int_0^{\tau_2} |u_2(t)|_{\mathrm{E}} dt.$$

We apply the Hölder inequality to obtain

$$\int_{0}^{\tau_{2}} \left| G_{2}(u_{2}) \right|_{\mathrm{H}}^{2} dt \leq \tilde{C} |u_{2}|_{X_{\tau_{2}}}^{3} \left[\int_{0}^{\tau_{2}} |u_{2}(t)|_{\mathrm{E}}^{2} dt \right]^{\frac{1}{2}} \left[\int_{0}^{\tau_{2}} dt \right]^{\frac{1}{2}}.$$

Now since $\int_0^{\tau_2} |u_2|_{\mathbf{E}}^2 dt \le |u_2|_{X_{\tau_2}}^2$ and $|u_2|_{X_{\tau_2}} \le 3n$,

$$\int_{0}^{\tau_{2}} \left| G_{2}(u_{2}) \right|_{\mathrm{H}}^{2} dt \leq \tilde{C} |u_{2}|_{X_{\tau_{2}}}^{3} |u_{2}|_{X_{\tau_{2}}} \tau_{2}^{\frac{1}{2}} \leq \tilde{C}(3n)^{4} \tau_{2}^{\frac{1}{2}}.$$

Hence, the inequality (4.2.11) takes the form

$$A_3^2 \le 9n^2 \tilde{C} |u_1 - u_2|_{X_T}^2 (3n)^4 \tau_2^{\frac{1}{2}},$$

from where we deduce

(4.2.12)
$$A_3 \leq (3n)^3 \tilde{C} |u_1 - u_2|_{X_T} \tau_2^{\frac{1}{4}}.$$

Since $\theta_n(|u_1|_{X_t}) = 0$ for $t > \tau_1$ and $\tau_1 < \tau_2$, we have

$$A_{4} = \left[\int_{0}^{\tau_{2}} \left|\theta_{n}(|u_{1}|_{X_{t}})[G_{2}(u_{2}) - G_{2}(u_{1})]\right|_{H}^{2} dt\right]^{\frac{1}{2}} = \left[\int_{0}^{\tau_{1}} \left|\theta_{n}(|u_{1}|_{X_{t}})[G_{2}(u_{2}) - G_{2}(u_{1})]\right|_{H}^{2} dt\right]^{\frac{1}{2}}.$$

Since $\theta_n(|u_1|_{X_t}) \le 1$ for $t \in [0, \tau_1]$ and using (4.2.3), we have

$$\begin{split} A_4 &\leq \left[\int_0^{\tau_1} \left| G_2(u_2) - G_2(u_1) \right|_{\mathcal{H}}^2 dt \right]^{\frac{1}{2}} \\ &\leq \tilde{C} \left[\int_0^{\tau_1} \left[\|u_1\|_{\mathcal{V}}^{1/2} |u_1|_{\mathcal{E}}^{1/2} \|u_1 - u_2\|_{\mathcal{V}} + \|u_1 - u_2\|_{\mathcal{V}}^{1/2} |u_1 - u_2|_{\mathcal{E}}^{1/2} \|u_2\|_{\mathcal{V}} \right]^2 dt \right]^{\frac{1}{2}}. \end{split}$$

Now by the Minkowski inequality,

$$\begin{split} A_4 &\leq \tilde{C} \left[\left[\int_0^{\tau_1} |u_1|_{\mathbf{E}} \|u_1 - u_2\|_{\mathbf{V}}^2 \|u_1\|_{\mathbf{V}} dt \right]^{\frac{1}{2}} + \left[\int_0^{\tau_1} \|u_2\|_{\mathbf{V}}^2 |u_1 - u_2|_{\mathbf{E}}^{1/2} \|u_1 - u_2\|_{\mathbf{V}} dt \right]^{\frac{1}{2}} \right] \\ &\leq \tilde{C} \left[\sup_{t \in [0, \tau_1]} \|u_1 - u_2\|_{\mathbf{V}}^2 \left[\sup_{t \in [0, \tau_1]} \|u_1\|_{\mathbf{V}}^2 \right]^{\frac{1}{2}} \int_0^{\tau_1} |u_1|_{\mathbf{E}} dt \right]^{\frac{1}{2}} \\ &+ \tilde{C} \left[\sup_{t \in [0, \tau_1]} \|u_2\|_{\mathbf{V}}^2 \left[\sup_{t \in [0, \tau_1]} \|u_1 - u_2\|_{\mathbf{V}}^2 \right]^{\frac{1}{2}} \int_0^{\tau_1} |u_1 - u_2|_{\mathbf{E}} dt \right]^{\frac{1}{2}} . \end{split}$$

Since

$$\sup_{t \in [0,\tau_1]} \|u_i\|_{\mathbf{V}}^2 \le |u_i|_{X_{\tau_1}}^2, \qquad \int_0^{\tau_1} |u_1|_{\mathbf{E}}^2 dt \le |u_1|_{X_{\tau_1}}^2,$$

and by using the Hölder inequality, we obtain

$$\begin{split} A_4 &\leq \tilde{C} \left[|u_1 - u_2|_{X_T}^2 |u_1|_{X_{\tau_1}} \left[\int_0^{\tau_1} |u_1|_{\mathbf{E}}^2 dt \right]^{\frac{1}{2}} \left[\int_0^{\tau_1} dt \right]^{\frac{1}{2}} \right]^{\frac{1}{2}} \\ &+ \tilde{C} \left[|u_1 - u_2|_{X_T} |u_2|_{X_{\tau_1}}^2 \left[\int_0^{\tau_1} |u_1 - u_2|_{\mathbf{E}}^2 dt \right]^{\frac{1}{2}} \left[\int_0^{\tau_1} dt \right]^{\frac{1}{2}} \right]^{\frac{1}{2}} \\ &\leq \tilde{C} \left[|u_1 - u_2|_{X_T}^2 |u_1|_{X_{\tau_1}}^2 \tau_1^{\frac{1}{2}} \right]^{\frac{1}{2}} + \tilde{C} \left[|u_1 - u_2|_{X_T}^2 |u_2|_{X_{\tau_1}}^2 \tau_1^{\frac{1}{2}} \right]^{\frac{1}{2}}. \end{split}$$

For $i = 1, 2, |u_i|_{X_{\tau_1}} \le 3n$, thus

$$(4.2.13) A_4 \le 6n\tilde{C}|u_1 - u_2|_{X_T} \tau_1^{\frac{1}{4}}$$

Now using (4.2.9), (4.2.11), (4.2.12) and (4.2.13) in (4.2.7), we obtain

$$\begin{split} |\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2)|_{L^2(0,T;\mathrm{H})} \\ &\leq (3n)^4 C |u_1 - u_2|_{X_T} \tau_2^{\frac{1}{2}} + (6n)^2 C |u_1 - u_2|_{X_T} \tau_1^{\frac{1}{2}} + (3n)^3 \tilde{C} |u_1 - u_2|_{X_T} \tau_2^{\frac{1}{4}} + 6n \tilde{C} |u_1 - u_2|_{X_T} \tau_1^{\frac{1}{4}} \\ &\leq (3n)^4 C |u_1 - u_2|_{X_T} T^{\frac{1}{2}} + (6n)^2 C |u_1 - u_2|_{X_T} T^{\frac{1}{2}} + (3n)^3 \tilde{C} |u_1 - u_2|_{X_T} T^{\frac{1}{4}} + 6n \tilde{C} |u_1 - u_2|_{X_T} T^{\frac{1}{4}} \\ &= K(n,T) |u_1 - u_2|_{X_T} T^{\frac{1}{4}}, \end{split}$$

where

$$K(n,T) = 3n \left(27n^3T^{1/4} + 9n^2 + 12nT^{1/4} + 2 \right),$$

is a constant which depends only on *n* and *T*. Thus we have proved that $\Phi_{n,T}$ is a Lipschitz function and satisfies (4.2.6).

4.2.2 Definition of a solution

Let us recall that $E \hookrightarrow V \hookrightarrow H$. $S = (S(t))_{t \ge 0}$ is the Stokes semigroup. The following are well-known [39, 88, 90]

A1. For every T > 0 and $f \in L^2(0, T; H)$ a function u = S * f, defined by

$$u(t) = \int_0^T S(t-r)f(r)dr, \qquad t \in [0,T],$$

belongs to X_T and

$$(4.2.14) |u|_{X_T} \le C_1 |f|_{L^2(0,T;\mathbf{H})}$$

A2. For every T > 0 and $u_0 \in V$ a function $u = Su_0$ defined by

$$u(t) = S(t)u_0,$$

belongs to X_T and

$$(4.2.15) |u|_{X_T} \le C_2 ||u_0||_{\mathcal{V}}.$$

Definition 4.2.4. • A solution of (4.2.1) on [0,T], $T \in [0,\infty)$ is a function $u \in X_T$ satisfying

$$u(t) = S(t)u_0 + \int_0^t S(t-r)G(u(r))dr, \quad \forall t \in [0,T],$$

where $G: \mathbf{E} \to \mathbf{H}$ is defined by

$$G(u) = |\nabla u|_{L^2}^2 u - B(u, u), \qquad u \in \mathbf{E}$$

• Let $\tau \in [0,\infty]$. A function $u \in \mathscr{C}([0,\tau), \mathbb{V})$ is a solution to (4.2.1) on $[0,\tau)$ iff $\forall T < \tau, u|_{[0,T]} \in X_T$ is a solution of (4.2.1) on [0,T].

4.2.3 Local existence

Lemma 4.2.5. Let K(n,T) be as introduced in Lemma 4.2.3. Consider a map $\Psi_{n,T}: X_T \to X_T$ defined by

$$\Psi_{n,T}(u) = S u_0 + S * \Phi_{n,T}(u), \quad u \in X_T.$$

Then for every $u_0 \in V$, there exists a constant $C_1 > 0$ such that

$$(4.2.16) \qquad |\Psi_{n,T}(u_1) - \Psi_{n,T}(u_2)|_{X_T} \le C_1 K(n,T) |u_1 - u_2|_{X_T} T^{\frac{1}{4}}, \quad u_1, u_2 \in X_T.$$

Moreover, $\forall \epsilon \in (0,1) \exists T_0 = T_0(n,\epsilon)$ such that $\Psi_{n,T}$ is an ϵ -contraction for $T \leq T_0$.

Proof. The map $\Psi_{n,T}$ is evidently well defined. Now for any $u_1, u_2 \in X_T$

$$\begin{split} |\Psi_{n,T}(u_1) - \Psi_{n,T}(u_2)|_{X_T} &= \left| S(t)u_0 + S * \Phi_{n,T}(u_1) - S(t)u_0 - S * \Phi_{n,T}(u_2) \right|_{X_T} \\ &= \left| S * \left(\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2) \right) \right|_{X_T}, \end{split}$$

then by treating $S * (\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2))$ as u and $[\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2)] \in L^2(0,T;H)$ as f in inequality (4.2.14) and using Lemma 4.2.3 we get

$$\begin{aligned} |\Psi_{n,T}(u_1) - \Psi_{n,T}(u_2)|_{X_T} &\leq C_1 |\Phi_{n,T}(u_1) - \Phi_{n,T}(u_2)|_{L^2(0,T;\mathrm{H})} \\ &\leq C_1 K(n,T) |u_1 - u_2|_{X_T} T^{\frac{1}{4}} \,, \end{aligned}$$

which shows that $\Psi_{n,T}$ is globally Lipschitz and satisfies (4.2.16). Let us fix $n \in \mathbb{N}$ and $\varepsilon \in (0,1)$. Since the constant C_1 is independent of T, we can find a $T_0 = T_0(n,\varepsilon)$ such that

$$C_1 K(n,T_0) T_0^{\frac{1}{4}} = \varepsilon,$$

and thus $\Psi_{n,T}$ is an ε -contraction for $T \leq T_0$.

Let $\varepsilon \in (0,1)$ then from Lemma 4.2.5, $\Psi_{n,T}$ is an ε -contraction for $T = T_0(n,\varepsilon)$ and thus by Banach Fixed Point Theorem there exists a unique $u^n \in X_T^{-1}$ s.t.

$$u^n = \Psi_{n,T}(u^n).$$

This implies that

$$u^{n}(t) = [\Psi_{n,T}(u^{n})](t), \quad t \in [0, T_{0}].$$

Let us define

$$\tau_n = \inf \{ t \in [0, T_0] : |u^n|_{X_t} \ge n \}.$$

Remark 4.2.6. If $|u^n|_{X_t} < n$ for each $t \in [0, T_0^n]$ then $\tau_n = T_0^n$.

¹In fact u^n should have been denoted by $u^{n,T}$ but we have refrained from this.

Theorem 4.2.7. Let R > 0 be given then $\exists T_* = T_*(R)$ such that for every $u_0 \in V$ with $||u_0||_V \leq R$, there exists a unique local solution $u : [0, T_*] \to V$ of (4.2.1).

Proof. Let R > 0 and fix $\varepsilon \in (0, 1)$. Let us choose² $n = \lfloor \frac{C_2 R}{1-\varepsilon} \rfloor + 1$ where C_2 is as defined in (4.2.15). Now for these fixed n and ε , $\exists T_0(n,\varepsilon)$ such that $\Psi_{n,T}$ is an ε -contraction for all $T \leq T_0$. In particular, it is true for $T = T_0$ and hence by Banach Fixed Point Theorem $\exists! u^n \in X_{T_0}$ such that

$$u^n = \Psi_{n,T}(u^n).$$

Note that we have

$$|u^{n}|_{X_{T_{0}}} = |\Psi_{n,T}(u^{n})|_{X_{T_{0}}} = |Su_{0} + S * \Phi_{n,T}(u^{n})|_{X_{T_{0}}}$$

$$\leq |Su_{0}|_{X_{T_{0}}} + |S * \Phi_{n,T}(u^{n})|_{X_{T_{0}}}.$$

Now from (4.2.15) and Lemma 4.2.5, we have

$$|u^{n}|_{X_{T_{0}}} \leq C_{2} ||u_{0}||_{V} + \varepsilon |u^{n}|_{X_{T_{0}}}.$$

Since $||u_0||_V \leq R$, hence on rearranging we get

$$(1-\varepsilon)|u^n|_{X_{T_0}} \le C_2 R\,,$$

and so

$$|u^n|_{X_{T_0}} \le \frac{C_2 R}{1-\varepsilon} \le n \,.$$

Now since $t \mapsto |\cdot|_{X_t}$ is an increasing function, the following holds

$$|u^n|_{X_t} \le n, \qquad \forall \ t \in [0, T_0].$$

In particular, $|u^n|_{X_{T_0}} \le n$, i.e. $|u^n|_{X_{T_0}}$ is finite and thus $u^n \in X_{T_0}$. This implies

$$\theta_n(|u^n|_{X_t}) = 1, \quad t \in [0, T_0].$$

Thus for $t \in [0, T_0]$

$$u^{n}(t) = S(t)u_{0} + \int_{0}^{t} S(t-r)G(u^{n}(r))dr.$$

So u^n on $[0, T_*(R)]$, where $T_* = T_0(n, \varepsilon)$, solves (4.2.1) and T_* depends only on R. Thus, we have proved the existence of a unique local solution of (4.2.1) for every initial data $u_0 \in V$, and this unique solution is denoted by u.

²[*M*] denotes the largest integer less than or equal to *M*.

4.2.4 The local solution stays on the manifold \mathcal{M}

Lemma 4.2.8. If u is the solution of (4.2.1) on $[0,\tau)$ then $u' \in L^2(0,T;H)$, for every $T < \tau$, i.e. $u' \in L^2_{loc}([0,\tau);H)$. Moreover, $u \in L^2(0,T;D(A))$.

Proof. Let us fix $T < \tau$. Since *u* is the solution of (4.2.1) on $[0, \tau)$ it satisfies

(4.2.17)
$$\frac{du}{dt} = -Au + |\nabla u|_{L^2}^2 u - B(u, u).$$

We will show that RHS of (4.2.17) belongs to $L^2(0, T; H)$ and hence $u' \in L^2(0, T; H)$. Since $u \in L^2(0, T; E)$, $Au \in L^2(0, T; H)$. From (4.2.2) we have

$$\begin{split} \int_{0}^{T} \left| |\nabla u(t)|_{L^{2}}^{2} u(t) \right|_{H}^{2} dt &\leq \int_{0}^{T} C^{2} \|u(t)\|_{V}^{6} dt \leq C^{2} \sup_{t \in [0,T]} \|u(t)\|_{V}^{6} \int_{0}^{T} dt \\ &\leq C^{2} T \left[\sup_{t \in [0,T]} \|u(t)\|_{V}^{2} \right]^{3} \leq C^{2} T |u|_{X_{T}}^{6} < \infty, \end{split}$$

thus we have shown that $|\nabla u|_{L^2}^2 u \in L^2(0, T; H)$. From (4.2.3), we have

$$\begin{split} \int_{0}^{T} \left| B(u(t), u(t)) \right|_{\mathrm{H}}^{2} dt &\leq \tilde{C}^{2} \int_{0}^{T} \|u(t)\|_{\mathrm{V}}^{3} |u(t)|_{\mathrm{E}} dt \leq \tilde{C}^{2} \sup_{t \in [0, T]} \|u(t)\|_{\mathrm{V}}^{3} \int_{0}^{T} |u(t)|_{\mathrm{E}} dt \\ &\leq \tilde{C}^{2} \left[\sup_{t \in [0, T]} \|u(t)\|_{\mathrm{V}}^{2} \right]^{\frac{3}{2}} \left[\int_{0}^{T} |u(t)|_{\mathrm{E}}^{2} dt \right]^{\frac{1}{2}} \left[\int_{0}^{T} dt \right]^{\frac{1}{2}} \\ &\leq \tilde{C}^{2} |u|_{X_{T}}^{3} |u|_{X_{T}} T^{\frac{1}{2}} < \infty. \end{split}$$

Thus the convective term from Navier-Stokes also belongs to $L^2(0, T; H)$ and hence RHS of (4.2.17) belongs to $L^2(0, T; H)$ which implies that $u' \in L^2(0, T; H)$ for all $T < \tau$. The second conclusion can be inferred from property **A1**.

Let us recall that the inner product $\langle \cdot, \cdot \rangle_V$ was defined in Section 3.1 for \mathbb{R}^2 as well as \mathbb{T}^2 .

Remark 4.2.9. In the framework of Lemma 2.6.3, we can identify v with u and so we get

(4.2.18)
$$\frac{1}{2}|u(t)|_{\mathrm{H}}^2 = \frac{1}{2}|u_0|^2 + \int_0^t \langle u'(s), u(s) \rangle_{\mathrm{H}} \, ds \,, \qquad \text{for a.e. } t \in [0, \tau) \,.$$

Moreover, from Theorem 4.2.7 and Lemma 4.2.8

(4.2.19)
$$\frac{1}{2} \|u(t)\|_{\mathbf{V}}^2 = \frac{1}{2} \|u_0\|_{\mathbf{V}}^2 + \int_0^t \langle u'(s), u(s) \rangle_{\mathbf{V}} ds, \quad \text{for a.e. } t \in [0, \tau).$$

Theorem 4.2.10. If $\tau \in [0,\infty]$, $u_0 \in \mathcal{M} \cap V$ and u is a solution to (4.2.1) on $[0,\tau)$ then $u(t) \in \mathcal{M}$ for all $t \in [0,\tau)$.

Proof. Let *u* be the solution to (4.2.1) and $u_0 \in \mathcal{M} \cap V$. Let us define $\phi(t) = |u(t)|_{\mathrm{H}}^2 - 1$. Then ϕ is absolutely continuous and by Remark 4.2.9 and (4.2.1) we have a.e. on $[0, \tau)$

$$\begin{split} \frac{d}{dt}\phi(t) &= \frac{d}{dt}[|u(t)|_{\mathrm{H}}^2 - 1] = 2\langle u'(t), u(t) \rangle_{\mathrm{H}} \\ &= 2\langle -Au(t) + |\nabla u(t)|_{L^2}^2 u(t) - B(u(t), u(t)), u(t) \rangle_{\mathrm{H}} \\ &= -2\langle Au(t), u(t) \rangle_{\mathrm{H}} + 2|\nabla u(t)|_{L^2}^2 \langle u(t), u(t) \rangle_{\mathrm{H}} \\ &= -2|\nabla u(t)|_{L^2}^2 + 2|\nabla u(t)|_{L^2}^2 |u(t)|^2 \\ &= 2|\nabla u(t)|_{L^2}^2 (|u(t)|_{\mathrm{H}}^2 - 1) = |\nabla u(t)|_{L^2}^2 \phi(t). \end{split}$$

This on integration gives

$$\phi(t) = \phi(0) \exp\left[\int_0^t |\nabla u(s)|_{L^2}^2 ds\right], \qquad t \in [0, \tau).$$

Since $u_0 \in \mathcal{M}, \phi(0) = 0$ and also as $u \in X_T$ is the solution of (4.2.1),

$$\int_0^t |\nabla u(s)|_{L^2}^2 ds \le \int_0^t ||u(s)||_V^2 ds < \infty, \qquad t \in [0, \tau).$$

Hence, we infer that $|u(t)|_{\mathrm{H}}^2 = 1$ for every $t \in [0, \tau)$. Thus, $u(t) \in \mathcal{M}$ for every $t \in [0, \tau)$.

Corollary 4.2.11. Let the initial data $u_0 \in \mathcal{M} \cap V$ and u is the solution to (4.2.1) on $[0, \tau)$. Then u'(t) is orthogonal to u(t) in H for almost all $t \in [0, \tau)$.

Remark 4.2.12. We can also prove Theorem 4.2.7 and Theorem 4.2.10 for any general bounded domain. Thus, establishing the existence of a local solution to (4.2.1) for any general bounded domain and \mathbb{R}^2 .

4.3 Global solution: Existence and Uniqueness

The main result of this section is the proof of Theorem 4.1.1, i.e. we will show that the local solution obtained in Theorem 4.2.7 is indeed a global one. Lemma A.1 and the Remark 4.3.1 play crucial role in proving the global existence of the solution. We first show that the enstrophy (gradient norm) of the solution remains bounded (see Lemma 4.3.2) and then use stitching argument to extend our solution from [0, T], $T < \infty$ on to the whole real line.

We recall the orthogonality property of the Stokes-operator in the following remark.

Remark 4.3.1. Let $u \in D(A)$, then

$$\langle B(u,u), Au \rangle_{\mathrm{H}} = 0, \quad \forall u \in \mathrm{D}(\mathrm{A}),$$

on a bounded domain with periodic boundary conditions (i.e. on a torus) [90] or on \mathbb{R}^2 .

We define the energy of our system by

$$\mathscr{E}(u) = \frac{1}{2} |\nabla u|_{L^2}^2, \quad u \in \mathbf{V}.$$

Then, heuristically, for $u \in V \cap \mathcal{M}$,

$$\nabla_{\mathcal{M}} \mathscr{E}(u) = \Pi_u (\nabla \mathscr{E}) = \Pi_u (\mathbf{A}u)$$
$$= \mathbf{A}u - |\nabla u|_{L^2}^2 u \,.$$

Thus, for $u \in \mathcal{M}$

(4.3.1)
$$\begin{aligned} |\nabla_{\mathcal{M}}\mathscr{E}(u)|_{\mathrm{H}}^{2} &= |\mathrm{A}u|_{\mathrm{H}}^{2} + |\nabla u|_{L^{2}}^{4}|u|_{\mathrm{H}}^{2} - 2|\nabla u|_{L^{2}}^{2}\langle \mathrm{A}u, u\rangle_{\mathrm{H}} \\ &= |u|_{\mathrm{E}}^{2} + |\nabla u|_{L^{2}}^{4} - 2|\nabla u|_{L^{2}}^{4} = |u|_{\mathrm{E}}^{2} - |\nabla u|_{L^{2}}^{4}. \end{aligned}$$

In particular, the R.H.S. of (4.3.1) is ≥ 0 .

Lemma 4.3.2. Let $u_0 \in V$ and u be the local solution of (4.2.1) on $[0, \tau)$, then

$$\sup_{s \in [0,\tau)} \|u(s)\|_{\mathcal{V}} \le \|u_0\|_{\mathcal{V}}.$$

Proof. Let *u* be the solution of (4.2.1). Then, from (4.2.1), Remark 4.2.9 and Corollary 4.2.11, for any $t \in [0, \tau)$ we have

$$\begin{split} \frac{1}{2} \|u(t)\|_{\mathrm{V}}^2 &= \frac{1}{2} \|u_0\|_{\mathrm{V}}^2 + \int_0^t \langle u'(s), u(s) \rangle_{\mathrm{V}} ds \\ &= \frac{1}{2} \|u_0\|_{\mathrm{V}}^2 + \int_0^t \langle u'(s), u(s) \rangle_{\mathrm{H}} ds + \int_0^t \langle u'(s), \mathrm{A}u(s) \rangle_{\mathrm{H}} ds \\ &= \frac{1}{2} \|u_0\|_{\mathrm{V}}^2 + \int_0^t \langle -\mathrm{A}u(s) + |\nabla u(s)|_{L^2}^2 u(s) - B(u(s), u(s)), \mathrm{A}u(s) \rangle_{\mathrm{H}} ds \\ &= \frac{1}{2} \|u_0\|_{\mathrm{V}}^2 + \int_0^t \left[-\langle \mathrm{A}u(s), \mathrm{A}u(s) \rangle_{\mathrm{H}} + |\nabla u(s)|_{L^2}^2 \langle u(s), \mathrm{A}u(s) \rangle_{\mathrm{H}} \right] ds \\ &- \int_0^t \langle B(u(s), u(s)), \mathrm{A}u(s) \rangle_{\mathrm{H}} ds \\ &= \frac{1}{2} \|u_0\|_{\mathrm{V}}^2 + \int_0^t \left[-|u(s)|_{\mathrm{E}}^2 + |\nabla u(s)|_{L^2}^4 \right] ds \,. \end{split}$$

Now from Theorem 4.2.10 we know that $u(t) \in \mathcal{M}$ for every $t \in [0, \tau)$ and hence by using (4.3.1) we obtain,

$$\frac{1}{2} \|u(t)\|_{\mathbf{V}}^2 = \frac{1}{2} \|u_0\|_{\mathbf{V}}^2 - \int_0^t \left| [\nabla_{\mathcal{M}} \mathscr{E}(u)](s) \right|_{\mathbf{H}}^2 ds,$$

and thus

$$\frac{1}{2} \|u(t)\|_{\mathbf{V}}^{2} + \int_{0}^{t} \left| [\nabla_{\mathcal{M}} \mathscr{E}(u)](s) \right|_{\mathbf{H}}^{2} ds = \frac{1}{2} \|u_{0}\|_{\mathbf{V}}^{2}.$$

Hence, we have shown that

$$\|u(t)\|_{\mathcal{V}} \le \|u_0\|_{\mathcal{V}}, \quad t \in [0, \tau).$$

Remark 4.3.3. The boundedness of enstrophy (the square of the gradient norm) of the solution, as proved in the Lemma 4.3.2, will play a crucial role in proving the existence of a global-in-time solution to problem (4.2.1). Note that, in the proof of the Lemma 4.3.2, the orthogonality of the Stokes operator A to the convective term B(u, u) in H was essential, which as far as we know, holds only on \mathbb{R}^2 and on bounded domains with periodic boundary conditions (i.e. on a torus). This is the reason we were unable to prove the existence of a global-in-time solution to problem (4.2.1) on any general bounded domain.

Lemma 4.3.4. Let $0 \le a < b < c < \infty$ and $u \in X_{[a,b]}, v \in X_{[b,c]}$, such that $u(b^{-}) = v(b^{+})$. Then $z \in X_{[a,c]}$ where,

$$z(t) = \begin{cases} u(t), & t \in [a,b), \\ v(t), & t \in [b,c). \end{cases}$$

Proof. Let us take $0 \le a < b < c < \infty$ and $u \in X_{[a,b]}$, $v \in X_{[b,c]}$, such that $u(b^-) = v(b^+)$. Then for any $0 \le t_1 < t_2 < \infty$, using the definition of the norm $|\cdot|_{X_{[t_1,t_2]}}$, we have

$$\begin{aligned} |z|_{X_{[a,c]}}^2 &= \sup_{t \in [a,c]} \|z(t)\|_{\mathbf{V}}^2 + \int_a^c |z(t)|_{\mathbf{E}}^2 dt \\ &\leq \sup_{t \in [a,b]} \|z(t)\|_{\mathbf{V}}^2 + \sup_{t \in [b,c]} \|z(t)\|_{\mathbf{V}}^2 + \int_a^b |z(t)|_{\mathbf{E}}^2 dt + \int_b^c |z(t)|_{\mathbf{E}}^2 dt \end{aligned}$$

Now by the definition of *z* we have

$$\begin{split} |z|_{X_{[a,c]}}^{2} &\leq \sup_{t \in [a,b]} \|u(t)\|_{\mathcal{V}}^{2} + \sup_{t \in [b,c]} \|v(t)\|_{\mathcal{V}}^{2} + \int_{a}^{b} |u(t)|_{\mathcal{E}}^{2} dt + \int_{b}^{c} |v(t)|_{\mathcal{E}}^{2} dt \\ &= \sup_{t \in [a,b]} \|u(t)\|_{\mathcal{V}}^{2} + \int_{a}^{b} |u(t)|_{\mathcal{E}}^{2} dt + \sup_{t \in [b,c]} \|v(t)\|_{\mathcal{V}}^{2} + \int_{b}^{c} |v(t)|_{\mathcal{E}}^{2} dt \\ &= |u|_{X_{[a,b]}}^{2} + |v|_{X_{[b,c]}}^{2}. \end{split}$$

Now since $u \in X_{[a,b]}$ and $v \in X_{[b,c]}$ we have $|z|_{X_{[a,c]}} < \infty$, and thus, $z \in X_{[a,c]}$.

We will use the following lemma to prove the main result about existence of the global solution.

Lemma 4.3.5. Let τ be finite and the initial data $u_0 \in V \cap \mathcal{M}$. If $u : [0, \tau] \to V$ is the solution of (4.2.1) on $[0, \tau]$ and $v : [\tau, 2\tau] \to V$ is the solution of (4.2.1) on $[\tau, 2\tau]$ such that $u(\tau^-) = v(\tau^+)$, then $z : [0, 2\tau] \to V$ defined as

$$z(t) = \begin{cases} u(t), & t \in [0, \tau], \\ v(t), & t \in [\tau, 2\tau], \end{cases}$$

is the solution of (4.2.1) on $[0, 2\tau]$ and $z \in X_{[0, 2\tau]}$.

Proof. Since *u* is the solution of (4.2.1) on $[0,\tau]$ then $u \in X_{[0,\tau]}$ and similarly $v \in X_{[\tau,2\tau]} := \mathscr{C}([\tau,2\tau];V) \cap L^2(\tau,2\tau;E)$. Thus by Lemma 4.3.4 and the definition of $z, z \in X_{[0,2\tau]}$. Now we are left to show that $z:[0,2\tau] \to V$ defined as

$$z(t) = \begin{cases} u(t), & t \in [0, \tau], \\ v(t), & t \in [\tau, 2\tau], \end{cases}$$

is the solution of (4.2.1) on $[0, 2\tau]$. In order to achieve this we will have to show that *z* satisfies (4.3.2) for every $t \in [0, 2\tau]$.

(4.3.2)
$$z(t) = S(t)z(0) + \int_0^t S(t-r)G(z(r))dr.$$

For $t \in [0, \tau]$, z satisfies (4.3.2), since z(t) = u(t), $\forall t \in [0, \tau]$ and u is the solution of (4.2.1) on $[0, \tau]$.

For $t \in [\tau, 2\tau]$, z(t) = v(t) and since v is the solution to (4.2.1) on $[\tau, 2\tau]$,

$$z(t) = \mathbf{v}(t) = S(t-\tau)\mathbf{v}(\tau) + \int_{\tau}^{t} S(t-\tau)G(\mathbf{v}(\tau)) d\tau.$$

Now because of continuity of *u* and v, $v(\tau) = u(\tau)$,

$$z(t) = S(t-\tau) \Big[S(\tau)u_0 + \int_0^\tau S(\tau-r)G(u(r))dr \Big] + \int_\tau^t S(t-r)G(v(r))dr.$$

Now using the definition of z we obtain,

$$z(t) = S(t)z(0) + \int_0^\tau S(t-r)G(z(r))dr + \int_\tau^t S(t-r)G(z(r))dr$$

= $S(t)z(0) + \int_0^t S(t-r)G(z(r))dr$.

Thus z satisfies (4.3.2) on $[0,2\tau]$ and hence z is a solution to (4.2.1) on $[0,2\tau]$.

Proof of Theorem 4.1.1 Let us take $u_0 \in V$. Put $R = ||u_0||_V$. By Theorem 4.2.7 there exists a T > 0 such that there exists a unique function $u : [0,T] \to V$ which solves (4.2.1) on [0,T] and $u \in X_T$. Also by Lemma 4.3.2 $||u(T)||_V \le R$ thus again by Theorem 4.2.7 there exists a unique function $v : [T,2T] \to V$ which solves (4.2.1) on [T,2T] and $v \in X_{[T,2T]}$. Now if we define a new function $z : [0,2T] \to V$ as

$$z(t) = \begin{cases} u(t), & t \in [0,T], \\ v(t), & t \in [T,2T] \end{cases}$$

then by Lemma 4.3.5, z is also a solution of (4.2.1) and $z \in X_{2T}$. Moreover $||z(2T)||_V \leq R$. We can keep doing this and extend our solution further and hence obtaining a global solution of (4.2.1) still denoted by u such that $u \in X_T$ for every $T < \infty$. Each bit of the solution is unique on the respective domain and hence when we glue two unique bits we get a unique extension and thus obtain a unique global solution due to its construction.

4.4 Convergence to the Euler equation, i.e. the inviscous limit

In this section we are concerned with the convergence of the solution of the constrained Navier-Stokes equations, namely

(4.4.1)
$$\begin{cases} \frac{du}{dt} + vAu - v |\nabla u|_{L^2}^2 u + B(u, u) = 0, \\ u(0) = u_0^v \in \mathbf{V} \cap \mathcal{M}, \end{cases}$$

as v vanishes on a torus.

The curl of a vector field u is defined by

(4.4.2)
$$\operatorname{Curl}(u) := D_1 u_2 - D_2 u_1.$$

We will prove Theorem 4.1.2 after several preliminary results.

Remark 4.4.1. Curl is a linear isomorphism between V and $L^2_0(\mathbb{T}^2)$, where

$$L^2_0(\mathbb{T}^2) := \left\{ \omega \in L^2(\mathbb{T}^2) : \int_{\mathbb{T}^2} \omega(x) dx = 0 \right\} \,.$$

Moreover for $u \in V$ and some universal constants C > 0, $C_p > 0$

$$(4.4.3) \qquad \qquad |\Delta u|_{L^2(\mathbb{T}^2)} \le C |\nabla \operatorname{Curl}(u)|_{L^2(\mathbb{T}^2)}$$

(4.4.4)
$$\|\nabla u\|_{L^{p}(\mathbb{T}^{2})} \leq C_{p} \|\operatorname{Curl}(u)\|_{L^{\infty}(\mathbb{T}^{2})}.$$

This remark is proved in Appendix B.

Hereafter u^{ν} is the solution to (4.4.1), and $\omega^{\nu}(t,x) := \operatorname{Curl}(u^{\nu}(t))(x)$. In particular, due to Remark 4.4.1 and Theorem 4.2.10, $\omega^{\nu} \in \mathscr{C}([0,T];L_0^2(\mathbb{T}^2)) \cap L^2(0,T;H^1(\mathbb{T}^2))$. It is then easy to check that ω^{ν} is a weak solution to

(4.4.5)
$$\begin{cases} \frac{d\omega^{\nu}}{dt} + \nabla \cdot (u^{\nu} \omega^{\nu}) = \nu \Delta \omega^{\nu} + \nu |\nabla u^{\nu}|_{L^{2}}^{2} \omega^{\nu}, \\ \omega^{\nu}(0) = \omega_{0}^{\nu} := \operatorname{Curl}(u_{0}^{\nu}) \in L_{0}^{2}(\mathbb{T}^{2}). \end{cases}$$

Proposition 4.4.2. Let us fix T > 0, and assume that $\omega_0^{\vee} \in L^{\infty}(\mathbb{T}^2)$. Then

(4.4.6)
$$\sup_{t \in [0,T]} \|\omega^{\nu}(t)\|_{L^{\infty}(\mathbb{T}^2)} \le \|\omega_0^{\nu}\|_{L^{\infty}(\mathbb{T}^2)} \exp\left(\nu \|u_0^{\nu}\|_{V}^2 T\right),$$

(4.4.7)
$$v \int_0^T |\nabla \omega^{\nu}(t)|^2_{L^2(\mathbb{T}^2)} dt \leq \frac{1}{2} |\omega_0^{\nu}|^2_{L^2(\mathbb{T}^2)} + v T ||u_0^{\nu}||^2_{\mathcal{V}} ||\omega_0^{\nu}||^2_{L^{\infty}(\mathbb{T}^2)} \exp\left(2v ||u_0^{\nu}||^2_{\mathcal{V}} T\right).$$

,

Proof. Take $h \in C^2(\mathbb{R})$, convex, with bounded second derivative. Then, since $\omega \in \mathscr{C}([0,T]; L^2_0(\mathbb{T}^2))$

(4.4.8)

$$\langle h(\omega^{\nu}(t)), \mathbf{1} \rangle - \langle h(\omega_{0}^{\nu}), \mathbf{1} \rangle$$

$$= \nu \int_{0}^{t} \left[-\langle h''(\omega^{\nu}(s)), |\nabla \omega^{\nu}(s)|_{L^{2}}^{2} \rangle + |\nabla u^{\nu}(s)|_{L^{2}}^{2} \langle h'(\omega^{\nu}(s)), \omega^{\nu}(s) \rangle \right] ds$$

$$\leq \nu \int_{0}^{t} |\nabla u^{\nu}(s)|_{L^{2}}^{2} \langle h'(\omega^{\nu}(s)), \omega^{\nu}(s) \rangle ds.$$

For $p \ge 2, R > 0$, take

(4.4.9)
$$h(w) \equiv h_{p,R}(w) := \begin{cases} |w|^p, & \text{if } |w| \le R, \\ R^p + p R^{p-1}(|w| - R) + \frac{p(p-1)}{2} R^{p-2}(|w| - R)^2, & \text{if } |w| > R. \end{cases}$$

Then $|h'(w)w| \le p h(w)$ and, by Lemma 4.3.2 $\forall s \in [0, t], ||u^{\nu}(s)||_{V}^{2} \le ||u_{0}^{\nu}||_{V}^{2}$

(4.4.10)
$$\langle h(\omega^{\nu}(t)), \mathbf{1} \rangle \leq \langle h(\omega_{0}^{\nu}), \mathbf{1} \rangle + \nu p \int_{0}^{t} \|u_{0}^{\nu}\|_{\mathbf{V}}^{2} \langle h(\omega^{\nu}(s)), \mathbf{1} \rangle ds.$$

By the Gronwall Lemma

(4.4.11)
$$\langle h(\omega^{\nu}(t)), \mathbf{1} \rangle \leq \langle h(\omega_0^{\nu}), \mathbf{1} \rangle \exp\left(\nu p \| u_0^{\nu} \|_{\mathbf{V}}^2 t\right), \quad t \in [0, T].$$

Since

(4.4.12)
$$\|\omega^{\nu}\|_{L^{\infty}} = \sup_{p,R} \langle h_{p,R}(\omega^{\nu}), \mathbf{1} \rangle^{1/p},$$

we get (4.4.6).

On the other hand, from the first equality in (4.4.8), taking now $h(w) = w^2/2$

$$\begin{aligned} \frac{1}{2} |\omega^{\nu}(T)|^{2}_{L^{2}(\mathbb{T}^{2})} + \nu \int_{0}^{T} |\nabla\omega^{\nu}(t)|^{2}_{L^{2}(\mathbb{T}^{2})} dt &= \frac{1}{2} |\omega^{\nu}_{0}|^{2}_{L^{2}(\mathbb{T}^{2})} + \nu \int_{0}^{T} |\nabla u^{\nu}(t)|^{2}_{L^{2}} |\omega^{\nu}(t)|^{2}_{L^{2}(\mathbb{T}^{2})} dt \\ &\leq \frac{1}{2} |\omega^{\nu}_{0}|^{2}_{L^{2}(\mathbb{T}^{2})} + \nu T ||u^{\nu}_{0}||^{2}_{\mathbb{V}} ||\omega^{\nu}_{0}||^{2}_{L^{\infty}(\mathbb{T}^{2})} e^{2\nu T ||u^{\nu}_{0}||^{2}_{\mathbb{V}}} \end{aligned}$$

where in the last line we used (4.4.6). Hence (4.4.7).

Proposition 4.4.3. For each $\varphi \in H^2(\mathbb{T}^2)$, and v > 0

 $(4.4.13) \qquad \langle \omega^{\nu}(t) - \omega^{\nu}(s), \varphi \rangle \le (t-s) \left(|\omega^{\nu}|_{L^{\infty}([0,T] \times \mathbb{T}^{2})} + 2\nu ||u_{0}^{\nu}||_{V} (1 + ||u_{0}^{\nu}||_{V}^{2}) \right) |\varphi|_{H^{2}(\mathbb{T}^{2})}.$

Proposition 4.4.4. Suppose that, uniformly in v, u_0^v is bounded in V and $\operatorname{Curl}(u_0^v)$ is bounded in $L^\infty(\mathbb{T}^2)$. Then the sequence u^v is precompact in $\mathscr{C}([0,T];L^2(\mathbb{T}^2))$.

Proof. Let us take and fix $\varphi \in H^2(\mathbb{T}^2)$. Also fix $0 \le s < t \le T$. Then from the equation (4.4.5) and $\|u^v(t)\|_V^2 \le \|u_0^v\|_V^2$ we get,

$$(4.4.14) \qquad \left| \langle u^{\nu}(t) - u^{\nu}(s), \varphi \rangle \right| \leq \nu \left| \int_{s}^{t} \langle \Delta u^{\nu}, \varphi \rangle \, dr \right| + \nu \|u_{0}^{\nu}\|_{V}^{2} \int_{s}^{t} \left| \langle u^{\nu}, \varphi \rangle \right| \, dr + \left| \int_{s}^{t} \langle u^{\nu} \nabla u^{\nu}, \varphi \rangle \, dr \right|.$$

By (4.4.3), (4.4.7) and the hypotheses on the initial data, the first term in the R.H.S. is bounded by $C_T |\varphi|_{L^2} (t-s)^{1/2}$ for some constant C_T independent on v. The second term in the R.H.S. of (4.4.14) easily enjoys the same bound. As for the third term in the R.H.S., for any p > 2, $|u|_{L^{\infty}} \leq C_p (|u|_{L^2} + ||\nabla u||_{L^p})$, so that from (4.4.4) and (4.4.6), this term is still bounded by $C_T |\varphi|_{L^2} (t-s)^{1/2}$.

Therefore, since u_0^v is bounded uniformly in $L^2(\mathbb{T}^2)$ by Poincaré inequality, it follows that u^v is equibounded and equicontinuous in $L^2(\mathbb{T}^2)$ and, by Arzelà-Ascoli theorem (see Theorem 2.4.3), precompact in $\mathscr{C}([0,T];L^2(\mathbb{T}^2))$.

Proof of Theorem 4.1.2 Fix T > 0. Using Propositions 4.4.3 - 4.4.4, from each subsequence we can extract a further subsequence such that $\omega^{\vee} \to \omega$ in $\mathscr{C}([0,T]; H^{-2}(\mathbb{T}^2))$ and weakly in $L^{\infty}([0,T] \times \mathbb{T}^2)$, $u^{\vee} \to u$ weakly in $L^{\infty}([0,T]; V)$ and in $\mathscr{C}([0,T]; L^2(\mathbb{T}^2))$. It is immediate to check that $\omega = \operatorname{Curl}(u)$.

Notice that $\omega_0^{\nu} := \operatorname{Curl}(u_0^{\nu})$ converges weakly in $L^{\infty}(\mathbb{T}^2)$ to $\omega_0 := \operatorname{Curl}(u_0)$. Passing to the limit in the weak formulation of the equation one then has, for each $\varphi \in C^2([0,T] \times \mathbb{T}^2)$

(4.4.15)
$$\langle \omega(t), \varphi(t) \rangle - \langle \omega_0, \varphi(0) \rangle - \int_0^t \langle \omega(s), \partial_s \varphi(s) \rangle \, ds - \int_0^t \langle u(s) \omega(s), \nabla \varphi(s) \rangle \, ds = 0 \,,$$

and $\omega(0) = \omega_0$. Recalling that $\omega = \operatorname{Curl}(u)$

$$(4.4.16) \quad \langle u(t), \nabla^{\perp} \varphi(t) \rangle - \langle u_0, \nabla^{\perp} \varphi(0) \rangle - \int_0^t \langle u(s), \partial_s \nabla^{\perp} \varphi(s) \rangle \, ds - \int_0^t \langle u(s) \cdot \nabla u(s), \nabla^{\perp} \varphi(s) \rangle \, ds = 0.$$

Since $\langle u \, \omega, \nabla \varphi \rangle = \langle u \cdot \nabla u, \nabla^{\perp} \varphi \rangle$ holds.

By Bardos uniqueness theorem [5, 37], we conclude that $u^{\nu} \rightarrow u$.

4.5 CNSE in the fractional Sobolev spaces

We study 2D Navier-Stokes equations with a constraint on L^2 energy of the solution in fractional Sobolev spaces. In Theorem 4.1.1 we proved the existence of a unique global-in-time solution for the constrained Navier-Stokes equations (4.1.3) on \mathbb{R}^2 and \mathbb{T}^2 with initial data $u_0 \in V$ (see Section 3.1). In this section we consider more regular initial data, $u_0 \in \hat{V} = D(A^{\frac{\alpha}{2}})$, $\alpha \in (1,3/2) \cup$ {2} (see below for details), and prove the existence of a unique global-in-time solution for the constrained Navier-Stokes equations on a general bounded domain under condition (4.5.21). In particular, we show it for a bounded periodic domain.

Our proof heavily relies on the Lemma 4.5.2, which holds true only for $\gamma \in (0, \frac{1}{2})$. This restriction on γ , is the essential reason for not having an existence of the solution in the case of $\alpha \in [\frac{3}{2}, 2)$, included in this thesis.

We are interested in the initial value problem

(4.5.1)
$$\begin{cases} \frac{du}{dt} = -vAu + v|\nabla u|_{L^2}^2 u - B(u, u), & \text{on } \mathcal{O}, \\ u(0) = u_0, & \text{on } \mathcal{O}, \\ u \cdot \mathbf{n} = 0 & \text{on } \Gamma, \end{cases}$$

where $u \in H$, and H is a space of divergence free vector fields, see (4.5.2) below for a precise definition. Γ is the smooth boundary of the bounded and simply connected domain \mathcal{O} . **n** is the unit outward normal to Γ .

The above problem has a local *maximal* solution for each $u_0 \in \hat{V} \cap \mathcal{M}$, where \hat{V} is defined in (4.5.3) and

$$\mathcal{M} = \{ u \in \mathbf{H} : |u|_{\mathbf{H}} = 1 \} .$$

As in Theorem 4.2.10, the solution u of (4.5.4) stays on the manifold \mathcal{M} for all times t. We use a different approach to prove the existence of a global solution compared to the proof of Theorem 4.1.1. We first show existence of a local solution using Banach fixed point theorem and then use these local solutions to construct a maximal solution. Next using the maximality of the solution we show that either the solution is a global one or \hat{V} -norm blows up in finite time, Lemma 4.5.10. Instead of using the geometric structure of (4.5.1) we use Lemma 2.6.5 and the Gronwall Lemma to obtain the bound on \hat{V} -norm of the solution, Lemma 4.5.12 and Remark 4.5.13. Invoking contradiction by the use of Lemmas 4.5.10 and 4.5.12, we infer the existence of a global-in-time solution.

Let us fix T > 0 and set

$$X_T = \mathscr{C}([0,T]; \hat{\mathbf{V}}) \cap L^2(0,T; \hat{\mathbf{E}}).$$

The following theorem holds true for both, bounded domain with Dirichlet boundary conditions and bounded domain with periodic boundary conditions, i.e. on a torus.

Theorem 4.5.1. Let $\alpha \in (1, \frac{3}{2}) \cup \{2\}$ and $u_0 \in \hat{V} \cap \mathcal{M}$. If the function β defined in (4.5.21) belongs to $L^1([0,T])$ for every T > 0, then there exists a global and locally unique solution u of (4.5.1), such that for every T > 0, $u \in X_T$.

In Subsection 4.5.1, the space X_T with more details along with a precise definition of the solution is given, and existence of a local solution is proved, together with some basic properties of the solution. In Subsection 4.5.2, the maximal solution is defined and it's existence is proved. Finally, in Subsection 4.5.3, we define the function β and establish the existence of the global solution.

Let \mathcal{O} be a bounded simply connected domain in \mathbb{R}^2 with sufficiently regular boundary Γ . We introduce the following spaces:

(4.5.2)
$$H = \left\{ u \in L^2(\mathcal{O}, \mathbb{R}^2) : \operatorname{div} u = 0, u \cdot \mathbf{n} = 0 \text{ on } \Gamma \right\},$$
$$V = H_0^1 \cap H,$$

where **n** is the outward normal to Γ .

We endow H with the scalar product and norm of L^2 and denote it by $\langle u, v \rangle_{\rm H}$, $|u|_{\rm H}$ respectively for $u, v \in {\rm H}$. We equip the space V with the scalar product and norm of H^1 and will denote it by $\langle \cdot, \cdot \rangle_{\rm V}$ and $\|\cdot\|_{\rm V}$ respectively. Let us recall the Stokes operator $A: D(A) \rightarrow H$, which is defined by

$$D(\mathbf{A}) = \mathbf{H} \cap H^2(\mathbb{R}^2),$$
$$\mathbf{A}u = -\Pi \Delta u, \quad u \in D(\mathbf{A}).$$

It is well known that A is a self adjoint positive operator in H [93, Chapter 4]. Thus, the fractional powers A^{α} exist for $\alpha \in \mathbb{R}$, and

$$D(A^{\alpha}) = [H, D(A)]_{\alpha}$$

where $[\cdot, \cdot]_{\alpha}$ is the complex interpolation functor of order α (see [58, Chapters 2 & 4]). The norms in the space $D(A^{\alpha})$ are equivalent to the norms in the space $H^{2\alpha}$. We will be using the following spaces

(4.5.3)

$$\hat{V} := D(A^{\frac{\alpha}{2}}),$$

 $\hat{H} := D(A^{\frac{\alpha-1}{2}}),$
 $\hat{E} := D(A^{\frac{\alpha+1}{2}}).$

Moreover, for $\alpha > 1$, we have the following identities (see [92])

$$D(A^{\frac{\alpha}{2}}) = H^{\alpha}(\mathscr{O}) \cap V,$$
$$D(A^{\frac{\alpha-1}{2}}) = H^{\alpha-1}(\mathscr{O}) \cap H,$$
$$D(A^{\frac{\alpha+1}{2}}) = H^{\alpha+1}(\mathscr{O}) \cap V.$$

4.5.1 Local solution : Existence and Uniqueness

In this subsection we will establish the existence of a local solution to the problem (we have taken v = 1)

(4.5.4)
$$\begin{cases} \frac{du}{dt} + Au - |\nabla u|_{L^2}^2 u + B(u, u) = 0, \\ u(0) = u_0, \end{cases}$$

with $u_0 \in \hat{V} \cap \mathcal{M}$, by using the Banach fixed point theorem. We follow the same methodology as we did while establishing the existence of a local solution to the problem (4.2.1).

The following lemma [12] along with the Gagliardo-Nirenberg inequality (4.5.6) plays a crucial role in obtaining the bounds on the non-linear terms of (4.5.4)

Lemma 4.5.2. Assume that $\gamma \in (0, \frac{1}{2})$. Then for any $s \in (1, 2]$ there exists a constant C > 0 such that

$$\|B(u,\mathbf{v})\|_{D(\mathbf{A}^{\frac{\gamma}{2}})} \leq C \|u\|_{D(\mathbf{A}^{\frac{s}{2}})} \|\mathbf{v}\|_{D(\mathbf{A}^{\frac{\gamma+1}{2}})}, \qquad u, \mathbf{v} \in \mathbf{D}(\mathbf{A}).$$

In particular,

(4.5.5)
$$\|u\nabla v\|_{H^{\gamma}} \le C \|u\|_{H^s} \|v\|_{H^{\gamma+1}}, \qquad u, v \in H^2$$

Let $u \in H^s$ for any $s \in (1,2]$ and $v \in H^1$ then there exists a positive constant C depending on s such that

$$(4.5.6) \|uv\|_{H^1} \le C \|u\|_{H^s} \|v\|_{H^1}, u \in H^s, v \in H^1.$$

In what follows we assume that D(A), V, H, \hat{E} , \hat{V} and \hat{H} are spaces as defined above. The next lemma establishing the estimates on the non-linear term arising from the constraint can be proved in the similar way as Lemma 4.2.1. Thus, we will state the lemma without the proof.

Lemma 4.5.3. Let $G_1: \hat{V} \rightarrow \hat{H}$ be defined by

$$G_1(u) = |\nabla u|_{L^2}^2 u, \qquad u \in \hat{\mathbf{V}}.$$

Then there exists C > 0 such that for $u_1, u_2 \in \hat{V}$,

$$(4.5.7) |G_1(u_1) - G_1(u_2)|_{\hat{\mathbf{H}}} \le C \|u_1 - u_2\|_{\hat{\mathbf{V}}} \Big[\|u_1\|_{\hat{\mathbf{V}}} + \|u_2\|_{\hat{\mathbf{V}}} \Big]^2.$$

Lemma 4.5.4. Let $\alpha \in (1, \frac{3}{2}) \cup \{2\}$ and $G_2 \colon \hat{E} \to \hat{H}$ be defined by

 $G_2(u) = B(u, u), \qquad u \in \hat{\mathcal{E}}.$

Then there exists $\tilde{C} > 0$ such that for $u_1, u_2 \in \hat{E}$,

$$(4.5.8) |G_2(u_1) - G_2(u_2)|_{\hat{\mathbf{H}}} \le \tilde{C} ||u_1 - u_2||_{\hat{\mathbf{V}}} (||u_1||_{\hat{\mathbf{V}}} + ||u_2||_{\hat{\mathbf{V}}})$$

Proof. Let us take $u_1, u_2 \in \hat{E}$, then

$$\begin{split} |G_{2}(u_{1}) - G_{2}(u_{2})|_{\hat{H}} &= |B(u_{1}, u_{1}) - B(u_{2}, u_{2})|_{\hat{H}} \\ &= |B(u_{1}, u_{1}) - B(u_{2}, u_{1}) + B(u_{2}, u_{1}) - B(u_{2}, u_{2})|_{\hat{H}} \\ &= |B(u_{1} - u_{2}, u_{1}) + B(u_{2}, u_{1} - u_{2})|_{\hat{H}} \\ &= |\Pi[(u_{1} - u_{2}) \cdot \nabla u_{1}] + \Pi[u_{2} \cdot \nabla(u_{1} - u_{2})]|_{\hat{H}} \\ &\leq \|(u_{1} - u_{2}) \cdot \nabla u_{1}\|_{H^{\alpha - 1}} + \|u_{2} \cdot \nabla(u_{1} - u_{2})\|_{H^{\alpha - 1}}. \end{split}$$

Now for $\alpha \in (1, \frac{3}{2})$, we use (4.5.5) with $\gamma = \alpha - 1 \in (0, \frac{1}{2})$, and for $\alpha = 2$ we use (4.5.6). Since, (4.5.5) and (4.5.6) hold for any $s \in (1, 2]$; we choose $s = \alpha$, and hence we obtain

$$\begin{aligned} \|(u_1 - u_2) \cdot \nabla u_1\|_{H^{\alpha - 1}} + \|u_2 \nabla \cdot (u_1 - u_2)\|_{H^{\alpha - 1}} &\leq C \|u_1 - u_2\|_{H^{\alpha}} \|u_1\|_{H^{\alpha}} + C \|u_2\|_{H^{\alpha}} \|u_1 - u_2\|_{H^{\alpha}} \\ &\leq C \|u_1 - u_2\|_{H^{\alpha}} (\|u_1\|_{H^{\alpha}} + \|u_2\|_{H^{\alpha}}) \\ &\leq C \|u_1 - u_2\|_{\hat{V}} (\|u_1\|_{\hat{V}} + \|u_2\|_{\hat{V}}). \end{aligned}$$

$$(4.5.9)$$

In case of $\alpha = 2$ the above inequalities take the form

$$\begin{aligned} \|(u_1 - u_2) \cdot \nabla u_1\|_{H^1} + \|u_2 \nabla \cdot (u_1 - u_2)\|_{H^1} &\leq C \|u_1 - u_2\|_{H^2} \|\nabla u_1\|_{H^1} + C \|u_2\|_{H^2} \|\nabla (u_1 - u_2)\|_{H^1} \\ &\leq C \|u_1 - u_2\|_{H^2} (\|u_1\|_{H^2} + \|u_2\|_{H^2}) \\ &\leq C \|u_1 - u_2\|_{\hat{V}} (\|u_1\|_{\hat{V}} + \|u_2\|_{\hat{V}}), \end{aligned}$$

$$(4.5.10)$$

where the last inequality holds since for $\alpha = 2$, $D(A^{\frac{\alpha}{2}}) = H^2 \cap V$.

Thus for $\alpha \in (1, \frac{3}{2}) \cup \{2\}$, from (4.5.9) and (4.5.10), we obtain

$$|G_2(u_1) - G_2(u_2)|_{\hat{\mathbf{H}}} \le C ||u_1 - u_2||_{\hat{\mathbf{V}}} (||u_1||_{\hat{\mathbf{V}}} + ||u_2||_{\hat{\mathbf{V}}}).$$

Let us recall that $\hat{E} \hookrightarrow \hat{V} \hookrightarrow \hat{H}$. Let $S = (S(t))_{t \ge 0}$ be the semigroup on \hat{H} generated by the Stokes operator. Then the following are well-known [39, 88, 90]:

A1. For every T > 0 and $f \in L^2(0, T; \hat{H})$ a function u = S * f, defined by

$$u(t) = \int_0^T S(t-r)f(r)dr \qquad t \in [0,T],$$

belongs to $X_T := \mathscr{C}([0,T];\hat{V}) \cap L^2(0,T;\hat{E})$ and

 $(4.5.11) |u|_{X_T} \le C_1 |f|_{L^2(0,T;\hat{H})},$

where $|u|_{X_T}^2 := \sup_{t \in [0,T]} ||u(t)||_{\hat{\mathbf{V}}}^2 + \int_0^T |u(t)|_{\hat{\mathbf{E}}}^2 dt$.

A2. For every T > 0 and $u_0 \in \hat{V}$ a function $u = Su_0$ defined by

$$u(t) = S(t)u_0,$$

belongs to X_T and

$$(4.5.12) |u|_{X_T} \le C_2 ||u_0||_{\hat{\mathbf{V}}}$$

Definition 4.5.5. • A solution of (4.5.4) on [0, T], $T \in [0, \infty)$ is a function $u \in X_T$ satisfying

$$u(t) = S(t)u_0 + \int_0^t S(t-r)G(u(r))dr \qquad t \in [0,T],$$

where $G: \hat{\mathbf{E}} \to \hat{\mathbf{H}}$ is defined by

$$G(u) = |\nabla u|_{L^2}^2 u - B(u, u), \qquad u \in \hat{\mathbf{E}}$$

• Let $\tau \in [0,\infty]$. A function $u \in \mathscr{C}([0,\tau), \hat{\mathbf{V}})$ is a solution to (4.5.4) on $[0,\tau)$ iff $\forall T < \tau, u|_{[0,T]} \in X_T$ is a solution of (4.5.4) on [0,T].

Now we state the main result of this section, which can be proved as Theorem 4.2.7 using Lemmas 4.5.3 - 4.5.4.

Theorem 4.5.6. Let R > 0 be given then $\exists T_* = T_*(R)$ such that for every $u_0 \in \hat{V}$ with $||u_0||_{\hat{V}} \leq R$ there exists a unique local solution $u : [0, T_*] \rightarrow \hat{V}$ of (4.5.4).

The following theorem states that the local solution of (4.5.4) stays on the manifold \mathcal{M} .

Theorem 4.5.7. If $\tau \in [0,\infty]$, $u_0 \in \mathcal{M} \cap \hat{V}$ and u is a solution to (4.5.4) on $[0,\tau)$ then $u(t) \in \mathcal{M}$ for all $t \in [0,\tau)$.

The proof of Theorem 4.5.7 is essentially the same as of Theorem 4.2.10 and hence, we have skipped it here.

4.5.2 Maximal solution

Let \mathscr{G} be a set of all local solutions, whose existence was established in Theorem 4.5.6. Let $u^1, u^2 \in \mathscr{G}$ defined on $[0, \tau_1)$ and $[0, \tau_2)$ respectively. We define an order " \leq " on \mathscr{G} by

$$u^1 \leq u^2$$
 iff $\tau_1 \leq \tau_2$ and $u^2|_{[0,\tau_1)} = u^1$

Lemma 4.5.8. If \mathcal{G} and \leq are as described above. Then, \mathcal{G} has a maximal element.

Proof. In order to show that \mathscr{G} has a maximal element we will prove that $\{\mathscr{G}, \leq\}$ is a partially ordered set (poset) and every chain in \mathscr{G} has an upper bound in \mathscr{G} .

Claim $\{\mathcal{G}, \leq\}$ is a poset.

Let $u^1, u^2, u^3 \in \mathcal{G}$ s.t. u^i is a solution of (4.5.4) on $[0, \tau_i)$, i = 1, 2, 3.

(a) \leq is reflexive.

 $u^1 \leq u^1$ since $\tau_1 = \tau_1$ and $u^1|_{[0,\tau_1)} = u^1$. Hence \leq is reflexive.

(b) \leq is anti-symmetric.

Let $u^1 \leq u^2$ and $u^2 \leq u^1$.

(4.5.13)
$$u^1 \leq u^2 \implies \tau_1 \leq \tau_2 \text{ and } u^2|_{[0,\tau_1)} = u^1$$

(4.5.14)
$$u^2 \le u^1 \implies \tau_2 \le \tau_1 \text{ and } u^1|_{[0,\tau_2)} = u^2$$

(4.5.13) and (4.5.14) $\implies \tau_1 = \tau_2$ and

$$u^1 = u^2|_{[0,\tau_1)} = u^2|_{[0,\tau_2)} = u^2$$

Hence \leq is anti-symmetric. (c) \leq is transitive. Let $u^1 \leq u^2$ and $u^2 \leq u^3$.

$$(4.5.15) u1 \le u2 \implies \tau_1 \le \tau_2 \text{ and } u2|_{[0,\tau_1)} = u1,$$

(4.5.16)
$$u^2 \leq u^3 \implies \tau_2 \leq \tau_3 \text{ and } u^3|_{[0,\tau_2)} = u^2,$$

(4.5.15) and $(4.5.16) \implies \tau_1 \le \tau_3$ and

$$u^{3}|_{[0,\tau_{1})} = u^{2}|_{[0,\tau_{1})} = u^{1}.$$

Thus $u^1 \leq u^3$. Hence \leq is transitive.

Thus from (a), (b) and (c), we conclude that $\{\mathcal{G}, \leq\}$ is a poset.

Let $u^{i_1} \leq u^{i_2} \leq u^{i_3} \leq \cdots$ be a chain in \mathscr{G} , where each u^{i_n} is a solution of (4.5.4) on $[0, \tau_{i_n})$ and $\tau_{i_1} \leq \tau_{i_2} \leq \tau_{i_3} \leq \cdots$. Also, let $\tau = \sup_{i_n, n \in \mathbb{N}} \tau_{i_n}$ and we define $u : [0, \tau) \to \hat{V}$ by

$$u(t) = u^{i_n}(t)$$
 if $t \in [0, \tau_{i_n})$.

Since $[0, \tau_{i_n}) \subset [0, \tau)$ for every $n \in \mathbb{N}$, u is the upper bound of the chain. Since each of u^{i_n} is a solution of (4.5.4), $u|_{[0,\tau)} \in X_T$ and from the definition of u it is clear that u solves (4.5.4) on $[0, \tau)$, thus $u \in \mathscr{G}$.

Hence every chain in 𝔅 has an upper bound in 𝔅. Thus by the Kuratowski-Zorn Lemma (see Lemma 2.6.12) 𝔅 has at least one maximal element. ■

Definition 4.5.9. We say that a function u is a maximal solution on $[0, \tau)$ if u is a solution to (4.5.4) on $[0, \tau)$ and if there exists another function $v \in X_T$ satisfying (4.5.4) on $[0, \tau_v)$ such that $u \leq v$ then $\tau_v = \tau$ and $u(t) = v(t) \forall t \in [0, \tau)$.

Lemma 4.5.10. Suppose *u* is a local maximal solution of (4.5.4) on $[0, \tau)$ and $\tau < \infty$. Then

$$\forall R > 0, \exists \delta > 0 : \|u(t)\|_{\hat{\mathcal{V}}} > R \text{ if } t \in (\tau - \delta, \tau).$$

Proof. We will prove the proposition by contradiction, i.e. we assume that $\exists R > 0$ such that $\forall \delta > 0, \exists t_{\delta} \in (\tau - \delta, \tau) : \|u(t_{\delta})\|_{\hat{V}} < R$.

Let us choose $\delta > 0$ such that $\delta < \frac{T_*}{2} \wedge \tau$, where T_* will be chosen later in the proof. Then for this δ , there exists a $t_{\delta} \in (\tau - \delta, \tau)$ such that $\|u(t_{\delta})\|_{\hat{V}} < R$.

Now since $||u(t_{\delta})||_{\hat{V}} < R$, then by Theorem 4.5.6, there exists $T_* = T_*(R)$ such that there exists a unique solution v of (4.5.4) on $[t_{\delta}, t_{\delta} + T_*]$ and $v(t_{\delta}) = u(t_{\delta})$. So on the common domain v = u, i.e.

$$\mathbf{v}(t) = u(t), \qquad t \in [t_{\delta}, \tau).$$

Now let us define $z : [0, \hat{\tau}] \rightarrow \hat{V}$ as follows:

$$z(t) = \begin{cases} u(t), & t \in [0, t_{\delta}] \\ v(t), & t \in [t_{\delta}, \hat{\tau}], \end{cases}$$

where $\hat{\tau} := t_{\delta} + T_*$.

Claim $u \leq z$ and $u \neq z$.

Since $[0, \tau] \subsetneq [0, \hat{\tau}]$, i.e. the domain of z is bigger than that of u and thus by the definition of z, $u \neq z$.

Now we need to show that z is a solution of (4.5.4). It is clear from Lemma 4.3.4 that $z \in X_{\hat{\tau}}$. Now it remains to show that z satisfies (4.5.17) for $t \in [0, \hat{\tau}]$.

(4.5.17)
$$z(t) = S(t)z(0) + \int_0^t S(t-r)G(z(r)) dr$$

For $t \in [0, t_{\delta}]$, *z* satisfies (4.5.17), since z(t) = u(t), $\forall t \in [0, t_{\delta}]$.

For $t \in [t_{\delta}, \hat{\tau}]$, z(t) = v(t) and since v is the solution of (4.5.4) on $[t_{\delta}, \hat{\tau}]$,

$$z(t) = \mathbf{v}(t) = S(t - t_{\delta})\mathbf{v}(t_{\delta}) + \int_{t_{\delta}}^{t} S(t - r)G(\mathbf{v}(r)) dr$$
Now since $v(t_{\delta}) = u(t_{\delta})$,

$$z(t) = S(t-t_{\delta}) \left[S(t_{\delta})u_0 + \int_0^{t_{\delta}} S(t_{\delta}-r)G(u(r))dr \right] + \int_{t_{\delta}}^t S(t-r)G(v(r))dr$$

Now using the definition of z we obtain,

$$z(t) = S(t)z(0) + \int_0^{t_{\delta}} S(t-r)G(z(r))dr + \int_{t_{\delta}}^t S(t-r)G(z(r))dr$$

= $S(t)z(0) + \int_0^t S(t-r)G(z(r))dr$.

Thus z satisfies (4.5.17) on $[0, \hat{\tau}]$ and hence z is a solution to (4.5.4) on $[0, \hat{\tau}]$.

Hence, $u \leq z$, but u is the maximal solution and thus we have the contradiction. This implies $\lim_{t \to \tau^-} \|u(t)\|_{\hat{V}} = \infty$.

4.5.3 Global solution: Existence and Uniqueness

In this subsection we will prove the existence of a global solution of (4.5.4) with $u_0 \in \mathcal{M} \cap \hat{V}$ for both, bounded domain with Dirichlet boundary conditions (see Theorem 4.5.14) and bounded domain with periodic boundary conditions, i.e. on a torus (see Corollary 4.5.15). We use the stitching argument to extend our solution from [0, T], $T < \infty$ on to the whole real line.

Let us recall the functional spaces that we will be using

$$\hat{\mathbf{V}} := \mathbf{D}(\mathbf{A}^{\frac{\alpha}{2}}),$$
$$\hat{\mathbf{H}} := \mathbf{D}(\mathbf{A}^{\frac{\alpha-1}{2}}),$$
$$\hat{\mathbf{E}} := \mathbf{D}(\mathbf{A}^{\frac{\alpha+1}{2}}).$$

Lemma 4.5.11. Let $\alpha \in (1, \frac{3}{2})$, $\theta \in (0, 1)$. Then for every $\varepsilon > 0$ there exists a constant C > 0, such that

$$(4.5.18) \qquad |\langle B(u,u), \mathbf{A}^{\alpha}u\rangle_{\mathbf{H}}| \le \varepsilon |u|_{\hat{\mathbf{E}}}^2 + \frac{C}{\varepsilon} ||u|_{\hat{\mathbf{V}}}^2 ||u|_{\mathbf{V}}^{2-2\theta} |u|_{\mathbf{E}}^{2\theta}, \qquad u \in \mathbf{D}(\mathbf{A}^{\alpha}).$$

Moreover for $\alpha = 2$, for every $\varepsilon > 0$ there exists a constant $\tilde{C} > 0$, such that

(4.5.19)
$$|\langle B(u,u), A^2 u \rangle_{\mathrm{H}}| \le \varepsilon |u|_{\hat{\mathrm{E}}}^2 + \frac{C}{\varepsilon} |u|_{\mathrm{E}}^4, \qquad u \in \mathrm{D}(\mathrm{A}^2).$$

Proof. Let $\alpha \in (1, 3/2)$ and $u \in D(A^{\alpha})$, then

$$\begin{split} |\langle B(u,u), \mathbf{A}^{\alpha}u\rangle_{\mathbf{H}}| &= |\langle \mathbf{A}^{\frac{\alpha-1}{2}}B(u,u), \mathbf{A}^{\frac{\alpha+1}{2}}u\rangle_{\mathbf{H}}| \\ &\leq |\mathbf{A}^{\frac{\alpha-1}{2}}B(u,u)|_{\mathbf{H}}|\mathbf{A}^{\frac{\alpha+1}{2}}u|_{\mathbf{H}}. \end{split}$$

Now using Lemma 4.5.2 with $\gamma = \alpha - 1$, Interpolation Theorem and the Young inequality we get for $u \in D(A^{\alpha})$

$$\begin{split} |\langle B(u,u), \mathbf{A}^{\alpha}u\rangle_{\mathbf{H}}| &\leq C \|u\|_{H^{s}} \|u\|_{\hat{\mathbf{V}}} |u|_{\hat{\mathbf{E}}} \\ &\leq C \|u\|_{H^{1}}^{1-\theta} \|u\|_{H^{2}}^{\theta} \|u\|_{\hat{\mathbf{V}}} |u|_{\hat{\mathbf{E}}} \\ &\leq \varepsilon |u|_{\hat{\mathbf{E}}}^{2} + \frac{C}{\varepsilon} \|u\|_{\hat{\mathbf{V}}}^{2} \|u\|_{\mathbf{V}}^{2-2\theta} |u|_{\mathbf{E}}^{2\theta} \,, \end{split}$$

where $\theta(s) = s - 1 \in (0, 1]$.

Let $\alpha = 2$, then for $u \in D(A^2)$ using Cauchy-Schwarz inequality we get

$$|\langle B(u,u), \mathbf{A}^2 u \rangle_{\mathbf{H}}| = |\langle \mathbf{A}^{\frac{1}{2}} B(u,u), \mathbf{A}^{\frac{3}{2}} u \rangle_{\mathbf{H}} \le ||B(u,u)||_{\mathbf{V}}|u|_{\hat{\mathbf{E}}},$$

where $\hat{E} = D(A^{\frac{3}{2}})$. Using (4.5.10) and the Young inequality, we obtain

$$|\langle B(u,u), \mathbf{A}^2 u \rangle_{\mathbf{H}}| \leq \tilde{C} |u|_{\mathbf{E}}^2 |u|_{\mathbf{E}} \leq \varepsilon |u|_{\mathbf{E}}^2 + \frac{\tilde{C}}{\varepsilon} |u|_{\mathbf{E}}^4$$

Now since for $\alpha = 2$, $\hat{V} = E$, (4.5.19) can be rewritten as

(4.5.20)
$$|\langle B(u,u), A^2 u \rangle_{\mathrm{H}}| \le \varepsilon |u|_{\hat{\mathrm{E}}}^2 + \frac{\tilde{C}}{\varepsilon} |u|_{\hat{\mathrm{V}}}^4.$$

Lemma 4.5.12. Let $\alpha \in (1, \frac{3}{2})$ and u be a maximal solution of (4.5.4) on $[0, \tau)$ with $u_0 \in \mathcal{M} \cap \hat{V}$. If $\varepsilon \in (0, 1], \theta \in (0, 1)$, then there exists a C > 0 such that

$$\|u(t)\|_{\hat{V}}^{2} \leq \|u_{0}\|_{\hat{V}}^{2} + \|u_{0}\|_{\hat{V}}^{2} \int_{0}^{t} \beta(s) e^{\int_{s}^{t} \beta(r) dr} ds, \quad t \in [0, \tau),$$

where

(4.5.21)
$$\beta(s) = 2|\nabla u(s)|_{L^2}^2 + \frac{2C}{\varepsilon} ||u(s)|_{V}^{2-2\theta} |u(s)|_{E}^{2\theta}, \quad s \in [0,\tau).$$

Proof. Let $\alpha \in (1, \frac{3}{2})$, $u_0 \in \mathcal{M} \cap V$ and u be the maximal solution of (4.5.4) on $[0, \tau)$. Then using (4.5.4), we have

$$\begin{split} \frac{1}{2} \|u(t)\|_{\hat{V}}^2 &= \frac{1}{2} \|u_0\|_{\hat{V}}^2 + \int_0^t \langle u'(s), \mathbf{A}^{\alpha} u(s) \rangle_{\mathbf{H}} ds \\ &= \frac{1}{2} \|u_0\|_{\hat{V}}^2 + \int_0^t \langle |\nabla u(s)|_{L^2}^2 u(s) - \mathbf{A}u(s) - B(u(s), u(s)), \mathbf{A}^{\alpha} u(s) \rangle_{\mathbf{H}} ds \\ &= \frac{1}{2} \|u_0\|_{\hat{V}}^2 + \int_0^t \langle |\nabla u(s)|_{L^2}^2 u(s), \mathbf{A}^{\alpha} u(s) \rangle_{\mathbf{H}} ds - \int_0^t \langle \mathbf{A}u(s), \mathbf{A}^{\alpha} u(s) \rangle_{\mathbf{H}} ds \\ &- \int_0^t \langle B(u(s), u(s)), \mathbf{A}^{\alpha} u(s) \rangle_{\mathbf{H}} ds \\ &= \frac{1}{2} \|u_0\|_{\hat{V}}^2 + \int_0^t |\nabla u(s)|_{L^2}^2 \langle u(s), \mathbf{A}^{\alpha} u(s) \rangle_{\mathbf{H}} ds - \int_0^t \langle \mathbf{A}^{\frac{\alpha+1}{2}} u(s), \mathbf{A}^{\frac{\alpha+1}{2}} u(s) \rangle_{\mathbf{H}} ds \\ &- \int_0^t \langle B(u(s), u(s)), \mathbf{A}^{\alpha} u(s) \rangle_{\mathbf{H}} ds . \end{split}$$

On rearranging we get

$$\frac{1}{2}\|u(t)\|_{\hat{\mathcal{V}}}^2 + \int_0^t |u(s)|_{\hat{\mathcal{E}}}^2 ds = \frac{1}{2}\|u_0\|_{\hat{\mathcal{V}}}^2 + \int_0^t |\nabla u(s)|_{L^2}^2 \|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds - \int_0^t \langle B(u(s), u(s)), \mathcal{A}^{\alpha}u(s)\rangle_{\mathcal{H}} ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds + \int_0^t |u(s)|_{\hat{\mathcal{V}}}^2 ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds + \int_0^t |u(s)|_{\hat{\mathcal{V}}}^2 ds = \frac{1}{2}\|u(s)\|_{\hat{\mathcal{V}}}^2 ds + \int_0^t |u(s)|_{\hat{\mathcal{V}}}^2 ds +$$

Using the Cauchy-Schwarz inequality, the Young inequality and Lemma 4.5.11 for every $\varepsilon \in (0, 1]$ there exists a C > 0 such that for every $\theta \in (0, 1)$

$$\begin{split} \frac{1}{2} \|u(t)\|_{\hat{\mathcal{V}}}^2 &+ \int_0^t |u(s)|_{\hat{\mathcal{E}}}^2 \, ds \leq \frac{1}{2} \|u_0\|_{\hat{\mathcal{V}}}^2 + \int_0^t |\nabla u(s)|^2 \|u(s)\|_{\hat{\mathcal{V}}}^2 \, ds + \int_0^t |\langle B(u(s), u(s)), \mathcal{A}^{\alpha} u(s) \rangle_{\mathcal{H}} | \, ds \\ &\leq \frac{1}{2} \|u_0\|_{\hat{\mathcal{V}}}^2 + \int_0^t |\nabla u(s)|_{L^2}^2 \|u(s)\|_{\hat{\mathcal{V}}}^2 \, ds + \int_0^t \left[\varepsilon |u(s)|_{\hat{\mathcal{E}}}^2 + \frac{C}{\varepsilon} \|u(s)\|_{\hat{\mathcal{V}}}^2 \|u(s)\|_{\mathcal{V}}^{2-2\theta} |u(s)|_{\mathcal{E}}^{2\theta} \right] \, ds \, . \end{split}$$

Thus on rearranging we get

$$\frac{1}{2}\|u(t)\|_{\hat{\mathbf{V}}}^2 + (1-\varepsilon)\int_0^t |u(s)|_{\hat{\mathbf{E}}}^2 ds \le \frac{1}{2}\|u_0\|_{\hat{\mathbf{V}}}^2 + \int_0^t \left[|\nabla u(s)|_{L^2}^2 + \frac{C}{\varepsilon}\|u(s)\|_{\mathbf{V}}^{2-2\theta}|u(s)|_{\mathbf{E}}^2\right]\|u(s)\|_{\hat{\mathbf{V}}}^2 ds.$$

In particular

(4.5.22)
$$\frac{1}{2} \|u(t)\|_{\hat{V}}^2 \leq \frac{1}{2} \|u_0\|_{\hat{V}}^2 + \int_0^t \left[|\nabla u(s)|_{L^2}^2 + \frac{C}{\varepsilon} \|u(s)\|_{V}^{2-2\theta} |u(s)|_{E}^{2\theta} \right] \|u(s)\|_{\hat{V}}^2 ds.$$

Now we apply the Gronwall Lemma in integral form with

$$\beta(s) = 2|\nabla u(s)|_{L^2}^2 + \frac{2C}{\varepsilon} ||u(s)|_{V}^{2-2\theta} |u(s)|_{E}^{2\theta}, \quad s \in [0, t],$$

to obtain

(4.5.23)
$$\|u(t)\|_{\hat{V}}^2 \le \|u_0\|_{\hat{V}}^2 + \|u_0\|_{\hat{V}}^2 \int_0^t \beta(s) e^{\int_s^t \beta(r) dr} ds$$

Remark 4.5.13. Now for $\alpha = 2$, using (4.5.19), equation (4.5.22) transforms to

(4.5.24)
$$\frac{1}{2}|u(t)|_{\mathrm{E}}^{2} \leq \frac{1}{2}|u_{0}|_{\mathrm{E}}^{2} + \int_{0}^{t} \left[|\nabla u(s)|_{L^{2}}^{2} + \frac{C}{\varepsilon}|u(s)|_{\mathrm{E}}^{2}\right]|u(s)|_{\mathrm{E}}^{2} ds.$$

Thus, for every $\varepsilon \in (0, 1]$, Lemma 4.5.12 holds true for

$$\beta(s) = 2|\nabla u(s)|_{L^2}^2 + \frac{2C}{\varepsilon}|u(s)|_{\rm E}^2, \quad s \in [0,\tau).$$

Theorem 4.5.14. Let u be a local maximal solution of (4.5.4) on $[0, \tau)$ with $\tau < \infty$, $u_0 \in \mathcal{M} \cap \hat{V}$. If the function β defined in (4.5.21) belongs to $L^1([0, \tau))$, then $\tau = \infty$.

Proof. We will prove the theorem by contradiction. Assume $\tau < \infty$, then by Lemma 4.5.10

$$\lim_{t\to\tau^-}\|u(t)\|_{\hat{\mathbf{V}}}^2=\infty.$$

But on the other hand, if $\beta \in L^1([0, \tau))$ then by Lemma 4.5.12 there exists some K > 0 such that

$$\sup_{t \in [0,\tau)} \|u(t)\|_{\hat{v}}^2 < K$$
,

which is a contradiction. Thus we infer that $\tau = \infty$.

Corollary 4.5.15. Let \mathcal{O} be a bounded periodic domain, $\alpha \in (1, \frac{3}{2}) \cup \{2\}$ and $u_0 \in \mathcal{M} \cap D(A^{\frac{\alpha}{2}})$. Then there exists a unique global solution u of (4.5.4) such that for every T > 0, $u \in X_T$.

Proof. Let $u_0 \in \mathcal{M} \cap D(A^{\frac{\alpha}{2}})$. It is enough to show that for $\tau < \infty$, the function β defined by (4.5.21) belongs to $L^1([0,\tau)$, in case of a periodic bounded domain. Now by Theorem 4.1.1, for every $\tau > 0$, $u \in \mathscr{C}([0,\tau]; V) \cap L^2(0,\tau; E)$ and thus β belongs to $L^1([0,\tau))$. Hence from Theorem 4.5.14, we have a unique global solution of (4.5.4).

4.6 Lower bound on the regularity of the initial data

This section is dedicated to finding a lower bound on the regularity of the initial data u_0 such that problem (4.5.4) has a local solution. The analysis carried out in this section is on a formal level and some of the details need to be verified hence, remains an open problem.

In the previous sections we have established the existence of a local solution for any general bounded domain $\mathcal{O} \subset \mathbb{R}^2$ and hence we have focussed on such a case here too. We are interested in the existence of a local solution of the following problem:

(4.6.1)
$$\begin{cases} \frac{du}{dt} + Au = |\nabla u|_{L^2}^2 u - B(u, u) \\ u(0) = u_0 \in \hat{\mathbf{V}} \cap \mathcal{M}, \end{cases}$$

where the Stokes operator A, the bilinear map *B*, the manifold \mathcal{M} and the functional space $\hat{V} = D(A^{\frac{\alpha}{2}})$ are understood as in the previous section. After establishing the existence for $\alpha \in [1, \frac{3}{2}) \cup \{2\}$ we are specifically interested in $\alpha \in (0, 1)$.

If the initial data $u_0 \in \hat{V}$, then according to the maximal regularity principle [39], the solution u of the parabolic equation (4.6.1) should belong to $X_T := \mathscr{C}([0,T];\hat{V}) \cap L^2(0,T;\hat{E})$, which in turn is possible only if the R.H.S. of equation (4.6.1) belongs to $L^2(0,T;\hat{H})$.

Since we are interested in finding a lower bound on α we will take a bottom to top approach, i.e. we will assume that u, the solution of (4.6.1) belongs to X_T and then obtain bounds on α for which each term in the R.H.S. of (4.6.1) belongs to $L^2(0,T;\hat{H})$.

Lemma 4.6.1. Let $\alpha \in (0,1)$ and $u \in \hat{E}$. Then, there exists a constant C > 0 such that

$$(4.6.2) |B(u,u)|_{\hat{\mathbf{H}}} \le C \|u\|_{\hat{\mathbf{V}}} |u|_{\hat{\mathbf{E}}}$$

In particular, if $u \in X_T$ then for every $\alpha \in (0,1)$, $B(u,u) \in L^2(0,T;\hat{H})$.

Proof. Let $\alpha > 0$ and $\varphi \in H^{1-\alpha}$, then

$$|b(u,u,\varphi)| = |\langle B(u,u),\varphi\rangle| = \left| \int_{\mathscr{O}} (u(x) \cdot \nabla) u(x)\varphi(x) dx \right|,$$

$$(4.6.3) \leq |u|_{L^{\frac{2}{1-\alpha}}} |\nabla u|_{L^{2}} |\varphi|_{L^{\frac{2}{\alpha}}},$$

where $\langle \cdot, \cdot \rangle$ denotes the duality between $H^{\alpha-1}$ and $H^{1-\alpha}$ and we have used the Hölder inequality to obtain the last relation.

Using the Sobolev embedding theorem for a bounded domain in \mathbb{R}^2 , we have

$$H^{1-lpha} \subset L^{rac{2}{lpha}} \quad ext{and} \quad H^{lpha} \subset L^{rac{2}{1-lpha}},$$

where $\alpha \in (0, 1)$. Thus (4.6.3) can be rewritten as

$$\begin{split} |b(u, u, \varphi)| &\leq C \|u\|_{H^{\alpha}} |\nabla u|_{L^{2}} |\varphi|_{H^{1-\alpha}} \\ &\leq C \|u\|_{H^{\alpha}} \|u\|_{V} |\varphi|_{H^{1-\alpha}} \,, \end{split}$$

where C > 0 is a generic constant. Since $\varphi \in H^{1-\alpha}$ and $\hat{\mathbf{E}} \subset \mathbf{V}$, we have

$$(4.6.4) |B(u,u)|_{\hat{H}} \le C ||u||_{\hat{V}} |u|_{\hat{E}},$$

where we have used the definition of functional spaces from (4.5.3).

Using (4.6.4) and the Hölder inequality, we get

$$\begin{split} \int_{0}^{T} |B(u(t), u(t))|_{\hat{H}}^{2} dt &\leq C \int_{0}^{T} |u(t)|_{\hat{E}}^{2} ||u(t)|_{\hat{V}}^{2} dt \\ &\leq C \sup_{t \in [0, T]} ||u(t)||_{\hat{V}}^{2} \int_{0}^{T} |u(t)|_{\hat{E}}^{2} dt \\ &\leq C |u|_{X_{T}}^{2} |u|_{X_{T}}^{2}. \end{split}$$

Since $u \in X_T$, we infer that $B(u, u) \in L^2(0, T; \hat{H})$.

Lemma 4.6.2. Let $u \in X_T$, then for $\alpha \in [\frac{1}{2}, 1]$, $|\nabla u|_{L^2}^2 u \in L^2(0, T; \hat{H})$.

Proof. Let $u \in X_T$. Note that for $\alpha \in (0, 1)$ we have the following inclusion of functional spaces $\hat{V} \subset H \subset \hat{H}$. Then by the Hölder inequality, we have

(4.6.5)
$$\begin{aligned} \left| |\nabla u|_{L^2}^2 u \right|_{L^2(0,T;\hat{H})}^2 &= \int_0^T |\nabla u(t)|_{L^2}^4 |u(t)|_{\hat{H}}^2 dt \\ &\leq \sup_{t \in [0,T]} \|u(t)\|_{\hat{V}}^2 \int_0^T |\nabla u(t)|_{L^2}^4 dt \end{aligned}$$

Using the interpolation between \hat{V} and \hat{H} , for every $v \in \hat{V}$ there exists a constant C > 0 such that

$$|\mathbf{v}|_{\mathbf{H}} \leq C |\mathbf{v}|_{\hat{\mathbf{H}}}^{\alpha} \|\mathbf{v}\|_{\hat{\mathbf{V}}}^{1-\alpha}.$$

Therefore on using the above relation in (4.6.5), we obtain

$$\begin{split} \left| |\nabla u|_{L^{2}}^{2} u \right|_{L^{2}(0,T;\hat{H})}^{2} &\leq C \sup_{t \in [0,T]} \|u(t)\|_{\hat{V}}^{2} \int_{0}^{T} |\nabla u(t)|_{\hat{H}}^{4\alpha} \|\nabla u(t)\|_{\hat{V}}^{4(1-\alpha)} dt \\ &\leq C |u|_{X_{T}}^{2} \int_{0}^{T} \|u(t)\|_{\hat{V}}^{4\alpha} |u(t)|_{\hat{E}}^{4(1-\alpha)} dt \leq C |u|_{X_{T}}^{2} \sup_{t \in [0,T]} \|u(t)\|_{\hat{V}}^{4\alpha} \int_{0}^{T} |u(t)|_{\hat{E}}^{4(1-\alpha)} dt \\ &\leq C |u|_{X_{T}}^{2} |u|_{X_{T}}^{4\alpha} \left[\int_{0}^{T} |u(t)|_{\hat{E}}^{2} dt \right]^{2(1-\alpha)} \left[\int_{0}^{T} dt \right]^{2\alpha-1} \leq C |u|_{X_{T}}^{2(1+2\alpha)} |u|_{X_{T}}^{4(1-\alpha)} T^{2\alpha-1} \\ &\leq C |u|_{X_{T}}^{2} |u|_{X_{T}}^{4\alpha} \left[\int_{0}^{T} |u(t)|_{\hat{E}}^{2} dt \right]^{2(1-\alpha)} \left[\int_{0}^{T} dt \right]^{2\alpha-1} \leq C |u|_{X_{T}}^{2(1+2\alpha)} |u|_{X_{T}}^{4(1-\alpha)} T^{2\alpha-1} \end{split}$$

Thus, we infer that for $\alpha \in \left[\frac{1}{2}, 1\right)$, $|\nabla u|_{L^2}^2 u \in L^2(0, T; \hat{H})$.

77

Hence, from Lemmas 4.6.1 - 4.6.2 and maximal regularity principle we can conclude that the minimum regularity required for initial data u_0 such that the problem (4.6.1) has a local solution is $D(A^{1/4})$.

As mentioned at the beginning of this section, the analysis carried out here is at a formal level, thus one needs to make sure that this holds rigorously too.



STOCHASTIC CONSTRAINED NAVIER-STOKES EQUATIONS

Stokes equations which were introduced in Chapter 3. In this chapter we study constrained Navier-Stokes equations driven by a multiplicative Gaussian noise in the Stratonovich form. In the deterministic case (see Chapter 4) we showed the existence of the global solution on a two dimensional torus and hence we concentrate on such a case here. We prove the existence of a martingale solution and later using Schmalfuss idea [82] we show the pathwise uniqueness of the solutions. We also establish the existence of a strong solution using results from Ondreját [68].

5.1 Introduction

In this chapter we consider the stochastic Navier-Stokes equations

(5.1.1)
$$du + [(u \cdot \nabla)u - v\Delta u + \nabla p] dt = \sum_{j=1}^{m} (c_j \cdot \nabla)u \circ dW_j(t), \quad t \in [0,\infty)$$

in $\mathcal{O} = [0, 2\pi]^2$ with periodic boundary conditions and with the incompressibility condition

$$\operatorname{div} u = 0.$$

This problem can be seen as a problem on a two-dimensional torus \mathbb{T}^2 what we will assume to be our case. Here $u : [0,\infty) \times \mathcal{O} \to \mathbb{R}^2$ and $p : [0,\infty) \times \mathcal{O} \to \mathbb{R}$ represent the velocity and the pressure of the fluid. Furthermore $\sum_{j=1}^{m} (c_j \cdot \nabla) u \circ dW_j(t)$ stands for the random forcing, where c_j , $j = 1, \dots, m$, are divergence free \mathbb{R}^2 -valued vectors (so that the corresponding transport operators $\tilde{C}_j u := (c_j \cdot \nabla) u$ are skew symmetric in $L^2(\mathbb{T}^2, \mathbb{R}^2)$) and W_j , $j = 1, \dots, m$ are independent \mathbb{R} -valued standard Brownian Motions. The above problem projected on $H \cap \mathcal{M}$ can be written in an abstract form as the following initial value problem

(5.1.2)
$$\begin{cases} du(t) + vAu(t)dt + B(u(t))dt = v|\nabla u(t)|_{L^2}^2 u(t)dt + \sum_{j=1}^m C_j u(t) \circ dW_j(t), t \in [0,T], \\ u(0) = u_0, \end{cases}$$

where H is the space of square integrable, divergence free and mean zero vector fields on $\mathcal O$ and

$$\mathcal{M} = \{ u \in \mathbf{H} : |u|_{\mathbf{H}} = 1 \}.$$

Here A and *B* are appropriate maps corresponding to the Laplacian and the nonlinear term, respectively in the Navier-Stokes equations, see Chapter 3 and $C_j = \Pi(\tilde{C}_j)$, where $\Pi: L^2(\mathbb{T}^2, \mathbb{R}^2) \to H$ is the Leray-Helmholtz projection operator [88] that projects the square integrable vector fields onto the divergence free vector field.

We prove the existence and uniqueness of a strong solution. The construction of a solution is based on the classical Faedo-Galerkin approximation, i.e.

(5.1.3)
$$\begin{cases} du_n(t) = -\left[P_n A u_n(t) + P_n B(u_n(t)) - |\nabla u_n(t)|_{L^2}^2 u_n(t)\right] dt \\ + \sum_{j=1}^m P_n C_j u_n(t) \circ dW_j(t), \quad t \in [0, T], \\ u_n(0) = \frac{P_n u_0}{|P_n u_0|} \end{cases}$$

given in Section 5.5. Let us point out that without the normalisation of the initial condition in the above problem (5.1.3), the solution may not be a global one, even in the deterministic case. The crucial point is to prove suitable uniform a'priori estimates on the sequence u_n . We will prove that the following estimates hold

$$\sup_{n\geq 1} \mathbb{E}\left[\int_0^T |u_n(s)|^2_{\mathrm{D}(\mathrm{A})} ds\right] < \infty,$$

and

$$\sup_{n\geq 1} \mathbb{E}\left(\sup_{0\leq s\leq T} \|u_n(s)\|_{\mathbf{V}}^{2p}\right) < \infty,$$

for $p \in [1, 1 + \frac{1}{K_c^2})$, where D(A) is the domain of the Stokes operator and V = D(A^{1/2}), see Section 3.1 for precise definitions and the positive constant K_c is defined in (5.3.1).

In Theorem 5.3.3 we prove the existence of a martingale solution using the tightness criterion in the topological space $\mathcal{Z}_T = \mathscr{C}([0,T];\mathrm{H}) \cap L^2_\mathrm{w}(0,T;D(\mathrm{A})) \cap L^2(0,T;\mathrm{V}) \cap \mathscr{C}([0,T];\mathrm{V}_\mathrm{w})$ showing that the trajectories of the solution lie in $\mathscr{C}([0,T];\mathrm{V}_\mathrm{w})$ but later on in Lemma 5.3.5 we show that in fact the trajectories lie in $\mathscr{C}([0,T];\mathrm{V})$.

This chapter is an extension of Chapter 4 from the deterministic to a stochastic setting. More information and motivation can also be found therein. Let us recall that already in the deterministic setting, we have been able to prove the global existence of solutions for CNSE only on a bounded domain with periodic boundary conditions and this is why we have concentrated here on such a case. A similar problem for stochastic heat equation with polynomial drift but with a different type of noise has recently been a subject of a PhD thesis by Javed Hussain [48]. It's remarkable that in that case the result holds for Dirichlet boundary conditions as well.

We consider the noise of gradient type in the Stratonovich form (5.1.1). The structure of noise is such that it is tangent to the manifold \mathcal{M} just like the non-linear part from Navier-Stokes and hence, there is no contribution to the equation (5.1.2) because of the constraint. In the deterministic setting (see Chapter 4) we proved the existence of a global solution by proving the existence of a local solution using the Banach Fixed Point Theorem and no explosion principle, i.e enstrophy (V-norm) of the solution remains bounded. We can't take the similar approach in the stochastic setting as one can't prove the existence of a local solution using the Banach Fixed Point Theorem and hence we switch to more classical approach of proving the existence of a solution using the Faedo-Galerkin approximation.

We consider the Faedo-Galerkin approximation (5.1.3) of (5.1.2). We prove that each approximating equation has a global solution. One can show that for every $n \in \mathbb{N}$ global solution to (5.1.3) exist for all domains, in particular for Dirichlet boundary conditions. But in order to obtain a'priori estimates, Lemma 5.5.4, we need to consider the Navier-Stokes Equations (NSE) on a two dimensional torus \mathbb{T}^2 (i.e. the NSEs with periodic boundary conditions).

In order to prove that the laws of the solution of these approximating equations are tight on \mathcal{Z}_T (defined in (5.4.3)), apart from a'priori estimates we also need the Aldous condition, Definition 2.9.10. After proving that the laws are tight in Lemma 5.5.5, by the application of Jakubowski-Skorohod Theorem and the martingale representation theorem we prove Theorem 5.3.3. The chapter is organised in the following way:

Stochastic constrained Navier-Stokes equations (SCNSE) are introduced in Section 5.2. The definitions of a martingale solution and strong solution and all the important results of this chapter are given in Section 5.3. Section 5.4 contains the well-known and already established results regarding compactness. In Section 5.5 we establish certain estimates on the way to prove Theorem 5.3.3. Existence and uniqueness of a strong solution using the results from Ondreját [68] is proved in Section 5.6. In Section 5.7, we prove the continuous dependence of the solution of (5.1.2) on the initial data. We conclude the chapter by showing that the semigroup $\{T_t\}_{t\geq 0}$ on $\mathscr{B}_b(V)$ (defined by (5.8.1)) generated by the solution of SCNSE are sequentially weakly Feller in V.

5.2 Stochastic constrained Navier-Stokes equations

The 2D Navier-Stokes equations driven by multiplicative Gaussian noise in the Stratonovich form are given by:

(5.2.1)
$$\begin{cases} du(t) + [(u(t) \cdot \nabla)u(t) - v \Delta u(t) + \nabla p(t)] dt = \sum_{j=1}^{m} \left[(c_j \cdot \nabla) u(t) \right] \circ dW_j(t), & t \in [0, \infty), \\ \text{div} u(\cdot, t) = 0, \ t \in [0, \infty), \\ u(x, 0) = u_0(x), \ x \in \mathcal{O}, \end{cases}$$

where $u: [0,\infty) \times \mathcal{O} \to \mathbb{R}^2$ and $p: [0,\infty) \times \mathcal{O} \to \mathbb{R}$ are velocity and pressure of the fluid respectively. v is the viscosity of the fluid (with no loss of generality, v will be taken equal to 1 for the rest of the article). Here we assume that c_j are divergence free \mathbb{R}^2 -valued vectors, W_j are \mathbb{R} -valued i.i.d. standard Brownian motions and \circ denotes the Stratonovich form. Note that the operators \tilde{C}_j , $j \in \{1, \ldots, m\}$, defined by $\tilde{C}_j u := (c_j \cdot \nabla) u$, for $u \in V$ are skew-symmetric on $L^2(\mathbb{T}^2, \mathbb{R}^2)$, i.e. $\tilde{C}_j^* = -\tilde{C}_j$, where \tilde{C}_j^* denotes the adjoint of \tilde{C}_j on $L^2(\mathbb{T}^2, \mathbb{R}^2)$.

We will be frequently using the following short-cut notation

$$Cu \circ dW(t) = \sum_{j=1}^{m} C_j u(t) \circ dW_j(t),$$

where $C_j = \prod(\tilde{C}_j)$ and \prod is the Leray-Helmholtz projection operator, as introduced in Chapter 3.

With all the notations as defined above and in Chapter 3, the stochastic Navier-Stokes equations (5.2.1) projected on divergence free vector field is given by

(5.2.2)
$$\begin{cases} du(t) + [Au(t) + B(u(t))] dt = Cu(t) \circ dW(t), \\ u(0) = u_0. \end{cases}$$

Let us recall the orthogonal projection map $\pi_u : H \to T_u \mathcal{M}$, which is given by

$$\pi_u(\mathbf{v}) = \mathbf{v} - \langle \mathbf{v}, u \rangle_{\mathrm{H}} u, \quad \text{for } u \in \mathcal{M},$$

where $T_u \mathcal{M}$ is the tangent space corresponding to the manifold \mathcal{M} as introduced earlier in Chapter 3.

Since for every $j \in \{1, ..., m\}$, $C_j^* = -C_j$ in H we infer that

$$(5.2.3) \qquad \langle C_j u, u \rangle_{\mathrm{H}} = 0, \quad u \in \mathrm{V}, \ j \in \{1, \dots, m\}.$$

In particular, if $u \in V \cap \mathcal{M}$, then $C_j u \in T_u \mathcal{M}$ for every $j \in \{1, ..., m\}$ and hence won't produce any correction terms when projected onto the tangent space $T_u \mathcal{M}$, which is shown explicitly below.

Let

$$F(u) = Au + B(u, u) - Cu \circ dW(t)$$

and $\hat{F}(u)$ be the projection of F(u) onto the tangent space $T_u \mathcal{M}$, then

$$\begin{split} \hat{F}(u) &= \pi_u(F(u)) = F(u) - \langle F(u), u \rangle_{\mathrm{H}} u \\ &= \mathrm{A}u + B(u) - Cu \circ dW - \langle \mathrm{A}u + B(u) - Cu \circ dW, u \rangle_{\mathrm{H}} u \\ &= \mathrm{A}u - \langle \mathrm{A}u, u \rangle_{\mathrm{H}} u + B(u) - \langle B(u), u \rangle_{\mathrm{H}} u - Cu \circ dW + \langle Cu, u \rangle_{\mathrm{H}} u \circ dW \\ &= \mathrm{A}u - |\nabla u|_{L^2}^2 u + B(u) - Cu \circ dW. \end{split}$$

The last equality follows from (5.2.3) and the identity that $\langle B(u), u \rangle_{\rm H} = 0$.

Thus by projecting NSEs (5.2.2) onto the tangent space $T_u \mathcal{M}$, we obtain the following stochastic constrained Navier-Stokes equations (SCNSE)

(5.2.4)
$$\begin{cases} du(t) + [Au(t) + B(u(t))] dt = |\nabla u(t)|_{L^2}^2 u(t) dt + Cu(t) \circ dW(t), \\ u(0) = u_0. \end{cases}$$

5.3 Assumptions, definitions and results

From now on we will assume that c_j are constant vector fields. Whether our results are true in a more general setting is an open problem.

Assumptions. We assume that

(A.1) Vectors c_1, \ldots, c_m belong to \mathbb{R}^2 such that $K_c^2 < 1$ where

(5.3.1) $K_c := \max_{j \in \{1, \cdots, m\}} |c_j|_{\mathbb{R}^2},$

 $|\cdot|_{\mathbb{R}^2}$ is the Euclidean norm in \mathbb{R}^2 .

(A.2) $u_0 \in V \cap \mathcal{M}$.

Definition 5.3.1. We say that problem (5.2.4) has a strong solution iff for every stochastic basis $(\Omega, \mathscr{F}, \mathbb{F}, \mathbb{P})$ and every \mathbb{R}^m – valued \mathbb{F} –Wiener process $W = (W(t))_{t \ge 0}$, there exists a \mathbb{F} –progressively measurable process $u : [0, T] \times \Omega \rightarrow D(A)$ with \mathbb{P} -a.e. paths

$$u(\cdot,\omega) \in \mathscr{C}([0,T]; \mathbf{V}) \cap L^2(0,T; \mathbf{D}(\mathbf{A})),$$

such that for all $t \in [0, T]$ and all $v \in V \mathbb{P}$ -a.s.

(5.3.2)
$$\langle u(t), \mathbf{v} \rangle - \langle u_0, \mathbf{v} \rangle + \int_0^t \langle \mathbf{A}u(s), \mathbf{v} \rangle \, ds + \int_0^t \langle B(u(s)), \mathbf{v} \rangle \, ds$$
$$= \int_0^t |\nabla u(s)|_{L^2}^2 \langle u(s), \mathbf{v} \rangle \, ds + \frac{1}{2} \int_0^t \sum_{j=1}^m \langle C_j^2 u(s), \mathbf{v} \rangle \, ds + \int_0^t \sum_{j=1}^m \langle C_j u(s), \mathbf{v} \rangle \, d\hat{W}_j(s).$$

Definition 5.3.2. We say that there exists a martingale solution of (5.2.4) iff there exist

• a stochastic basis $(\hat{\Omega}, \hat{\mathscr{F}}, \hat{\mathbb{F}}, \hat{\mathbb{P}})$,

- an \mathbb{R}^m -valued $\hat{\mathbb{F}}$ -Wiener process \hat{W} ,
- and a $\hat{\mathbb{F}}$ -progressively measurable process $u: [0,T] \times \hat{\Omega} \to D(A)$ with $\hat{\mathbb{P}}$ -a.e. paths

$$u(\cdot,\omega) \in \mathscr{C}([0,T]; \mathbf{V}_w) \cap L^2(0,T; \mathbf{D}(\mathbf{A})),$$

such that for all $t \in [0, T]$ and all $v \in V$ the identity (5.3.2) holds $\hat{\mathbb{P}}$ -a.s.

Next we state some important results of this chapter which will be proved in further sections.

Theorem 5.3.3. Let assumptions (A.1) - (A.2) be satisfied. Then there exists a martingale solution $(\hat{\Omega}, \hat{\mathscr{F}}, \hat{\mathbb{P}}, \hat{\mathbb{W}}, u)$ of problem (5.2.4) such that

(5.3.3)
$$\hat{\mathbb{E}}\left[\sup_{t\in[0,T]}\|u(t)\|_{\mathrm{V}}^{2} + \int_{0}^{T}|u(t)|_{\mathrm{D}(\mathrm{A})}^{2}dt\right] < \infty.$$

Remark 5.3.4. The solution obtained in the above theorem is weak in probabilistic sense and strong in PDE sense.

The next lemma shows that almost all the trajectories of the solution obtained in Theorem 5.3.3 are almost everywhere equal to a continuous V-valued function defined on [0, T].

Lemma 5.3.5. Assume that the assumptions (A.1) - (A.2) are satisfied. Let $(\hat{\Omega}, \hat{\mathscr{F}}, \hat{\mathbb{F}}, \hat{\mathbb{P}}, \hat{\mathbb{W}}, u)$ be a martingale solution of (5.2.4) such that

(5.3.4)
$$\hat{\mathbb{E}}\left[\sup_{t\in[0,T]}\|u(t)\|_{\mathrm{V}}^{2} + \int_{0}^{T}|u(s)|_{\mathrm{D}(\mathrm{A})}^{2}ds\right] < \infty.$$

Then for $\hat{\mathbb{P}}$ almost all $\omega \in \hat{\Omega}$ the trajectory $u(\cdot, \omega)$ is almost everywhere equal to a continuous V-valued function defined on [0, T]. Moreover, the following equality in H holds for every $t \in [0, T]$, $\hat{\mathbb{P}}$ -a.s.

(5.3.5)
$$u(t) = u_0 - \int_0^t \left[Au(s) + B(u(s)) - |\nabla u(s)|_{L^2}^2 u(s) \right] ds + \frac{1}{2} \int_0^t \sum_{j=1}^m C_j^2 u(s) ds + \int_0^t \sum_{j=1}^m C_j u(s) d\hat{W}(s).$$

Definition 5.3.6. Let $(\Omega, \mathscr{F}, \mathbb{F}, \mathbb{P}, W, u^i)$, i = 1, 2 be the martingale solutions of (5.2.4) with $u^i(0) = u_0$, i = 1, 2. Then we say that the solutions are pathwise unique if for all $t \in [0, T]$, \mathbb{P} -a.s. $u^1(t) = u^2(t)$.

In Lemma 5.6.1 we will show that the pathwise uniqueness property for our problem holds. This will enable us to deduce the following theorem that summarises the main result of this chapter: **Theorem 5.3.7.** For every $u_0 \in V$ there exists a pathwise unique strong solution u of stochastic constrained Navier-Stokes equations (5.2.4) such that

(5.3.6)
$$\mathbb{E}\left[\int_{0}^{T} |u(t)|^{2}_{\mathrm{D}(\mathrm{A})} dt + \sup_{t \in [0,T]} ||u(t)||^{2}_{\mathrm{V}}\right] < \infty.$$

Remark 5.3.8. The solution of (5.2.4) obtained in previous theorem is strong in both probabilistic and PDE sense.

5.4 Compactness

Let us consider the following functional spaces for fixed T > 0: $\mathscr{C}([0,T];H) :=$ the space of continuous functions $u : [0,T] \to H$ with the topology \mathscr{T}_1 induced by the norm $|u|_{\mathscr{C}([0,T];H)} := \sup_{t \in [0,T]} |u(t)|_H$,

 $L^2_{\mathrm{w}}(0,T;\mathrm{D}(\mathrm{A})) :=$ the space $L^2(0,T;\mathrm{D}(\mathrm{A}))$ with the weak topology \mathscr{T}_2 , $L^2(0,T;\mathrm{V}) :=$ the space of measurable functions $u : [0,T] \to \mathrm{V}$ such that

$$|u|_{L^{2}(0,T;V)} = \left(\int_{0}^{T} ||u(t)||_{V}^{2} dt\right)^{\frac{1}{2}} < \infty,$$

with the topology \mathcal{T}_3 induced by the norm $|u|_{L^2(0,T;V)}$.

Let V_w denote the Hilbert space V endowed with the weak topology.

 $\mathscr{C}([0,T]; V_w) :=$ the space of weakly continuous functions $u : [0,T] \to V$ endowed with the weakest topology \mathscr{T}_4 such that for all $h \in V$ the mappings

$$\mathscr{C}([0,T]; \mathbf{V}_w) \ni u \to \langle u(\cdot), h \rangle_{\mathbf{V}} \in \mathscr{C}([0,T]; \mathbb{R})$$

are continuous. In particular, $u_n \rightarrow u$ in $\mathscr{C}([0,T]; V_w)$ iff for all $h \in V$:

$$\lim_{n\to\infty}\sup_{t\in[0,T]}|\langle u_n(t)-u(t),h\rangle_{\rm V}|=0.$$

Consider the ball

(5

$$\mathbb{B} := \{ x \in \mathbf{V} : \|x\|_{\mathbf{V}} \le r \}.$$

Let q be the metric compatible with the weak topology on \mathbb{B} . Let us consider the following subspace of the space $\mathscr{C}([0,T]; V_w)$

$$\mathscr{C}([0,T];\mathbb{B}_w) = \text{the space of weakly continuous functions } u:[0,T] \to V$$
4.1) such that $\sup_{t \in [0,T]} \|u(t)\|_V \le r.$

The space $\mathscr{C}([0,T];\mathbb{B}_w)$ is metrizable (see [9, 23]) with metric

(5.4.2)
$$\varrho(u,\mathbf{v}) = \sup_{t \in [0,T]} q(u(t),\mathbf{v}(t)).$$

Since by the Banach-Alaoglu theorem [80] \mathbb{B}_w is compact, $(\mathscr{C}([0,T];\mathbb{B}_w),\varrho)$ is a complete metric space.

The following lemma [24, Lemma 2.1] says that any sequence $(u_n)_{n \in \mathbb{N}} \subset \mathscr{C}([0, T]; \mathbb{B})$ convergent in $\mathscr{C}([0, T]; \mathbb{H})$ is also convergent in the space $\mathscr{C}([0, T]; \mathbb{B}_w)$.

Lemma 5.4.1. Let $u_n : [0,T] \to V, n \in \mathbb{N}$ be functions such that

- (i) $\sup_{n \in \mathbb{N}} \sup_{s \in [0,T]} \|u_n(s)\|_{\mathcal{V}} \leq r$,
- (*ii*) $u_n \rightarrow u$ in $\mathscr{C}([0,T];H)$.

Then $u, u_n \in \mathscr{C}([0,T]; \mathbb{B}_w)$ and $u_n \to u$ in $\mathscr{C}([0,T]; \mathbb{B}_w)$ as $n \to \infty$.

Let

$$(5.4.3) \qquad \qquad \mathcal{Z}_T = \mathcal{C}([0,T];\mathbf{H}) \cap L^2_{\mathbf{w}}(0,T;\mathbf{D}(\mathbf{A})) \cap L^2(0,T;\mathbf{V}) \cap \mathcal{C}([0,T];\mathbf{V}_w),$$

and let \mathcal{T} be the supremum of the corresponding topologies.

Now we formulate the compactness criterion analogous to the result due to Mikulevicus and Rozowskii [65], Brzeźniak and Motyl [24] for the space \mathcal{Z}_T .

Lemma 5.4.2. Let \mathcal{Z}_T , \mathcal{T} be as defined in (5.4.3). Then a set $\mathcal{K} \subset \mathcal{Z}_T$ is \mathcal{T} -relatively compact if the following three conditions hold

- (a) $\sup_{u \in \mathcal{K}} \sup_{s \in [0,T]} \|u(s)\|_{\mathcal{V}} < \infty$,
- (b) $\sup_{u \in \mathcal{K}} \int_0^T |u(s)|^2_{D(A)} ds < \infty$, i.e. \mathcal{K} is bounded in $L^2(0,T;D(A))$,
- (c) $\lim_{\delta \to 0} \sup_{u \in \mathcal{K}} \sup_{s,t \in [0,T]} |u(t) u(s)|_{\mathcal{H}} = 0.$

Proof. Let \mathscr{K} be a subset of \mathscr{Z}_T . Because of the assumption (*a*) we may consider the metric space $\mathscr{C}([0,T];\mathbb{B}_w) \subset \mathscr{C}([0,T];\mathbb{V}_w)$ defined by (5.4.1) and (5.4.2) with $r = \sup_{u \in \mathscr{K}} \sup_{s \in [0,T]} ||u(s)||_{\mathbb{V}}$. Because of the assumption (*b*) the restriction to \mathscr{K} of the weak topology in $L^2(0,T;\mathbb{D}(A))$ is metrizable. Since the restrictions to \mathscr{K} of the four topologies considered in \mathscr{Z}_T are metrizable, compactness of a subset of \mathscr{Z}_T is equivalent to its sequential compactness.

Let (u_n) be a sequence in \mathcal{K} . By the Banach-Alaoglu Theorem [80], condition (b) yields that $\overline{\mathcal{K}}$ is compact in $L^2_w(0,T;\mathbb{D}(A))$. Condition (c) implies that the functions u_n are equicontinuous in $\mathscr{C}([0,T], \mathbb{H})$. Since the embeddings $\mathbb{D}(A) \hookrightarrow \mathbb{V} \hookrightarrow \mathbb{H}$ are continuous and the embedding $\mathbb{D}(A) \hookrightarrow \mathbb{V}$ is compact, then Dubinsky Theorem (see Theorem 2.4.5) with conditions (b) and (c) imply that \mathcal{K} is compact in $L^2(0,T;\mathbb{V}) \cap \mathscr{C}([0,T];\mathbb{H})$. Hence in particular, there exists a subsequence, still denoted by (u_n) , convergent in \mathbb{H} . Therefore by Lemma 5.4.1 (u_n) is convergent in $\mathscr{C}([0,T];\mathbb{B}_w)$. This completes the proof of the lemma.

5.4.1 Tightness

Using Section 2.9 and the compactness criterion from Lemma 5.4.2 we obtain the following corollary which we will use to prove tightness of the laws defined by the Galerkin approximations.

Corollary 5.4.3 (Tightness criterion). Let $(X_n)_{n \in \mathbb{N}}$ be a sequence of continuous \mathbb{F} -adapted H-valued processes such that

(a) there exists a constant $C_1 > 0$ such that

$$\sup_{n \in \mathbb{N}} \mathbb{E} \left[\sup_{s \in [0,T]} \|X_n(s)\|_{\mathbf{V}}^2 \right] \leq C_1,$$

(b) there exists a constant $C_2 > 0$ such that

$$\sup_{n\in\mathbb{N}}\mathbb{E}\left[\int_{0}^{T}|X_{n}(s)|^{2}_{\mathrm{D}(\mathrm{A})}ds\right]\leq C_{2},$$

(c) $(X_n)_{n \in \mathbb{N}}$ satisfies the Aldous condition [A] in H.

Let $\tilde{\mathbb{P}}_n$ be the law of X_n on \mathcal{Z}_T . Then for every $\varepsilon > 0$ there exists a compact subset K_{ε} of \mathcal{Z}_T such that

$$\sup_{n\in\mathbb{N}}\tilde{\mathbb{P}}_n(K_{\varepsilon})\geq 1-\varepsilon.$$

Proof. Let $\varepsilon > 0$. By the Chebyshev inequality and (a), we infer that for any $n \in \mathbb{N}$ and any r > 0

$$\tilde{\mathbb{P}}_n\left(\sup_{s\in[0,T]}\|X_n(s)\|_{\mathbf{V}}^2>r\right)\leq \frac{\tilde{\mathbb{E}}_n\left[\sup_{s\in[0,T]}\|X_n(s)\|_{\mathbf{V}}^2\right]}{r}\leq \frac{C_1}{r}$$

Let R_1 be such that $\frac{C_1}{R_1} \leq \frac{\varepsilon}{3}$. Then

$$\sup_{n\in\mathbb{N}}\tilde{\mathbb{P}}_n\left(\sup_{s\in[0,T]}\|X_n(s)\|_{\mathrm{V}}^2>R_1\right)\leq\frac{\varepsilon}{3}.$$

Let $B_1 := \left\{ u \in \mathcal{Z}_T : \operatorname{sup}_{s \in [0,T]} \| u(s) \|_{V}^2 \leq R_1 \right\}.$

By the Chebyshev inequality and (b), we infer that for any $n \in \mathbb{N}$ and any r > 0

$$\tilde{\mathbb{P}}_n(|X_n|_{L^2(0,T;\mathbf{D}(\mathbf{A}))} > r) \le \frac{\mathbb{E}_n[|X_n|_{L^2(0,T;\mathbf{D}(\mathbf{A}))}^2]}{r^2} \le \frac{C_2}{r^2}.$$

Let R_2 be such that $\frac{C_2}{R_2^2} \leq \frac{\varepsilon}{3}$. Then

$$\sup_{n\in\mathbb{N}}\tilde{\mathbb{P}}_n\big(|X_n|_{L^2(0,T;\mathrm{D}(\mathrm{A}))}>R_2\big)\leq\frac{\varepsilon}{3}.$$

Let $B_2 := \{ u \in \mathcal{Z}_T : |u|_{L^2(0,T; D(A))} \le R_2 \}.$

By Lemmas 2.9.9 and 2.9.11 there exists a subset $A_{\frac{\varepsilon}{3}} \subset \mathscr{C}([0,T], \mathbb{H})$ such that $\tilde{\mathbb{P}}_n(A_{\frac{\varepsilon}{3}}) \ge 1 - \frac{\varepsilon}{3}$ and

$$\lim_{\delta \to 0} \sup_{u \in A_{\frac{\varepsilon}{3}}} \sup_{\substack{s,t \in [0,T]\\|t-s| \le \delta}} |u(t) - u(s)|_{\mathrm{H}} = 0.$$

It is sufficient to define K_{ε} as the closure of the set $B_1 \cap B_2 \cap A_{\frac{\varepsilon}{3}}$ in \mathcal{Z}_T . By Lemma 5.4.2, K_{ε} is compact in \mathcal{Z}_T . The proof is thus complete.

5.4.2 The Skorohod Theorem

Let us recall the Jakubowski's generalisation of the Skorohod Theorem as given by Brzeźniak and Ondreját [26, Theorem C.1], see also [49].

Theorem 5.4.4. Let \mathscr{X} be a topological space such that there exists a sequence $\{f_m\}_{m \in \mathbb{N}}$ of continuous functions $f_m : \mathscr{X} \to \mathbb{R}$ that separates points of \mathscr{X} . Let us denote by \mathscr{S} the σ -algebra generated by the maps $\{f_m\}$. Then

- (a) every compact subset of $\mathscr X$ is metrizable,
- (b) if $(\mu_m)_{m \in \mathbb{N}}$ is a tight sequence of probability measures on $(\mathcal{X}, \mathcal{S})$, then there exists a subsequence $(m_k)_{k \in \mathbb{N}}$, a probability space $(\Omega, \mathcal{F}, \mathbb{P})$ with \mathcal{X} -valued Borel measurable variables ξ_k, ξ such that μ_{m_k} is the law of ξ_k and ξ_k converges to ξ almost surely on Ω . Moreover, the law of ξ is a Radon measure.

Lemma 5.4.5. The topological space \mathcal{Z}_T satisfies the assumptions of Theorem 5.4.4.

Proof. We want to prove that on each space appearing in the definition (5.4.3) of the space \mathcal{Z}_T there exists a countable set of continuous real-valued functions separating points.

Since the spaces $\mathscr{C}([0,T];H)$ and $L^2(0,T;V)$ are separable, metrizable and complete, this condition is satisfied, see [3], exposé 8.

For the space $L^2_w(0,T; D(A))$ it is sufficient to put

$$f_m(u) := \int_0^T \langle u(t), \mathbf{v}_m(t) \rangle_{\mathrm{D}(\mathrm{A})} dt \in \mathbb{R}, \qquad u \in L^2_w(0, T; \mathrm{D}(\mathrm{A})), \quad m \in \mathbb{N},$$

where $\{v_m, m \in \mathbb{N}\}$ is a dense subset of $L^2(0, T; D(A))$.

Let us consider the space $\mathscr{C}([0,T]; V_w)$. Let $\{h_m, m \in \mathbb{N}\}$ be any dense subset of H and let \mathbb{Q}_T be the set of rational numbers belonging to the interval [0,T]. Then the family $\{f_{m,t}, m \in \mathbb{N}, t \in \mathbb{Q}_T\}$ defined by

$$f_{m,t}(u) := \langle u(t), h_m \rangle_{\mathcal{V}} \in \mathbb{R}, \qquad u \in \mathscr{C}([0,T]; \mathcal{V}_w), \quad m \in \mathbb{N}, \quad t \in \mathbb{Q}_T$$

consists of continuous functions separating points in $\mathscr{C}([0,T]; V_w)$, thus concluding the proof of the lemma.

Using Theorem 5.4.4 and Lemma 5.4.5, we obtain the following corollary which we will apply to construct a martingale solution to the stochastic constrained Navier-Stokes equations (5.2.4).

Corollary 5.4.6. Let $(\eta_n)_{n \in \mathbb{N}}$ be a sequence of \mathcal{Z}_T -valued random variables such that their laws $\mathcal{L}(\eta_n)$ on $(\mathcal{Z}_T, \mathcal{T})$ form a tight sequence of probability measures. Then there exists a subsequence (n_k) , a probability space $(\tilde{\Omega}, \tilde{\mathcal{F}}, \tilde{\mathbb{P}})$ and \mathcal{Z}_T -valued random variables $\tilde{\eta}, \tilde{\eta}_k, k \in \mathbb{N}$ such that the variables η_k and $\tilde{\eta}_k$ have the same laws on \mathcal{Z}_T and $\tilde{\eta}_k$ converges to $\tilde{\eta}$ almost surely on $\tilde{\Omega}$.

5.5 Faedo-Galerkin approximation and existence of a martingale solution

As mentioned in the introduction, the proof of the existence of a martingale solution is based on the Faedo-Galerkin approximation. In this section we first talk about the basic ingredients required for the approximation and then obtain the a'priori estimates, which we later use in the Subsection 5.5.1 to prove the tightness of laws induced by the solutions of the approximating equations (5.5.2).

Let $\{e_i\}_{i=1}^{\infty}$ be the orthonormal basis in H composed of eigenvectors of A. Let $H_n := \text{span}\{e_1, \dots, e_n\}$ be the subspace with the norm inherited from H, then $P_n : H \to H_n$ given by

(5.5.1)
$$P_n u := \sum_{i=1}^n \langle u, e_i \rangle_{\mathrm{H}} e_i, \quad u \in \mathrm{H}$$

is the orthogonal projection onto H_n .

Let us consider the classical Faedo-Galerkin approximation of (5.2.4) in the space H_n :

(5.5.2)
$$\begin{cases} du_n(t) = -\left[P_nAu_n(t) + P_nB(u_n(t)) + |\nabla u_n(t)|^2_{L^2}u_n(t)\right]dt \\ + \sum_{j=1}^m P_nC_ju_n(t) \circ dW_j(t), \qquad t \in [0,T], \\ u_n(0) = \frac{P_nu_0}{|P_nu_0|}. \end{cases}$$

Using the idea from [48] and the Banach Fixed Point Theorem we can show that the SDE (5.5.2) has a local maximal solution up to some stopping time $\tau \leq T$. In the following lemma we show that this local solution stays on the manifold \mathcal{M} if we start from the manifold, i.e. if the initial data $u_n(0) \in \mathcal{M}$ then $u_n(t) \in \mathcal{M}$ for every $t \in [0, \tau)$.

Lemma 5.5.1. Let $u_0 \in V \cap \mathcal{M}$ then the solution of (5.5.2) stays on the manifold \mathcal{M} , i.e. for all $t \in [0, \tau), u_n(t) \in \mathcal{M}$.

Proof. Let u_n be the solution of (5.5.2). Then applying Itô formula to the function $|x|_{\text{H}}^2$ and the process u_n along (5.5.2), (3.2.5) and assumption (A.1), we get

$$\begin{split} \frac{1}{2}d|u_{n}(t)|_{\mathrm{H}}^{2} &= \langle u_{n}(t), -P_{n}\mathrm{A}u_{n}(t) - P_{n}B(u_{n}(t)) + |\nabla u_{n}(t)|_{L^{2}}^{2}u_{n}(t)\rangle_{\mathrm{H}}dt \\ &+ \frac{1}{2}\sum_{j=1}^{m} \langle u_{n}(t), (P_{n}C_{j})^{2}u_{n}(t)\rangle_{\mathrm{H}}dt + \frac{1}{2}\sum_{j=1}^{m} \langle P_{n}C_{j}u_{n}(t), P_{n}C_{j}u_{n}(t)\rangle_{\mathrm{H}}dt \\ &+ \sum_{j=1}^{m} \langle u_{n}(t), P_{n}C_{j}u_{n}(t)dW_{j}(t)\rangle_{\mathrm{H}} \\ &= -\|u_{n}(t)\|_{\mathrm{V}}^{2}dt + |\nabla u_{n}(t)|_{L^{2}}^{2}|u_{n}(t)|_{\mathrm{H}}^{2}dt + \frac{1}{2}\sum_{j=1}^{m} \langle C_{j}^{*}u_{n}(t), C_{j}u_{n}(t)\rangle_{\mathrm{H}}dt \\ &+ \frac{1}{2}\sum_{j=1}^{m} |C_{j}u_{n}(t)|_{\mathrm{H}}^{2}dt \\ &= \|u_{n}(t)\|_{\mathrm{V}}^{2} \left[|u_{n}(t)|_{\mathrm{H}}^{2} - 1\right]dt + \frac{1}{2}\sum_{j=1}^{m} \left[|C_{j}u_{n}(t)|_{\mathrm{H}}^{2} - |C_{j}u_{n}(t)|_{\mathrm{H}}^{2}\right]dt. \end{split}$$

Thus, we have

$$d\left[|u_n(t)|_{\mathrm{H}}^2 - 1\right] = 2||u_n(t)||_{\mathrm{V}}^2\left[|u_n(t)|_{\mathrm{H}}^2 - 1\right]dt.$$

Integrating on both sides from 0 to t, we obtain

$$|u_n(t)|^2 - 1 = [|u_n(0)|_{\mathrm{H}}^2 - 1] \exp\left[2\int_0^t ||u_n(s)|_{\mathrm{V}}^2 ds\right].$$

Now since $|u_n(0)|_{\mathcal{H}} = 1$ and $\int_0^t ||u_n(s)||_{\mathcal{V}}^2 ds < \infty$, we get $|u_n(t)|_{\mathcal{H}} = 1$ for all $t \in [0, \tau)$, i.e $u_n(t) \in \mathcal{M}$ for every $t \in [0, \tau)$.

Since on the finite dimensional space H_n the H and V norm are equivalent, we can infer from the previous lemma that the V-norm of the solution stays bounded. Hence using this nonexplosion result as in the case of deterministic setting (see Chapter 4) we can prove the following lemma:

Lemma 5.5.2. For each $n \in \mathbb{N}$, there exists a continuous H_n -valued global solution u_n of (5.5.2). Moreover for every T > 0, for any $q \in [2, \infty)$

$$\mathbb{E}\left[\int_0^T |u_n(s)|_{\mathrm{H}}^q ds\right] < \infty.$$

We will require the following lemma to obtain a'priori bounds.

Lemma 5.5.3. Let $c \in \mathbb{R}^2$ and let $\mathbf{c} : \mathbb{T}^2 \to \mathbb{R}^2$ be the corresponding constant vector field. Put, for $u \in H^{1,2}(\mathbb{T}^2, \mathbb{R}^2)$

$$\tilde{C}u = \mathbf{c} \cdot \nabla u$$
 and $Cu = \Pi(\tilde{C}u)$

If the vector field $u \in H^{2,2}(\mathbb{T}^2,\mathbb{R}^2)$ is divergence free, then $\tilde{C}u$ is divergence free as well and in particular,

(5.5.3)
$$ACu - CAu = 0, \quad u \in H^{3,2}(\mathbb{T}^2, \mathbb{R}^2).$$

Proof. Let $c = (c_1, c_2)$ then $\tilde{C}u = (c_1D_1 + c_2D_2)u$. We have

$$div(Cu) = D_1((c_1D_1 + c_2D_2)u_1) + D_2((c_1D_1 + c_2D_2)u_2)$$

= $c_1D_1D_1u_1 + c_2D_1D_2u_1 + c_1D_2D_1u_2 + c_2D_2D_2u_2$
= $c_1D_1(D_1u_1 + D_2u_2) + c_2D_2(D_1u_1 + D_2u_2)$
= $(c_1D_1 + c_2D_2)(divu) = 0$,

where we used that vector *c* is constant and *u* is divergence free respectively. In order to establish the equality (5.5.3) we start by considering ACu - CAu. Since Au is divergence free, from first part we have $\Pi(\tilde{C}Au) = \tilde{C}Au$ and so $CAu = \tilde{C}Au$. Thus

$$ACu - CAu = -\Delta ((c_1D_1 + c_2D_2)u) - (c_1D_1 + c_2D_2)(-\Delta u)$$

= -[c_1\Delta D_1u + c_2\Delta D_2u] + [c_1\Delta D_1u + c_2\Delta D_2u] = 0

since c is a constant vector, completing the proof.

Lemma 5.5.4. Let T > 0 and u_n be the solution of (5.5.2). Then under the assumptions (A.1) - (A.2), for all $\rho > 0$ and $p \in [1, 1 + \frac{1}{K_c^2})$, there exist positive constants $C_1(p, \rho)$, $C_2(p, \rho)$ and $C_3(\rho)$ such that if $||u_0||_V \le \rho$, then

(5.5.4)
$$\sup_{n \ge 1} \mathbb{E} \left(\sup_{r \in [0,T]} \|u_n(r)\|_{V}^{2p} \right) \le C_1(p,\rho)$$

(5.5.5)
$$\sup_{n\geq 1} \mathbb{E} \int_0^T \|u_n(s)\|_{\mathbf{V}}^{2(p-1)} |\mathbf{A}u_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathbf{H}}^2 \, ds \le C_2(p,\rho),$$

and

(5.5.6)
$$\sup_{n \ge 1} \mathbb{E} \int_0^T |u_n(s)|_{\mathrm{D}(\mathrm{A})}^2 \, ds \le C_3(\rho)$$

Proof. Let $u_n(t)$ be the solution of (5.5.2) then applying the Itô formula to $\phi(x) = ||x||_V^2$ and the process $u_n(t)$, we get

$$\begin{split} d\|u_n(t)\|_{\mathbf{V}}^2 &= 2\langle \mathbf{A}u_n(t), -P_n\mathbf{A}u_n(t) - P_nB(u_n(t), u_n(t)) + |\nabla u_n(t)|_{L^2}^2 u_n(t)\rangle_{\mathbf{H}} dt \\ &+ 2\times \frac{1}{2}\sum_{j=1}^m \langle \mathbf{A}u_n(t), (P_nC_j)^2 u_n(t)\rangle_{\mathbf{H}} dt + 2\times \frac{1}{2}\sum_{j=1}^m \langle \mathbf{A}P_nC_j u_n(t), P_nC_j u_n(t)\rangle_{\mathbf{H}} dt \\ &+ 2\sum_{j=1}^m \langle \mathbf{A}u_n(t), P_nC_j u_n(t) dW_j(t)\rangle_{\mathbf{H}}. \end{split}$$

Now since $\langle |\nabla u_n(t)|_{L^2}^2 u_n(t), Au_n(t) - |\nabla u_n(t)|_{L^2}^2 u_n(t) \rangle = 0$, using (3.2.5), we have

$$\begin{split} d\|u_{n}(t)\|_{\mathcal{V}}^{2} &= -2\langle \mathrm{A}u_{n}(t) - |\nabla u_{n}(t)|_{L^{2}}^{2}u_{n}(t), \mathrm{A}u_{n}(t) - |\nabla u_{n}(t)|^{2}u_{n}(t)\rangle_{\mathrm{H}}dt \\ &+ 2\langle |\nabla u_{n}(t)|^{2}u_{n}(t), \mathrm{A}u_{n}(t) - |\nabla u_{n}(t)|_{L^{2}}^{2}u_{n}(t)\rangle_{\mathrm{H}}dt \\ &- 2\langle \mathrm{A}u_{n}(t), B(u_{n}(t), u_{n}(t))\rangle_{\mathrm{H}}dt + \sum_{j=1}^{m} \langle \mathrm{A}u_{n}(t), C_{j}^{2}u_{n}(t)\rangle_{\mathrm{H}}dt \\ &+ \sum_{j=1}^{m} \langle \mathrm{A}C_{j}u_{n}(t), C_{j}u_{n}(t)\rangle_{\mathrm{H}}dt + 2\sum_{j=1}^{m} \langle \mathrm{A}u_{n}(t), C_{j}u_{n}(t)dW_{j}(t)\rangle_{\mathrm{H}} \\ &= -2|\mathrm{A}u_{n}(t) - |\nabla u_{n}(t)|_{L^{2}}^{2}u_{n}(t)|_{\mathrm{H}}^{2}dt + \sum_{j=1}^{m} \langle \mathrm{A}C_{j}u_{n}(t) - C_{j}\mathrm{A}u_{n}(t), C_{j}u_{n}(t)\rangle_{\mathrm{H}}dt \\ &+ 2\sum_{j=1}^{m} \langle \mathrm{A}u_{n}(t), C_{j}u_{n}(t)dW_{j}(t)\rangle_{\mathrm{H}}. \end{split}$$

By Assumption (A.1) and Lemma 5.5.3 we have $AC_j u - C_j A u = 0$ for every $j \in \{1, \dots, m\}$. Thus, integrating on both sides we get

(5.5.7)
$$\begin{aligned} \|u_{n}(t)\|_{V}^{2} + 2\int_{0}^{t} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2}u_{n}(s)|_{H}^{2}ds \\ &= \|u_{n}(0)\|_{V}^{2} + 2\sum_{j=1}^{m}\int_{0}^{t} \langle Au_{n}(s), C_{j}u_{n}(s)dW_{j}(s)\rangle_{H} \\ &\leq \|u(0)\|_{V}^{2} + 2\sum_{j=1}^{m}\int_{0}^{t} \langle Au_{n}(s), C_{j}u_{n}(s)dW_{j}(s)\rangle_{H}. \end{aligned}$$

By Lemma 5.5.2, we infer that the process

$$\mu_n(t) = \sum_{j=1}^m \int_0^t \langle \operatorname{Au}_n(s), C_j u_n(s) \, dW_j(s) \rangle_{\mathrm{H}}, \quad t \in [0, T]$$

is a \mathbb{R} -valued \mathbb{F} -martingale and that $\mathbb{E}[\mu_n(t)] = 0$ for $t \in [0, T]$. Thus

(5.5.8)
$$\mathbb{E}\|u_n(t)\|_{\mathbf{V}}^2 + 2\mathbb{E}\int_0^t |\mathbf{A}u_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathbf{H}}^2 ds \le \mathbb{E}\|u(0)\|_{\mathbf{V}}^2, \quad t \in [0,T].$$

Hence

(5.5.9)
$$\sup_{n \ge 1} \sup_{t \in [0,T]} \mathbb{E} \| u_n(t) \|_{\mathbf{V}}^2 \le \mathbb{E} \| u(0) \|_{\mathbf{V}}^2.$$

Note that using (5.5.9) in (5.5.8), we also have the following estimate

(5.5.10)
$$\sup_{n\geq 1} \mathbb{E} \int_0^T |Au_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{H}^2 \, ds \le \mathbb{E} \|u(0)\|_{V}^2.$$

Let $\xi(t) = ||u_n(t)||_V^2$, $t \in [0, T]$ and $\phi(x) = x^p$, for some fixed $p \in [1, \infty)$. Using the Itô formula and (5.5.7), we obtain

$$\begin{aligned} \|u_{n}(t)\|_{V}^{2p} &= \|u_{n}(0)\|_{V}^{2p} - 2p \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds \\ &+ 2p(p-1) \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-2)} \langle Au_{n}(s), C_{j}u_{n}(s) \rangle_{H}^{2} ds \\ &+ 2p \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} \langle Au_{n}(s), C_{j}u_{n}(s) dW_{j}(s) \rangle_{H}. \end{aligned}$$

$$(5.5.11)$$

Since C is skew symmetric, $\langle Cu_n(s), u_n(s) \rangle = 0$ and thus, we get

$$\begin{split} \|u_{n}(t)\|_{V}^{2p} + 2p \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds \\ &= \|u_{n}(0)\|_{V}^{2p} + 2p \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} \langle Au_{n}(s), C_{j}u_{n}(s) dW_{j}(s) \rangle_{H} \\ &+ 2p(p-1) \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-2)} \langle Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s), C_{j}u_{n}(s) \rangle_{H}^{2} ds \end{split}$$

Using the Hölder inequality we have

$$\begin{split} \|u_{n}(t)\|_{V}^{2p} + 2p \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds \\ &\leq \|u_{n}(0)\|_{V}^{2p} + 2p \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} \langle Au_{n}(s), C_{j}u_{n}(s)dW_{j}(s) \rangle_{H} \\ &+ 2p(p-1) \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-2)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} |C_{j}u_{n}(s)|_{H}^{2} ds. \end{split}$$

On rearranging we get

$$\begin{split} \|u_{n}(t)\|_{V}^{2p} + 2p \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds \\ &\leq \|u_{n}(0)\|_{V}^{2p} + 2p \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|^{2(p-1)} \langle Au_{n}(s), C_{j}u_{n}(s) dW_{j}(s) \rangle_{H} \\ &+ 2p(p-1)K_{c}^{2} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds, \end{split}$$

where K_c is the positive constant defined in equality (5.3.1).

For $p \in [1, 1 + \frac{1}{K_c^2})$, $K_p = 2p \left[1 - K_c^2(p-1)\right] > 0$, thus

$$\|u_{n}(t)\|_{V}^{2p} + K_{p} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds$$

$$\leq \|u_{n}(0)\|_{V}^{2p} + 2p \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} \langle Au_{n}(s), C_{j}u_{n}(s) dW_{j}(s) \rangle_{H}.$$

Using Lemma 5.5.2 we infer that the process

$$\eta_n(t) = \sum_{j=1}^m \int_0^t \|u_n(s)\|_{V}^{2(p-1)} \langle Au_n(s), C_j u_n(s) dW_j(s) \rangle_{H}, \quad t \in [0,T],$$

is a martingale and $\mathbb{E}[\eta_n(t)] = 0$. Thus

(5.5.13)
$$\mathbb{E}\|u_n(t)\|_{\mathbf{V}}^{2p} + K_p \mathbb{E} \int_0^t \|u_n(s)\|_{\mathbf{V}}^{2(p-1)} |\mathbf{A}u_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathbf{H}}^2 ds \le \mathbb{E}\|u_n(0)\|_{\mathbf{V}}^{2p}.$$

In particular

(5.5.14)
$$\sup_{n \ge 1} \sup_{t \in [0,T]} \mathbb{E} \|u_n(t)\|^{2p} \le \mathbb{E} \|u_0\|_{\mathbf{V}}^{2p}.$$

Note that using (5.5.14) in (5.5.13), we also have the following estimate,

(5.5.15)
$$\sup_{n\geq 1} \mathbb{E} \int_0^T \|u_n(s)\|_{\mathbf{V}}^{2(p-1)} |\mathbf{A}u_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathbf{H}}^2 \, ds \leq \frac{1}{K_p} \mathbb{E} \|u_0\|_{\mathbf{V}}^{2p}.$$

In order to prove (5.5.4) we start from (5.5.11),

$$\begin{split} \|u_{n}(t)\|_{\mathcal{V}}^{2p} &= \|u_{n}(0)\|_{\mathcal{V}}^{2p} - 2p \int_{0}^{t} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{\mathcal{H}}^{2} ds \\ &+ 2p(p-1) \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-2)} \langle Au_{n}(s), C_{j}u_{n}(s) \rangle_{\mathcal{H}}^{2} ds \\ &+ 2p \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \langle Au_{n}(s), C_{j}u_{n}(s) dW_{j}(s) \rangle_{\mathcal{H}}. \end{split}$$

Since for every $j \in \{1, \dots, m\}, \langle C_j u_n(s), u_n(s) \rangle_{\mathrm{H}} = 0$, hence

$$\begin{split} \|u_{n}(t)\|_{V}^{2p} + 2p \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds &= \|u_{n}(0)\|_{V}^{2p} \\ &+ 2p(p-1) \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-2)} \langle Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s), C_{j}u_{n}(s) \rangle_{H}^{2} ds \\ &+ 2p \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} \langle Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s), C_{j}u_{n}(s) dW_{j}(s) \rangle_{H}. \end{split}$$

Taking the mathematical expectation and using the Hölder inequality, we have

$$\mathbb{E} \sup_{r \in [0,t]} \|u_{n}(r)\|_{V}^{2p} + 2p\mathbb{E} \sup_{r \in [0,t]} \int_{0}^{r} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds \leq \mathbb{E} \|u_{n}(0)\|_{V}^{2p} + 2p(p-1)K_{c}^{2}\mathbb{E} \sup_{r \in [0,t]} \left[\int_{0}^{r} \|u_{n}(s)\|^{2(p-2)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} |\nabla u_{n}(s)|_{L^{2}}^{2} ds \right]$$

$$(5.5.16) \qquad + 2p\mathbb{E} \sup_{r \in [0,t]} \left[\sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{V}^{2(p-1)} \langle Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s), C_{j}u_{n}(s) dW_{j}(s) \rangle_{H} \right].$$

Using the Burkholder-Davis-Gundy inequality, we get

$$\begin{split} \mathbb{E} \sup_{r \in [0,t]} \left| \sum_{j=1}^{m} \int_{0}^{r} \|u_{n}(s)\|_{\mathrm{V}}^{2(p-1)} \langle \mathrm{A}u_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s), C_{j} u_{n}(s) dW_{j}(s) \rangle_{\mathrm{H}} \right| \\ &\leq 3 \mathbb{E} \left| \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{\mathrm{V}}^{4(p-1)} \langle \mathrm{A}u_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s), C_{j} u_{n}(s) \rangle_{\mathrm{H}}^{2} ds \right|^{1/2} \\ &\leq 3 \mathbb{E} \left| \sum_{j=1}^{m} \int_{0}^{t} \|u_{n}(s)\|_{\mathrm{V}}^{4(p-1)} |\mathrm{A}u_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{\mathrm{H}}^{2} |C_{j} u_{n}(s)|_{\mathrm{H}}^{2} ds \right|^{1/2} \\ &\leq 3 \mathbb{E} K_{c} \left[\int_{0}^{t} \|u_{n}(s)\|_{\mathrm{V}}^{2p} \|u_{n}(s)\|_{\mathrm{V}}^{2(p-1)} |\mathrm{A}u_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{\mathrm{H}}^{2} ds \right]^{1/2}. \end{split}$$

Using the Hölder inequality and the Young inequality, we obtain

$$\begin{split} & \mathbb{E} \sup_{r \in [0,t]} \left| \sum_{j=1}^{m} \int_{0}^{r} \|u_{n}(s)\|_{\mathrm{V}}^{2(p-1)} \langle \mathrm{A}u_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s), C_{j}u_{n}(s) dW_{j}(s) \rangle_{\mathrm{H}} \right| \\ & \leq 3 \mathbb{E} \left[K_{c} \left(\sup_{r \in [0,t]} \|u_{n}(r)\|_{\mathrm{V}}^{2p} \right)^{1/2} \left(\int_{0}^{t} \|u_{n}(s)\|_{\mathrm{V}}^{2(p-1)} |\mathrm{A}u_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{\mathrm{H}}^{2} ds \right)^{1/2} \right] \\ & \leq 3 \mathbb{E} \left[\varepsilon \sup_{r \in [0,t]} \|u_{n}(r)\|_{\mathrm{V}}^{2p} + \frac{K_{c}^{2}}{4\varepsilon} \int_{0}^{t} \|u_{n}(s)\|_{\mathrm{V}}^{2(p-1)} |\mathrm{A}u_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{\mathrm{H}}^{2} ds \right]. \end{split}$$

Thus using this in (5.5.16), we get

$$\begin{split} \mathbb{E} \sup_{r \in [0,t]} \|u_n(r)\|_{\mathcal{V}}^{2p} + 2\mathbb{E} \sup_{r \in [0,t]} \int_0^r \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |Au_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds \\ & \leq \mathbb{E} \|u_n(0)\|_{\mathcal{V}}^{2p} + 2p(p-1)K_c^2 \mathbb{E} \sup_{r \in [0,t]} \int_0^r \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |Au_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds \\ (5.5.17) \qquad + \frac{3pK_c^2}{2\varepsilon} \mathbb{E} \int_0^t \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |Au_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds \, . \end{split}$$

Hence for $\varepsilon = \frac{1}{12p}$, Eq. (5.5.17) reduces to

$$\begin{split} \mathbb{E} \sup_{r \in [0,t]} \|u_n(r)\|_{\mathcal{V}}^{2p} + 4\mathbb{E} \sup_{r \in [0,t]} \int_0^r \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |Au_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds &\leq 2\mathbb{E} \|u_n(0)\|_{\mathcal{V}}^{2p} \\ &+ 4p(p-1)K_c^2 \mathbb{E} \sup_{r \in [0,t]} \int_0^r \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |Au_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds \\ &+ 36p^2 K_c^2 \mathbb{E} \int_0^t \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |Au_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds. \end{split}$$

Since $\int_0^r |\mathrm{A} u_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathrm{H}}^2 \, ds$ is an increasing function, we have

$$\begin{split} \mathbb{E} \sup_{r \in [0,t]} \|u_n(r)\|_{\mathcal{V}}^{2p} + 4\mathbb{E} \int_0^t \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |\mathcal{A}u_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds \\ \leq 2\mathbb{E} \|u_n(0)\|_{\mathcal{V}}^{2p} + 4pK_c^2 [10p-1]\mathbb{E} \int_0^t \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |\mathcal{A}u_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds. \end{split}$$

In particular

$$\begin{split} \mathbb{E} \sup_{r \in [0,t]} \|u_n(r)\|_{\mathcal{V}}^{2p} \leq & 4pK_c^2 [10p-1] \mathbb{E} \int_0^t \|u_n(s)\|_{\mathcal{V}}^{2(p-1)} |Au_n(s) - |\nabla u_n(s)|_{L^2}^2 u_n(s)|_{\mathcal{H}}^2 \, ds \\ &+ 2\mathbb{E} \|u_n(0)\|_{\mathcal{V}}^{2p} \, . \end{split}$$

Since $\mathbb{E} \|u_n(0)\|_{\mathbf{V}}^{2p} \leq \mathbb{E} \|u_0\|_{\mathbf{V}}^{2p}$ and using (5.5.15), for $p \in [1, 1 + \frac{1}{K_c^2})$

$$\mathbb{E}\int_{0}^{T} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{H}^{2} ds$$

is uniformly bounded in n, thus

$$\sup_{n\geq 1} \mathbb{E} \sup_{r\in[0,T]} \|u_n(r)\|_{\mathbf{V}}^{2p} \leq C_1(p,\rho).$$

Now we will establish (5.5.6). Note that

$$\mathbb{E}\int_{0}^{T} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds = \mathbb{E}\int_{0}^{T} |\mathrm{A}u_{n}(s) - |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s)|_{\mathrm{H}}^{2} ds + \mathbb{E}\int_{0}^{T} ||u_{n}(s)||^{4} ds.$$

Using (5.5.4) for p = 2 and (5.5.5) for p = 1, we get

$$\sup_{n\geq 1} \mathbb{E} \int_0^T |u_n(s)|^2_{\mathcal{D}(\mathcal{A})} ds \le C_2(1,\rho) + C_1(2,\rho)T =: C_3(\rho).$$

5.5.1 Tightness of the laws of approximating solutions

In this subsection using the a'priori estimates from the Lemma 5.5.4 and the Corollary 5.4.3 we will prove that for every $n \in \mathbb{N}$ the measures $\mathscr{L}(u_n)$ on $(\mathscr{Z}_T, \mathscr{T})$ defined by the solutions of the stochastic ODE (5.5.2) are tight. The following is the main result of this subsection.

Lemma 5.5.5. The set of measures $\{\mathscr{L}(u_n), n \in \mathbb{N}\}$ is tight on $(\mathscr{Z}_T, \mathscr{T})$.

Proof. We apply Corollary 5.4.3. According to the a'priori estimates (5.5.4) (for p = 1) and (5.5.6), conditions (*a*) and (*b*) of Corollary 5.4.3 are satisfied. Thus it is sufficient to prove that the sequence $(u_n)_{n \in \mathbb{N}}$ satisfies the Aldous condition [**A**] in H. Let $(\tau_n)_{n \in \mathbb{N}}$ be a sequence of stopping times such that $0 \le \tau_n \le T$. By (5.5.2), for $t \in [0, T]$ we have

$$\begin{split} u_n(t) &= u_n(0) - \int_0^t P_n A u_n(s) \, ds - \int_0^t P_n B(u_n(s)) \, ds + \int_0^t |\nabla u_n(s)|_{L^2}^2 u_n(s) \, ds \\ &+ \frac{1}{2} \int_0^t (P_n C)^2 u_n(s) \, ds + \int_0^t P_n C u_n(s) \, dW(s) \\ &:= J_1^n + J_2^n(t) + J_3^n(t) + J_4^n(t) + J_5^n(t) + J_6^n(t), \quad t \in [0, T]. \end{split}$$

Let $\theta > 0$. We start by estimating each term in the R.H.S. of the above equality.

Ad. J_2^n . Since $A: D(A) \to H$ is a bounded linear map, then by the Hölder inequality and estimate (5.5.6), we have the following inequalities

$$\mathbb{E}\left[|J_{2}^{n}(\tau_{n}+\theta)-J_{2}^{n}(\tau_{n})|_{\mathrm{H}}\right] = \mathbb{E}\left|\int_{\tau_{n}}^{\tau_{n}+\theta}P_{n}\mathrm{A}u_{n}(s)\,ds\right|_{\mathrm{H}} \leq c\mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta}|\mathrm{A}u_{n}(s)|_{\mathrm{H}}\,ds$$

$$\leq c\mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta}|u_{n}(s)|_{\mathrm{D}(\mathrm{A})}\,ds \leq c\theta^{\frac{1}{2}}\left(\mathbb{E}\left[\int_{0}^{T}|u_{n}(s)|_{D(\mathrm{A})}^{2}ds\right]\right)^{\frac{1}{2}} \leq cC_{3}^{\frac{1}{2}}\cdot\theta^{\frac{1}{2}} =:c_{2}\cdot\theta^{\frac{1}{2}}$$

Ad. J_3^n . Since $B: V \times V \to H$ is bilinear and continuous, then using (3.2.3), the Cauchy-Schwarz inequality, estimates (5.5.4) (for p = 1) and (5.5.6), we have the following estimates

$$\mathbb{E}[|J_{3}^{n}(\tau_{n}+\theta)-J_{3}^{n}(\tau_{n})|_{\mathrm{H}}] = \mathbb{E}\left|\int_{\tau_{n}}^{\tau_{n}+\theta} P_{n}B(u_{n}(s))ds\right|_{\mathrm{H}} \leq c\mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta} |B(u_{n}(s),u_{n}(s))|_{\mathrm{H}}ds$$

$$\leq c\mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta} |u_{n}(s)|_{\mathrm{H}}^{\frac{1}{2}}||u_{n}(s)||_{\mathrm{V}}|u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{\frac{1}{2}}ds \leq c\mathbb{E}\left[\int_{\tau_{n}}^{\tau_{n}+\theta} ||u_{n}(s)||_{\mathrm{V}}^{\frac{3}{2}}|u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{1/2}ds\right]$$

$$\leq c\mathbb{E}\left(\left[\int_{\tau_{n}}^{\tau_{n}+\theta} ||u_{n}(s)||_{\mathrm{V}}^{2}ds\right]^{\frac{3}{4}}\left[\int_{\tau_{n}}^{\tau_{n}+\theta} ||u_{n}(s)||_{\mathrm{D}(\mathrm{A})}^{2}ds\right]^{\frac{1}{4}}\right)$$

$$(5.5.19) \leq c\theta^{\frac{3}{4}}\left[\mathbb{E}\sup_{s\in[0,T]} ||u_{n}(s)||_{\mathrm{V}}^{2}\right]^{\frac{3}{4}}\left[\mathbb{E}\int_{0}^{T} ||u_{n}(s)||_{\mathrm{D}(\mathrm{A})}^{2}ds\right]^{\frac{1}{4}} \leq cC_{1}(1)^{\frac{3}{4}}C_{3}^{\frac{1}{4}} \cdot \theta^{\frac{3}{4}} =: c_{3} \cdot \theta^{\frac{3}{4}}.$$

Ad. J_4^n . Using Lemma 5.5.1 and estimate (5.5.4) (for p = 1), we have

$$\mathbb{E}[|J_{4}^{n}(\tau_{n}+\theta) - J_{4}^{n}(\tau_{n})|_{\mathrm{H}}] = \mathbb{E}\left|\int_{\tau_{n}}^{\tau_{n}+\theta} |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s) \, ds\right|_{\mathrm{H}}$$

$$(5.5.20) \leq \mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta} |\nabla u_{n}(s)|_{L^{2}}^{2} |u_{n}(s)|_{\mathrm{H}} \, ds \leq \mathbb{E}\sup_{s \in [0,T]} \|u_{n}(s)\|_{\mathrm{V}}^{2} \theta \leq C_{1}(1) \cdot \theta =: c_{4} \cdot \theta.$$

Ad. J_5^n . Since C is linear and continuous, then using the Cauchy-Schwarz inequality, assumption **(A.1)** and (5.5.6), we have the following

$$\mathbb{E}[|J_{5}^{n}(\tau_{n}+\theta) - J_{5}^{n}(\tau_{n})|_{\mathrm{H}}] = \mathbb{E}\left|\frac{1}{2}\sum_{j=1}^{m}\int_{\tau_{n}}^{\tau_{n}+\theta} (P_{n}C_{j})^{2}u_{n}(s)ds\right|_{\mathrm{H}}$$

$$\leq \frac{1}{2}c\mathbb{E}\left(\sum_{j=1}^{m}\int_{\tau_{n}}^{\tau_{n}+\theta} |C_{j}^{2}u_{n}(s)|_{\mathrm{H}}ds\right) \leq \frac{1}{2}cK_{c}^{2}\mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}ds$$

$$\leq \frac{1}{2}cK_{c}^{2}\left[\mathbb{E}\int_{0}^{T} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2}ds\right]^{\frac{1}{2}}\theta^{\frac{1}{2}} \leq \frac{cK_{c}^{2}}{2}C_{3}^{\frac{1}{2}}\cdot\theta^{\frac{1}{2}} =:c_{5}\cdot\theta^{\frac{1}{2}}.$$
(5.5.21)

Ad. J_6^n . Using the Itô isometry, assumption (**A.1**) and estimate (5.5.4) (for p = 1), we obtain the following

$$\mathbb{E}\left[|J_{6}^{n}(\tau_{n}+\theta)-J_{6}^{n}(\tau_{n})|_{\mathrm{H}}^{2}\right] = \mathbb{E}\left|\int_{\tau_{n}}^{\tau_{n}+\theta} P_{n}Cu_{n}(s)dW(s)\right|_{\mathrm{H}}^{2} \leq c\mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta} |Cu_{n}(s)|_{\mathrm{H}}^{2}ds \\ \leq cK_{c}^{2}\mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta} \|u_{n}(s)\|_{\mathrm{V}}^{2}ds \leq cK_{c}^{2}\mathbb{E}\sup_{s\in[0,T]} \|u_{n}(s)\|_{\mathrm{V}}^{2}\theta \leq cK_{c}^{2}C_{1}(1)\cdot\theta =: c_{6}\cdot\theta.$$

$$(5.5.22) \qquad \leq cK_{c}^{2}\mathbb{E}\int_{\tau_{n}}^{\tau_{n}+\theta} \|u_{n}(s)\|_{\mathrm{V}}^{2}ds \leq cK_{c}^{2}\mathbb{E}\sup_{s\in[0,T]} \|u_{n}(s)\|_{\mathrm{V}}^{2}\theta \leq cK_{c}^{2}C_{1}(1)\cdot\theta =: c_{6}\cdot\theta.$$

Let us fix $\kappa > 0$ and $\varepsilon > 0$. By the Chebyshev's inequality and estimates (5.5.18) - (5.5.21), we obtain

$$\mathbb{P}(\{|J_i^n(\tau_n+\theta) - J_i^n(\tau_n)|_{\mathbf{H}} \ge \kappa\}) \le \frac{1}{\kappa} \mathbb{E}\left[|J_i^n(\tau_n+\theta) - J_i^n(\tau_n)|_{\mathbf{H}}\right] \le \frac{c_i\theta}{\kappa}; \quad n \in \mathbb{N},$$

where $i = 1, \dots, 5$. Let $\delta_i = \frac{\kappa}{c_i} \varepsilon$. Then

$$\sup_{n \in \mathbb{N}} \sup_{0 \le \theta \le \delta_i} \mathbb{P}(\{|J_i^n(\tau_n + \theta) - J_i^n(\tau_n)|_{\mathcal{H}} \ge \kappa\}) \le \varepsilon, \quad i = 1...5$$

By the Chebyshev inequality and (5.5.22), we have

$$\mathbb{P}(\{|J_6^n(\tau_n+\theta) - J_6^n(\tau_n)|_{\mathbf{H}} \ge \kappa\}) \le \frac{1}{\kappa^2} \mathbb{E}\left[|J_i^n(\tau_n+\theta) - J_i^n(\tau_n)|_{\mathbf{H}}^2\right]$$
$$\le \frac{c_6\theta}{\kappa^2}, \quad n \in \mathbb{N}.$$

Let $\delta_6 = \frac{\kappa^2}{C_6} \varepsilon$. Then

$$\sup_{n \in \mathbb{N}} \sup_{0 \le \theta \le \delta_6} \mathbb{P}(\{|J_6^n(\tau_n + \theta) - J_6^n(\tau_n)|_{\mathcal{H}} \ge \kappa\}) \le \varepsilon.$$

Since [**A**] holds for each term J_i^n , $i = 1, 2, \dots, 6$; we infer that it holds also for (u_n) . Therefore we can conclude the proof of the lemma by invoking Corollary 5.4.3.

5.5.2 Proof of Theorem 5.3.3

We prove certain pointwise convergence in Lemma 5.5.8 which is later used to construct a continuous H-valued martingale. Martingale representation theorem then guarantees the existence of a martingale solution of problem (5.2.4), proving Theorem 5.3.3.

By Lemma 5.5.5 the set of measures $\{\mathscr{L}(u_n), n \in \mathbb{N}\}$ is tight on the space $(\mathscr{Z}_T, \mathscr{T})$, defined by (5.4.3). Hence by Corollary 5.4.6 there exist a subsequence $(n_k)_{k \in \mathbb{N}}$, a probability space $(\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{P}})$ and, on this space, \mathscr{Z}_T -valued random variables $\tilde{u}, \tilde{u}_{n_k}, k \geq 1$ such that

(5.5.23) \tilde{u}_{n_k} has the same law as u_{n_k} and $\tilde{u}_{n_k} \to \tilde{u}$ in \mathcal{Z}_T , $\tilde{\mathbb{P}}$ -a.s.

 $\tilde{u}_{n_k} \to \tilde{u}$ in \mathcal{Z}_T , $\tilde{\mathbb{P}}$ – a.s. precisely means that

$$\begin{split} \tilde{u}_{n_k} &\to \tilde{u} \text{ in } \mathscr{C}([0,T]; \mathbf{H}), \\ \tilde{u}_{n_k} &\to \tilde{u} \text{ in } L^2(0,T; \mathbf{D}(\mathbf{A})), \\ \tilde{u}_{n_k} &\to \tilde{u} \text{ in } L^2(0,T; \mathbf{V}), \\ \tilde{u}_{n_k} &\to \tilde{u} \text{ in } \mathscr{C}([0,T]; \mathbf{V}_w). \end{split}$$

Let us denote the subsequence (\tilde{u}_{n_k}) again by $(\tilde{u}_n)_{n \in \mathbb{N}}$.

Since $u_n \in \mathscr{C}([0,T]; H_n)$, \mathbb{P} -a.s. and $\mathscr{C}([0,T]; H_n)$ is a Borel subset of $\mathscr{C}([0,T]; H) \cap L^2(0,T; V)$ and also \tilde{u}_n , u_n have the same laws on \mathcal{Z}_T we can make the following inferences

$$\begin{split} \mathcal{L}(\tilde{u}_n)(\mathcal{C}([0,T];\mathbf{H}_n) &= 1, \quad n \geq 1, \\ |\tilde{u}_n(t)|_{\mathbf{H}} &= |u_n(t)|_{\mathbf{H}}, \quad \text{a.s.} \end{split}$$

Also from (5.5.23) $\tilde{u}_n \to \tilde{u}$ in $\mathscr{C}([0,T];H)$ and by Lemma 5.5.1 $u_n(t) \in \mathscr{M}$ for every $t \in [0,T]$. Therefore we can conclude that

(5.5.24)
$$\tilde{u}(t) \in \mathcal{M}, \quad t \in [0,T].$$

Moreover by (5.5.4) and (5.5.6), for $p \in [1, 1 + \frac{1}{K^2}]$

(5.5.25)
$$\sup_{n \in \mathbb{N}} \tilde{\mathbb{E}} \left(\sup_{0 \le s \le T} \| \tilde{u}_n(s) \|_{V}^{2p} \right) \le C_1(p),$$

(5.5.26)
$$\sup_{n \in \mathbb{N}} \tilde{\mathbb{E}}\left[\int_0^1 |\tilde{u}_n(s)|^2_{\mathrm{D}(\mathrm{A})} ds\right] \le C_3.$$

By inequality (5.5.26) we infer that the sequence (\tilde{u}_n) contains a subsequence, still denoted by (\tilde{u}_n) convergent weakly in the space $L^2([0,T] \times \tilde{\Omega}; D(A))$. Since by (5.5.23) $\tilde{u}_n \to \tilde{u}$ in $\mathcal{Z}_T \tilde{\mathbb{P}}$ -a.s., we conclude that $\tilde{u} \in L^2([0,T] \times \tilde{\Omega}; D(A))$, i.e.

(5.5.27)
$$\tilde{\mathbb{E}}\left[\int_0^T |\tilde{u}(s)|^2_{\mathrm{D}(\mathrm{A})} ds\right] < \infty.$$

Similarly by inequality (5.5.25) we can choose a subsequence of (\tilde{u}_n) convergent weak star in the space $L^2(\tilde{\Omega}; L^{\infty}(0,T; V))$ and, using (5.5.23), we infer that

(5.5.28)
$$\tilde{\mathbb{E}}\left(\sup_{0\leq s\leq T}\|\tilde{u}(s)\|_{V}^{2}\right) < \infty.$$

For each $n \ge 1$, let us consider a process \tilde{M}_n with trajectories in $\mathscr{C}([0,T];H_n)$, in particular in $\mathscr{C}([0,T];H)$ defined by

(5.5.29)
$$\tilde{M}_{n}(t) = \tilde{u}_{n}(t) - P_{n}\tilde{u}(0) + \int_{0}^{t} P_{n}A\tilde{u}_{n}(s)ds + \int_{0}^{t} P_{n}B(\tilde{u}_{n}(s))ds - \int_{0}^{t} |\nabla\tilde{u}_{n}(s)|^{2}\tilde{u}_{n}(s)ds - \frac{1}{2}\sum_{j=1}^{m}\int_{0}^{t} (P_{n}C_{j})^{2}\tilde{u}_{n}(s)ds, \quad t \in [0,T].$$

Lemma 5.5.6. \tilde{M}_n is a square integrable martingale with respect to the filtration $\tilde{\mathbb{F}}_n = (\tilde{\mathscr{F}}_{n,t})$, where $\tilde{\mathscr{F}}_{n,t} = \sigma\{\tilde{u}_n(s), s \leq t\}$, with the quadratic variation

(5.5.30)
$$\langle \langle \tilde{M}_n \rangle \rangle_t = \int_0^t \sum_{j=1}^m |P_n C_j \tilde{u}_n(s)|_{\mathrm{H}}^2 ds$$

Proof. Indeed since \tilde{u}_n and u_n have the same laws, for all $s, t \in [0, T], s \leq t$, for all bounded continuous functions h on $\mathscr{C}([0,s]; H)$, and all $\psi, \zeta \in H$, we have

(5.5.31)
$$\tilde{\mathbb{E}}\left[\langle \tilde{M}_n(t) - \tilde{M}_n(s), \psi \rangle_{\mathrm{H}} h(\tilde{u}_{n|[0,s]})\right] = 0$$

and

(5.5.32)
$$\widetilde{\mathbb{E}} \left[\left(\langle \tilde{M}_n(t), \psi \rangle_{\mathrm{H}} \langle \tilde{M}_n(t), \zeta \rangle_{\mathrm{H}} - \langle \tilde{M}_n(s), \psi \rangle_{\mathrm{H}} \langle \tilde{M}_n(s), \zeta \rangle_{\mathrm{H}} - \sum_{j=1}^m \int_s^t \langle \left(C_j \tilde{u}_n(\sigma) \right)^* P_n \psi, \left(C_j \tilde{u}_n(\sigma) \right)^* P_n \zeta \rangle_{\mathbb{R}} \, d\sigma \right) \cdot h(\tilde{u}_{n \mid [0,s]}) \right] = 0.$$

Lemma 5.5.7. Let us define a process \tilde{M} for $t \in [0, T]$ by

(5.5.33)
$$\tilde{M}(t) = \tilde{u}(t) - \tilde{u}(0) + \int_0^t A\tilde{u}(s) ds + \int_0^t B(\tilde{u}(s)) ds$$

(5.5.34)
$$-\int_0^t |\nabla \tilde{u}(s)|_{L^2}^2 \tilde{u}(s) ds - \frac{1}{2} \sum_{j=1}^m \int_0^t C_j^2 \tilde{u}(s) ds.$$

Then \tilde{M} is an H-valued continuous process.

Proof. Since $\tilde{u} \in \mathscr{C}([0,T];V)$ we just need to show that each of the remaining four terms on the RHS of (5.5.33) are H-valued and well defined.

Using the Cauchy-Schwarz inequality repeatedly and by (5.5.27) we have the following inequalities

$$\tilde{\mathbb{E}}\int_0^T |\mathrm{A}\tilde{u}(s)|_{\mathrm{H}}\,ds \leq T^{1/2} \left(\tilde{\mathbb{E}}\int_0^T |\tilde{u}(s)|_{\mathrm{D}(\mathrm{A})}^2\,ds\right)^{1/2} < \infty.$$

Using (3.2.3), the Hölder inequality, (5.5.24) and the estimates (5.5.27) and (5.5.28) we obtain the following:

$$\begin{split} &\tilde{\mathbb{E}} \int_{0}^{T} |B(\tilde{u}(s))|_{\mathrm{H}} ds \leq 2\tilde{\mathbb{E}} \int_{0}^{T} |\tilde{u}(s)|_{\mathrm{H}}^{1/2} |\nabla \tilde{u}(s)|_{L^{2}} |\tilde{u}(s)|_{\mathrm{D(A)}}^{1/2} ds \\ &\leq 2\tilde{\mathbb{E}} \left[\left(\int_{0}^{T} \|\tilde{u}(s)\|_{\mathrm{V}}^{4/3} ds \right)^{3/4} \left(\int_{0}^{T} |\tilde{u}(s)|_{\mathrm{D(A)}}^{2} ds \right)^{1/4} \right] \\ &\leq 2T^{3/4} \left(\tilde{\mathbb{E}} \sup_{s \in [0,T]} \|\tilde{u}(s)\|_{\mathrm{V}}^{4/3} \right)^{3/4} \left(\tilde{\mathbb{E}} \int_{0}^{T} |\tilde{u}(s)|_{\mathrm{D(A)}}^{2} ds \right)^{1/4} < \infty. \end{split}$$

Using the Hölder inequality, (5.5.24) and inequality (5.5.28) we have

$$\tilde{\mathbb{E}}\int_0^T |\nabla \tilde{u}(s)|_{L^2}^2 |\tilde{u}(s)|_{\mathcal{H}} ds \leq \tilde{\mathbb{E}}\int_0^T \|\tilde{u}(s)\|_{\mathcal{V}}^2 ds \leq \tilde{\mathbb{E}}\left(\sup_{s\in[0,T]} \|\tilde{u}(s)\|_{\mathcal{V}}^2\right) T < \infty.$$

Now we are left to deal with the last term on the RHS. Using assumption (A.1) and the estimate (5.5.27), we have the following inequalities for every $j \in \{1, \dots, m\}$,

$$\tilde{\mathbb{E}} \int_{0}^{T} |C_{j}^{2} \tilde{u}(s)|_{\mathrm{H}} \leq K_{c} T^{1/2} \left(\tilde{\mathbb{E}} \int_{0}^{T} |\tilde{u}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds \right)^{1/2} < \infty$$

This concludes the proof of the lemma.

Lemma 5.5.8. For all $s, t \in [0, T]$ such that $s \le t$, we have :

(a)
$$\lim_{n\to\infty} \langle \tilde{u}_n(t), P_n \psi \rangle_{\mathrm{H}} = \langle \tilde{u}(t), \psi \rangle_{\mathrm{H}}, \quad \tilde{\mathbb{P}}\text{-}a.s. \quad \psi \in \mathrm{H}$$

(b)
$$\lim_{n\to\infty}\int_s^t \langle A\tilde{u}_n(\sigma), P_n\psi \rangle_H d\sigma = \int_s^t \langle A\tilde{u}(\sigma), \psi \rangle_H d\sigma$$
, $\tilde{\mathbb{P}}$ -a.s. $\psi \in H$,

- (c) $\lim_{n\to\infty}\int_s^t \langle B(\tilde{u}_n(\sigma), \tilde{u}_n(\sigma)), P_n\psi\rangle_{\mathrm{H}} d\sigma = \int_s^t \langle B(\tilde{u}(\sigma), \tilde{u}(\sigma)), \psi\rangle_{\mathrm{H}} d\sigma, \quad \tilde{\mathbb{P}}\text{-}a.s. \quad \psi \in \mathrm{V},$
- (d) $\lim_{n\to\infty}\int_s^t |\nabla \tilde{u}_n(\sigma)|_{L^2}^2 \langle \tilde{u}_n(\sigma), P_n\psi\rangle_{\mathrm{H}} \, d\sigma = \int_s^t |\nabla \tilde{u}(\sigma)|_{L^2}^2 \langle \tilde{u}(\sigma), \psi\rangle_{\mathrm{H}} \, d\sigma, \quad \tilde{\mathbb{P}}\text{-}a.s. \ \psi \in \mathrm{H},$

(e)
$$\lim_{n\to\infty} \langle \int_s^t C_j^2 \tilde{u}_n(\sigma), P_n \psi \rangle_{\mathrm{H}} d\sigma = \int_s^t \langle C_j^2 \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma, \quad \tilde{\mathbb{P}}\text{-}a.s. \quad \psi \in \mathrm{H}.$$

Proof. Let us fix $s, t \in [0, T]$, $s \le t$. By (5.5.23) we know that

$$(5.5.35) \qquad \tilde{u}_n \to \tilde{u} \text{ in } \mathscr{C}([0,T];\mathrm{H}) \cap L^2_w(0,T;\mathrm{D}(\mathrm{A})) \cap L^2(0,T;\mathrm{V}) \cap \mathscr{C}([0,T];\mathrm{V}_w), \quad \tilde{\mathbb{P}}\text{-a.s.}$$

Let $\psi \in H$. Since $\tilde{u}_n \to \tilde{u}$ in $\mathscr{C}([0,T];H) \tilde{\mathbb{P}}$ -a.s. and $P_n \psi \to \psi$ in H, we have

$$\begin{split} &\lim_{n\to\infty} \langle \tilde{u}_n(t), P_n\psi\rangle_{\mathrm{H}} - \langle \tilde{u}(t), \psi\rangle_{\mathrm{H}} \\ &= \lim_{n\to\infty} \langle \tilde{u}_n(t) - \tilde{u}(t), P_n\psi\rangle_{\mathrm{H}} + \lim_{n\to\infty} \langle \tilde{u}(t), P_n\psi - \psi\rangle_{\mathrm{H}} = 0 \quad \tilde{\mathbb{P}}\text{-a.s.} \end{split}$$

Thus we infer that assertion (a) holds.

Let $\psi \in \mathbf{H}$, then

$$\begin{split} &\int_{s}^{t} \langle \mathrm{A}\tilde{u}_{n}(\sigma), P_{n}\psi \rangle_{\mathrm{H}} \, d\sigma - \int_{s}^{t} \langle \mathrm{A}\tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} \, d\sigma \\ &= \int_{s}^{t} \langle \mathrm{A}\tilde{u}_{n}(\sigma) - \mathrm{A}\tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} \, d\sigma + \int_{s}^{t} \langle \mathrm{A}\tilde{u}_{n}(\sigma), P_{n}\psi - \psi \rangle_{\mathrm{H}} \, d\sigma \\ &\leq \int_{s}^{t} \langle \tilde{u}_{n}(\sigma) - \tilde{u}(\sigma), \mathrm{A}^{-1}\psi \rangle_{\mathrm{D}(\mathrm{A})} \, d\sigma + \int_{s}^{t} |\tilde{u}_{n}(\sigma)|_{\mathrm{D}(\mathrm{A})} |P_{n}\psi - \psi|_{\mathrm{H}} \, d\sigma \\ &\leq \int_{s}^{t} \langle \tilde{u}_{n}(\sigma) - \tilde{u}(\sigma), \mathrm{A}^{-1}\psi \rangle_{\mathrm{D}(\mathrm{A})} \, d\sigma + |P_{n}\psi - \psi|_{\mathrm{H}} |\tilde{u}_{n}|_{L^{2}(0,T;\mathrm{D}(\mathrm{A}))} T^{1/2} \end{split}$$

By (5.5.35) $\tilde{u}_n \to \tilde{u}$ weakly in $L^2(0,T; D(A)) \tilde{\mathbb{P}}$ -a.s., \tilde{u}_n is a uniformly bounded sequence in $L^2(0,T; D(A))$ and $P_n \psi \to \psi$ in H. Hence we have $\tilde{\mathbb{P}}$ -a.s.

$$\lim_{n \to \infty} \int_{s}^{t} \langle \tilde{u}_{n}(\sigma) - \tilde{u}(\sigma), \mathbf{A}^{-1}\psi \rangle_{\mathbf{D}(\mathbf{A})} \ d\sigma \to 0,$$

and

$$\lim_{n\to\infty}|P_n\psi-\psi|_{\rm H}\to 0.$$

Thus, we have shown that assertion (b) is true.

We will now prove assertion (*c*). Let $\psi \in V$. Then we have the following estimates:

$$\begin{split} &\int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)), P_{n}\psi \rangle_{\mathrm{H}} - \int_{s}^{t} \langle B(\tilde{u}(\sigma)), \psi \rangle_{\mathrm{H}} d\sigma \\ &= \int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)) - B(\tilde{u}(\sigma)), \psi \rangle_{\mathrm{H}} d\sigma + \int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)), P_{n}\psi - \psi \rangle_{\mathrm{H}} d\sigma \\ &= \int_{s}^{t} \left[b(\tilde{u}_{n}(\sigma), \tilde{u}_{n}(\sigma), \psi) - b(\tilde{u}(\sigma), \tilde{u}(\sigma), \psi) \right] d\sigma + \int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)), P_{n}\psi - \psi \rangle_{\mathrm{H}} d\sigma \end{split}$$

Using (3.2.1), we get

$$\begin{split} &\int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)), P_{n}\psi\rangle_{\mathrm{H}} - \int_{s}^{t} \langle B(\tilde{u}(\sigma)),\psi\rangle_{\mathrm{H}} d\sigma \\ &= \int_{s}^{t} b(\tilde{u}_{n}(\sigma) - \tilde{u}(\sigma), \tilde{u}_{n}(\sigma), \psi) d\sigma + \int_{s}^{t} b(\tilde{u}(\sigma), \tilde{u}_{n}(\sigma) - \tilde{u}(\sigma), \psi) d\sigma \\ &+ \int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)), P_{n}\psi - \psi\rangle_{\mathrm{H}} d\sigma \\ &\leq \int_{s}^{t} \|\tilde{u}_{n}(\sigma) - \tilde{u}(\sigma)\|_{\mathrm{V}} \|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}} \|\psi\|_{\mathrm{V}} d\sigma + \int_{s}^{t} \|\tilde{u}(\sigma)\|_{\mathrm{V}} \|\tilde{u}_{n}(\sigma) - \tilde{u}(\sigma)\|_{\mathrm{V}} \|\psi\|_{\mathrm{V}} d\sigma \\ &+ \int_{s}^{t} \|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2} \|P_{n}\psi - \psi\|_{\mathrm{V}} d\sigma. \end{split}$$

Now since by (5.5.35) $\tilde{u}_n \to \tilde{u}$ in $L^2(0,T;V)$ strongly, the sequence (\tilde{u}_n) is uniformly bounded in $L^2(0,T;V)$. Thus using the Cauchy-Schwarz inequality and the convergence of $P_n\psi \to \psi$ in V, we have $\tilde{\mathbb{P}}$ -a.s.

$$\begin{split} &\lim_{n \to \infty} \int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)), P_{n}\psi \rangle_{\mathrm{H}} - \int_{s}^{t} \langle B(\tilde{u}(\sigma)), \psi \rangle_{\mathrm{H}} d\sigma \\ &\leq \lim_{n \to \infty} |\tilde{u}_{n} - \tilde{u}|_{L^{2}(0,T;\mathrm{V})} \Big[|\tilde{u}_{n}|_{L^{2}(0,T;\mathrm{V})} \\ &+ |\tilde{u}|_{L^{2}(0,T;\mathrm{V})} \Big] \|\psi\|_{\mathrm{V}} + \lim_{n \to \infty} |\tilde{u}_{n}|_{L^{2}(0,T;\mathrm{V})}^{2} \|P_{n}\psi - \psi\|_{\mathrm{V}} \to 0. \end{split}$$

Next we deal with (*d*). Let $\psi \in H$, then

$$\begin{split} &\int_{s}^{t} |\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}_{n}(\sigma), P_{n}\psi \rangle_{\mathrm{H}} d\sigma - \int_{s}^{t} |\nabla \tilde{u}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma \\ &= \int_{s}^{t} \left[|\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} - |\nabla \tilde{u}(\sigma)|_{L^{2}}^{2} \right] \langle \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma + \int_{s}^{t} |\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}_{n}(\sigma) - \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma \\ &+ \int_{s}^{t} |\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}_{n}(\sigma), P_{n}\psi - \psi \rangle_{\mathrm{H}} d\sigma \\ &= \int_{s}^{t} \left[|\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} - |\nabla \tilde{u}(\sigma)|_{L^{2}}^{2} \right] \left[|\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} + |\nabla \tilde{u}(\sigma)|_{L^{2}}^{2} \right] \langle \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma \\ &+ \int_{s}^{t} |\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}_{n}(\sigma) - \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma + \int_{s}^{t} |\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}_{n}(\sigma), P_{n}\psi - \psi \rangle_{\mathrm{H}} d\sigma. \end{split}$$

Thus, by the Cauchy-Schwarz inequality we get

$$\begin{split} &\int_{s}^{t} |\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}_{n}(\sigma), P_{n}\psi \rangle_{\mathrm{H}} d\sigma - \int_{s}^{t} |\nabla \tilde{u}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma \\ &\leq \int_{s}^{t} [\|\tilde{u}_{n}(\sigma) - \tilde{u}(\sigma)\|_{\mathrm{V}}] [\|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}} + \|\tilde{u}(\sigma)\|_{\mathrm{V}}] |\tilde{u}(\sigma)|_{\mathrm{H}} |\psi|_{\mathrm{H}} d\sigma \\ &+ \int_{s}^{t} \|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2} |\tilde{u}_{n}(\sigma) - \tilde{u}(\sigma)|_{\mathrm{H}} |\psi|_{\mathrm{H}} d\sigma + \int_{s}^{t} \|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2} |\tilde{u}_{n}(\sigma)|_{\mathrm{H}} |P_{n}\psi - \psi|_{\mathrm{H}} d\sigma \end{split}$$

By (5.5.35) $\tilde{u}_n \to \tilde{u}$ strongly in $\mathscr{C}([0,T]; \mathbf{H}) \cap L^2(0,T; \mathbf{V})$, in particular $\tilde{u} \in L^2(0,T; \mathbf{V})$, the sequence (\tilde{u}_n) is uniformly bounded in $L^2(0,T; \mathbf{V})$ and $P_n \psi \to \psi$ in H. Thus, we have $\tilde{\mathbb{P}}$ -a.s.

$$\begin{split} &\lim_{n \to \infty} \int_{s}^{t} |\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}_{n}(\sigma), P_{n}\psi \rangle_{\mathrm{H}} d\sigma - \int_{s}^{t} |\nabla \tilde{u}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma \\ &\leq \lim_{n \to \infty} \left[|\tilde{u}_{n}|_{L^{2}(0,T;V)} + |\tilde{u}|_{L^{2}(0,T;V)} \right] |\tilde{u}_{n}|_{L^{\infty}(0,T;H)} |\tilde{u}_{n} - \tilde{u}|_{L^{2}(0,T;V)} |\psi|_{\mathrm{H}} \\ &+ \lim_{n \to \infty} |\tilde{u}_{n}|_{L^{2}(0,T;V)}^{2} |\tilde{u}_{n} - \tilde{u}|_{L^{\infty}(0,T;H)} |\psi|_{\mathrm{H}} + \lim_{n \to \infty} |\tilde{u}_{n}|_{L^{2}(0,T;V)}^{2} |\tilde{u}_{n}|_{L^{\infty}(0,T;H)} |P_{n}\psi - \psi|_{\mathrm{H}} \to 0. \end{split}$$

Hence we infer that assertion (d) holds.

Now we are left to show that (*e*) holds. Let $\psi \in H$, then

$$\begin{split} &\int_{s}^{t} \langle C^{2} \tilde{u}_{n}(\sigma), P_{n} \psi \rangle_{\mathrm{H}} d\sigma - \int_{s}^{t} \langle C^{2} \tilde{u}(\sigma), \psi \rangle_{\mathrm{H}} d\sigma \\ &= \int_{s}^{t} \langle C^{2} (\tilde{u}_{n}(\sigma) - \tilde{u}(\sigma)), \psi \rangle_{\mathrm{H}} d\sigma + \int_{s}^{t} \langle C^{2} \tilde{u}_{n}(\sigma), P_{n} \psi - \psi \rangle_{\mathrm{H}} d\sigma \\ &\leq \int_{s}^{t} \langle C^{2} A^{-1} \mathcal{A}(\tilde{u}_{n}(\sigma) - \tilde{u}(\sigma)), \psi \rangle_{\mathrm{H}} d\sigma + K_{c}^{2} \int_{s}^{t} |\tilde{u}_{n}(\sigma)|_{\mathrm{D}(\mathrm{A})} |P_{n} \psi - \psi|_{\mathrm{H}} d\sigma, \end{split}$$

where K_c is defined in (5.3.1). Since (\tilde{u}_n) is a uniformly bounded sequence in $L^2(0, T; D(A))$ and $C^2 A^{-1}$ is a bounded operator thus by (5.5.35), we have $\tilde{\mathbb{P}}$ -a.s.

$$\begin{split} &\lim_{n\to\infty}\int_{s}^{t}\langle C^{2}\tilde{u}_{n}(\sigma),P_{n}\psi\rangle_{\mathrm{H}}d\sigma - \int_{s}^{t}\langle C^{2}\tilde{u}(\sigma),\psi\rangle_{\mathrm{H}}d\sigma \\ &\leq \lim_{n\to\infty}\int_{s}^{t}\langle \mathrm{A}(\tilde{u}_{n}(\sigma)-\tilde{u}(\sigma)),(C^{2}\mathrm{A}^{-1})^{*}\psi\rangle_{\mathrm{H}}d\sigma + \lim_{n\to\infty}K_{c}^{2}|\tilde{u}|_{L^{2}(0,T;\mathrm{D}(\mathrm{A}))}|P_{n}\psi-\psi|_{\mathrm{H}}T^{1/2} \\ &= \lim_{n\to\infty}\int_{s}^{t}\langle \tilde{u}_{n}(\sigma)-\tilde{u}(\sigma),\mathrm{A}^{-1}(C^{2}\mathrm{A}^{-1})^{*}\psi\rangle_{\mathrm{D}(\mathrm{A})}d\sigma + \lim_{n\to\infty}K_{c}^{2}|\tilde{u}|_{L^{2}(0,T;\mathrm{D}(\mathrm{A}))}|P_{n}\psi-\psi|_{\mathrm{H}}T^{\frac{1}{2}} \to 0, \end{split}$$

where to establish the convergence we have used that $P_n \psi \rightarrow \psi$ in H.

This completes the proof of Lemma 5.5.8.

Let *h* be a bounded continuous function on $\mathscr{C}([0,T];H)$ and $\tilde{\mathbb{F}} = (\tilde{\mathscr{F}}_t) = \sigma\{\tilde{u}(s), s \leq t\}$ be the filtration of sigma fields generated by the process \tilde{u} .

Lemma 5.5.9. For all $s, t \in [0, T]$, such that $s \le t$ and all $\psi \in V$:

$$(5.5.36) \quad \lim_{n \to \infty} \tilde{\mathbb{E}} \left[\langle \tilde{M}_n(t) - \tilde{M}_n(s), \psi \rangle h(\tilde{u}_{n|[0,s]}) \right] = \tilde{\mathbb{E}} \left[\langle \tilde{M}(t) - \tilde{M}(s), \psi \rangle h(\tilde{u}_{|[0,s]}) \right], \quad h \in \mathscr{C}([0,T]; \mathbf{H}),$$
where $\langle \cdot, \cdot \rangle$ denotes the duality between V and V'.

Proof. Let us fix $s, t \in [0, T], s \le t$ and $\psi \in V$. By (5.5.29), we have

$$\begin{split} \langle \tilde{M}_{n}(t) - \tilde{M}_{n}(s), \psi \rangle &= \langle \tilde{u}_{n}(t), P_{n}\psi \rangle_{\mathrm{H}} - \langle \tilde{u}_{n}(s), P_{n}\psi \rangle_{\mathrm{H}} + \int_{s}^{t} \langle \mathrm{A}\tilde{u}_{n}(\sigma), P_{n}\psi \rangle_{\mathrm{H}} \, d\sigma \\ &+ \int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)), P_{n}\psi \rangle \, d\sigma - \int_{s}^{t} |\nabla \tilde{u}_{n}(\sigma)|_{L^{2}}^{2} \langle \tilde{u}_{n}(\sigma), P_{n}\psi \rangle_{\mathrm{H}} \, d\sigma \\ &- \frac{1}{2} \int_{s}^{t} \langle C^{2}\tilde{u}_{n}(\sigma), P_{n}\psi \rangle_{\mathrm{H}} \, d\sigma. \end{split}$$

By Lemma 5.5.8, we infer that

(5.5.37)
$$\lim_{n \to \infty} \langle \tilde{M}_n(t) - \tilde{M}_n(s), \psi \rangle = \langle \tilde{M}(t) - \tilde{M}(s), \psi \rangle, \quad \tilde{\mathbb{P}}\text{-a.s.}$$

In order to prove (5.5.36) we first observe that since $\tilde{u}_n \to \tilde{u}$ in \mathcal{Z}_T , in particular in $\mathscr{C}([0,T];H)$ and h is a bounded continuous function on $\mathscr{C}([0,T];H)$, we get

(5.5.38)
$$\lim_{n \to \infty} h(\tilde{u}_{n|[0,s]}) = h(\tilde{u}_{|[0,s]}) \quad \tilde{\mathbb{P}} - a.s.$$

and

$$(5.5.39) \qquad \qquad \sup_{n \in \mathbb{N}} |h(\tilde{u}_{n|[0,s]})|_{L^{\infty}} < \infty.$$

Let us define a sequence of \mathbb{R} -valued random variables:

$$f_n(\omega) := \left[\langle \tilde{M}_n(t,\omega), \psi \rangle - \langle \tilde{M}_n(s,\omega), \psi \rangle \right] h(\tilde{u}_{n|[0,s]}), \quad \omega \in \tilde{\Omega}.$$

We will prove that the functions $\{f_n\}_{n \in \mathbb{N}}$ are uniformly integrable in order to apply the Vitali theorem later on. We claim that

(5.5.40)
$$\sup_{n\geq 1} \tilde{\mathbb{E}}[|f_n|^2] < \infty.$$

By the Cauchy-Schwarz inequality and the embedding $V' \hookrightarrow H$, for each $n \in \mathbb{N}$ there exists a positive constant *c* such that

(5.5.41)
$$\tilde{\mathbb{E}}[|f_n|^2] \le 2c|h|_{L^{\infty}}^2 |\psi|_{V}^2 \tilde{\mathbb{E}}\left[|\tilde{M}_n(t)|_{H}^2 + |\tilde{M}_n(s)|_{H}^2\right].$$

Since \tilde{M}_n is a continuous martingale with quadratic variation defined in (5.5.30), by the Burkholder-Davis-Gundy inequality we obtain

(5.5.42)
$$\tilde{\mathbb{E}}\left[\sup_{t\in[0,T]}|\tilde{M}_{n}(t)|_{\mathrm{H}}^{2}\right] \leq c\tilde{\mathbb{E}}\left[\sum_{j=1}^{m}\int_{0}^{T}|P_{n}C_{j}\tilde{u}_{n}(\sigma)|_{\mathrm{H}}^{2}d\sigma\right].$$

Since $P_n: H \to H$ is a contraction then by assumption (A.1) and (5.5.25) for p = 1, we have

(5.5.43)
$$\begin{split} \tilde{\mathbb{E}}\left[\sum_{j=1}^{m}\int_{0}^{T}|P_{n}C_{j}\tilde{u}_{n}(\sigma)|_{\mathrm{H}}^{2}\,d\sigma\right] &\leq \tilde{\mathbb{E}}\left[mK_{c}^{2}\int_{0}^{T}\|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2}d\sigma\right] \\ &\leq mK_{c}^{2}\tilde{\mathbb{E}}\left[\sup_{\sigma\in[0,T]}\|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2}\right]T < \infty. \end{split}$$

Then, by (5.5.41) and (5.5.43) we see that (5.5.40) holds. Since the sequence $\{f_n\}_{n\in\mathbb{N}}$ is uniformly integrable and by (5.5.37) it is $\tilde{\mathbb{P}}$ -a.s. point-wise convergent, then application of the Vitali Theorem (see Theorem 2.5.14) completes the proof of the lemma.

From Lemma 5.5.6 and Lemma 5.5.9 we have the following corollary.

Corollary 5.5.10. For all $s, t \in [0, T]$ such that $s \le t$:

$$\mathbb{E}\left(\tilde{M}(t)-\tilde{M}(s)\big|\tilde{\mathscr{F}}_t\right)=0.$$

Lemma 5.5.11. For all $s, t \in [0,T]$ such that $s \le t$ and all $\psi, \zeta \in V$:

$$\begin{split} &\lim_{n\to\infty} \tilde{\mathbb{E}}\Big[\Big(\langle \tilde{M}_n(t),\psi\rangle\langle \tilde{M}_n(t),\zeta\rangle-\langle \tilde{M}_n(s),\psi\rangle\langle \tilde{M}_n(s),\zeta\rangle\Big)h(\tilde{u}_{n\mid[0,s]})\Big]\\ &=\tilde{\mathbb{E}}\Big[\Big(\langle \tilde{M}(t),\psi\rangle\langle \tilde{M}(t),\zeta\rangle-\langle \tilde{M}(s),\psi\rangle\langle \tilde{M}(s),\zeta\rangle\Big)h(\tilde{u}_{\mid[0,s]})\Big],\quad h\in\mathscr{C}([0,T];\mathbf{H}), \end{split}$$

where $\langle \cdot, \cdot \rangle$ denotes the dual pairing between V and V'.

Proof. Let us fix $s, t \in [0, T]$ such that $s \le t$ and $\psi, \zeta \in V$ and define the random variables f_n and f by

$$\begin{split} f_{n}(\omega) &:= \Big(\langle \tilde{M}_{n}(t,\omega),\psi\rangle \langle \tilde{M}_{n}(t,\omega),\zeta\rangle - \langle \tilde{M}_{n}(s,\omega),\psi\rangle \langle \tilde{M}_{n}(s,\omega),\zeta\rangle \Big) h(\tilde{u}_{n\mid[0,s]}(\omega)), \\ f(\omega) &:= \Big(\langle \tilde{M}(t,\omega),\psi\rangle \langle \tilde{M}(t,\omega),\zeta\rangle - \langle \tilde{M}(s,\omega),\psi\rangle \langle \tilde{M}(s,\omega),\zeta\rangle \Big) h(\tilde{u}_{\mid[0,s]}(\omega)), \quad \omega \in \tilde{\Omega}. \end{split}$$

By (5.5.37) and (5.5.38) we infer that $\lim_{n\to\infty} f_n(\omega) = f(\omega)$, for $\tilde{\mathbb{P}}$ almost all $\omega \in \tilde{\Omega}$. We will prove that the functions $\{f_n\}_{n\in\mathbb{N}}$ are uniformly integrable. We claim that for some r > 1,

$$(5.5.44) \qquad \qquad \sup_{n\geq 1} \tilde{\mathbb{E}}\left[|f_n|^r\right] < \infty.$$

For each $n \in \mathbb{N}$, as before we have

(5.5.45)
$$\tilde{\mathbb{E}}\left[\left|f_{n}\right|^{r}\right] \leq C \|h\|_{L^{\infty}}^{r} \|\psi\|_{V}^{r} \|\zeta\|_{V}^{r} \tilde{\mathbb{E}}\left[\left|\tilde{M}_{n}(t)\right|^{2r} + |\tilde{M}_{n}(s)|^{2r}\right].$$

Since \tilde{M}_n is a continuous martingale with quadratic variation defined in (5.5.29), by the Burkholder-Davis-Gundy inequality we obtain

(5.5.46)
$$\tilde{\mathbb{E}}\left[\sup_{t\in[0,T]}|\tilde{M}_{n}(t)|^{2r}\right] \leq c\tilde{\mathbb{E}}\left[\sum_{j=1}^{m}\int_{0}^{T}|P_{n}C_{j}\tilde{u}_{n}(\sigma)|_{\mathrm{H}}^{2}d\sigma\right]^{r}.$$

Since $P_n: H \to H$ is a contraction, by assumption (A.1), we have

(5.5.47)
$$\begin{split} \tilde{\mathbb{E}}\left[\sum_{j=1}^{m}\int_{0}^{T}|P_{n}C_{j}\tilde{u}_{n}(\sigma)|_{\mathrm{H}}^{2}d\sigma\right]^{r} &\leq \tilde{\mathbb{E}}\left[mK_{c}^{2}\int_{0}^{T}\|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2}d\sigma\right]^{r} \\ &\leq (mK_{c}^{2})^{r}\,\tilde{\mathbb{E}}\left(\sup_{\sigma\in[0,T]}\|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2r}\right)T^{r}. \end{split}$$

Thus for $r \in (1, 1 + \frac{1}{K_c^2})$, by (5.5.45), (5.5.46), (5.5.47) and (5.5.25) we infer that condition (5.5.44) holds. By the Vitali theorem

(5.5.48)
$$\lim_{n \to \infty} \tilde{\mathbb{E}}[f_n] = \tilde{\mathbb{E}}[f].$$

The proof of the lemma is thus complete.

Lemma 5.5.12 (Convergence of quadratic variations). For any $s, t \in [0,T]$ and $\psi, \zeta \in V$, for all $h \in \mathcal{C}([0,T];H)$ we have

$$\lim_{n\to\infty} \tilde{\mathbb{E}}\left[\left(\sum_{j=1}^{m} \int_{s}^{t} \left\langle \left(C_{j}\tilde{u}_{n}(\sigma)\right)^{*} P_{n}\psi, \left(C_{j}\tilde{u}_{n}(\sigma)\right)^{*} P_{n}\zeta\right\rangle_{\mathbb{R}} d\sigma\right) \cdot h(\tilde{u}_{n\mid[0,s]})\right]$$
$$= \tilde{\mathbb{E}}\left[\left(\sum_{j=1}^{m} \int_{s}^{t} \left\langle \left(C_{j}\tilde{u}(\sigma)\right)^{*}\psi, \left(C_{j}\tilde{u}(\sigma)\right)^{*}\zeta\right\rangle_{\mathbb{R}} d\sigma\right) \cdot h(\tilde{u}_{\mid[0,s]})\right].$$

Proof. Let us fix $\psi, \zeta \in V$ and define a sequence of random variables by

$$f_{n}(\omega) := \left(\sum_{j=1}^{m} \int_{s}^{t} \left\langle \left(C_{j}\tilde{u}_{n}(\sigma,\omega)\right)^{*} P_{n}\psi, \left(C_{j}\tilde{u}_{n}(\sigma,\omega)\right)^{*} P_{n}\zeta\right\rangle_{\mathbb{R}} d\sigma \right) \cdot h(\tilde{u}_{n\mid[0,s]}), \quad \omega \in \tilde{\Omega}.$$

We will prove that these random variables are uniformly integrable and convergent $\tilde{\mathbb{P}}$ -a.s. to some random variable *f*. In order to do that we will show that for some r > 1,

(5.5.49)
$$\sup_{n\geq 1} \tilde{\mathbb{E}}|f_n|^r < \infty.$$

Since $P_n: H \to H$ is a contraction, by the Cauchy-Schwarz inequality, and assumption (A.1) there exists a positive constant *c* such that

$$\begin{split} \left| \left(C_{j} \tilde{u}_{n}(\sigma, \omega) \right)^{*} P_{n} \psi \right|_{\mathbb{R}} &\leq \left| \left(C_{j} \tilde{u}_{n}(\sigma, \omega) \right)^{*} \right|_{\mathcal{L}(\mathrm{H};\mathbb{R})} |P_{n} \psi|_{\mathrm{H}} \leq |C_{j} \tilde{u}_{n}(\sigma, \omega)|_{\mathcal{L}(\mathbb{R};\mathrm{H})} |\psi|_{\mathrm{H}} \\ &\leq K_{c} \left\| \tilde{u}_{n}(\sigma, \omega) \right\|_{\mathrm{V}} |\psi|_{\mathrm{H}}, \qquad j \in \{1, \cdots, m\}, \end{split}$$

where $\mathscr{L}(X,Y)$ denotes the operator norm of the linear operators from X to Y. Thus using the Hölder inequality, we obtain

$$\begin{split} \tilde{\mathbb{E}} |f_{n}|^{r} &= \tilde{\mathbb{E}} \left| \left(\sum_{j=1}^{m} \int_{s}^{t} \left\langle \left(C_{j} \tilde{u}_{n}(\sigma) \right)^{*} P_{n} \psi, \left(C \tilde{u}_{n}(\sigma) \right)^{*} P_{n} \zeta \right\rangle_{\mathbb{R}} d\sigma \right) \cdot h(\tilde{u}_{n \mid [0,s]}) \right|^{r} \\ &\leq |h|_{L^{\infty}}^{r} \tilde{\mathbb{E}} \left(\sum_{j=1}^{m} \int_{s}^{t} \left| \left(C_{j} \tilde{u}_{n}(\sigma) \right)^{*} P_{n} \psi \right|_{\mathbb{R}} \cdot \left| \left(C_{j} \tilde{u}_{n}(\sigma) \right)^{*} P_{n} \zeta \right|_{\mathbb{R}} d\sigma \right)^{r} \\ &\leq (m K_{c}^{2})^{r} |h|_{L^{\infty}}^{r} |\psi|_{\mathrm{H}}^{r} |\zeta|_{\mathrm{H}}^{r} \tilde{\mathbb{E}} \left(\int_{s}^{t} \|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2} d\sigma \right)^{r} \\ &\leq (m K_{c}^{2})^{r} |h|_{L^{\infty}}^{r} |\psi|_{\mathrm{H}}^{r} |\zeta|_{\mathrm{H}}^{r} \tilde{\mathbb{E}} \left(\sup_{\sigma \in [0,T]} \|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2r} \right) T^{r}. \end{split}$$

$$(5.5.50)$$

Therefore using (5.5.50) and (5.5.25) we infer that (5.5.49) holds for every $r \in (1, 1 + \frac{1}{K_c^2})$.

Now for pointwise convergence we will show that for a fix $\omega \in \tilde{\Omega}$,

(5.5.51)
$$\lim_{n \to \infty} \int_{s}^{t} \sum_{j=1}^{m} \left\langle \left(C_{j} \tilde{u}_{n}(\sigma, \omega) \right)^{*} P_{n} \psi, \left(C_{j} \tilde{u}_{n}(\sigma, \omega) \right)^{*} P_{n} \zeta \right\rangle_{\mathbb{R}} d\sigma$$
$$= \int_{s}^{t} \sum_{j=1}^{m} \left\langle \left(C_{j} \tilde{u}(\sigma, \omega) \right)^{*} \psi, \left(C_{j} \tilde{u}(\sigma, \omega) \right)^{*} \zeta \right\rangle_{\mathbb{R}} d\sigma.$$

Let us fix $\omega \in \tilde{\Omega}$ such that

- (i) $\tilde{u}_n(\cdot,\omega) \to \tilde{u}(\cdot,\omega)$ in $L^2(0,T;\mathbf{V})$,
- (ii) and the sequence $(\tilde{u}_n(\cdot,\omega))_{n\geq 1}$ is uniformly bounded in $L^2(0,T;\mathbf{V})$.

Note that to prove (5.5.51), it is sufficient to prove that

(5.5.52)
$$(C_j \tilde{u}_n(\sigma, \omega))^* P_n \psi \to (C_j \tilde{u}(\sigma, \omega))^* \psi \text{ in } L^2(s, t; \mathbb{R}),$$

for every $j \in \{1, \dots, m\}$. Using the Cauchy-Schwarz inequality we have

$$\begin{split} &\int_{s}^{t} \left| \left(C_{j} \tilde{u}_{n}(\sigma, \omega) \right)^{*} P_{n} \psi - \left(C_{j} \tilde{u}(\sigma, \omega) \right)^{*} \psi \right|_{\mathbb{R}}^{2} d\sigma \\ &\leq \int_{s}^{t} \left(\left| \left(C_{j} \tilde{u}_{n}(\sigma, \omega) \right)^{*} \left(P_{n} \psi - \psi \right) \right|_{\mathbb{R}} + \left| \left(C_{j} \tilde{u}_{n}(\sigma, \omega) - C_{j} \tilde{u}(\sigma, \omega) \right)^{*} \psi \right|_{\mathbb{R}} \right)^{2} d\sigma \\ &\leq 2 \int_{s}^{t} \left| C_{j} \tilde{u}_{n}(\sigma, \omega) \right|_{\mathscr{L}(\mathbb{R}; \mathrm{H})}^{2} \left| P_{n} \psi - \psi \right|_{\mathrm{H}}^{2} d\sigma + 2 \int_{s}^{t} \left| C_{j} \tilde{u}_{n}(\sigma, \omega) - C_{j} \tilde{u}(\sigma, \omega) \right|_{\mathscr{L}(\mathbb{R}; \mathrm{H})}^{2} \left| \psi \right|_{\mathrm{H}}^{2} d\sigma \\ &=: I_{n}^{1}(t) + I_{n}^{2}(t). \end{split}$$

We will deal with each of the terms individually. We start with $I_n^1(t)$. Since

$$\lim_{n\to\infty}|P_n\psi-\psi|_{\rm H}=0,\quad\psi\in{\rm V},$$

and by assumption (A.1), (ii) there exists a positive constant K such that

$$\sup_{n\geq 1}\int_{s}^{t}|C\tilde{u}_{n}(\sigma,\omega)|_{\mathscr{L}(\mathbb{R};\mathrm{H})}^{2}\,d\sigma\leq K_{c}^{2}\sup_{n\geq 1}\int_{s}^{t}\|\tilde{u}_{n}(\sigma,\omega)\|_{\mathrm{V}}^{2}\,d\sigma\leq K.$$

Thus we infer that

 $\lim_{n\to\infty}I_n^1(t)=0.$

Next we consider $I_n^2(t)$. Using assumption (A.1) and (i) we can show that for every $j \in \{1, \dots, m\}$,

$$\begin{split} \lim_{n \to \infty} & \int_{s}^{t} \left| C_{j} \tilde{u}_{n}(\sigma, \omega) - C_{j} \tilde{u}_{(\sigma, \omega)} \right|_{\mathscr{L}(\mathbb{R}; \mathrm{H})}^{2} |\psi|_{\mathrm{H}}^{2} d\sigma \\ & \leq \lim_{n \to \infty} |\psi|_{\mathrm{H}}^{2} K_{c}^{2} \int_{s}^{t} \|\tilde{u}_{n}(\sigma, \omega) - \tilde{u}(\sigma, \omega)\|_{\mathrm{V}}^{2} d\sigma = 0. \end{split}$$

Hence, we have proved (5.5.52), finishing the proof of lemma.

By Lemma 5.5.9 we can pass to the limit in (5.5.31). By Lemmas 5.5.11 and 5.5.12 we can pass to the limit in (5.5.32) as well. After passing to the limits we infer that for all $\psi, \zeta \in V$ and $h \in \mathscr{C}([0, T]; H)$:

(5.5.53)
$$\tilde{\mathbb{E}}\left[\langle \tilde{M}(t) - \tilde{M}(s), \psi \rangle h(\tilde{u}_{|[0,s]})\right] = 0,$$

and

(5.5.54)
$$\widetilde{\mathbb{E}}\Big[\Big(\langle \tilde{M}(t),\psi\rangle\langle \tilde{M}(t),\zeta\rangle-\langle \tilde{M}(s),\psi\rangle\langle \tilde{M}(s),\zeta\rangle\\ -\sum_{j=1}^{m}\int_{s}^{t}\langle \big(C_{j}\tilde{u}(\sigma)\big)^{*}\psi,\big(C_{j}\tilde{u}(\sigma)\big)^{*}\zeta\rangle_{\mathbb{R}}d\sigma\Big)\cdot h(\tilde{u}_{|[0,s]})\Big]=0.$$

From the two previous lemmas and Lemma 5.5.6, we infer the following corollary.

Corollary 5.5.13. *For* $t \in [0, T]$

$$\langle\langle \tilde{M} \rangle\rangle_t = \int_0^t \sum_{j=1}^m \left| C_j \tilde{u}(s) \right|_{\mathrm{H}}^2 ds, \qquad t \in [0,T].$$

Theorem 5.3.3 proof continued. Now we apply the idea analogous to that used by Da Prato and Zabczyk, see [38, Section 8.3]. By Lemma 5.5.7 and Corollary 5.5.10, we infer that $\tilde{M}(t)$, $t \in [0, T]$ is an H-valued continuous square integrable martingale with respect to the filtration $\tilde{\mathbb{F}} = (\tilde{\mathscr{F}}_t)_{t\geq 0}$. Moreover, by Corollary 5.5.13 the quadratic variation of \tilde{M} is given by

$$\langle \langle \tilde{M} \rangle \rangle_t = \int_0^t \sum_{j=1}^m \left| C_j \tilde{u}(s) \right|_{\mathrm{H}}^2 ds, \qquad t \in [0,T].$$

Therefore by the martingale representation theorem (see Theorem 2.8.7), there exist

- a stochastic basis $(\tilde{\tilde{\Omega}}, \tilde{\tilde{\mathscr{F}}}, \tilde{\tilde{\mathscr{F}}}_{t \ge 0}, \tilde{\tilde{\mathbb{P}}}),$
- a \mathbb{R}^m -valued $\tilde{\mathbb{F}}$ -Wiener process $\tilde{W}(t)$ defined on this basis,
- and a progressively measurable process $\tilde{\tilde{u}}(t)$ such that for all $t \in [0, T]$ and $v \in V$:

$$\begin{split} \langle \tilde{\tilde{u}}(t), \mathbf{v} \rangle &- \langle \tilde{\tilde{u}}(0), \mathbf{v} \rangle + \int_0^t \langle \mathbf{A} \tilde{\tilde{u}}(s), \mathbf{v} \rangle \, ds + \int_0^t \langle B(\tilde{\tilde{u}}(s)), \mathbf{v} \rangle \, ds \\ &= \int_0^t |\nabla \tilde{\tilde{u}}(s)|_{L^2}^2 \langle \tilde{\tilde{u}}(s), \mathbf{v} \rangle \, ds + \frac{1}{2} \int_0^t \sum_{j=1}^m \langle C_j^2 \tilde{\tilde{u}}(s), \mathbf{v} \rangle \, ds + \int_0^t \sum_{j=1}^m \langle C_j \tilde{\tilde{u}}(s), \mathbf{v} \rangle \, d\tilde{W}(s). \end{split}$$

Thus the conditions from Definition 5.3.2 hold with $(\hat{\Omega}, \hat{\mathscr{F}}, \{\hat{\mathscr{F}}_t\}_{t\geq 0}, \hat{\mathbb{P}}) = (\tilde{\tilde{\Omega}}, \tilde{\tilde{\mathscr{F}}}, \{\tilde{\tilde{\mathscr{F}}}_t\}_{t\geq 0}, \tilde{\mathbb{P}}), \hat{W} = \tilde{\tilde{W}}$ and $\hat{u} = \tilde{\tilde{u}}$. Hence the proof of Theorem 5.3.3 is complete.

5.6 Pathwise uniqueness and strong solution

In this section we will show that the solutions of (5.2.4) are pathwise unique and that problem (5.2.4) has a strong solution in PDE as well as in probabilistic sense. In the previous section we showed that paths of martingale solution u of (5.2.4) belong to $\mathscr{C}([0,T]; V_w) \cap L^2(0,T; D(A))$. We start by proving Lemma 5.3.5, i.e. showing that the trajectories of the solution $u \in \mathscr{C}([0,T]; V) \cap L^2(0,T; D(A))$. *Proof of Lemma 5.3.5* u is a martingale solution of (5.2.4) thus, $u \in \mathscr{C}([0, T]; V_w) \cap L^2(0, T; D(A))$ $\hat{\mathbb{P}}$ -a.s. We start by showing that RHS of (5.3.5) makes sense. In order to do so we will show that each term on the RHS is well defined.

Firstly we consider the non-linear term arising from Navier-Stokes. Using (3.2.3), the Hölder inequality, (5.5.24) and the estimate (5.3.4), we have the following bounds :

$$\hat{\mathbb{E}} \int_{0}^{T} |B(u(s))|_{\mathrm{H}}^{2} ds \leq 2 \hat{\mathbb{E}} \int_{0}^{T} |u(s)|_{\mathrm{H}} |\nabla u(s)|_{L^{2}}^{2} |u(s)|_{\mathrm{D}(\mathrm{A})} ds$$

$$\leq 2T^{1/2} \left(\hat{\mathbb{E}} sup_{s \in [0,T]} \|u(s)\|_{\mathrm{V}}^{4} \right)^{1/2} \left(\hat{\mathbb{E}} \int_{0}^{T} |u(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds \right)^{1/2} < \infty.$$

Using (5.5.24), the Hölder inequality, (5.5.23), estimates (5.5.25) and (5.3.4) we have the following inequalities for the non-linear term generated from the projection of the Stokes operator,

$$\hat{\mathbb{E}} \int_{0}^{T} \left| |\nabla u(s)|_{L^{2}}^{2} u(s) \right|_{H}^{2} ds = \hat{\mathbb{E}} \int_{0}^{T} |\nabla u(s)|_{L^{2}}^{4} ds \leq T \left(\hat{\mathbb{E}} \sup_{s \in [0,T]} \|u(s)\|_{V}^{4} \right) < \infty.$$

Next we deal with the correction term arising from the Stratonovich integral. Using assumption (A.1) and (5.3.4), for every $j \in \{1, \dots, m\}$ we have

$$\hat{\mathbb{E}} \int_{0}^{T} |C_{j}^{2} u(s)|_{\mathrm{H}}^{2} \leq K_{c}^{4} \hat{\mathbb{E}} \int_{0}^{T} |u(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds < \infty,$$

where K_c is defined in (5.3.1).

We are left to show that the Itô integral belongs to $L^2(\Omega \times [0,T]; V)$. Due to Itô isometry it is enough to show that for every $j \in \{1, \dots, m\}$

Using assumption (A.1) and (5.3.4), we have

$$\hat{\mathbb{E}} \int_{0}^{T} \|C_{j}u(s)\|_{V}^{2} ds \leq K_{c} \hat{\mathbb{E}} \int_{0}^{T} |u(s)|_{D(A)}^{2} ds < \infty.$$

Thus we have shown that each term in (5.3.5) is well defined. Now we will show that the equality holds.

Since *u* is a martingale solution of (5.2.4), for every $v \in V$ and $t \in [0, T]$ it satisfies the equality (5.3.2), i.e. $\hat{\mathbb{P}}$ -a.s.

$$\begin{aligned} \langle u(t), \mathbf{v} \rangle - \langle u_0, \mathbf{v} \rangle + \int_0^t \langle Au(s), \mathbf{v} \rangle \, ds + \int_0^t \langle B(u(s)), \mathbf{v} \rangle \, ds \\ &= \int_0^t |\nabla u(s)|_{L^2}^2 \langle u(s), \mathbf{v} \rangle \, ds + \frac{1}{2} \int_0^t \sum_{j=1}^m \langle C_j^2 u(s), \mathbf{v} \rangle \, ds + \int_0^t \sum_{j=1}^m \langle C_j u(s), \mathbf{v} \rangle \, d\hat{W}_j(s). \end{aligned}$$

Note that the above equation holds true for every $v \in V$ (as defined in (3.1.2)) and hence (5.3.5) holds in the distribution sense. But since V is dense in V, equality (5.3.5) holds true almost everywhere, which justifies Remark 5.3.4.
We use [71, Lemma 4.1] to prove the first part of the lemma. We work with the $D(A) \subset V \subset H$ space triple. Let us rewrite (5.3.5) in the following form

$$u(t) = u_0 + \int_0^t g(s) ds + N(t),$$

where *g* contains all the deterministic terms and *N* corresponds to the noise term. We have shown that $g \in L^2(\Omega; L^2(0, T; H))$ and $N \in L^2(\Omega; L^2(0, T; V))$. Thus from [71, Lemma 4.1] we infer that $u \in L^2(\Omega; \mathscr{C}([0, T]; V))$. This concludes the proof of lemma.

In the following lemma we will prove that the solutions of (5.2.4) are pathwise unique. The proof uses the Schmalfuss idea of application of the Itô formula for appropriate function (see [82]).

Lemma 5.6.1. Assume that the assumptions (A.1) - (A.2) are satisfied. If u_1, u_2 are two martingale solutions of (5.2.4) defined on the same filtered probability space $(\hat{\Omega}, \hat{\mathscr{F}}, \hat{\mathbb{F}}, \hat{\mathbb{P}})$ then $\hat{\mathbb{P}}$ -a.s. for all $t \in [0, T]$, $u_1(t) = u_2(t)$.

Proof. Let us denote the difference of the two solutions by $U := u_1 - u_2$. Then U satisfies the following equation

(5.6.2)
$$dU(t) + [AU(t) + B(u_{2}(t)) - B(u_{1}(t))] dt = \left[|\nabla u_{1}(t)|_{L^{2}}^{2} u_{1}(t) - |\nabla u_{2}(t)|_{L^{2}}^{2} u_{2}(t) \right] dt + \sum_{j=1}^{m} C_{j}U(t) \circ dW_{j}(t), \quad t \in [0, T].$$

Let us define the stopping time

(5.6.3)
$$\tau_N := T \wedge \inf\{t \in [0,T] : \|u_1(t)\|_V^2 \lor \|u_2(t)\|_V^2 > N\}, \quad N \in \mathbb{N}$$

Since $\hat{\mathbb{E}}\left[\sup_{t\in[0,T]} \|u_i(t)\|_V^2\right] < \infty \hat{\mathbb{P}}$ -a.s. for i = 1, 2, $\lim_{N\to\infty} \tau_N = T$.

We apply the Itô formula to the function

$$F(t,x) = e^{-r(t)} |x|_{\mathrm{H}}^2, \quad t \in [0,T], \ x \in \mathrm{V}$$

where r(t), $t \in [0, T]$, is a real valued \mathscr{C}^1 -class function which will be defined precisely later in the proof.

Since

$$\frac{\partial F}{\partial t} = -r'(t)e^{-r(t)}|x|_{\mathrm{H}}^2, \quad \frac{\partial F}{\partial x}(\cdot) = 2e^{-r(t)}\langle x, \cdot \rangle_{\mathrm{H}},$$

we obtain for all $t \in [0, T]$

$$\begin{split} e^{-r(t\wedge\tau_{N})}|U(t\wedge\tau_{N})|_{\mathrm{H}}^{2} &= \int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \left(-r'(s)|U(s)|_{\mathrm{H}}^{2} + 2\langle -\mathrm{A}U(s) + B(u_{1}(s)) - B(u_{2}(s)), U(s)\rangle_{\mathrm{H}}\right) ds \\ &+ \int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \left(2\langle |\nabla u_{1}(s)|_{L^{2}}^{2} u_{1}(s) - |\nabla u_{2}(s)|_{L^{2}}^{2} u_{2}(s), U(s)\rangle_{\mathrm{H}} + \sum_{j=1}^{m} \langle C_{j}^{2}U(s), U(s)\rangle_{\mathrm{H}}\right) ds \\ &+ \frac{1}{2} \int_{0}^{t\wedge\tau_{N}} \sum_{j=1}^{m} Tr \left[C_{j}U(s)\frac{\partial^{2}F}{\partial x^{2}}(C_{j}U(s))^{*}\right] ds + 2 \int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \sum_{j=1}^{m} \langle C_{j}U(s), U(s)\rangle_{\mathrm{H}} dW(s). \end{split}$$

Thus using the assumption (A.1), we obtain the following simplified expression

$$\begin{split} e^{-r(t\wedge\tau_{N})}|U(t\wedge\tau_{N})|_{\mathrm{H}}^{2} &\leq \int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \left(-r'(s)|U(s)|_{\mathrm{H}}^{2} - 2\|U(s)\|_{\mathrm{V}}^{2} - 2b(U(s), u_{1}(s), U(s)) \right) ds \\ &+ 2\int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \left((|\nabla u_{1}(s)|_{L^{2}}^{2} - |\nabla u_{2}(s)|_{L^{2}}^{2}) \langle u_{1}(s), U(s) \rangle_{\mathrm{H}} + |\nabla u_{2}(s)|_{L^{2}}^{2} |U(s)|_{\mathrm{H}}^{2} \right) ds \\ &+ \int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \sum_{j=1}^{m} \left(\langle C_{j}^{2}U(s), U(s) \rangle_{\mathrm{H}} + \frac{1}{2} \times 2 \langle C_{j}U(s), C_{j}U(s) \rangle_{\mathrm{H}} \right) ds. \end{split}$$

Using (3.2.1) and the Cauchy-Schwarz inequality, we get

$$\begin{split} e^{-r(t\wedge\tau_{N})}|U(t\wedge\tau_{N})|_{\mathrm{H}}^{2} + 2\int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \|U(s)\|_{\mathrm{V}}^{2} ds \\ &\leq \int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \left(-r'(s)|U(s)|_{\mathrm{H}}^{2} + 4|U(s)|_{\mathrm{H}} \|U(s)\|_{\mathrm{V}} \|u_{1}(s)\|_{\mathrm{V}}\right) ds \\ &\quad + 2\int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \|U(s)\|_{\mathrm{V}} \left(|\nabla u_{1}(s)|_{L^{2}} + |\nabla u_{2}(s)|_{L^{2}}\right) |u_{1}(s)|_{\mathrm{H}} |U(s)|_{\mathrm{H}} ds \\ &\quad + 2\int_{0}^{t\wedge\tau_{N}} e^{-r(s)} |\nabla u_{2}(t)|_{L^{2}}^{2} |U(s)|_{\mathrm{H}}^{2} ds. \end{split}$$

Using the Young inequality, we obtain

$$e^{-r(t\wedge\tau_{N})}|U(t\wedge\tau_{N})|_{\mathrm{H}}^{2} + 2\int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \|U(s)\|_{\mathrm{V}}^{2} ds \leq \int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \left[-r'(s) + 8\|u_{1}(s)\|_{\mathrm{V}}^{2}\right] |U(s)|_{\mathrm{H}}^{2} ds$$

+ $2\int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \left(|\nabla u_{1}(s)|_{L^{2}} + |\nabla u_{2}(s)|_{L^{2}}\right)^{2} |u_{1}(s)|_{\mathrm{H}}^{2} |U(s)|_{\mathrm{H}}^{2} ds$
(5.6.4) $+ \int_{0}^{t\wedge\tau_{N}} e^{-r(s)} \|U(s)\|_{\mathrm{V}}^{2} ds.$

Now choosing

$$r(t) := \int_0^t \left[8 \|u_1(s)\|_{\mathbf{V}}^2 + 2 \left(|\nabla u_1(s)|_{L^2} + |\nabla u_2(s)|_{L^2} \right)^2 |u_1(s)|_{\mathbf{H}}^2 \right] ds,$$

inequality (5.6.4) reduces to

$$e^{-r(t\wedge\tau_N)}|U(t\wedge\tau_N)|_{\rm H}^2 + \int_0^{t\wedge\tau_N} e^{-r(s)} \|U(s)\|_{\rm V}^2 ds \le 0.$$

In particular

(5.6.5)
$$\sup_{t \in [0,T]} \left[e^{-r(t \wedge \tau_N)} |U(t \wedge \tau_N)|_{\mathrm{H}}^2 \right] = 0.$$

Note that since u_1 and u_2 are the martingale solutions of (5.2.4) satisfying the estimates (5.5.4) and (5.5.6) and because of the Lemma 5.5.1, r is well defined for all $t \in [0, T]$.

Since $\hat{\mathbb{P}}$ -a.s. $\lim_{N\to\infty} \tau_N = T$ and $\hat{\mathbb{E}}[r(T)] < \infty$, thus from (5.6.5) we infer that $\hat{\mathbb{P}}$ -a.s. for all $t \in [0, T]$, U(t) = 0. The proof of the lemma is thus complete.

Definition 5.6.2. Let $(\Omega^i, \mathscr{F}^i, \mathbb{P}^i, \mathbb{P}^i, W^i, u^i)$, i = 1, 2 be the martingale solutions of (5.2.4) with $u^i(0) = u_0$, i = 1, 2. Then we say that the solutions are unique in law if

$$\operatorname{Law}_{\mathbb{P}^1}(u^1) = \operatorname{Law}_{\mathbb{P}^2}(u^2) \text{ on } \mathscr{C}([0,\infty); V_w) \cap L^2([0,\infty); D(A)),$$

where $\text{Law}_{\mathbb{P}^{i}}(u^{i})$, i = 1, 2 are by definition probability measures on $\mathscr{C}([0,\infty); V_{w}) \cap L^{2}([0,\infty); D(A))$.

Corollary 5.6.3. Assume that assumptions (A.1) - (A.2) are satisfied. Then

- (1) There exists a pathwise unique strong solution of (5.2.4).
- (2) Moreover, if (Ω, ℱ, ℙ, ℙ, W, u) is a strong solution of (5.2.4) then for ℙ-almost all ω ∈ Ω the trajectory u(·,ω) is equal almost everywhere to a continuous V-valued function defined on [0, T].
- (3) The martingale solution of (5.2.4) are unique in law.

Proof. By Theorem 5.3.3 there exists a martingale solution and in the Lemma 5.6.1 we showed it is pathwise unique, thus assertion (1) follows from [68, Theorem 2]. Assertion (2) is a direct consequence of Lemma 5.3.5. Assertion (3) follows from [68, Theorems 2,11].

Using Theorem 5.3.3, Lemma 5.6.1 and Corollary 5.6.3 one can infer Theorem 5.3.7.

5.7 The continuous dependence of solutions on the initial data

This section deals with the continuous dependence of martingale solutions of (5.2.4) on the initial data. Roughly speaking, we will show that if $(u_{0,n})_{n \in \mathbb{N}} \subset V \cap \mathcal{M}$ is a sequence of initial conditions approaching in V topology to $u_0 \in V \cap \mathcal{M}$, then the sequence $(u_n)_{n \in \mathbb{N}}$ of martingale solutions of (5.2.4) corresponding to initial data $(u_{0,n})$, satisfying inequalities (5.5.4) – (5.5.6), on a changed probability basis, converges to a martingale solution with the initial condition u_0 . Note that existence of such solutions u_n , $n \in \mathbb{N}$, is guaranteed by Theorem 5.3.3. Let us recall that for a fixed T > 0,

 $\mathcal{Z}_T = \mathscr{C}([0,T];\mathbf{H}) \cap L^2_w(0,T;\mathbf{D}(\mathbf{A})) \cap L^2(0,T;\mathbf{V}) \cap \mathscr{C}([0,T];\mathbf{V}_w).$

The following auxiliary result which is needed in the proof of Theorem 5.7.7, cannot be deduced directly from the Kuratowski Theorem.

Lemma 5.7.1. Assume that T > 0. Then the following sets $\mathscr{C}([0,T];V) \cap \mathscr{Z}_T$ and $L^2(0,T;D(A)) \cap \mathscr{Z}_T$ are Borel subsets of \mathscr{Z}_T .

Proof. First of all $\mathscr{C}([0,T];V) \subset \mathscr{C}([0,T];H) \cap L^2(0,T;V)$. Secondly, $\mathscr{C}([0,T];V)$ and $\mathscr{C}([0,T];H) \cap L^2(0,T;V)$ are Polish spaces. And finally, since V is continuously embedded in H, the map

$$i: \mathscr{C}([0,T]; \mathbf{V}) \to \mathscr{C}([0,T]; \mathbf{H}) \cap L^2(0,T; \mathbf{V}),$$

is continuous and hence Borel. Thus by application of the Kuratowski Theorem (see Theorem 2.5.17) $\mathscr{C}([0,T];V)$ is a Borel subset of $\mathscr{C}([0,T];H) \cap L^2(0,T;V)$. Therefore by Lemma C.1 $\mathscr{C}([0,T];V) \cap \mathscr{Z}_T$ is a Borel subset of $\mathscr{C}([0,T];H) \cap L^2(0,T;V) \cap \mathscr{Z}_T$ which is equal to \mathscr{Z}_T .

Similarly we can show that $L^2(0,T; D(A)) \cap \mathcal{Z}_T$ is a Borel subset of \mathcal{Z}_T . $L^2(0,T; D(A)) \hookrightarrow L^2(0,T; V)$ and both are Polish spaces thus by application of the Kuratowski Theorem, $L^2(0,T; D(A))$ is a Borel subset of $L^2(0,T; V)$. Finally, we can conclude the proof of theorem by Lemma C.1.

5.7.1 Tightness criterion and the Jakubowski-Skorohod Theorem

One of the main tools in this section is the tightness criterion in the space \mathcal{Z}_T . We will use a slight generalization of the criterion stated in Corollary 5.4.3. Namely, we will consider the sequence of stochastic processes defined on their own probability spaces in contrast to one common probability space. Let $(\Omega_n, \mathscr{F}_n, \mathbb{P}_n), n \in \mathbb{N}$, be a sequence of probability spaces with the filtration $\mathbb{F}_n = (\mathscr{F}_{n,t})_{t\geq 0}$.

Corollary 5.7.2. (Tightness criterion) Assume that $(X_n)_{n \in \mathbb{N}}$ is a sequence of continuous \mathbb{F}_n -adapted H-valued processes defined on Ω_n such that

(5.7.1)
$$\sup_{n \in \mathbb{N}} \mathbb{E}_n \left[\sup_{s \in [0,T]} \|X_n(s)\|^2 \right] < \infty,$$

(5.7.2)
$$\sup_{n\in\mathbb{N}}\mathbb{E}_n\left[\int_0^T |X_n(s)|^2_{\mathrm{D}(\mathrm{A})}ds\right] < \infty,$$

(a) and for every $\varepsilon > 0$ and for every $\eta > 0$ there exists $\delta > 0$ such that for every sequence $(\tau_n)_{n \in \mathbb{N}}$ of [0, T]-valued \mathbb{F}_n -stopping times one has

(5.7.3)
$$\sup_{n \in \mathbb{N}} \sup_{0 \le \theta \le \delta} \mathbb{P}_n \{ |X_n(\tau_n + \theta) - X_n(\tau_n)|_{\mathcal{H}} \ge \eta \} \le \varepsilon.$$

Let $\tilde{\mathbb{P}}_n$ be the law of X_n on the Borel σ -field $\mathscr{B}(\mathscr{Z}_T)$. Then for every $\varepsilon > 0$ there exists a compact subset K_{ε} of \mathscr{Z}_T such that

$$\sup_{n\in\mathbb{N}}\tilde{\mathbb{P}}_n(K_{\varepsilon})\geq 1-\varepsilon.$$

The proof of Corollary 5.7.2 is essentially same as the proof of Corollary 5.4.3.

If the sequence $(X_n)_{n \in \mathbb{N}}$ satisfies condition (a) then we say that it satisfies the Aldous condition [A] in H on [0,T]. If it satisfies condition (a) for each T > 0, we say that it satisfies the Aldous condition [A] in H (see Definition 2.9.10).

Below we will formulate a sufficient condition for the Aldous condition. This idea has been used in the proof of Lemma 5.5.5, but has not been formulated in such a way.

Lemma 5.7.3. Assume that Y is a separable Banach space, $\sigma \in (0,1]$ and that $(u_n)_{n \in \mathbb{N}}$ is a sequence of continuous \mathbb{F}_n -adapted Y-valued processes indexed by [0,T] for some T > 0, such that

(**a**') there exists C > 0 such that for every $\theta > 0$ and for every sequence $(\tau_n)_{n \in \mathbb{N}}$ of [0, T]-valued \mathbb{F}_n -stopping times one has

(5.7.4)
$$\mathbb{E}_n\left[|u_n(\tau_n+\theta)-u_n(\tau_n)|_Y\right] \le C\theta^{\sigma}.$$

Then the sequence $(u_n)_{n \in \mathbb{N}}$ satisfies the Aldous condition [A] in Y on [0,T].

Proof. Let us fix $\eta > 0$ and $\varepsilon > 0$. By the Chebyshev inequality and the estimate (5.7.4) we obtain

$$\mathbb{P}_n\big(\big\{|u_n(\tau_n+\theta)-u_n(\tau_n)|_Y \ge \eta\big\}\big) \le \frac{1}{\eta} \mathbb{E}_n\big[|u_n(\tau_n+\theta)-u_n(\tau_n)|_Y\big] \le \frac{C \cdot \theta^{\sigma}}{\eta}, \qquad n \in \mathbb{N}.$$

Let us choose $\delta := \left[\frac{\eta \cdot \varepsilon}{C}\right]^{\frac{1}{\sigma}}$. Then we have

$$\sup_{n \in \mathbb{N}} \sup_{1 \le \theta \le \delta} \mathbb{P}_n \{ |u_n(\tau_n + \theta) - u_n(\tau_n)|_Y \ge \eta \} \le \varepsilon.$$

This completes the proof.

We restate the version of the Skorohod Theorem that we stated in Theorem 5.4.4 in a slightly different way.

Theorem 5.7.4. Let (\mathcal{X}, τ) be a topological space such that there exists a sequence (f_m) of continuous functions $f_m : \mathcal{X} \to \mathbb{R}$ that separates points of \mathcal{X} . Let (X_n) be a sequence of \mathcal{X} -valued Borel random variables. Suppose that for every $\varepsilon > 0$ there exists a compact subset $K_{\varepsilon} \subset \mathcal{X}$ such that

$$\sup_{n\in\mathbb{N}}\mathbb{P}(\{X_n\in K_{\varepsilon}\})>1-\varepsilon.$$

Then there exists a subsequence $(n_k)_{k \in \mathbb{N}}$, a sequence $(Y_k)_{k \in \mathbb{N}}$ of \mathscr{X} -valued Borel random variables and an \mathscr{X} -valued Borel random variable Y defined on some probability space $(\Omega, \mathscr{F}, \mathbb{P})$ such that

$$\mathscr{L}(X_{n_k}) = \mathscr{L}(Y_k), \qquad k = 1, 2, \dots$$

and for all $\omega \in \Omega$:

 $Y_k(\omega) \xrightarrow{\tau} Y(\omega)$ as $k \to \infty$.

Note that the sequence (f_m) defines another, weaker topology on \mathscr{X} . However, this topology restricted to σ -compact subsets of \mathscr{X} is equivalent to the original topology τ . Let us emphasize that thanks to the assumption on the tightness of the set of laws $\{\mathscr{L}(X_n), n \in \mathbb{N}\}$ on the space \mathscr{X} the maps Y and $Y_k, k \in \mathbb{N}$, in Theorem 5.7.4 are measurable with respect to the Borel σ -field in the space \mathscr{X} .

In Lemma 5.4.5 we have already shown that the topological space \mathcal{Z}_T satisfies the assumptions of Theorem 5.7.4.

5.7.2 The continuous dependence

We prove the following result related to the continuous dependence on the deterministic initial condition.

Theorem 5.7.5. Let T > 0. Assume that $(u_{0,n})_{n \in \mathbb{N}}$ is a $V \cap \mathcal{M}$ -valued sequence bounded in V and

$$(\hat{\Omega}_n, \hat{\mathscr{F}}_n, \hat{\mathbb{F}}_n, \hat{\mathbb{P}}_n, \hat{W}_n, u_n)$$

be a martingale solution of problem (5.2.4) with the initial data $u_{0,n}$ and satisfying inequalities (5.5.4) – (5.5.6). Then, the set of Borel measures $\{\mathscr{L}(u_n), n \in \mathbb{N}\}$ is tight on the space $(\mathscr{Z}_T, \mathscr{T})$.

Proof. Let us fix T > 0. Let $(u_{0,n})_{n \in \mathbb{N}}$ be a $V \cap \mathcal{M}$ -valued sequence. Let

$$(\hat{\Omega}_n, \hat{\mathscr{F}}_n, \hat{\mathbb{F}}_n, \hat{\mathbb{P}}_n, \hat{W}_n, u_n)$$

be the martingale solution of problem (5.2.4) with the initial data u_0^n and satisfying inequalities (5.5.4) – (5.5.6). Such a solution exists by Theorem 5.3.3.

To show that the set of measures $\{\mathscr{L}(u_n), n \in \mathbb{N}\}\$ are tight on the space $(\mathscr{Z}_T, \mathscr{T})$, we argue as in the proof of Lemma 5.5.3 using Corollary 5.7.2. We first observe that due to estimates (5.5.4) (with p = 1) and (5.5.6), conditions (5.7.1) and (5.7.2) of Corollary 5.7.2 are satisfied. Thus, we are left to prove condition (**a**), i.e. the sequence $(u_n)_{n\in\mathbb{N}}$ satisfies the Aldous condition [**A**]. By Lemma 5.7.3 it is sufficient to prove the condition (**a**').

Note that we have to choose our steps very carefully as we no longer treat strong solutions to an SDE in a finite dimensional Hilbert space but instead a strong solution to an SPDE in an infinite dimensional Hilbert space.

Let $(\tau_n)_{n \in \mathbb{N}}$ be a sequence of stopping times taking values in [0, T]. Since each process satisfies equation (5.3.2), by Lemma 5.3.5 we have

$$u_{n}(t) = u_{0,n} - \int_{0}^{t} Au_{n}(s) ds - \int_{0}^{t} B(u_{n}(s)) ds + \int_{0}^{t} |\nabla u_{n}(s)|_{L^{2}}^{2} u_{n}(s) ds$$
$$+ \frac{1}{2} \int_{0}^{t} C^{2} u_{n}(s) ds + \int_{0}^{t} C u_{n}(s) dW(s)$$
$$=: J_{1}^{n} + J_{2}^{n}(t) + J_{3}^{n}(t) + J_{4}^{n}(t) + J_{5}^{n}(t) + J_{6}^{n}(t), \quad t \in [0, T],$$

where the above equality is understood in the space V. Let us choose $\theta > 0$. It is sufficient to show that each sequence J_i^n of processes, $i = 1, \dots, 6$ satisfies the sufficient condition (\mathbf{a}') from Lemma 5.7.3. Now the rest of the proof is identical to that of Lemma 5.5.3.

Remark 5.7.6. It is easy to be convinced that u_n take values in \mathcal{Z}_T but it's not so obvious to see that in fact u_n are Borel measurable functions. Indeed, this is so because our construction of the martingale solution is based on the Jakubowski-Skorohod Theorem, see Theorem 5.7.4 for details.

The main result about the continuous dependence of the solutions of the stochastic constrained Navier-Stokes equations on the initial state is expressed in the following theorem.

Theorem 5.7.7. Assume that $(u_{0,n})_{n \in \mathbb{N}}$ is a $V \cap \mathcal{M}$ -valued sequence that is convergent weakly to $u_0 \in V \cap \mathcal{M}$. Let

$$(\hat{\Omega}_n, \hat{\mathscr{F}}_n, \hat{\mathbb{F}}_n, \hat{\mathbb{P}}_n \hat{W}_n, u_n)$$

be a martingale solution of problem (5.2.4) on $[0,\infty)$ with the initial data u_0^n and satisfying inequalities (5.5.4) – (5.5.6). Then for every T > 0 there exist

- a subsequence $(n_k)_k$,
- a stochastic basis (Ω̃, ℱ, ℙ̃, ℙ̃),
- $a \mathbb{R}^m$ -valued $\tilde{\mathbb{F}}$ -Wiener process \tilde{W}
- and $\tilde{\mathbb{F}}$ -progressively measurable processes \tilde{u} , $(\tilde{u}_{n_k})_{k\geq 1}$ (defined on this basis) with laws supported in \mathcal{Z}_T such that

(5.7.5) \tilde{u}_{n_k} has the same law as u_{n_k} on \mathcal{Z}_T and $\tilde{u}_{n_k} \to \tilde{u}$ in \mathcal{Z}_T , $\tilde{\mathbb{P}}$ - a.s.

and the system

$$(\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{F}}, \tilde{\mathbb{P}}, \tilde{W}, \tilde{u})$$

is a martingale solution to problem (5.2.4) on the interval [0,T] with the initial data u_0 . In particular, for all $t \in [0,T]$ and all $v \in V$

$$\begin{split} \langle \tilde{u}(t), v \rangle_{\mathrm{V}} - \langle \tilde{u}(0), v \rangle_{\mathrm{V}} + \int_{0}^{t} \langle \mathrm{A}\tilde{u}(s), v \rangle_{\mathrm{V}} ds + \int_{0}^{t} \langle B(\tilde{u}(s)), v \rangle_{\mathrm{V}} ds \\ &= \int_{0}^{t} \langle |\nabla \tilde{u}(s)|_{\mathrm{H}}^{2} \tilde{u}(s), v \rangle_{\mathrm{V}} ds + \langle \int_{0}^{t} C \tilde{u}(s) \circ d \tilde{W}(s), v \rangle_{\mathrm{V}}, \quad \tilde{\mathbb{P}}\text{-}a.s. \end{split}$$

Moreover, the process \tilde{u} satisfies the following inequality for every $p \in [1, 1 + \frac{1}{\kappa^2})$

(5.7.6)
$$\tilde{\mathbb{E}}\left[\sup_{s\in[0,T]}\|\tilde{u}(s)\|_{\mathrm{V}}^{2p} + \int_{0}^{T}|\tilde{u}(s)|_{\mathrm{D}(\mathrm{A})}^{2}ds\right] < \infty.$$

Proof. Since the product topological space $\mathscr{Z}_T \times \mathscr{C}([0,T],\mathbb{R}^m)$ satisfies the assumptions of Theorem 5.7.4, by applying it together with Theorem 5.7.5, there exists a subsequence (n_k) , a probability space $(\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{P}})$ and $\mathscr{Z}_T \times \mathscr{C}([0,T],\mathbb{R}^m)$ -valued Borel random variables $(\tilde{u}, \tilde{W}), (\tilde{u}_k, \tilde{W}_k), k \in \mathbb{N}$ such that \tilde{W} and $\tilde{W}_k, k \in \mathbb{N}$ are \mathbb{R}^m -valued Wiener processes such that

(5.7.7) the laws on
$$\mathscr{B}(\mathscr{Z}_T \times \mathscr{C}([0,T],\mathbb{R}^m))$$
 of (u_{n_k}, W) and $(\tilde{u}_k, \tilde{W}_k)$ are equal.

where $\mathscr{B}(\mathscr{Z}_T \times \mathscr{C}([0,T],\mathbb{R}^m))$ is the Borel σ -algebra on $\mathscr{Z}_T \times \mathscr{C}([0,T],\mathbb{R}^m)$, and

(5.7.8)
$$(\tilde{u}_k, \tilde{W}_k)$$
 converges to (\tilde{u}, \tilde{W}) in $\mathcal{Z}_T \times \mathscr{C}([0, T], \mathbb{R}^m)$ $\tilde{\mathbb{P}}$ -almost surely on $\tilde{\Omega}$.

Note that since $\mathscr{B}(\mathscr{Z}_T \times \mathscr{C}([0,T],\mathbb{R}^m)) \subset \mathscr{B}(\mathscr{Z}_T) \times \mathscr{B}(\mathscr{C}([0,T],\mathbb{R}^m))$, the function *u* is \mathscr{Z}_T Borel random variable.

Define a corresponding sequence of filtrations by

(5.7.9)
$$\widetilde{\mathbb{F}}_k = (\widetilde{\mathscr{F}}_k(t))_{t \ge 0}, \text{ where } \widetilde{\mathscr{F}}_k(t) = \sigma\big(\{(\widetilde{u}_k(s), \widetilde{W}_k(s)), s \le t\}\big), t \in [0, T].$$

To conclude the proof, we need to show that the random variable \tilde{u} gives rise to a martingale solution. The proof of this claim is very similar to the proof of Theorem 2.3 in [66]. Let us denote the subsequence $(\tilde{u}_{n_k})_k$ again by $(\tilde{u}_n)_n$.

The few differences are:

(i) The finite dimensional space H_n is replaced by the whole space H. But now since the space $\mathscr{C}([0,T];H)$ is a Borel subset of \mathscr{Z}_T and \tilde{u}_n and u_n have the same laws on \mathscr{Z}_T , we infer that

$$\tilde{u}_n \in \mathscr{C}([0,T];\mathbf{H}) \qquad n \ge 1, \quad \mathbb{P}\text{-a.s.}$$

(ii) The operator P_n has to be replaced by the identity. But this is rather a simplification.

In addition to point (i) above, we have that for every $p \in [1, 1 + \frac{1}{K^2})$

(5.7.10)
$$\sup_{n\in\mathbb{N}} \tilde{\mathbb{E}} \left(\sup_{0\leq s\leq T} \|\tilde{u}_n(s)\|_{\mathrm{V}}^{2p} \right) \leq C_1(p),$$

Similarly,

$$\tilde{u}_n \in L^2(0,T; \mathbf{D}(\mathbf{A}))$$
 $n \ge 1$, \mathbb{P} -a.s.

and

(5.7.11)
$$\sup_{n \in \mathbb{N}} \tilde{\mathbb{E}}\left[\int_0^T |\tilde{u}_n(s)|^2_{\mathrm{D}(\mathrm{A})} ds\right] \le C_2.$$

By inequality (5.7.11) we infer that the sequence (\tilde{u}_n) contains a subsequence, still denoted by (\tilde{u}_n) , convergent weakly in the space $L^2([0,T] \times \tilde{\Omega}; D(A))$. Since by (5.7.8) $\tilde{\mathbb{P}}$ -a.s. $\tilde{u}_n \to \tilde{u}$ in \mathcal{Z}_T , we conclude that $\tilde{u} \in L^2([0,T] \times \tilde{\Omega}; D(A))$, i.e.

(5.7.12)
$$\widetilde{\mathbb{E}}\left[\int_0^T |\tilde{u}(s)|^2_{\mathrm{D}(\mathrm{A})} ds\right] < \infty.$$

Similarly, by inequality (5.7.10) we can choose a subsequence of (\tilde{u}_n) convergent weak star in the space $L^p(\tilde{\Omega}; L^{\infty}(0, T; V))$ and, using (5.7.8), infer that

(5.7.13)
$$\tilde{\mathbb{E}}\left[\sup_{0\leq s\leq T}\|\tilde{u}(s)\|_{V}^{2p}\right]<\infty.$$

The remaining proof will be done in two steps.

Step 1. Let us fix T > 0. We will first prove the following Lemma.

Lemma 5.7.8. *For all* $t \in (0, T]$ *and* $\varphi \in V$

 $(a) \lim_{n \to \infty} \tilde{\mathbb{E}} \left[\int_0^T \left| \langle \tilde{u}_n(t) - \tilde{u}(t), \varphi \rangle_{\mathrm{H}} \right|^2 dt \right] = 0,$ $(b) \lim_{n \to \infty} \tilde{\mathbb{E}} \left[\left| \langle \tilde{u}_n(0) - \tilde{u}(0), \varphi \rangle_{\mathrm{H}} \right|^2 \right] = 0,$ $(c) \lim_{n \to \infty} \tilde{\mathbb{E}} \left[\int_0^T \left| \int_0^t \langle A \tilde{u}_n(s) - A \tilde{u}(s), \varphi \rangle ds \right| dt \right] = 0,$ $(d) \lim_{n \to \infty} \tilde{\mathbb{E}} \left[\int_0^T \left| \int_0^t \langle B (\tilde{u}_n(s)) - B (\tilde{u}(s)), \varphi \rangle ds \right| dt \right] = 0,$ $(e) \lim_{n \to \infty} \tilde{\mathbb{E}} \left[\int_0^T \left| \int_0^t \langle |\nabla \tilde{u}_n(s)|_{L^2}^2 \tilde{u}_n(s) - |\nabla \tilde{u}(s)|_{L^2}^2 \tilde{u}(s), \varphi \rangle ds \right| dt \right] = 0,$ $(f) \lim_{n \to \infty} \tilde{\mathbb{E}} \left[\int_0^T \left| \int_0^t \langle C^2 \tilde{u}_n(s) - C^2 \tilde{u}(s), \varphi \rangle ds \right| dt \right] = 0,$ $(g) \lim_{n \to \infty} \tilde{\mathbb{E}} \left[\int_0^T \left| \langle \int_0^t [C \tilde{u}_n(s) - C \tilde{u}(s)] d \tilde{W}(s), \varphi \rangle \right|^2 dt \right] = 0.$

Proof. Let us fix $\varphi \in V$.

Ad (a). Since by (5.7.8) $\tilde{u}_n \to \tilde{u}$ in $\mathscr{C}([0,T];\mathrm{H}) \tilde{\mathbb{P}}$ -a.s., $\langle \tilde{u}_n(\cdot), \varphi \rangle_{\mathrm{H}} \to \langle \tilde{u}(\cdot), \varphi \rangle_{\mathrm{H}}$ in $\mathscr{C}([0,T];\mathbb{R}), \tilde{\mathbb{P}}$ -a.s. Hence, in particular, for all $t \in [0,T]$

$$\lim_{n \to \infty} \langle \tilde{u}_n(t), \varphi \rangle_{\mathrm{H}} = \langle \tilde{u}(t), \varphi \rangle_{\mathrm{H}}, \qquad \tilde{\mathbb{P}}\text{-a.s.}$$

Since $\tilde{u}_n(t) \in \mathcal{M}$ for all $t \in [0,T]$, $\sup_{t \in [0,T]} |\tilde{u}_n(t)|_{\mathrm{H}}^2 < \infty$, $\tilde{\mathbb{P}}$ -a.s., using the dominated convergence theorem we infer that

(5.7.14)
$$\lim_{n \to \infty} \int_0^T |\langle \tilde{u}_n(t) - \tilde{u}(t), \varphi \rangle_{\mathrm{H}}|^2 dt = 0 \qquad \tilde{\mathbb{P}}\text{-a.s.}$$

Since $\tilde{u}_n(t), \tilde{u}(t) \in \mathcal{M}$ for all $t \in [0, T]$, by the Hölder inequality for every $n \in \mathbb{N}$ and $r \in [1, \infty)$

(5.7.15)
$$\tilde{\mathbb{E}}\left[\left|\int_{0}^{T} |\tilde{u}_{n}(t) - \tilde{u}(t)|_{\mathrm{H}}^{2} dt\right|^{r}\right] \leq c \tilde{\mathbb{E}}\left[\int_{0}^{T} \left(|\tilde{u}_{n}(t)|_{\mathrm{H}}^{2r} + |\tilde{u}(t)|_{\mathrm{H}}^{2r}\right) dt\right] = 2cT,$$

where c is some positive constant. To conclude the proof of assertion (a) it is sufficient to use (5.7.14), (5.7.15) and the Vitali Theorem.

Ad (b). Since by (5.7.8) $\tilde{u}_n \to \tilde{u}$ in $\mathscr{C}(0,T;\mathrm{H}) \tilde{\mathbb{P}}$ -a.s. and \tilde{u} is continuous at t = 0, we infer that $\langle \tilde{u}_n(0), \varphi \rangle_{\mathrm{H}} \to \langle \tilde{u}(0), \varphi \rangle_{\mathrm{H}}, \tilde{\mathbb{P}}$ -a.s. Now, assertion (b) follows from the Vitali Theorem. **Ad** (c). Since by (5.7.8) $\tilde{u}_n \to \tilde{u}$ in $L^2_w(0,T;\mathrm{D}(\mathrm{A})), \tilde{\mathbb{P}}$ -a.s., we infer that $\tilde{\mathbb{P}}$ -a.s.

(5.7.16)
$$\lim_{n \to \infty} \int_0^t \langle A \tilde{u}_n(s), \varphi \rangle \, ds = \lim_{n \to \infty} \int_0^t \langle \tilde{u}_n(s), A^{-1} \varphi \rangle_{\mathrm{D}(\mathrm{A})} \, ds$$
$$= \int_0^t \langle \tilde{u}(s), A^{-1} \varphi \rangle_{\mathrm{D}(\mathrm{A})} \, ds = \int_0^t \langle A \tilde{u}(s), \varphi \rangle \, ds$$

By the Hölder inequality and estimate (5.7.11) we infer that for all $t \in [0, T]$ and $n \in \mathbb{N}$

(5.7.17)
$$\widetilde{\mathbb{E}}\left[\left|\int_{0}^{t} \langle A\tilde{u}_{n}(s), \varphi \rangle \, ds\right|^{2}\right] \leq c \left|\varphi\right|_{H}^{2} \widetilde{\mathbb{E}}\left[\int_{0}^{T} \left|\tilde{u}_{n}(s)\right|_{D(A)}^{2} ds\right] \leq \tilde{c}C_{2},$$

where $c, \tilde{c} > 0$ are some constants. By (5.7.16), (5.7.17) and the Vitali Theorem we conclude that for all $t \in [0, T]$

$$\lim_{n\to\infty} \tilde{\mathbb{E}}\Big[\Big|\int_0^t \langle \mathrm{A}\tilde{u}_n(s) - \mathrm{A}\tilde{u}(s), \varphi \rangle \, ds\Big|\Big] = 0.$$

Assertion (c) follows now from (5.7.11) and the dominated convergence theorem. **Ad** (d). By (5.7.8), $\tilde{u}_n \to \tilde{u}$ in $\mathscr{C}([0,T];H) \cap L^2(0,T;V)$, $\tilde{\mathbb{P}}$ -a.s. and since $\tilde{u}_n(t) \in \mathscr{M}$ for all $t \in [0,T]$ hence by (3.2.1) we infer that $\tilde{\mathbb{P}}$ -a.s. for all $t \in [0,T]$ and $\varphi \in V$

$$\begin{split} \lim_{n \to \infty} \left[\int_0^t \langle B(\tilde{u}_n(s)), \varphi \rangle_{\mathrm{H}} \, ds - \int_0^t \langle B(\tilde{u}(s)), \varphi \rangle_{\mathrm{H}} \, ds \right] \\ &= \lim_{n \to \infty} \int_0^t \left[b(\tilde{u}_n(s), \tilde{u}_n(s), \varphi) - b(\tilde{u}(s), \tilde{u}(s), \varphi) \right] \, ds \\ &= \lim_{n \to \infty} \left[\int_0^t b(\tilde{u}_n(s) - \tilde{u}(s), \tilde{u}_n(s), \varphi) \, ds + \int_0^t b(\tilde{u}(s), \tilde{u}_n(s) - \tilde{u}(s), \varphi) \, ds \right] \\ &\leq \lim_{n \to \infty} \int_0^t (\|\tilde{u}_n(s)\|_{\mathrm{V}}\| + \|\tilde{u}(s)\|_{\mathrm{V}}) \|\tilde{u}_n(s) - \tilde{u}(s)\|_{\mathrm{V}} \|\varphi\|_{\mathrm{V}} \, ds \\ (5.7.18) \qquad \leq \lim_{n \to \infty} \left(|\tilde{u}_n|_{L^2(0,T;\mathrm{V})} + |\tilde{u}|_{L^2(0,T;\mathrm{V})}| \right) |\tilde{u}_n - \tilde{u}|_{L^2(0,T;\mathrm{V})} \|\varphi\|_{\mathrm{V}} = 0. \end{split}$$

Using the Hölder inequality, (3.2.1), and the estimate (5.7.10) we infer that for all $t \in [0,T]$, $r \in (1, 1 + \frac{1}{K_c^2})$ and $n \in \mathbb{N}$ the following inequalities hold

(5.7.19)

$$\widetilde{\mathbb{E}}\left[\left|\int_{0}^{t} \langle B(\tilde{u}_{n}(s)), \varphi \rangle \, ds \right|^{r}\right] \leq \widetilde{\mathbb{E}}\left[\left(\int_{0}^{t} |B(\tilde{u}_{n}(s))|_{V'} \|\varphi\|_{V} \, ds\right)^{r}\right] \\
\leq (c \|\varphi\|_{V})^{r} \widetilde{\mathbb{E}}\left(\int_{0}^{t} \|\tilde{u}_{n}(s)\|_{V}^{2r} \, ds\right) \\\leq \widetilde{c}\widetilde{\mathbb{E}}\left[\sup_{s \in [0,T]} \|\tilde{u}_{n}(s)\|_{V}^{2r}\right] \leq \widetilde{C}C_{1}(r).$$

By (5.7.18), (5.7.19) and the Vitali Theorem we obtain for all $t \in [0, T]$

(5.7.20)
$$\lim_{n \to \infty} \tilde{\mathbb{E}}\Big[\Big|\int_0^t \langle B(\tilde{u}_n(s)) - B(\tilde{u}(s)), \varphi \rangle \, ds\Big|\Big] = 0$$

Hence by (5.7.20) and the dominated convergence theorem, we infer that assertion (d) holds. **Ad** (e). By (5.7.8), $\tilde{u}_n \to \tilde{u}$ in $\mathscr{C}([0,T];\mathrm{H}) \cap L^2(0,T;\mathrm{V})$, $\tilde{\mathbb{P}}$ -a.s. and since $\tilde{u}_n(t) \in \mathcal{M}$ for all $t \in [0,T]$ hence we infer that $\tilde{\mathbb{P}}$ -a.s. for all $t \in [0,T]$

$$\begin{split} \lim_{n \to \infty} \int_{0}^{t} |\nabla \tilde{u}_{n}(s)|_{L^{2}}^{2} \langle \tilde{u}_{n}(s), \varphi \rangle_{\mathrm{H}} \, ds - \int_{0}^{t} |\nabla \tilde{u}(s)|_{L^{2}}^{2} \langle \tilde{u}(s), \varphi \rangle_{\mathrm{H}} \, ds \\ &= \lim_{n \to \infty} \int_{0}^{t} \left[|\nabla \tilde{u}_{n}(s)|_{L^{2}}^{2} - |\nabla \tilde{u}(s)|_{L^{2}}^{2} \right] \langle \tilde{u}_{n}(s), \varphi \rangle_{\mathrm{H}} \, ds + \int_{0}^{t} |\nabla \tilde{u}(s)|_{L^{2}}^{2} \langle \tilde{u}_{n}(s) - \tilde{u}(s), \varphi \rangle_{\mathrm{H}} \, ds \\ &= \lim_{n \to \infty} \int_{0}^{t} \left[|\nabla \tilde{u}_{n}(s)|_{L^{2}} - |\nabla \tilde{u}(s)|_{L^{2}} \right] \left[|\nabla \tilde{u}_{n}(s)|_{L^{2}} + |\nabla \tilde{u}(s)|_{L^{2}} \right] \langle \tilde{u}_{n}(s), \varphi \rangle_{\mathrm{H}} \, ds \\ &+ \lim_{n \to \infty} \int_{0}^{t} |\nabla \tilde{u}(s)|_{L^{2}}^{2} \langle \tilde{u}_{n}(s) - \tilde{u}(s), \varphi \rangle_{\mathrm{H}} \, ds \\ &\leq \lim_{n \to \infty} \int_{0}^{t} (\|\tilde{u}_{n}(s) - \tilde{u}(s)\|_{\mathrm{V}}) (\|\tilde{u}_{n}(s)\|_{\mathrm{V}} \|\tilde{u}(s)\|_{\mathrm{V}}) |\tilde{u}_{n}(s)|_{\mathrm{H}} |\varphi|_{\mathrm{H}} \, ds \\ &+ \lim_{n \to \infty} \int_{0}^{t} \|\tilde{u}(s)\|^{2} |\tilde{u}_{n}(s) - \tilde{u}(s)|_{\mathrm{H}} |\varphi|_{\mathrm{H}} \, ds \\ &\leq \lim_{n \to \infty} \tilde{c}_{1} \left[|\tilde{u}_{n}|_{L^{2}(0,T;\mathrm{V})} + |\tilde{u}|_{L^{2}(0,T;\mathrm{V})} \right] |\tilde{u}|_{L^{\infty}(0,T;\mathrm{H})} |\tilde{u}_{n} - \tilde{u}|_{L^{2}(0,T;\mathrm{V})} \\ (5.7.21) \qquad + \lim_{n \to \infty} \tilde{c}_{2} |\tilde{u}(s)|_{L^{2}(0,T;\mathrm{V})}^{2} |\tilde{u}_{n} - \tilde{u}|_{L^{\infty}(0,T;\mathrm{H})} = 0. \end{split}$$

Using the Hölder inequality, (5.7.10) and the fact that $u_n(t) \in \mathcal{M}$ for all $t \in [0, T]$ we infer that for all $t \in [0, T]$, $r \in (1, 1 + \frac{1}{K_c^2})$ and $n \in \mathbb{N}$ the following inequalities hold

$$\widetilde{\mathbb{E}}\left[\left|\int_{0}^{t}|\nabla \tilde{u}_{n}(s)|_{L^{2}}^{2}\langle \tilde{u}_{n}(s),\varphi\rangle_{\mathrm{H}}ds\right|^{r}\right] \leq \widetilde{\mathbb{E}}\left[\left(\int_{0}^{t}\|\tilde{u}_{n}(s)\|_{\mathrm{V}}^{2}|\tilde{u}_{n}(s)|_{\mathrm{H}}|\varphi|_{\mathrm{H}}ds\right)^{r}\right]$$
$$\leq (c|\varphi|_{\mathrm{H}})^{r}\widetilde{\mathbb{E}}\left[\left(\int_{0}^{t}|\tilde{u}_{n}(s)|_{\mathrm{H}}^{\frac{r}{r-1}}\right)^{r-1}\int_{0}^{t}\|\tilde{u}_{n}(s)\|_{\mathrm{V}}^{2r}ds\right]$$
$$\leq \tilde{c}t^{r-1}\widetilde{\mathbb{E}}\left[\sup_{s\in[0,T]}\|\tilde{u}_{n}(s)\|_{\mathrm{V}}^{2r}\right] \leq \tilde{C}C_{1}(r).$$

(5.7)

(5

By (5.7.21), (5.7.22) and the Vitali Theorem we obtain for all $t \in [0, T]$

(5.7.23)
$$\lim_{n \to \infty} \tilde{\mathbb{E}}\left[\left| \int_0^t \left\langle |\nabla \tilde{u}_n(s)|_{\mathrm{H}}^2 \tilde{u}_n(s) - |\nabla \tilde{u}(s)|_{\mathrm{H}}^2 \tilde{u}(s), \varphi \right\rangle_{\mathrm{H}} ds \right| \right] = 0.$$

Hence by (5.7.23) and the dominated convergence theorem, we infer that assertion (e) holds. Ad (f). Since by (5.7.8), $\tilde{u}_n \to \tilde{u}$ in $L^2(0,T; D(A))$, $\tilde{\mathbb{P}}$ -a.s., using (5.7.11) we infer that $\tilde{\mathbb{P}}$ -a.s.

$$\begin{split} &\lim_{n \to \infty} \int_0^t \langle C^2 \tilde{u}_n(s) - C^2 \tilde{u}(s), \varphi \rangle_{\mathrm{H}} \, ds = \lim_{n \to \infty} \int_0^t \langle C^2 (\tilde{u}_n(s) - \tilde{u}(s)), \varphi \rangle_{\mathrm{H}} \, ds \\ &= \lim_{n \to \infty} \int_0^t \langle C^2 \mathrm{A}^{-1} \mathrm{A}(\tilde{u}_n(s) - \tilde{u}(s)), \varphi \rangle_{\mathrm{H}} \, ds. \end{split}$$

Now since $C^2 A^{-1}$ is a bounded operator

$$\lim_{n \to \infty} \int_0^t \langle C^2 \tilde{u}_n(s) - C^2 \tilde{u}(s), \varphi \rangle_{\mathcal{H}} ds = \lim_{n \to \infty} \int_0^t \langle \mathcal{A}(\tilde{u}_n(s) - \tilde{u}(s)), (C^2 \mathcal{A}^{-1})^* \varphi \rangle_{\mathcal{H}} ds$$

$$(5.7.24) \qquad = \lim_{n \to \infty} \int_0^t \langle \tilde{u}_n(s) - \tilde{u}(s), \mathcal{A}^{-1}(C^2 \mathcal{A}^{-1})^* \varphi \rangle_{\mathcal{D}(\mathcal{A})} ds = 0.$$

By the Hölder inequality and estimate (5.7.11) we infer that for all $t \in [0, T]$ and $n \in \mathbb{N}$

(5.7.25)
$$\begin{split} \tilde{\mathbb{E}}\left[\left|\int_{0}^{t} \langle C^{2}\tilde{u}_{n}(s),\varphi\rangle_{\mathrm{H}} ds\right|^{2}\right] &= \tilde{\mathbb{E}}\left[\left|\int_{0}^{t} \langle \tilde{u}_{n}(s),\mathrm{A}^{-1}(C^{2}A^{-1})^{*}\varphi\rangle_{\mathrm{D}(\mathrm{A})} ds\right|^{2}\right] \\ &\leq \tilde{c}|\varphi|_{\mathrm{H}}^{2}\tilde{\mathbb{E}}\left[\int_{0}^{t} |\tilde{u}_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds\right] \leq \tilde{c}C_{2}, \end{split}$$

where $\tilde{c} > 0$ is some constant. By (5.7.24), (5.7.25) and the Vitali theorem we conclude that for all $t \in [0,T]$

$$\lim_{n\to\infty} \tilde{\mathbb{E}}\left[\left| \int_0^t \langle C^2 \tilde{u}_n(s) - C^2 \tilde{u}(s), \varphi \rangle_{\mathrm{H}} ds \right| \right] = 0.$$

Assertion (f) follows from (5.7.11) and the dominated convergence theorem. **Ad** (g) Since by (5.7.8) $\tilde{u}_n \to \tilde{u}$ in $L^2(0,T;V), \tilde{\mathbb{P}}$ -a.s., we infer that for all $t \in [0,T]$ and $\varphi \in H$

$$\lim_{n \to \infty} \int_0^t |\langle C\tilde{u}_n(s) - C\tilde{u}(s), \varphi \rangle|_{\mathcal{T}_2(\mathbb{R}^m, \mathbb{R})}^2 ds \le \lim_{n \to \infty} |\varphi|_{\mathrm{H}}^2 \int_0^t |C\tilde{u}_n(s) - C\tilde{u}(s)|_{\mathrm{H}}^2 ds$$

(7.26)
$$\le \lim_{n \to \infty} K_c^2 |\varphi|_{\mathrm{H}}^2 \int_0^t \|\tilde{u}_n(s) - \tilde{u}(s)\|_{\mathrm{V}}^2 ds = 0,$$

where c > 0 is some constant.

By the uniform estimates (5.7.10) and (5.7.13) we obtain the following inequalities for every $t \in [0, T], r \in (1, 1 + \frac{1}{K_c^2})$ and $n \in \mathbb{N}$

$$\begin{split} \tilde{\mathbb{E}}\left[\left|\int_{0}^{t}\left|\langle C\tilde{u}_{n}(s) - C\tilde{u}(s), \varphi\rangle\right|_{\mathcal{F}_{2}(\mathbb{R}^{m};\mathbb{R})}^{2} ds\right|^{r}\right] &\leq c\tilde{\mathbb{E}}\left[\left|\varphi\right|_{\mathrm{H}}^{2r} \int_{0}^{t}\left[\left|C\tilde{u}_{n}(s)\right|_{\mathrm{H}}^{2r} + \left|C\tilde{u}(s)\right|_{\mathrm{H}}^{2r}\right] ds\right] \\ &\leq \tilde{c}\tilde{\mathbb{E}}\left[\sup_{s\in[0,T]}\left\|\tilde{u}_{n}(s)\right\|_{\mathrm{V}}^{2r} + \sup_{s\in[0,T]}\left\|\tilde{u}(s)\right\|_{\mathrm{V}}^{2r}\right] &\leq 2\tilde{c}C_{1}(r), \end{split}$$

where c, \tilde{c} are some positive constants. Using the Vitali theorem, by (5.7.26) and (5.7.27) we infer that for all $\varphi \in H$

(5.7.28)
$$\lim_{n \to \infty} \tilde{\mathbb{E}}\left[\int_0^t |\langle C\tilde{u}_n(s) - C\tilde{u}(s), \varphi \rangle|_{\mathcal{T}_2(\mathbb{R}^m, \mathbb{R})}^2 ds\right] = 0$$

Hence, by the properties of the Itô integral we infer that for all $t \in [0, T]$ and $\varphi \in H$

(5.7.29)
$$\lim_{n \to \infty} \tilde{\mathbb{E}}\left[\left| \left\langle \int_0^t \left[C \tilde{u}_n(s) - C \tilde{u}(s) \right] d \tilde{W}(s), \varphi \right\rangle \right|^2 \right] = 0$$

By the Itô isometry, and estimates (5.7.10), (5.7.13) we have for all $\varphi \in H$, $t \in [0,T]$ and $n \in \mathbb{N}$

$$\begin{split} \tilde{\mathbb{E}}\big[\Big| \Big\langle \int_{0}^{t} [C\tilde{u}_{n}(s) - C\tilde{u}(s)] d\tilde{W}(s), \varphi \Big\rangle \Big|^{2} \big] \\ &= \tilde{\mathbb{E}}\big[\int_{0}^{t} |\langle C\tilde{u}_{n}(s) - C\tilde{u}(s), \varphi \rangle |_{\mathcal{T}_{2}(\mathbb{R}^{m};\mathbb{R})}^{2} ds \big] \\ &\leq \tilde{\mathbb{E}}\big[|\varphi|_{\mathrm{H}}^{2} \int_{0}^{t} |C\tilde{u}_{n}(s) - C\tilde{u}(s)|_{\mathrm{H}}^{2} ds \big] \\ &\leq c \tilde{\mathbb{E}}\big[\sup_{s \in [0,T]} \|\tilde{u}_{n}(s)\|_{\mathrm{V}}^{2} + \sup_{s \in [0,T]} \|\tilde{u}(s)\|_{\mathrm{V}}^{2} \big] \leq 2cC_{1}(1), \end{split}$$

where c > 0 is some constant. Thus by (5.7.29), (5.7.30) and the Lebesgue Dominated Convergence theorem we infer that for all $\varphi \in \mathbf{H}$

(5.7.31)
$$\lim_{n \to \infty} \int_0^T \tilde{\mathbb{E}}\left[\left| \left\langle \int_0^t \left[C \tilde{u}_n(s) - C \tilde{u}(s) \right] d\tilde{W}(s), \varphi \right\rangle \right|^2 \right] = 0.$$

As a direct consequence of Lemma 5.7.8 we get the following corollary which we precede by introducing some auxiliary notation. Analogously to [20] and [66], let us denote

(5.7.32)
$$\Lambda_{n}(\tilde{u}_{n},\tilde{W}_{n},\varphi)(t) := \langle \tilde{u}_{n}(0),\varphi \rangle_{\mathrm{H}} - \int_{0}^{t} \langle \mathrm{A}\tilde{u}_{n}(s),\varphi \rangle ds - \int_{0}^{t} \langle B(\tilde{u}_{n}(s)),\varphi \rangle ds + \int_{0}^{t} \langle |\nabla \tilde{u}_{n}(s)|_{L^{2}}^{2} \tilde{u}_{n}(s),\varphi \rangle ds + \langle \int_{0}^{t} C \tilde{u}_{n}(s) \circ d\tilde{W}_{n}(s),\varphi \rangle, \quad t \in [0,T],$$

and

(5.7.33)
$$\Lambda(\tilde{u},\tilde{W},\varphi)(t) := \langle \tilde{u}(0),\varphi \rangle_{\mathrm{H}} - \int_{0}^{t} \langle \mathrm{A}\tilde{u}(s),\varphi \rangle ds - \int_{0}^{t} \langle B(\tilde{u}(s)),\varphi \rangle ds + \int_{0}^{t} \langle |\nabla \tilde{u}(s)|_{L^{2}}^{2} \tilde{u}(s),\varphi \rangle ds + \langle \int_{0}^{t} C\tilde{u}(s) \circ d\tilde{W}(s),\varphi \rangle, \quad t \in [0,T].$$

Corollary 5.7.9. For every $\varphi \in V$,

(5.7.34)
$$\lim_{n \to \infty} |\langle \tilde{u}_n(\cdot), \varphi \rangle_{\mathrm{H}} - \langle \tilde{u}(\cdot), \varphi \rangle_{\mathrm{H}}|_{L^2([0,T] \times \tilde{\Omega})} = 0$$

and

(5.7.35)
$$\lim_{n \to \infty} |\Lambda_n(\tilde{u}_n, \tilde{W}_n, \varphi) - \Lambda(\tilde{u}, \tilde{W}, \varphi)|_{L^1([0,T] \times \tilde{\Omega})} = 0.$$

Proof. Assertion (5.7.34) follows from the equality

$$\left|\langle \tilde{u}_{n}(\cdot), \varphi \rangle_{\mathrm{H}} - \langle \tilde{u}(\cdot), \varphi \rangle_{\mathrm{H}} \right|_{L^{2}([0,T] \times \tilde{\Omega})}^{2} = \tilde{\mathbb{E}} \left[\int_{0}^{T} \left| \langle \tilde{u}_{n}(t) - \tilde{u}(t), \varphi \rangle_{\mathrm{H}} \right|^{2} dt \right]$$

and Lemma 5.7.8 (a). Let us move to the proof of assertion (5.7.35). Note that by the Fubini theorem, we have

$$\begin{split} &|\Lambda_n(\tilde{u}_n,\tilde{W}_n,\varphi) - \Lambda(\tilde{u},\tilde{W},\varphi)|_{L^1([0,T]\times\tilde{\Omega})} \\ &= \int_0^T \tilde{\mathbb{E}} \Big[|\Lambda_n(\tilde{u}_n,\tilde{W}_n,\varphi)(t) - \Lambda(\tilde{u},\tilde{W},\varphi)(t)| \, \Big] dt. \end{split}$$

To conclude the proof of Corollary 5.7.9 it is sufficient to note that by Lemma 5.7.8 (b) – (g), each term on the right hand side of (5.7.32) tends at least in $L^1([0,T] \times \tilde{\Omega})$ to the corresponding term in (5.7.33).

Step 2. Since u_n is a solution of the stochastic constrained Navier-Stokes equations (5.2.4), for all $t \in [0,T]$ and $\varphi \in V$

$$\langle u_n(t), \varphi \rangle_{\mathrm{H}} = \Lambda_n(u_n, W, \varphi)(t), \qquad \mathbb{P}\text{-a.s.}$$

In particular,

$$\int_{0}^{T} \mathbb{E}\left[\left|\langle u_{n}(t), \varphi \rangle_{\mathrm{H}} - \Lambda_{n}(u_{n}, W, \varphi)(t)\right|\right] dt = 0.$$

Since $\mathscr{L}(u_n, W) = \mathscr{L}(\tilde{u}_n, \tilde{W}_n)$,

$$\int_0^1 \tilde{\mathbb{E}}\big[|\langle \tilde{u}_n(t), \varphi \rangle_{\mathrm{H}} - \Lambda_n(\tilde{u}_n, \tilde{W}_n, \varphi)(t)| \,\big] \, dt = 0.$$

Moreover, by (5.7.34) and (5.7.35)

$$\int_0^T \tilde{\mathbb{E}}\big[|\langle \tilde{u}(t), \varphi \rangle_{\mathrm{H}} - \Lambda(\tilde{u}, \tilde{W}, \varphi)(t)| \, \big] \, dt = 0.$$

Hence for Lebesgue-almost all $t \in [0, T]$ and $\tilde{\mathbb{P}}$ -almost all $\omega \in \tilde{\Omega}$

$$\langle \tilde{u}(t), \varphi \rangle_{\mathrm{H}} - \Lambda(\tilde{u}, \tilde{W}, \varphi)(t) = 0,$$

i.e. for Lebesgue-almost all $t \in [0, T]$ and $\tilde{\mathbb{P}}$ -almost all $\omega \in \tilde{\Omega}$

(5.7.36)
$$\langle \tilde{u}(t), \varphi \rangle_{\mathrm{H}} + \int_{0}^{t} \langle \mathrm{A}\tilde{u}(s), \varphi \rangle \, ds + \int_{0}^{t} \langle B(\tilde{u}(s)), \varphi \rangle \, ds$$
$$= \langle \tilde{u}(0), \varphi \rangle_{\mathrm{H}} + \int_{0}^{t} \langle |\nabla u(s)|_{L^{2}}^{2} u(s), \varphi \rangle \, ds + \langle \int_{0}^{t} Cu(s) \circ d\tilde{W}(s), \varphi \rangle.$$

Putting $\tilde{\mathscr{U}} := (\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{P}}, \tilde{\mathbb{F}})$, we infer that the system $(\tilde{\mathscr{U}}, \tilde{W}, \tilde{u})$ is a martingale solution of equation (5.2.4). By (5.7.12) and (5.7.13) the process \tilde{u} satisfies inequality (5.7.6). The proof of Theorem 5.7.7 is thus complete.

5.8 Sequentially weak Feller property

In this section we show that the family $\{T_t\}_{t\geq 0}$ defined by formula (5.8.1) is sequentially weakly Feller. We show that the weak convergence of the solutions of SCNSE in V is sufficient to establish the sequentially weak Feller property of $\{T_t\}_{t\geq 0}$. This property, along with some a'priori estimates (e.g. boundedness in probability) implies existence of an invariant measure, see [62] for a generalised result and [25] for a particular case, i.e. NSEs in two dimensional unbounded domains. However, so far we have been unable to find such a'priori bounds.

Let us fix a stochastic basis $(\Omega, \mathscr{F}, \mathbb{F}, \mathbb{P})$ with the filtration $\mathbb{F} = \{\mathscr{F}_t\}_{t\geq 0}$ and an \mathbb{R}^m -valued standard Wiener process W on this stochastic basis. By $u(t, u_0)$ we denote the pathwise unique strong solution to equation (5.2.4), defined on the above stochastic basis (which exists by Theorem 5.3.7). For any bounded Borel function $\varphi \in \mathscr{B}_b(V)$, $t \geq 0$, we define a function $T_t \varphi \colon V \to \mathbb{R}$ by

(5.8.1)
$$(T_t\varphi)(u_0) := \mathbb{E}\left[\varphi(u(t,u_0))\right], \quad u_0 \in \mathbf{V}$$

It follows from Lemma 5.6.1 and Ondrejat [69] (see also [17]) that $T_t \varphi \in \mathscr{B}_b(V)$ and $\{T_t\}_{t\geq 0}$ is a semigroup on $\mathscr{B}_b(V)$. Moreover, $\{T_t\}_{t\geq 0}$ is a Feller semigroup, i.e. T_t maps $C_b(V)$ into itself.

We also have a different version of the Feller property, which is proved in the following theorem.

Theorem 5.8.1. The semigroup $\{T_t\}_{t\geq 0}$ is sequentially weakly Feller, i.e., if $\varphi : \mathbb{V} \to \mathbb{R}$ is a bounded sequentially weakly continuous function, then for t > 0, $T_t \varphi : \mathbb{V} \to \mathbb{R}$ is also a bounded sequentially weakly continuous function. In particular, if $\xi_n \to \xi$ weakly in \mathbb{V} then,

(5.8.2)
$$T_t \varphi(\xi_n) \to T_t \varphi(\xi)$$

Proof. Let us choose and fix $0 < t \le T, \xi \in V$ and $\varphi : V \to \mathbb{R}$ be a bounded weakly continuous function. Need to show that $T_t \varphi$ is sequentially weakly Feller in V. For this aim let us choose an V-valued sequence (ξ_n) weakly convergent in V to ξ . Since the function $T_t \varphi : V \to \mathbb{R}$ is bounded, we only need to prove (5.8.2).

Let $u_n(\cdot) = u(\cdot, \xi_n)$ be the strong solution of (5.2.4) on [0, T] with the initial data ξ_n and let $u(\cdot) = u(\cdot, \xi)$ be the strong solution of (5.2.4) with the initial data ξ on the same stochastic basis $(\Omega, \mathscr{F}, \mathbb{F}, \mathbb{P}, W)$, which exist by Theorem 5.3.7. By Theorem 5.7.7, about the continuous dependence on the initial data, there exist

• a subsequence $(n_k)_k$,

- a stochastic basis $(\tilde{\Omega}, \tilde{\mathbb{F}}, \tilde{\mathscr{F}}, \tilde{\mathbb{P}})$, where $\tilde{\mathbb{F}} = {\{\tilde{\mathscr{F}}_s\}_{s \in [0,T]}}$,
- an \mathbb{R}^m -valued $\tilde{\mathbb{F}}$ -Wiener process \tilde{W} ,
- and progressively measurable processes $\tilde{u}(s)$, $(\tilde{u}_{n_k}(s))_{k\geq 1}$, $s \in [0, T]$ (defined on this basis) with laws supported in \mathcal{Z}_T such that
 - (5.8.3) \tilde{u}_{n_k} has the same law as u_{n_k} on \mathcal{Z}_T and $\tilde{u}_{n_k} \to \tilde{u}$ in \mathcal{Z}_T , $\tilde{\mathbb{P}}$ a.s.

and the system

$$(5.8.4) \qquad \qquad (\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{F}}, \tilde{\mathbb{P}}, \tilde{W}, \tilde{u})$$

is a martingale solution to (5.2.4) on the interval [0, T] with the initial data ξ .

In particular, by (5.8.3), $\tilde{\mathbb{P}}$ -almost surely¹

$$\tilde{u}_{n_k}(t) \rightarrow \tilde{u}(t)$$
 weakly in V.

Since the function $\varphi: V \to \mathbb{R}$ is sequentially weakly continuous, we infer that $\tilde{\mathbb{P}}$ -a.s.,

$$\varphi(\tilde{u}_{n_k}(t)) \to \varphi(\tilde{u}(t))$$
 in \mathbb{R} .

Since the function φ is also bounded, by the Lebesgue dominated convergence theorem we infer that

(5.8.5)
$$\lim_{k \to \infty} \tilde{\mathbb{E}} \left[\varphi(\tilde{u}_{n_k}(t)) \right] = \tilde{\mathbb{E}} \left[\varphi(\tilde{u}(t)) \right].$$

From the equality of laws of \tilde{u}_{n_k} and u_{n_k} , $k \in \mathbb{N}$, on the space \mathcal{Z}_T we infer that \tilde{u}_{n_k} and u_{n_k} have the same laws on V_w and so

(5.8.6)
$$\widetilde{\mathbb{E}}\left[\varphi(\tilde{u}_{n_k}(t))\right] = \mathbb{E}\left[\varphi(u_{n_k}(t))\right].$$

On the other hand, R.H.S. of (5.8.6) is equal by (5.8.1), to $T_t \varphi(\xi_{n_k})$.

Since u, by assumption, is a martingale solution of (5.2.4) with the initial data ξ and by the above, \tilde{u} is also a martingale solution with the initial data ξ . Thus, by Corollary 5.6.3, we infer that

the processes u and \tilde{u} have the same law on the space \mathcal{Z}_T .

Hence

(5.8.7)
$$\tilde{\mathbb{E}}\left[\varphi(\tilde{u}(t))\right] = \mathbb{E}\left[\varphi(u(t))\right].$$

As before, the R.H.S. of (5.8.7) is equal by (5.8.1), to $T_t \varphi(\xi)$.

¹Let us observe that it would be sufficient to have strong convergence below. But we have been unable to get such a stronger result. The power of our method lies in the fact that the weak convergence is sufficient for our purposes.

Thus by (5.8.5), (5.8.6) and (5.8.7), we infer

$$\lim_{k\to\infty} T_t \varphi(\xi_{n_k}) = T_t \varphi(\xi).$$

Using the subsequence argument, we can conclude that the whole sequence $(T_t \varphi(\xi_n))_{n \in \mathbb{N}}$ is convergent and

$$\lim_{n\to\infty}T_t\varphi(\xi_n)=T_t\varphi(\xi).$$

C H A P T E R

STOCHASTIC TAMED NAVIER-STOKES EQUATIONS ON \mathbb{R}^3

he tamed Navier-Stokes equations on \mathbb{R}^3 were introduced by Röckner and Zhang [75], where they proved the existence and uniqueness of a smooth solution to tamed 3D Navier-Stokes equations in the whole space. Later on in [76] they proved the existence of a unique strong solution to stochastic tamed 3D Navier-Stokes equations in the whole space and for the periodic boundary case using a result from Stroock and Varadhan [87]. In this chapter we reprove their results for a slightly simplified system using a self-contained approach. We generalise Röckner and Zhang result corresponding to estimate on L^4 -norm of the solution from torus to the Euclidean space \mathbb{R}^3 . We also establish the existence of an invariant measure on \mathbb{R}^3 for time homogeneous damped tamed 3D Navier-Stokes equations, given by (6.6.1).

6.1 Introduction

We are interested in the study of the stochastic tamed Navier-Stokes equations (NSE) on \mathbb{R}^3 which were introduced by Röckner and Zhang [76]. We consider the following stochastic tamed NSEs with viscosity v (assumed to be positive), on \mathbb{R}^3 :

$$du(t,x) = \left[v \Delta u(t,x) - (u(t,x) \cdot \nabla) u(t,x) - \nabla p(t,x) - g(|u(t,x)|^2) u(t) + f(x,u(t,x)) \right] dt$$

$$(6.1.1) \qquad + \sum_{j=1}^{\infty} \left[(\sigma_j(t,x) \cdot \nabla) u(t) + \nabla \tilde{p}_j(t,x) \right] dW_t^j, \qquad (t,x) \in [0,T] \times \mathbb{R}^3,$$

subject to the incompressibility condition

(6.1.2)
$$\operatorname{div} u(t,x) = 0, \quad (t,x) \in [0,T] \times \mathbb{R}^3,$$

and the initial condition

(6.1.3)
$$u(0,x) = u_0(x), \quad x \in \mathbb{R}^3,$$

where p(t,x) and $\tilde{p}_k(t,x)$ are unknown scalar functions, and the taming function $g : \mathbb{R}_+ \to \mathbb{R}_+$ is smooth and satisfies for some $N \in \mathbb{N}$

(6.1.4)
$$\begin{cases} g(r) = 0, & \text{if } r \le N, \\ g(r) = (r - N)/\nu, & \text{if } r \ge N + 1, \\ 0 \le g'(r) \le 2/(\nu \land 1), \ r \in [N, N + 1]. \end{cases}$$

 $\{W_t^j; t \ge 0, j = 1, 2, ...\}$ is a sequence of independent one-dimensional standard $\mathbb{F} = (\mathscr{F}_t)_{t\ge 0}$ -Brownian Motions on the complete filtered probability space $(\Omega, \mathscr{F}, \mathbb{F}, \mathbb{P})$. The stochastic integral is understood as Itô integral. The arguments of the coefficients are given as follows:

$$\mathbb{R}^{3} \times \mathbb{R}^{3} \ni (x, u) \mapsto f(x, u) \in \mathbb{R}^{3}$$
$$\mathbb{R}_{+} \times \mathbb{R}^{3} \ni (t, x) \mapsto \sigma(t, x) \in \mathbb{R}^{3} \times \ell^{2},$$

where ℓ^2 is the Hilbert space consisting of all sequences of square summable numbers with standard norm $\|\cdot\|_{\ell^2}$. In the following f and σ are always assumed to be measurable with respect to all their variables.

In classical Navier-Stokes equations on \mathbb{R}^3 with $u_0 \in V$ (see Section 6.2) there is only existence of local solution [88]. The addition of tamed term enables to prove the global existence [75]. The non-explosion of the solution is due to the tamed term. Röckner and Zhang [76] proved the existence of a martingale solution to (6.1.1) (for more generalised noise) in the absence of compact Sobolev embeddings. They use the localization method to prove tightness, a method introduced by Stroock and Varadhan [87]. In this chapter we present a self-contained proof of the same. In order to prove the existence of a martingale solution they use the Faedo-Galerkin approximation with the non-classical finite dimensional space $H_n^1 = \operatorname{span}\{e_i, i = 1 \cdots n\}$ where $\mathscr{E} = \{e_i\}_{i \in \mathbb{N}} \subset \mathcal{V}$ (see Section 6.2) is the orthonormal basis of H^1 . They also require that in the case of periodic boundary conditions \mathscr{E} is an orthogonal basis of H^0 which was essential in obtaining the L^4 -estimate of the solution. We generalise this result to the Euclidean space \mathbb{R}^3 . Another reason for them to choose periodic boundary conditions was the compactness of $H^2 \hookrightarrow H^1$ embedding, which along with the L^4 -estimate of the solution was crucial in establishing the existence of invariant measures. We do not require this embedding and hence are able to obtain the existence of invariant measures for time homogeneous damped tamed Navier-Stokes equations (6.6.1) on \mathbb{R}^3 .

In the present chapter we prove the existence of a unique strong solution to the stochastic tamed 3D Navier-Stokes equations (6.1.1) under some natural assumptions (A1) - (A2) on f and σ (see Section 6.2). To prove the existence of strong solution we use the Yamada-Watanabe theorem [95] which states that the existence of martingale solutions plus pathwise uniqueness implies the existence of a unique strong solution. In order to establish the existence of martingale solutions, instead of using the standard Faedo-Galerkin approximations we use a different approach motivated from [42] and [61]. We study a truncated SPDE on an infinite dimensional space H_n , defined in the Section 6.4 and then use the tightness criterion, the Jakubowski-Skorohod Theorem

and the martingale representaion theorem to prove the existence of martingale solutions. The essential reason, for us to incorporate this approximation scheme was the non-commutativity of gradient operator (∇) with the standard Faedo-Galerkin projection operator P_n [13, Section 5]. The commutativity is essential for us to obtain a'priori bounds. We also prove the existence of invariant measures, Theorem 6.6.1, for time homogeneous damped tamed Navier-Stokes equations (6.6.1) under the assumptions (A1)' - (A3)' (see Section 6.6). We use the technique (Theorem 6.6.4) of Maslowski and Seidler [62] working with weak topologies to establish the existence of invariant measures. We show the two conditions of Theorem 6.6.4, boundedness in probability and sequentially weak Feller property are satisfied for the semigroup $(T_t)_{t\geq 0}$, defined by (6.6.2). In contrast to Röckner and Zhang [76], a'priori bound on L^4 -norm of the solution plays an essential role in the existence of martingale solutions and not in the existence of invariant measures.

This chapter is organised as follows: in Section 6.2, we recall some standard notations and results and set the assumptions on f and σ . We also establish certain estimates on the tamed term which we use later in Sections 6.4 and 6.5. In Section 6.3, we establish the tightness criterion and state Skorohod's theorem which we use along with a'priori estimates obtained in the Section 6.5 to prove the existence of a martingale solution and path-wise uniqueness of the solution. In Section 6.4 we introduce our truncated SPDE and describe the approximation scheme motivated from [42, 61], along with all the machinery required. Finally in Section 6.6 we establish the existence of an invariant measure for time homogeneous damped tamed 3D Navier-Stokes equations (6.6.1).

6.2 Functional setting

6.2.1 Notations

Let $C_0^{\infty}(\mathbb{R}^3, \mathbb{R}^3)$ denote the set of all smooth functions from \mathbb{R}^3 to \mathbb{R}^3 with compact supports. For $p \ge 1$, let $L^p(\mathbb{R}^3, \mathbb{R}^3)$ be the vector valued L^p -space in which the norm is denoted by $\|\cdot\|_{L^p}$. If p = 2, then $L^2(\mathbb{R}^3, \mathbb{R}^3)$ is a Hilbert space with the scalar product given by

$$\langle u, \mathbf{v} \rangle_{L^2} := \int_{\mathbb{R}^3} u(x) \cdot \mathbf{v}(x) dx, \quad u, \mathbf{v} \in L^2(\mathbb{R}^3, \mathbb{R}^3).$$

Let $H^1(\mathbb{R}^3, \mathbb{R}^3)$ stand for the Sobolev space of all $u \in L^2(\mathbb{R}^3, \mathbb{R}^3)$ for which there exist weak derivatives $D_i u \in L^2(\mathbb{R}^3, \mathbb{R}^3)$, i = 1, ..., 3. It is a Hilbert space with the scalar product given by

$$\langle u, \mathbf{v} \rangle_{H^1} := \langle u, \mathbf{v} \rangle_{L^2} + ((u, \mathbf{v})), \quad u, \mathbf{v} \in H^1(\mathbb{R}^3, \mathbb{R}^3),$$

where

(6.2.1)
$$((u, \mathbf{v})) := \langle \nabla u, \nabla \mathbf{v} \rangle_{L^2} = \sum_{i=1}^3 \int_{\mathbb{R}^3} \frac{\partial u}{\partial x_i} \cdot \frac{\partial \mathbf{v}}{\partial x_i} dx, \quad u, \mathbf{v} \in H^1(\mathbb{R}^3, \mathbb{R}^3).$$

Let

$$\begin{split} \mathcal{V} &:= \left\{ u \in C_0^{\infty}(\mathbb{R}^3, \mathbb{R}^3) : \operatorname{div} u = 0 \right\}, \\ \mathrm{H} &:= \mathrm{the\ closure\ of}\ \ \mathcal{V} \ \mathrm{in}\ \ L^2(\mathbb{R}^3, \mathbb{R}^3), \\ \mathrm{V} &:= \mathrm{the\ closure\ of}\ \ \mathcal{V} \ \mathrm{in}\ \ H^1(\mathbb{R}^3, \mathbb{R}^3), \\ \mathrm{D}(\mathrm{A}) &:= \mathrm{H} \cap H^2(\mathbb{R}^3, \mathbb{R}^3). \end{split}$$

On H we consider the scalar product and the norm inherited from $L^2(\mathbb{R}^3,\mathbb{R}^3)$ and denote them by $\langle\cdot,\cdot\rangle_{\mathrm{H}}$ and $\|\cdot\|_{\mathrm{H}}$ respectively, i.e.

$$\langle u, \mathbf{v} \rangle_{\mathbf{H}} := \langle u, \mathbf{v} \rangle_{L^2}, \qquad |u|_{\mathbf{H}} := |u|_{L^2}, \quad u, \mathbf{v} \in \mathbf{H}.$$

On V we consider the scalar product and norm inherited from $H^1(\mathbb{R}^3, \mathbb{R}^3)$, i.e.

(6.2.2)
$$\langle u, v \rangle_{\mathbf{V}} := \langle u, v \rangle_{L^2} + ((u, v)), \qquad \|u\|_{\mathbf{V}}^2 := |u|_{\mathbf{H}}^2 + |\nabla u|_{L^2}^2, \quad u, v \in \mathbf{V},$$

where $((\cdot, \cdot))$ is defined in (6.2.1). D(A) is a Hilbert space under the graph norm

$$|u|_{\mathcal{D}(\mathcal{A})}^2 := |u|_{\mathcal{H}}^2 + |\mathcal{A}u|_{L^2}^2, \qquad u \in \mathcal{D}(\mathcal{A}),$$

where the inner product is given by

$$\langle u, v \rangle_{\mathrm{D}(\mathrm{A})} := \langle u, v \rangle_{\mathrm{H}} + \langle \mathrm{A}u, \mathrm{A}v \rangle_{L^2}, \qquad u, v \in \mathrm{D}(\mathrm{A})$$

6.2.2 Some operators

Let us recall the tri-linear form $b: L^p \times W^{1,q} \times L^r \to \mathbb{R}$ which was introduced earlier in Chapter 3

(6.2.3)
$$b(u,w,\mathbf{v}) = \int_{\mathbb{R}^3} (u \cdot \nabla w) \mathbf{v} \, dx, \qquad u \in L^p, \, w \in W^{1,q}, \, \mathbf{v} \in L^r,$$

where $p, q, r \in [1, \infty]$, satisfy

$$\frac{1}{p} + \frac{1}{q} + \frac{1}{r} \le 1$$

We will recall the fundamental properties of the form b which are valid in unbounded domains.

By the Sobolev embedding theorem and Hölder inequality, we obtain the following estimates

$$(6.2.4) |b(u,w,v)| \le ||u||_{L^4} ||w||_V ||v||_{L^4}, \quad u,v \in L^4, w \in V$$

$$(6.2.5) \leq c \|u\|_{\mathcal{V}} \|w\|_{\mathcal{V}} \|v\|_{\mathcal{V}}, \quad u, v, w \in \mathcal{V}$$

for some positive constant *c*. Thus the form *b* is continuous on V. Moreover, if we define a bilinear map *B* by $B(u,w) := b(u,w,\cdot)$ then by inequality (6.2.5) we infer that $B(u,w) \in V'$ for all $u, w \in V$ and that the following inequality holds

$$(6.2.6) ||B(u,w)||_{V'} \le c ||u||_{V} ||w||_{V}, \quad u,w \in V.$$

Moreover, the mapping $B: V \times V \rightarrow V'$ is bilinear and continuous.

Let us, for any s > 0, define the following standard scale of Hilbert spaces (see Rudin [80] for the definition of $H^s(\mathbb{R}^3, \mathbb{R}^3)$ space)

$$V_s :=$$
 the closure of \mathcal{V} in $H^s(\mathbb{R}^3, \mathbb{R}^3)$.

If $s > \frac{d}{2} + 1$, then by the Sobolev Embedding theorem

(6.2.7)
$$H^{s-1}(\mathbb{R}^3, \mathbb{R}^3) \hookrightarrow C_b(\mathbb{R}^3, \mathbb{R}^3) \hookrightarrow L^{\infty}(\mathbb{R}^3, \mathbb{R}^3).$$

Here $C_b(\mathbb{R}^3, \mathbb{R}^3)$ denotes the space of continuous and bounded \mathbb{R}^3 -valued functions defined on \mathbb{R}^3 . If $u, w \in V$ and $v \in V_s$ with $s > \frac{d}{2} + 1$ then

$$|b(u, w, \mathbf{v})| = |b(u, \mathbf{v}, w)| \le |u|_{L^2} |w|_{L^2} ||\nabla \mathbf{v}||_{L^\infty} \le c |u|_{L^2} ||w|_{L^2} ||\mathbf{v}||_{V_s}$$

for some constant c > 0. Thus *b* can be uniquely extended to the tri-linear form (denoted by the same letter)

$$b: \mathbf{H} \times \mathbf{H} \times \mathbf{V}_s \to \mathbb{R}$$

and

$$|b(u,w,\mathbf{v})| \le c |u|_{\mathbf{H}} |w|_{\mathbf{H}} ||\mathbf{v}||_{\mathbf{V}_s}, \quad u,w \in \mathbf{H}, \mathbf{v} \in \mathbf{V}_s.$$

At the same time, the operator B can be uniquely extended to a bounded bilinear operator

$$B: H \times H \rightarrow V'_{s}$$

In particular, it satisfies the following estimate

(6.2.8)
$$||B(u,w)||_{V'_{a}} \le c|u|_{H}|w|_{H}, \quad u,w \in H.$$

We will also use the notation, B(u) := B(u, u).

Let us assume that s > 1. It is clear that V_s is dense in V and the embedding $j_s : V_s \hookrightarrow V$ is continuous. Then there exists [23, Lemma C.1] a Hilbert space U such that $U \subset V_s$, U is dense in V_s and

the natural embedding $i_s: U \hookrightarrow V_s$ is compact.

The following Gagliardo-Nirenberg interpolation inequality will be used frequently. Let $q \in [1,\infty]$ and $m \in \mathbb{N}$. If

$$\frac{1}{q} = \frac{1}{2} - \frac{m\alpha}{3}, \quad 0 \le \alpha \le 1,$$

then for any $u \in H^m$ there exists a constant $C_{m,q}$ depending on m and q such that

(6.2.9)
$$\|u\|_{L^q} \le C_{m,q} \|u\|_{H^m}^{\alpha} \|u\|_{L^2}^{1-\alpha}$$

Let Π be the orthogonal projection from $L^2(\mathbb{R}^3, \mathbb{R}^3)$ to H, famously known as the Leray-Helmholtz projection [88]. For any $u \in H$ and $v \in L^2(\mathbb{R}^3, \mathbb{R}^3)$, we have

$$\langle u, \mathbf{v} \rangle_{\mathbf{H}} := \langle u, \Pi \mathbf{v} \rangle_{\mathbf{H}} = \langle u, \mathbf{v} \rangle_{L^2}.$$

The Stokes operator A: $D(A) \rightarrow H$, is given by

$$Au = -\Pi(\Delta u), \qquad u \in D(A).$$

The bilinear map $B: H \times H \rightarrow H$ will be given by

$$B(u, \mathbf{v}) = \Pi((u \cdot \nabla)\mathbf{v}), \quad u, \mathbf{v} \in \mathbf{H}.$$

6.2.3 Assumptions

We now introduce the assumptions on the coefficients f and σ :

(A1) A function $f : \mathbb{R}^3 \times \mathbb{R}^3 \to \mathbb{R}^3$ is of C^1 class and for any T > 0 there exist a constant $C_{T,f} > 0$ such that for any $x \in \mathbb{R}^3, u \in \mathbb{R}^3$,

$$|\partial_{x^j} f(x,u)|^2 + |f(x,u)|^2 \le C_{T,f} \cdot (1+|u|^2), \quad j=1,2,3,$$

 $|\partial_{u^j} f(x,u)| \le C_{T,f}.$

(A2) A measurable function $\sigma : [0,\infty) \times \mathbb{R}^3 \to \mathbb{R}^3$ of C^1 class with respect to the *x*-variable and for any T > 0 there exist a constant $C_{\sigma,T} > 0$ such that for all $t \in [0,T]$, $x \in \mathbb{R}^3$

$$\|\partial_{x^{j}}\sigma(t,x)\|_{\ell^{2}} \le C_{\sigma,T}, \quad j=1,2,3$$

and, for all $t \in [0,\infty)$, $x \in \mathbb{R}^3$

(6.2.10)
$$\|\sigma(t,x)\|_{\ell^2}^2 \le \frac{1}{4}.$$

Below for the sake of simplicity the variable "x" in the coefficients will be dropped.

Define, for $k \in \mathbb{N}$, $G_j : [0, T] \times H \rightarrow H$ by

(6.2.11)
$$G_j(t,u) := \Pi[(\sigma_j(t) \cdot \nabla)u], \quad t \in [0,T], u \in \mathbf{H}.$$

Then a function $G: \mathbf{H} \to \mathcal{T}_2(\ell^2; \mathbf{H})$ is defined by

(6.2.12)
$$G(u)(k) = \sum_{j=1}^{\infty} k_j G_j(u), \quad u \in \mathbf{H}.$$

Let $\{e_j\}_{j=1}^{\infty}$ be the orthonormal basis of ℓ^2 then we see that (6.2.12) implies

$$G(u)(e_j) = G_j(u)$$

For simplicity we will assume that v = 1. In particular, the function *g* defined by (6.1.4) will from now on be given by

(6.2.13)
$$\begin{cases} g(r) = 0, & \text{if } r \le N, \\ g(r) = (r - N), & \text{if } r \ge N + 1, \\ 0 \le g'(r) \le 2, & r \in [N, N + 1]. \end{cases}$$

Observe that the function g defined in this way satisfies

$$(6.2.14) |g(r)| \le r, r \ge 0,$$

and

(6.2.15)
$$|g(r) - g(r')| \le 2|r - r'|, \quad r, r' \ge 0.$$

We are interested in proving the existence of solutions to (6.1.1) - (6.1.3). In particular, we want to prove the existence of divergence free vector fields u and scalar pressure p satisfying (6.1.1) and (6.1.3). Thus we project equation (6.1.1) using the orthogonal projection operator Π on the space H of the L^2 -valued, divergence free vector fields. On projecting, we obtain the following abstract stochastic evolution equation:

(6.2.16)
$$\begin{cases} du(t) = \left[-Au(t) - B(u(t)) - \Pi[g(|u(t)|^2)u(t)] + \Pi f(u(t))\right] dt + \sum_{j=1}^{\infty} G_j(t, u(t)) dW_j(t), \\ u(0) = u_0, \end{cases}$$

where we assume that $u_0 \in V$ and $W(t) = (W_j(t))_{j=1}^{\infty}$ is a cylindrical Wiener process on ℓ^2 and $\{W^j(t), t \ge 0, j \in \mathbb{N}\}$ is an infinite sequence of independent standard Brownian motions. We will repeatedly use the following notation

$$G(t,u)dW(t) = \sum_{j=1}^{\infty} G_j(t,u)dW_j(t).$$

We will need the following lemma in Section 6.4 to obtain the a'priori estimates.

Lemma 6.2.1. *i)* For any $u \in D(A)$

(6.2.17)
$$|\langle B(u), u \rangle_{\mathcal{V}}| \leq \frac{1}{2} |u|_{\mathcal{D}(\mathcal{A})}^2 + \frac{1}{2} ||u| \cdot |\nabla u||_{L^2}^2$$

ii) If $u \in H$, then

(6.2.18)
$$\begin{cases} ((-g(|u|^2)u, u)) \le C_N |\nabla u|_{L^2}^2 - 2 ||u| \cdot |\nabla u||_{L^2}^2, \\ \langle -g(|u|^2)u, u \rangle_{\mathrm{H}} \le - ||u||_{L^4}^4 + C_N |u|_{\mathrm{H}}^2, \end{cases}$$

where the semi-inner product $((\cdot, \cdot))$ is defined in (6.2.1).

iii) For any $u \in D(A)$,

(6.2.19)
$$\|G(t,u)\|_{\mathcal{T}_{2}(\ell^{2};\mathrm{H})}^{2} \leq \frac{1}{4} |\nabla u|_{L^{2}}^{2},$$

(6.2.20)
$$\|G(t,u)\|_{\mathcal{F}_2(\ell^2; \mathbb{V})}^2 \leq \frac{1}{2} |Au|_{L^2}^2 + C_{\sigma,T} |\nabla u|_{L^2}^2.$$

Proof. Let $u \in D(A)$. Since $\langle B(u), u \rangle_{H} = 0$, using the Cauchy-Schwartz and Young inequality we get

$$\begin{split} |\langle B(u), u \rangle_{\mathcal{V}}| &= |\langle B(u), (I - \Delta)u \rangle_{\mathcal{H}}| \le |-\Delta u|_{L^{2}} |(u \cdot \nabla)u|_{L^{2}} \\ &\le \frac{1}{2} |-\Delta u|_{L^{2}}^{2} + \frac{1}{2} |(u \cdot \nabla)u|_{L^{2}}^{2} \le \frac{1}{2} |u|_{\mathcal{D}(\mathcal{A})}^{2} + \frac{1}{2} ||u| \cdot |\nabla u||_{L^{2}}^{2}. \end{split}$$

Let us introduce a function $\phi \colon \mathbb{R}_+ \to \mathbb{R}$ such that $g(r) = r - \phi(r)$ which in particular satisfies

$$\phi(r) = \begin{cases} r, & r \leq N, \\ N, & r \geq N+1. \end{cases}$$

Since $\phi'(r) = 1 - g'(r)$, there exists a constant $C_N > 0$ such that $|\phi'(r)| \leq \tilde{C}_N$ for every $r \geq 0$. Moreover

$$\phi'(r) = \begin{cases} 1, & r \leq N, \\ 0, & r \geq N+1. \end{cases}$$

Hence, we infer that $|\phi'(r) \cdot r|$ is bounded by some positive constant C_N .

Let $u \in D(A)$. Using the definitions of g and of semi-norm $((\cdot, \cdot))$, we get

$$\begin{aligned} ((-g(|u|^2)u, u)) &= -\langle g(|u|^2)u, -\Delta u \rangle_{L^2} \\ &= -\int_{\mathbb{R}^3} g(|u(x)|^2)u(x) \cdot (-\Delta u(x)) \, dx \\ &= -\int_{\mathbb{R}^3} |u(x)|^2 u(x) (-\Delta u(x)) \, dx + \int_{\mathbb{R}^3} \phi(|u(x)|^2)u(x) (-\Delta u(x)) \, dx. \end{aligned}$$

Thus, on integration by parts we get

$$((-g(|u|^{2})u,u)) = -\left[\int_{\mathbb{R}^{3}} |u(x)|^{2} \cdot |\nabla u(x)|^{2} dx + 2 \int_{\mathbb{R}^{3}} |u(x)|^{2} \cdot |\nabla u(x)|^{2} dx\right]$$

(6.2.21)
$$+ \int_{\mathbb{R}^{3}} \phi(|u(x)|^{2}) \cdot |\nabla u(x)|^{2} dx + \sum_{j,k=1}^{3} \int_{\mathbb{R}^{3}} D_{k} \left(\phi(|u(x)|^{2})\right) u_{j}(x) \cdot D_{k} u_{j}(x) dx.$$

Using the bound on $|\phi'(r) \cdot r|$, we obtain

$$(6.2.22) \qquad \sum_{j,k=1}^{3} \int_{\mathbb{R}^{3}} D_{k} \left(\phi(|u(x)|^{2}) \right) u_{j}(x) \cdot D_{k} u_{j}(x) dx = 2 \sum_{k=1}^{3} \int_{\mathbb{R}^{3}} \phi'(|u(x)|^{2}) \langle u(x), D_{k} u(x) \rangle_{\mathbb{R}^{3}}^{2} dx$$
$$\leq 2 \int_{\mathbb{R}^{3}} |\phi'(|u(x)|^{2})| \cdot |u(x)|^{2} |\nabla u(x)|^{2} dx \leq C_{N} |\nabla u|_{L^{2}}^{2}.$$

Since $g(r) \ge 0$, $|\phi(r)| \le r$ for all $r \ge 0$. Thus using (6.2.22) in (6.2.21), we obtain

$$\begin{aligned} ((-g(|u|^{2})u, u)) &\leq -3 \big| |u| \cdot |\nabla u| \big|_{L^{2}}^{2} + C_{N} |\nabla u|_{L^{2}}^{2} + \big| |u| \cdot |\nabla u| \big|_{L^{2}}^{2} \\ &= C_{N} |\nabla u|_{L^{2}}^{2} - 2 \big| |u| \cdot |\nabla u| \big|_{L^{2}}^{2}. \end{aligned}$$

Now to prove the second inequality, we take the similar approach. Let $u \in H$, then

$$\langle -g(|u|^2)u, u \rangle_{\mathrm{H}} = -\int_{\mathbb{R}^3} |u(x)|^2 |u(x)|^2 dx + \int_{\mathbb{R}^3} \phi(|u(x)|^2) |u(x)|^2 dx.$$

By the definition of ϕ there exists a constant $C_N > 0$ such that $|\phi(r)| \le C_N$ for all r > 0, thus

$$\langle -g(|u|^2)u, u \rangle_{\mathrm{H}} \leq - \|u\|_{L^4}^4 + C_N |u|_{\mathrm{H}}^2.$$

This completes the proof of part (ii).

Now for (iii), by (A1) and (A2) we have

$$\begin{split} \|G(t,u)\|_{\mathcal{T}_{2}(\ell^{2};\mathrm{H})}^{2} &= \sum_{j=1}^{\infty} |G_{j}(t,u)|_{\mathrm{H}}^{2} = \sum_{j=1}^{\infty} \int_{\mathbb{R}^{3}} |G_{j}(t,x,u(x))|^{2} dx \\ &\leq \int_{\mathbb{R}^{3}} \|\sigma(t,x)\|_{\ell^{2}}^{2} |\nabla u(x)|^{2} dx \leq \sup_{x \in \mathbb{R}^{3}} \|\sigma(t,x)\|_{\ell^{2}}^{2} |\nabla u|_{L^{2}}^{2} \leq \frac{1}{4} |\nabla u|_{L^{2}}^{2}. \end{split}$$

Secondly, noting that

$$\|G(t,u)\|_{\mathcal{T}_{2}(\ell^{2};\mathbf{V})}^{2} = \|G(t,u)\|_{\mathcal{T}_{2}(\ell^{2};\mathbf{H})}^{2} + \|\nabla G(t,u)\|_{\mathcal{T}_{2}(\ell^{2};\mathbf{H})}^{2}$$

and

$$\partial_{x^j} G_k(t,u) = \Pi \partial_{x^j} [(\sigma_k(t,x) \cdot \nabla) u] = \Pi \left[(\partial_{x^j} \sigma_k(t,x) \cdot \nabla) u + (\sigma_k(t,x) \cdot \nabla) \partial_{x^j} u \right].$$

Thus

$$\begin{aligned} \|\nabla G(t,u)\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2} &= \sum_{k=1}^{\infty} \left| \sum_{j=1}^{3} \Pi \left[(\partial_{x^{j}} \sigma_{k}(t,x) \cdot \nabla) u + (\sigma_{k}(t,x) \cdot \nabla) \partial_{x^{j}} u \right] \right|_{\mathrm{H}}^{2} \\ &\leq 2 \int_{\mathbb{R}^{3}} \sum_{j=1}^{3} \|\partial_{x^{j}} \sigma(t,x)\|_{\ell^{2}}^{2} |\nabla u(x)|^{2} dx + 2 \int_{\mathbb{R}^{3}} \|\sigma(t,x)\|_{\ell^{2}}^{2} |\Delta u(x)|^{2} dx \end{aligned}$$

Hence, by assumptions (A1), (A2) and (6.2.19), we have

$$\|G(t,u)\|_{\mathcal{T}_{2}(\ell^{2};\mathbb{V})}^{2} \leq \frac{1}{2} |Au|_{L^{2}}^{2} + C_{T,\sigma} |\nabla u|_{L^{2}}^{2}.$$

Remark 6.2.2. On a purely heuristic level, by the application of Itô Lemma to the function $|u|_{\rm H}^2$ and a solution *u* to (6.1.1), using Lemma 6.2.1 one obtains the following inequality

$$(6.2.23) \qquad \frac{1}{2} d|u(t)|_{\mathrm{H}}^{2} = \langle u(t), -\mathrm{A}u(t) - B(u(t)) - g(|u(t)|^{2})u(t) + f(u(t))\rangle_{\mathrm{H}} + \langle u(t), G(s, u(t)) dW_{t} \rangle_{\mathrm{H}} + \frac{1}{2} \|G_{n}(s, u_{n}(s))\|_{\mathcal{T}_{2}(\ell^{2};\mathrm{H})}^{2} \leq -\frac{7}{8} |\nabla u(t)|_{L^{2}}^{2} - \|u(t)\|_{L^{4}}^{4} + C_{N,f} |u(t)|_{\mathrm{H}}^{2} + \langle u(t), G(s, u(t)) dW_{t} \rangle_{\mathrm{H}}$$

which could lead to a'priori estimates that can be used further to prove the existence of the solution.

6.3 Compactness

Let $(\mathcal{O}_R)_{R \in \mathbb{N}}$ be a sequence of bounded open subsets of \mathbb{R}^3 with regular boundaries $\partial \mathcal{O}_R$ such that $\mathcal{O}_R \subset \mathcal{O}_{R+1}$. Let us consider the following functional spaces:

 $\mathscr{C}([0,T]; \mathbf{U}') :=$ the space of continuous functions $u : [0,T] \to \mathbf{U}'$ with the topology \mathscr{T}_1 induced by the norm $|u|_{\mathscr{C}([0,T];\mathbf{U}')} := \sup_{t \in [0,T]} |u(t)|_{\mathbf{U}'}$,

 $L^2_w(0,T;\mathbb{D}(A)) :=$ the space $L^2(0,T;\mathbb{D}(A))$ with the weak topology \mathcal{T}_2 ,

 $L^{2}(0,T; H_{loc}) :=$ the space of measurable functions $u : [0,T] \rightarrow H$ such that for all $R \in \mathbb{N}$

(6.3.1)
$$q_{T,R}(u) := \|u\|_{L^2(0,T;H_{\mathscr{O}_R})} = \left(\int_0^T \int_{\mathscr{O}_R} |u(t,x)|^2 \, dx \, dt\right)^{1/2} < \infty,$$

with the topology \mathcal{T}_3 induced by the semi-norms $(q_{T,R})_{R \in \mathbb{N}}$.

The following lemma is inspired by the classical Dubinsky Theorem (see Theorem 2.4.5) and the compactness result due to Mikulevicus and Rozovskii [65, Lemma 2.7].

Lemma 6.3.1. Let

(6.3.2)
$$\tilde{\mathcal{Z}}_T := \mathscr{C}([0,T]; \mathbf{U}') \cap L^2_w(0,T; \mathbf{D}(\mathbf{A})) \cap L^2(0,T; \mathbf{H}_{loc})$$

and let $\tilde{\mathcal{T}}$ be the supremum of the corresponding topologies. Then a set $\mathcal{K} \subset \tilde{\mathcal{I}}_T$ is $\tilde{\mathcal{T}}$ -relatively compact if the following two conditions hold :

- i) $\sup_{u \in \mathcal{X}} \int_0^T |u(s)|^2_{D(A)} ds < \infty$, i.e. \mathcal{K} is bounded in $L^2(0,T;D(A))$,
- *ii)* $\lim_{\delta \to 0} \sup_{u \in \mathcal{X}} \sup_{\substack{s,t \in [0,T] \\ |t-s| \le \delta}} |u(t) u(s)|_{U'} = 0.$

The above lemma can be proved by modifying the proof of [23, Lemma 3.1], see also [94, Theorem IV.4.1].

Let V_w denote the Hilbert space V endowed with the weak topology.

 $\mathscr{C}([0,T]; V_w) :=$ the space of weakly continuous functions $u : [0,T] \to V$ endowed with the weakest topology \mathscr{T}_4 such that for all $h \in V$ the mappings

$$\mathscr{C}([0,T]; \mathbf{V}_w) \ni u \to \langle u(\cdot), h \rangle_{\mathbf{V}} \in \mathscr{C}([0,T]; \mathbb{R})$$

are continuous. In particular $u_n \rightarrow u$ in $\mathscr{C}([0, T]; V_w)$ iff for all $h \in V$:

$$\lim_{n\to\infty}\sup_{t\in[0,T]}|\langle u_n(t)-u(t),h\rangle_{\rm V}|=0.$$

Consider the ball

 $\mathbb{B} := \{x \in \mathbf{V} : \|x\|_{\mathbf{V}} \le r\}.$

Let q be the metric compatible with the weak topology on \mathbb{B} . Let us recall the following subspace of the space $\mathscr{C}([0,T]; V_w)$

$$\mathscr{C}([0,T];\mathbb{B}_w) :=$$
 the space of weakly continuous functions $u: [0,T] \to \mathbb{V}$

(6.3.3) such that
$$\sup_{t \in [0,T]} ||u(t)||_{V} \le r$$
.

The space $\mathscr{C}([0,T];\mathbb{B}_w)$ is metrizable with metric

(6.3.4)
$$\rho(u, \mathbf{v}) = \sup_{t \in [0,T]} q(u(t), \mathbf{v}(t)).$$

Since by the Banach-Alaoglu theorem [80], the set \mathbb{B}_w is compact, $(\mathscr{C}([0,T];\mathbb{B}_w), \varrho)$ is a complete metric space.

The following lemma says that any sequence $(u_n)_{n \in \mathbb{N}} \subset \mathscr{C}([0, T]; \mathbb{B})$ convergent in $\mathscr{C}([0, T]; H)$ is also convergent in the space $\mathscr{C}([0, T]; \mathbb{B}_w)$. The proof of the lemma is similar to the proof of [24, Lemma 2.1].

Lemma 6.3.2. Let $u_n : [0,T] \rightarrow V, n \in \mathbb{N}$ be functions such that

- (*i*) $\sup_{n \in \mathbb{N}} \sup_{s \in [0,T]} \|u_n(s)\|_{\mathcal{V}} \le r$,
- (*ii*) $u_n \rightarrow u$ in $\mathscr{C}([0,T];H)$.

Then $u, u_n \in \mathscr{C}([0,T]; \mathbb{B}_w)$ and $u_n \to u$ in $\mathscr{C}([0,T]; \mathbb{B}_w)$ as $n \to \infty$.

Let

(6.3.5)
$$\mathcal{Z}_T = \mathscr{C}([0,T]; \mathbf{U}') \cap L^2_w(0,T; \mathbf{D}(\mathbf{A})) \cap L^2(0,T; \mathbf{H}_{loc}) \cap \mathscr{C}([0,T]; \mathbf{V}_w),$$

and let $\mathcal T$ be the supremum of the corresponding topologies.

Now we formulate the compactness criterion analogous to the result due to Mikulevicus and Rozowskii [65], Brzeźniak and Motyl [23, Lemma 3.3] for the space \mathcal{Z}_T .

Lemma 6.3.3. Let $(\mathcal{Z}_T, \mathcal{T})$ be as defined in (6.3.5). Then a set $\mathcal{K} \subset \mathcal{Z}_T$ is \mathcal{T} -relatively compact if the following three conditions hold

- (a) $\sup_{u \in \mathcal{X}} \sup_{s \in [0,T]} \|u(s)\|_{\mathcal{V}} < \infty$,
- (b) $\sup_{u \in \mathcal{K}} \int_0^T |u(s)|^2_{D(A)} ds < \infty$, i.e. \mathcal{K} is bounded in $L^2(0,T;D(A))$,
- (c) $\lim_{\delta \to 0} \sup_{u \in \mathcal{K}} \sup_{\substack{s,t \in [0,T] \\ |t-s| \leq \delta}} |u(t) u(s)|_{\mathrm{H}} = 0.$

Proof. Let us notice that $\mathcal{Z}_T = \tilde{\mathcal{Z}}_T \cap \mathscr{C}([0,T]; V_w)$, where $\tilde{\mathcal{Z}}_T$ is defined by (6.3.2). Let \mathcal{K} be a subset of \mathcal{Z}_T . Because of the assumption (*a*) we may consider the metric space $\mathscr{C}([0,T]; \mathbb{B}_w) \subset \mathscr{C}([0,T]; V_w)$ defined by (6.3.3) and (6.3.4) with $r = \sup_{u \in \mathcal{K}} \sup_{s \in [0,T]} ||u(s)||_V$. Because of the

assumption (b) the restriction to \mathcal{K} of the weak topology in $L^2(0,T; D(A))$ is metrizable. Since the restrictions to \mathcal{K} of the four topologies considered in \mathcal{Z}_T are metrizable, compactness of a subset of \mathcal{Z}_T is equivalent to it's sequential compactness.

Let (u_n) be a sequence in \mathcal{K} . By Lemma 6.3.1, the boundedness of the set \mathcal{K} in $L^2(0,T;\mathbb{D}(A))$ and assumption (c) imply that \mathcal{K} is compact in $\tilde{\mathcal{Z}}_T$. Since the embeddings $\mathbb{D}(A) \hookrightarrow \mathbb{V} \hookrightarrow \mathbb{H}$ are continuous and the embedding $\mathbb{D}(A) \hookrightarrow \mathbb{V}$ is compact, by Dubinsky Theorem 2.4.5 assumptions (b) and (c) imply that \mathcal{K} is relatively compact in $L^2(0,T;\mathbb{V}) \cap \mathscr{C}([0,T];\mathbb{H})$. Hence in particular, there exists a subsequence, still denoted by (u_n) , convergent in \mathbb{H} . Therefore by Lemma 6.3.2 and assumption $(a), (u_n)$ is convergent in $\mathscr{C}([0,T];\mathbb{B}_w)$. This completes the proof of the lemma.

6.3.1 Tightness

Using Section 2.9 and the compactness criterion from Lemma 6.3.3 we obtain the following corollary which we will use to prove tightness of the laws defined by the truncated SPDE (6.4.26).

Corollary 6.3.4 (Tightness criterion). Let $(X_n)_{n \in \mathbb{N}}$ be a sequence of continuous \mathbb{F} -adapted H-valued processes such that

(a) there exists a constant $C_1 > 0$ such that

$$\sup_{n \in \mathbb{N}} \mathbb{E} \left[\sup_{s \in [0,T]} \|X_n(s)\|_{\mathcal{V}}^2 \right] \leq C_1,$$

(b) there exists a constant $C_2 > 0$ such that

$$\sup_{n\in\mathbb{N}}\mathbb{E}\left[\int_{0}^{T}|X_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2}ds\right]\leq C_{2},$$

(c) $(X_n)_{n \in \mathbb{N}}$ satisfies the Aldous condition [A] in H.

Let $\tilde{\mathbb{P}}_n$ be the law of X_n on \mathcal{Z}_T . Then for every $\varepsilon > 0, \exists$ a compact subset K_{ε} of \mathcal{Z}_T such that

$$\sup_{n\in\mathbb{N}}\tilde{\mathbb{P}}_n(K_{\varepsilon})\geq 1-\varepsilon.$$

Proof. Let $\varepsilon > 0$. By the Chebyshev inequality and (a), we infer that for any $n \in \mathbb{N}$ and any r > 0

$$\tilde{\mathbb{P}}_n\left(\sup_{s\in[0,T]} \|X_n(s)\|_{\mathbf{V}}^2 > r\right) \leq \frac{\tilde{\mathbb{E}}_n\left[\sup_{s\in[0,T]} \|X_n(s)\|_{\mathbf{V}}^2\right]}{r} \leq \frac{C_1}{r}.$$

Let R_1 be such that $\frac{C_1}{R_1} \leq \frac{\varepsilon}{3}$. Then

$$\sup_{n\in\mathbb{N}}\tilde{\mathbb{P}}_n\left(\sup_{s\in[0,T]}\|X_n(s)\|_{\mathbf{V}}^2>R_1\right)\leq\frac{\varepsilon}{3}.$$

Let $B_1 := \left\{ u \in \mathcal{Z}_T : \sup_{s \in [0,T]} \| u(s) \|_{\mathcal{V}}^2 \leq R_1 \right\}.$

By the Chebyshev inequality and (b), we infer that for any $n \in \mathbb{N}$ and any r > 0

$$\tilde{\mathbb{P}}_n(|X_n|_{L^2(0,T;\mathbf{D}(\mathbf{A}))} > r) \le \frac{\tilde{\mathbb{E}}_n[|X_n|_{L^2(0,T;\mathbf{D}(\mathbf{A}))}^2]}{r^2} \le \frac{C_2}{r^2}.$$

Let R_2 be such that $\frac{C_2}{R_2^2} \leq \frac{\varepsilon}{3}$. Then

$$\sup_{n \in \mathbb{N}} \tilde{\mathbb{P}}_n \left(|X_n|_{L^2(0,T; \mathcal{D}(\mathcal{A}))} > R_2 \right) \le \frac{\varepsilon}{3}$$

Let $B_2 := \{ u \in \mathcal{Z} : |u|_{L^2(0,T; \mathbf{D}(\mathbf{A}))} \le R_2 \}.$

By Lemmas 2.9.9, 2.9.11 there exists a subset $A_{\frac{\varepsilon}{3}} \subset \mathscr{C}([0,T], \mathbb{H})$ such that $\tilde{\mathbb{P}}_n(A_{\frac{\varepsilon}{3}}) \geq 1 - \frac{\varepsilon}{3}$ and

$$\lim_{\delta \to 0} \sup_{u \in A_{\frac{\epsilon}{3}}} \sup_{\substack{s,t \in [0,T]\\|t-s| \le \delta}} |u(t) - u(s)|_{\mathrm{H}} = 0$$

It is sufficient to define K_{ε} as the closure of the set $B_1 \cap B_2 \cap A_{\frac{\varepsilon}{3}}$ in \mathcal{Z}_T . By Lemma 6.3.3, K_{ε} is compact in \mathcal{Z}_T . The proof is thus complete.

6.3.2 The Skorohod Theorem

Let us recall the Jakubowski's generalisation of the Skorohod Theorem as given by Brzeźniak and Ondreját [26, Theorem C.1], see also [49].

Theorem 6.3.5. Let \mathscr{X} be a topological space such that there exists a sequence $\{f_m\}_{m \in \mathbb{N}}$ of continuous functions $f_m : \mathscr{X} \to \mathbb{R}$ that separates points of \mathscr{X} . Let us denote by \mathscr{S} the σ -algebra generated by the maps $\{f_m\}$. Then

- (a) every compact subset of \mathscr{X} is metrizable,
- (b) if $(\mu_m)_{m \in \mathbb{N}}$ is a tight sequence of probability measures on $(\mathcal{X}, \mathscr{S})$, then there exists a subsequence $(m_k)_{k \in \mathbb{N}}$, a probability space $(\Omega, \mathscr{F}, \mathbb{P})$ with \mathscr{X} -valued Borel measurable variables ξ_k, ξ such that μ_{m_k} is the law of ξ_k and ξ_k converges to ξ almost surely on Ω . Moreover, the law of ξ is a Radon measure.

Using Theorem 6.3.5, we obtain the following corollary which we will apply to construct a martingale solution to the tamed Navier-Stokes equations.

Corollary 6.3.6. Let $(\eta_n)_{n \in \mathbb{N}}$ be a sequence of \mathcal{Z}_T -valued random variables such that their laws $\mathcal{L}(\eta_n)$ on $(\mathcal{Z}_T, \mathcal{T})$ form a tight sequence of probability measures. Then there exists a subsequence (n_k) , a probability space $(\tilde{\Omega}, \tilde{\mathcal{F}}, \tilde{\mathbb{P}})$ and \mathcal{Z}_T -valued random variables $\tilde{\eta}, \tilde{\eta}_k, k \in \mathbb{N}$ such that the variables η_k and $\tilde{\eta}_k$ have the same laws on \mathcal{Z}_T and $\tilde{\eta}_k$ converges to $\tilde{\eta}$ almost surely on $\tilde{\Omega}$.

Proof. It is sufficient to prove that on each space appearing in the definition (6.3.5) of the space \mathcal{Z}_T , there exists a countable set of continuous real-valued functions separating points.

Since the spaces $\mathscr{C}([0,T]; U')$ and $L^2(0,T; H_{\text{loc}})$ are separable, metrizable and complete, this condition is satisfied, see [3], exposé 8.

For the space $L^2_w(0,T; D(A))$ it is sufficient to put

$$f_m(u) := \int_0^T \langle u(t), \mathbf{v}_m(t) \rangle_{\mathbf{D}(\mathbf{A})} dt \in \mathbb{R}, \qquad u \in L^2_w(0, T; \mathbf{D}(\mathbf{A})), \quad m \in \mathbb{N},$$

where $\{v_m, m \in \mathbb{N}\}$ is a dense subset of $L^2(0, T; D(A))$.

Let us consider the space $\mathscr{C}([0,T]; V_w)$. Let $\{h_m, m \in \mathbb{N}\}$ be any dense subset of H and let \mathbb{Q}_T be the set of rational numbers belonging to the interval [0,T]. Then the family $\{f_{m,t}, m \in \mathbb{N}, t \in \mathbb{Q}_T\}$ defined by

$$f_{m,t}(u) := \langle u(t), h_m \rangle_{\mathcal{V}} \in \mathbb{R}, \qquad u \in \mathscr{C}([0,T]; \mathcal{V}_w), \quad m \in \mathbb{N}, \quad t \in \mathbb{Q}_T$$

consists of continuous functions separating points in $\mathscr{C}([0,T]; V_w)$. The statement of the corollary follows from Theorem 6.3.5, concluding the proof.

6.3.3 Martingale and strong solution

We end this section by giving the definitions of a martingale and strong solution to (6.2.16).

Definition 6.3.7. We say that there exists a martingale solution of (6.2.16) iff there exist

- a stochastic basis $(\hat{\Omega}, \hat{\mathscr{F}}, \hat{\mathbb{F}}, \hat{\mathbb{P}})$,
- a cylindrical Wiener process $\hat{W}(t) = (\hat{W}_j(t))_{j=1}^{\infty}$ on ℓ^2 , where $\{\hat{W}_j(t), t \ge 0, j \in \mathbb{N}\}$ is an infinite sequence of independent standard $(\hat{\mathscr{F}}_t)$ -Brownian motions,
- and a progressively measurable process $u: [0,T] \times \hat{\Omega} \to D(A)$ with $\hat{\mathbb{P}}$ -a.e. paths

$$u(\cdot,\omega) \in \mathscr{C}([0,T]; \mathbf{V}_w) \cap L^2(0,T; \mathbf{D}(\mathbf{A})),$$

such that for all $t \in [0, T]$ and all $v \in \mathcal{V} \hat{\mathbb{P}}$ -a.s.

(6.3.6)
$$\langle u(t), \mathbf{v} \rangle + \int_0^t \langle Au(s), \mathbf{v} \rangle \, ds + \int_0^t \langle B(u(s)), \mathbf{v} \rangle \, ds + \int_0^t \langle g(|u(s)|^2) u(s), \mathbf{v} \rangle \, ds \\ = \langle u_0, \mathbf{v} \rangle + \int_0^t \langle f(u(s)), \mathbf{v} \rangle \, ds + \left\langle \int_0^t G(s, u(s)) \, dW(s), \mathbf{v} \right\rangle.$$

Definition 6.3.8. We say that problem (6.2.16) has a strong solution iff for every stochastic basis $(\Omega, \mathscr{F}, \mathbb{F}, \mathbb{P})$ and every cylindrical Wiener process $W(t) = (W_j(t))_{j=1}^{\infty}$ on ℓ^2 there exists a progressively measurable process $u : [0, T] \times \Omega \to D(A)$ with \mathbb{P} -a.e. paths

$$u(\cdot,\omega) \in \mathscr{C}([0,T];\mathbf{V}) \cap L^2(0,T;\mathbf{D}(\mathbf{A})),$$

such that for all $t \in [0, T]$ and all $v \in \mathcal{V}$ (6.3.6) holds \mathbb{P} -a.s.

6.4 Truncated SPDE

We will be using the following notations and spaces repeatedly in this section.

$$B_n := \{x \in \mathbb{R}^3 : |x| \le n\} \subset \mathbb{R}^3, \ n \in \mathbb{N}.$$

We will use $\mathscr{F}(u)$ and \hat{u} interchangeably to denote the Fourier transform of u. The inverse Fourier transform will be given by \mathscr{F}^{-1} .

We define H_n as the subspace of H

$$\mathbf{H}_n := \{ u \in \mathbf{H} : \operatorname{supp}(\hat{u}) \subset B_n \}.$$

The norm on H_n is inherited from H and will be denoted by $\|\cdot\|_{H_n}$.

Let

$$P_n: \mathbf{H} \to \mathbf{H}_n,$$

be the orthogonal projection i.e. $\forall u \in H, u - P_n u \perp H_n$ and

$$y = P_n u \Leftrightarrow y \in H_n \text{ and } u - y \perp H_n$$

One can show that P_n is given by

$$P_n u = \mathscr{F}^{-1}(\mathbb{I}_{B_n} \hat{u}).$$

Let us recall that $D(A) := H \cap H^{2,2}$ and the Stokes operator is given by

$$Au = -\Pi(\Delta u), \qquad u \in D(A),$$

and D(A) is a Hilbert space under the graph norm

$$|u|_{\mathrm{D}(\mathrm{A})}^2 := |u|_{\mathrm{H}}^2 + |\mathrm{A}u|_{\mathrm{H}}^2.$$

Lemma 6.4.1. Let P_n be the orthogonal projection given by (6.4.1), then $P_n : V \to V$ is a contraction.

Proof. Let $u \in V$, then by the definition of P_n and V

$$\begin{split} \|P_n u\|_{\mathcal{V}} &= \left[\int_{\mathbb{R}^3} (1+|\xi|^2) |\mathscr{F}(P_n u)(\xi)|^2 d\xi \right]^{1/2} = \left[\int_{\mathbb{R}^3} (1+|\xi|^2) |\mathbb{1}_{B_n}(\xi) \hat{u}(\xi)|^2 d\xi \right]^{1/2} \\ &= \left[\int_{|\xi| \le n} (1+|\xi|^2) |\hat{u}(\xi)|^2 d\xi \right]^{1/2} \le \left[\int_{\mathbb{R}^3} (1+|\xi|^2) |\hat{u}(\xi)|^2 d\xi \right]^{1/2} = \|u\|_{\mathcal{V}}. \end{split}$$

Thus we have shown that

$$\|P_n u\|_{\mathcal{V}} \leq \|u\|_{\mathcal{V}}.$$

Lemma 6.4.2. If $u \in D(A)$ then $\Delta u \in H$. In particular, if $u \in D(A)$ then $Au = -\Delta u$.

Proof. Since $u \in D(A)$, it is clear that $\Delta u \in L^2$. Thus we are left to show that $div(\Delta u) = 0$ in the weak sense. Let $\varphi \in C_0^{\infty}(\mathbb{R}^3)$, then using the definition of div and Δ , we get

$$\langle \operatorname{div}(\Delta u) | \varphi \rangle = -\langle \Delta u | \nabla \varphi \rangle$$
$$= -\langle u | \Delta (\nabla \varphi) \rangle$$
$$= \langle \operatorname{div} u | \Delta \varphi \rangle = 0$$

By definition $Au = -\Pi(\Delta u)$ but since $\Delta u \in H$, and $\Pi : L^2 \to H$ is an orthogonal projection, $\Pi(\Delta u) = \Delta u$ and hence,

$$(6.4.2) Au = -\Delta u, u \in D(A)$$

Lemma 6.4.3. $H_n \subset D(A)$ and

$$(6.4.3) P_n(\mathbf{A}u) = \mathbf{A}u, u \in \mathbf{H}_n$$

Proof. We start with proving the first statement. Let $u \in H_n$. By definition

$$\mathbf{D}(\mathbf{A}) = \{ u \in \mathbf{H} : u \in H^{2,2} \} = \left\{ u \in \mathbf{H} : \int_{\mathbb{R}^3} \left(1 + |\xi|^2 \right)^2 |\hat{u}(\xi)|^2 d\xi < \infty \right\}.$$

Since $u \in \mathbf{H}_n$, supp $(\hat{u}) \subset B_n$,

$$\begin{split} \int_{\mathbb{R}^3} \left(1+|\xi|^2\right)^2 |\hat{u}(\xi)|^2 d\xi &= \int_{|\xi| \le n} \left(1+|\xi|^2\right)^2 |\hat{u}(\xi)|^2 d\xi \le (1+n^2)^2 \int_{|\xi| \le n} |\hat{u}(\xi)|^2 d\xi \\ &= (1+n^2)^2 \int_{\mathbb{R}^3} |\hat{u}(\xi)|^2 d\xi = (1+n^2)^2 \|u\|_{\mathcal{H}_n}^2 < \infty. \end{split}$$

Thus we have proved that $u \in D(A)$ and hence $H_n \subset D(A)$. Moreover we showed that there exists a constant $C_n > 0$, depending on n such that

(6.4.4)
$$|u|_{\mathrm{D}(\mathrm{A})} \le C_n ||u||_{\mathrm{H}_n}, \quad u \in \mathrm{H}_n.$$

Now in order to establish the equality (6.4.3), we just need to show that $Au \in H_n$. Since $u \in H_n$, $u \in D(A)$. Hence, Lemma 6.4.1 implies $Au = -\Delta u$. We are left to show that $supp(\mathscr{F}(Au)) \subset B_n$. Using the definition of Au, we get following equalities

$$\mathscr{F}(\mathrm{A}u)(\xi) = -\mathscr{F}(\Delta u)(\xi) = -|\xi|^2 \hat{u}(\xi).$$

Thus

$$\operatorname{supp}(\mathscr{F}(\operatorname{Au})) \subset \operatorname{supp}(|\cdot|^2) \cap \operatorname{supp}(\hat{u}) \subset B_n$$

Hence $Au \in H_n$. Since $P_n : H \to H_n$ is an orthogonal projection, we infer that

$$P_n(\mathrm{A}u) = \mathrm{A}u.$$

Lemma 6.4.4. $A_n := A|_{H_n} : H_n \to H_n$, is linear and bounded.

Proof. In Lemma 6.4.2 we showed that A_n is well defined and it's straightforward to show it is linear. We are left to show that it is bounded. Let $u \in H_n$, then by the Plancherel Theorem (see Theorem 2.6.7) and the definition of H_n

$$\|\mathbf{A}_{n}u\|_{\mathbf{H}_{n}} = |-\Delta u|_{L^{2}} = \left[\int_{\mathbb{R}^{3}} |\xi|^{2} |\hat{u}(\xi)|^{2} d\xi\right]^{1/2} = \left[\int_{|\xi| \le n} |\xi|^{2} |\hat{u}(\xi)|^{2} d\xi\right]^{1/2}$$
$$\leq \left[n^{2} \int_{|\xi| \le n} |\hat{u}(\xi)|^{2} d\xi\right]^{1/2} = \left[n^{2} \int_{\mathbb{R}^{3}} |\hat{u}(\xi)|^{2} d\xi\right]^{1/2} = n \|u\|_{\mathbf{H}_{n}}.$$

Thus

$$(6.4.5) ||A_n u||_{\mathbf{H}_n} \le n ||u||_{\mathbf{H}_n}.$$

Lemma 6.4.5. The map B_n defined by

$$(6.4.6) B_n: \mathbf{H}_n \times \mathbf{H}_n \ni (u, \mathbf{v}) \mapsto P_n(B(u, \mathbf{v})) \in \mathbf{H}_n$$

is well defined and Lipschitz on balls. Moreover

(6.4.7)
$$\langle B_n(u), u \rangle_{\mathrm{H}} = 0, \quad u \in \mathrm{H}_n,$$

(6.4.8)
$$|((B_n(u), u))| \le \frac{1}{2} |u|_{\mathcal{D}(\mathcal{A})}^2 + \frac{1}{2} ||u| \cdot |\nabla u||_{L^2}^2, \quad u \in \mathcal{H}_n,$$

where $B_n(u) := B_n(u, u)$ and $((\cdot, \cdot))$ is defined in (6.2.1).

Proof. We will show that $\forall u, v \in H_n$, $B(u, v) \in H$. Since $u, v \in H_n$, $u, v \in D(A)$. Thus, by the Hölder inequality

$$|B(u,\mathbf{v})|_{\mathbf{H}} = |\Pi(u \cdot \nabla \mathbf{v})|_{\mathbf{H}} \le |u \cdot \nabla \mathbf{v}|_{L^2} \le ||u||_{L^{\infty}} |\nabla \mathbf{v}|_{L^2}.$$

From (6.2.7), $H^{s,2} \hookrightarrow L^{\infty}$ for every $s > \frac{d}{2}$. Therefore, there exists a constant C > 0 such that

$$||u||_{L^{\infty}} \leq C ||u||_{H^{s,2}}.$$

In particular it holds true for s = 2. Thus, we have

$$|B(u,v)|_{\mathrm{H}} \leq C ||u||_{H^2} ||v||_{H^1}.$$

Now by (6.4.4) and (6.4.15)

(6.4.9)
$$|B(u,v)|_{\mathrm{H}} \leq K_n ||u||_{\mathrm{H}_n} ||v||_{\mathrm{H}_n} < \infty.$$

Hence $B(u, v) \in H$, which implies $B_n(u, v) \in H_n$ and is well defined.

Let $u, v \in \mathbb{B}_R$, where

(6.4.10)
$$\mathbb{B}_{R} := \left\{ u \in \mathcal{H}_{n} : \|u\|_{\mathcal{H}_{n}} \le R \right\}.$$

Then, as before

$$\begin{split} \|B_{n}(u) - B_{n}(v)\|_{\mathcal{H}_{n}} &\leq |B(u) - B(v)|_{\mathcal{H}} \leq |u \cdot \nabla u - v \cdot \nabla v|_{L^{2}} \\ &\leq |(u - v) \cdot \nabla u|_{L^{2}} + |v \cdot \nabla (u - v)|_{L^{2}} \\ &\leq \|u - v\|_{L^{\infty}} |\nabla u|_{L^{2}} + \|v\|_{L^{\infty}} |\nabla (u - v)|_{L^{2}} \\ &\leq \|u - v\|_{H^{2}} \|u\|_{H^{1}} + \|v\|_{H^{2}} \|u - v\|_{H^{1}}. \end{split}$$

Since $u, v \in \mathbb{B}_R$, using (6.4.4) and (6.4.15), we get

(6.4.11)
$$\|B_n(u) - B_n(v)\|_{H_n} \le C_{n,R} \|u - v\|_{H_n}, \quad u, v \in \mathbb{B}_R.$$

Since $u \in H_n$ and P_n is the orthogonal projection on H,

$$\langle B_n(u), u \rangle_{\mathrm{H}} = \langle P_n(B(u, u)), u \rangle_{\mathrm{H}} = \langle B(u, u), P_n u \rangle_{\mathrm{H}} = \langle B(u, u), u \rangle_{\mathrm{H}} = 0$$

Also by using the definition of $((\cdot, \cdot))$ and the Cauchy-Schwartz inequality we get

$$\begin{aligned} |\langle (B_n(u), u)\rangle| &= |\langle B_n(u), -\Delta u\rangle_{\mathcal{H}}| = |\langle B(u, u), -P_n(\Delta u)\rangle_{\mathcal{H}}| = |\langle B(u, u), -\Delta u\rangle_{\mathcal{H}}| \\ &\leq |B(u, u)|_{\mathcal{H}} |(-\Delta u)|_{\mathcal{H}} \leq \frac{1}{2} |u|_{\mathcal{D}(\mathcal{A})}^2 + \frac{1}{2} ||u| \cdot |\nabla u||_{L^2}^2. \end{aligned}$$

Lemma 6.4.6. The map g_n defined by

(6.4.12)
$$g_n: \mathbf{H}_n \ni u \mapsto P_n\left[\Pi(g(|u|^2)u)\right] \in \mathbf{H}_n,$$

is well defined and Lipschitz on balls. Moreover

(6.4.13)
$$\begin{cases} ((-g_n(u), u)) \le C_N |\nabla u|_{L^2}^2 - 2 ||u| \cdot |\nabla u||_{L^2}^2, & u \in \mathbf{H}_n, \\ \langle -g_n(u), u \rangle_{\mathbf{H}} \le - ||u||_{L^4}^4 + C_N |u|_{\mathbf{H}}^2, & u \in \mathbf{H}_n. \end{cases}$$

Proof. Let $u \in H_n$, then by the definition of g (6.2.13), the estimate (6.2.14) and the embedding of $H^1 \hookrightarrow L^6$, we have

$$\begin{split} \|g_{n}(u)\|_{\mathcal{H}_{n}} &= \left\|P_{n}\left[\Pi(g(|u|^{2})u)\right]\right\|_{\mathcal{H}_{n}} \leq |\Pi(g(|u|^{2})u)|_{\mathcal{H}} \leq |g(|u|^{2})u|_{L^{2}} \\ &= \left[\int_{\mathbb{R}^{3}} \left|g(|u(x)|^{2})\right|^{2} |u(x)|^{2} dx\right]^{1/2} \leq \left[\int_{\mathbb{R}^{3}} |u(x)|^{6} dx\right]^{1/2} = \|u\|_{L^{6}}^{3} \\ &\leq C\|u\|_{H^{1}}^{3} = C\left[\int_{\mathbb{R}^{3}} (1+|\xi|^{2})|\hat{u}(\xi)|^{2} d\xi\right]^{3/2} = C\left[\int_{|\xi| \leq n} (1+|\xi|^{2})|\hat{u}(\xi)|^{2} d\xi\right]^{3/2} \\ &\leq C(1+n^{2})^{3/2} \left[\int_{|\xi| \leq n} |\hat{u}(\xi)|^{2} d\xi\right]^{3/2} = C(1+n^{2})^{3/2} \left[\int_{\mathbb{R}^{3}} |\hat{u}(\xi)|^{2} d\xi\right]^{3/2} \\ &\leq C(1+n^{2})^{3/2} \left[\int_{|\xi| \leq n} |\hat{u}\|_{L^{2}}^{3} = C_{n} \|u\|_{\mathcal{H}_{n}}^{3} < \infty. \end{split}$$

$$(6.4.14)$$

Therefore $g_n : H_n \to H_n$ is well defined. From above we can also infer that there exists a constant $C_n > 0$ depending on n such that

$$(6.4.15) ||u||_{H^1} \le C_n ||u||_{H_n}, u \in H_n.$$

Let $u, v \in \mathbb{B}_R$, then as before using (6.2.13), we have

$$\begin{split} \|g_{n}(u) - g_{n}(v)\|_{H_{n}} &\leq |\Pi(g(|u|^{2})u) - \Pi(g(|v|^{2})v)|_{H} \leq |g(|u|^{2})u - g(|v|^{2})v|_{L^{2}} \\ &\leq |g(|u|^{2})(u - v)|_{L^{2}} + |(g(|u|^{2}) - g(|v|^{2}))v|_{L^{2}} \\ &\leq \left[\int_{\mathbb{R}^{3}} |u(x)|^{4} |u(x) - v(x)|^{2} dx\right]^{1/2} + 4 \left[\int_{\mathbb{R}^{3}} |u(x) - v(x)|^{2} \left[|u(x)|^{2} + |v(x)|^{2}\right] |v(x)|^{2} dx\right]^{1/2} . \end{split}$$

Since $H^1 \hookrightarrow L^6$, we obtain

$$\begin{split} \|g_{n}(u) - g_{n}(v)\|_{H_{n}} &\leq \left[\int_{\mathbb{R}^{3}} |u(x)|^{6} dx\right]^{1/3} \left[\int_{\mathbb{R}^{3}} |u(x) - v(x)|^{6} dx\right]^{1/6} \\ &+ 4 \left[\int_{\mathbb{R}^{3}} |u(x) - v(x)|^{6} dx\right]^{1/6} \left[\left[\int_{\mathbb{R}^{3}} |u(x)|^{6} dx\right]^{1/3} + \left[\int_{\mathbb{R}^{3}} |v(x)|^{6} dx\right]^{1/3}\right]^{1/2} \left[\int_{\mathbb{R}^{3}} |v(x)|^{6} dx\right]^{1/6} \\ &= \left[\|u\|_{L^{6}}^{2} \|u - v\|_{L^{6}} + 4\|u - v\|_{L^{6}} \left(\|u\|_{L^{6}}^{2} + \|v\|_{L^{6}}^{2}\right)^{1/2} \|v\|_{L^{6}}\right] \\ &\leq C\|u - v\|_{H^{1}} \left[\|u\|_{H^{1}}^{2} + 4 \left(\|u\|_{H^{1}}^{2} + \|v\|_{H^{1}}^{2}\right)^{1/2} \|v\|_{H^{1}}\right]. \end{split}$$

Since $u, v \in \mathbb{B}_R$, using (6.4.15), we get

$$\|g_{n}(u) - g_{n}(v)\|_{H_{n}} \leq \hat{C}_{n} \|u - v\|_{H_{n}} \left[\|u\|_{H_{n}}^{2} + 4\left(\|u\|_{H_{n}}^{2} + \|v\|_{H_{n}}^{2} \right)^{1/2} \|v\|_{H_{n}} \right]$$

$$\leq C_{n,R} \|u - v\|_{H_{n}}.$$

Let $u \in H_n$, then using Lemmas 6.4.2 and 6.4.3, the definitions of g_n and $((\cdot, \cdot))$ we get

$$((-g_n(u), u)) = -\langle g_n(u), -\Delta u \rangle_{\mathcal{H}} = -\langle \Pi(g(|u|^2)u, -P_n(\Delta u)) \rangle_{\mathcal{H}}$$
$$= -\langle g(|u|^2)u, -\Pi(\Delta u) \rangle_{\mathcal{H}} = -\langle g(|u|^2)u, -\Delta u \rangle_{L^2}.$$

Also note that for $u \in \mathbf{H}_n$

$$\langle -g_n(u), u \rangle_{\mathrm{H}} = -\langle \Pi(g(|u|^2))u, P_n u \rangle_{\mathrm{H}} = -\langle g(|u|^2)u, \Pi(u) \rangle_{L^2} = -\langle g(|u|^2)u, u \rangle_{L^2}.$$

Hence the inequalities (6.4.13) can be established with the help of the above two relations and Lemma 6.2.1 (ii). This completes proof of the lemma.

Lemma 6.4.7. Let us assume that the function f satisfies the assumption (A1). Then the map

(6.4.17)
$$f_n: \mathbf{H}_n \ni u \mapsto P_n[\Pi(f(u))] \in \mathbf{H}_n$$

is well defined and Lipschitz.

Proof. Let $u \in H_n$, then by the assumption (A1),

$$||f_n(u)||_{\mathcal{H}_n} \le |\Pi(f(u))|_{\mathcal{H}} \le |f(u)|_{L^2} \le C_f ||u||_{L^2} = C_f ||u||_{\mathcal{H}_n} < \infty.$$

Therefore $f_n : H_n \to H_n$ is well defined. Let $u, v \in H_n$, then

(6.4.18)
$$\|f_n(u) - f_n(v)\|_{H_n} \le |\Pi f(u) - \Pi f(v)|_H \le |f(u) - f(v)|_{L^2}$$
$$\le C_f |u - v|_{L^2} = C_f ||u - v||_{H_n}.$$

Lemma 6.4.8. Let σ satisfy the assumption (A2). Then the map

(6.4.19)
$$G_n: \mathbf{H}_n \ni u \mapsto P_n \circ (G(u)) \in \mathcal{T}_2(\ell^2; \mathbf{H}_n)$$

is well defined and Lipschitz.

Proof. Let $u \in H_n$, then

$$\begin{split} \|G_n(u)\|_{\mathscr{T}_2(\ell^2;\mathbf{H}_n)} &\leq \|(G(u))\|_{\mathscr{T}_2(\ell^2;\mathbf{H})} \leq \left[\int_{\mathbb{R}^3} \|\sigma(x)\|_{\ell^2}^2 |\nabla u(x)|^2 \, dx\right]^{1/2} \\ &\leq \left[\sup_{x \in \mathbb{R}^3} \|\sigma(x)\|_{\ell^2}^2\right]^{1/2} |\nabla u|_{L^2} \leq \frac{1}{2} \|u\|_{H^1}. \end{split}$$

Using (6.4.15), we infer

(6.4.20) $\|G_n(u)\|_{\mathcal{T}_2(\ell^2;\mathbf{H}_n)} \le C_n \|u\|_{\mathbf{H}_n} < \infty.$

Thus $G_n: \mathbf{H}_n \to \mathcal{T}_2(\ell^2; \mathbf{H}_n)$ is well defined. Let $u, v \in \mathbf{H}_n$, then

$$\begin{split} \|G_{n}(u) - G_{n}(\mathbf{v})\|_{\mathcal{F}_{2}^{2}(\ell^{2};\mathbf{H}_{n})} &\leq \|G(u) - G(\mathbf{v})\|_{\mathcal{F}_{2}^{2}(\ell^{2};\mathbf{H})} \\ &\leq \left[\int_{\mathbb{R}^{3}} \sum_{j=1}^{\infty} |\sigma_{j}(x)|^{2} |\nabla(u - \mathbf{v})(x)|^{2} dx\right]^{1/2} \\ &= \left[\int_{\mathbb{R}^{3}} \|\sigma(x)\|_{\ell^{2}}^{2} |\nabla(u - \mathbf{v})(x)|^{2} dx\right]^{1/2} \\ &\leq \left(\sup_{x \in \mathbb{R}^{3}} \|\sigma(x)\|_{\ell^{2}}^{2}\right)^{1/2} |\nabla(u - \mathbf{v})|_{L^{2}} \leq \frac{1}{2} \|u - \mathbf{v}\|_{H^{1}} \end{split}$$

Using (6.4.15), we infer

(6.4.21)
$$\|G_n(u) - G_n(v)\|_{\mathcal{F}_2(\ell^2; \mathbf{H}_n)} \le C_n \|u - v\|_{\mathbf{H}_n}.$$

Proposition 6.4.9. L^2 , H^1 and D(A)-norms on H_n are equivalent.
Proof. Let $u \in H_n$, then using the Plancherel Theorem

(6.4.22)
$$|u|_{L^2} = \left[\int_{\mathbb{R}^3} |\hat{u}(\xi)|^2 d\xi\right]^{1/2} = \left[\int_{|\xi| \le n} |\hat{u}(\xi)|^2 d\xi\right]^{1/2} = ||u||_{\mathrm{H}_n}.$$

Thus if $u \in H_n$ then L^2 and H_n have equal norms. The equivalence of H^1 and H_n norms is established from (6.4.15). Using (6.4.5) and (6.4.22) we can establish equivalence of D(A) and H_n norms.

As discussed earlier in the introduction instead of using standard Galerkin approximation of SPDE on the finite dimensional space we will look at the truncated SPDE on an infinite dimensional space H_n . We will establish the existence of a unique global solution to the truncated SPDE and obtain a priori estimates in order to prove the tightness of measures on a suitable space.

In order to study the truncated SPDE on H_n we project the SPDE (6.2.16) on H_n using P_n . The projected SPDE on H_n is given by

(6.4.23)
$$\begin{cases} du_n(t) = -[A_n u_n(t) + B_n(u_n(t)) + g_n(u_n(t)) - f_n(u_n(t))] dt + G_n(u_n(t)) dW(t), \\ u_n(0) = P_n(u_0), \end{cases}$$

where $u_n \in H_n$, $u_0 \in V$ and other operators B_n , g_n , f_n and G_n are as defined in Lemmas 6.4.4 - 6.4.8.

Lemma 6.4.10. Let us define $F : H_n \to \mathbb{R}$ by

(6.4.24)
$$F(u) := \|G_n(u)\|_{\mathcal{F}_2(\ell^2; \mathbf{H}_n)} + 2\langle u, -\mathbf{A}_n u - B_n(u) - g_n(u) + f_n(u) \rangle_{\mathbf{H}}, \quad u \in \mathbf{H}_n.$$

Then for every $u \in H_n$ there exists $K_1 > 0$ such that

(6.4.25)
$$F(u) \le K_1 (1 + \|u\|_{\mathrm{H}_n}^2).$$

Proof. From the definition of A_n , B_n , g_n and f_n , we have

$$\begin{split} \|G_n(u)\|_{\mathcal{T}_2(\ell^2;\mathbf{H}_n)} + 2\langle u, -\mathbf{A}_n u - B_n(u) - g_n(u) + f_n(u) \rangle_{\mathbf{H}} \\ &= \|G_n(u)\|_{\mathcal{T}_2(\ell^2;\mathbf{H}_n)} + 2\langle u, -\Pi(\Delta u) - P_n(B(u)) - P_n[\Pi(g(|u|^2)u)] + P_n[\Pi(f(u))] \rangle_{\mathbf{H}}. \end{split}$$

Since $u \in H_n$, using Lemma 6.4.8, we get

$$F(u) \leq \frac{1}{4} \|u\|_{H_{n}}^{2} - 2|\nabla u|_{L^{2}}^{2} - 2\langle u, B(u) \rangle_{H} - 2\langle u, g(|u|^{2})u \rangle_{H} + 2\langle u, f(u) \rangle_{H}$$

$$\leq \frac{1}{4} \|u\|_{H_{n}}^{2} - 2|\nabla u|_{L^{2}}^{2} - 2|\sqrt{g(|u|^{2})}|u||_{L^{2}}^{2} + C_{f} \|u\|_{H_{n}}^{2}$$

$$F(u) + 2|\nabla u|_{L^2}^2 + 2|\sqrt{g(|u|^2)|u|}|_{L^2}^2 \le \frac{1}{4}||u||_{H_n}^2 + C_f ||u||_{H_n}^2 \le K_1(1 + ||u||_{H_n}^2),$$

for appropriately chosen K_1 . Thus, in particular

$$F(u) \le K_1(1 + \|u\|_{\mathbf{H}_n}^2).$$

We will need the following theorem to prove Theorem 6.4.12. We have modified it in the way (compared to the statement in Theorem 2.8.8, see also [1, Theorem 3.1]) we will use it.

Theorem 6.4.11. Let X be a separable, possibly infinite dimensional, Hilbert space. Assume that σ and b satisfy the following conditions

(i) For any R > 0 there exists a constant C > 0 such that

$$\|\sigma(u) - \sigma(v)\|_{\mathcal{F}_{2}(\ell^{2};X)} + \|b(u) - b(v)\|_{X} \le C \|u - v\|_{X}^{2}, \quad \|u\|_{X}, \|v\|_{X} \le R$$

(ii) There exists a constant $K_1 > 0$ such that

$$\|\sigma(u)\|_{\mathcal{T}_{2}(\ell^{2};X)}^{2}+2\langle u,b(u)\rangle_{L^{2}}\leq K_{1}(1+\|u\|_{X}^{2}), \qquad u\in X.$$

Then for any X-valued ξ , there exists a unique global solution $u = (u(t))_{t \ge 0}$ to

$$u(t) = \xi + \int_0^t \sigma(u(s)) dW(s) + \int_0^t b(u(s)) ds.$$

Theorem 6.4.12. Let the assumptions (A1) and (A2) hold. Then for every $u_0 \in V$ there exists a unique global solution $u_n = (u_n(t))_{t \ge 0}$ to (6.4.26)

$$\begin{cases} u_n(t) + \int_0^t [A_n u_n(s) + B_n(u_n(s)) + g_n(u_n(s))] \, ds = \int_0^t f_n(u_n(s)) \, ds + \int_0^t G_n(u_n(s)) \, dW(s), \\ u_n(0) = P_n u_0. \end{cases}$$

Proof. The proof is direct application of Theorem 6.4.11. Using Lemmas 6.4.4 - 6.4.8, we can show that the condition (*i*) of Theorem 6.4.11 is satisfied. In Lemma 6.4.10 we proved that the condition (*ii*) is satisfied. Thus we have the existence of a unique global solution u_n to (6.4.26).

By Lemma 6.4.4 the map A_n is linear and bounded on H_n and thus, $A_n = A$ on H_n .

6.5 Existence of solution

6.5.1 A'priori estimates

In this subsection we will obtain certain a'priori estimates for the solution u_n of (6.4.26). We will use these a'priori estimates in Lemma 6.5.3 to prove the tightness of measures on the space \mathcal{Z}_T , defined in (6.3.5). We will also establish certain higher order estimates which will be required to prove the convergence of non-linear terms in later sections.

Let us fix T > 0. For any R > 0, define the stopping time

(6.5.1)
$$\tau_{R}^{n} := \inf\{t \in [0,T] : \|u_{n}(t)\|_{V} \ge R\},\$$

where u_n is the solution of (6.4.26). By the definition of martingale solution one can infer that for every $n \ge 1$, $\tau_R^n \nearrow \infty$ as $R \nearrow \infty$.

Lemma 6.5.1. Let u_n be the solution of (6.4.26). For all $\rho > 0$ there exist positive constants $C_1(\rho), C_2(\rho)$ such that if $||u_0||_V \le \rho$, then

(6.5.2)
$$\sup_{n \in \mathbb{N}} \mathbb{E} \left(\sup_{t \in [0,T]} \|u_n(t \wedge \tau_R^n)\|_{\mathbf{V}}^2 \right) \le C_1(\rho),$$

(6.5.3)
$$\sup_{n\in\mathbb{N}}\mathbb{E}\int_0^{T\wedge \tau_R^n}|u_n(t)|^2_{\mathrm{D}(\mathrm{A})}dt \leq C_2(\rho),$$

Moreover, for every $\delta > 0$ there exists a constant $C(\delta) > 0$ such that if $|u_0|_H \le \delta$, then

(6.5.4)
$$\sup_{n \in \mathbb{N}} \mathbb{E} \int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{L^{4}}^{4} \leq C_{3}(\delta).$$

Proof. Let u_n be the solution of (6.4.26) then applying the Itô formula to $\phi(x) = |x|_{\rm H}^2$ and the process u_n , we get

$$|u_{n}(t \wedge \tau_{R}^{n})|_{\mathrm{H}}^{2} = |P_{n}u_{0}|_{\mathrm{H}}^{2} + 2\int_{0}^{t \wedge \tau_{R}^{n}} \langle u_{n}(s), -\mathrm{A}u_{n}(s) - B_{n}(u_{n}(s)) - g_{n}(u_{n}(s)) \rangle_{\mathrm{H}} ds + 2\int_{0}^{t \wedge \tau_{R}^{n}} \langle u_{n}(s), f_{n}(u_{n}(s)) \rangle_{\mathrm{H}} ds + 2\int_{0}^{t \wedge \tau_{R}^{n}} \langle u_{n}(s), G_{n}(s, u_{n}(s)) dW_{s} \rangle_{\mathrm{H}} + \int_{0}^{t \wedge \tau_{R}^{n}} \|G_{n}(s, u_{n}(s))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2} ds.$$
(6.5.5)

Using Lemma 6.2.1, assumptions (A1) and (A2), boundedness of P_n in H, the Cauchy-Schwarz and the Young inequality, we get

$$|u_{n}(t \wedge \tau_{R}^{n})|_{H}^{2} \leq |u_{0}|_{H}^{2} - 2\int_{0}^{t \wedge \tau_{R}^{n}} |\nabla u_{n}(s)|_{L^{2}}^{2} ds - 2\int_{0}^{t \wedge \tau_{R}^{n}} \langle g(|u_{n}(s)|^{2})u_{n}(s), u_{n}(s) \rangle_{H} ds$$

$$(6.5.6) + 2C_{f} \int_{0}^{t \wedge \tau_{R}^{n}} |u_{n}(s)|_{H}^{2} ds + 2\int_{0}^{t \wedge \tau_{R}^{n}} \langle u_{n}(s), G(s, u_{n}(s)) dW_{s} \rangle_{H} + \frac{1}{4} \int_{0}^{t \wedge \tau_{R}^{n}} |\nabla u_{n}(s)|_{L^{2}}^{2} ds.$$

Since $u_n \in \mathbf{H}_n$, we have the following identities

$$\begin{split} \langle g_n(u_n), u_n \rangle_{\mathcal{H}} &= \langle \Pi \left(g(|u_n|^2) u_n \right), P_n u_n \rangle_{\mathcal{H}} = \langle g(|u_n|^2) u_n, \Pi u_n \rangle_{L^2} \\ &= \langle g(|u_n|^2) u_n, u_n \rangle_{L^2}. \end{split}$$

Thus, using the second part of the inequality (6.4.13), we get

$$\begin{split} |u_{n}(t \wedge \tau_{R}^{n})|_{\mathrm{H}}^{2} &\leq |u_{0}|_{\mathrm{H}}^{2} - 2\int_{0}^{t \wedge \tau_{R}^{n}} |\nabla u_{n}(s)|_{L^{2}}^{2} ds - 2\int_{0}^{t \wedge \tau_{R}^{n}} |u_{n}(s)|_{L^{4}}^{4} ds + 2C_{N} \int_{0}^{t \wedge \tau_{R}^{n}} |u_{n}(s)|_{\mathrm{H}}^{2} ds \\ &+ 2C_{f} \int_{0}^{t \wedge \tau_{R}^{n}} |u_{n}(s)|_{\mathrm{H}}^{2} ds + 2\int_{0}^{t \wedge \tau_{R}^{n}} \langle u_{n}(s), G(s, u_{n}(s)) dW_{s} \rangle_{\mathrm{H}} \\ &+ \frac{1}{4} \int_{0}^{t \wedge \tau_{R}^{n}} |\nabla u_{n}(s)|_{L^{2}}^{2} ds. \end{split}$$

On rearranging we have

$$(6.5.7) \qquad |u_n(t \wedge \tau_R^n)|_{\mathrm{H}}^2 + \frac{7}{4} \int_0^{t \wedge \tau_R^n} |\nabla u_n(s)|_{L^2}^2 ds + 2 \int_0^{t \wedge \tau_R^n} |u_n(s)|_{L^4}^4 ds \\ \leq |u_0|_{\mathrm{H}}^2 + C_{f,N} \int_0^{t \wedge \tau_R^n} |u_n(s)|_{\mathrm{H}}^2 ds + 2 \int_0^{t \wedge \tau_R^n} \langle u_n(s), G(s, u_n(s)) dW_s \rangle_{\mathrm{H}}.$$

Now since the process $\mu_n(t \wedge t \wedge \tau_R^n), t \in [0, T]$

$$\mu_n(t \wedge \tau_R^n) = \int_0^{t \wedge \tau_R^n} \langle u_n(s), G(s, u_n(s)) dW_s \rangle_{\mathrm{H}}, \quad t \in [0, T]$$

is a \mathbb{F} -martingale, as by Lemma 6.2.1 and (6.5.1) we have the following inequalities

$$\begin{split} \mathbb{E} \int_{0}^{t\wedge\tau_{R}^{n}} \left| \langle u_{n}(s), G(s, u_{n}(s)) \rangle_{\mathrm{H}} \right|^{2} ds &\leq \mathbb{E} \int_{0}^{t\wedge\tau_{R}^{n}} |u_{n}(s)|_{\mathrm{H}}^{2} \|G(s, u_{n}(s))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2} ds \\ &\leq \frac{1}{4} \mathbb{E} \int_{0}^{t\wedge\tau_{R}^{n}} |u_{n}(s)|_{\mathrm{H}}^{2} |\nabla u_{n}(s)|_{L^{2}}^{2} ds < \infty, \end{split}$$

where to establish the last inequality we have used the equivalences of norm from Proposition 6.4.9. Thus $\mathbb{E}[\mu_n(t)] = 0$.

Using Lemma 2.7.20 for the following three processes:

$$X(t) = |u_n(t \wedge \tau_R^n)|_{\mathrm{H}}^2, \qquad Y(t) = \frac{7}{4} \int_0^{t \wedge \tau_R^n} |\nabla u_n(s)|_{L^2}^2 \, ds + 2 \int_0^{t \wedge \tau_R^n} ||u_n(s)||_{L^4}^4 \, ds$$

and

$$I(t) = 2\mu_n(t) = 2\int_0^{t\wedge\tau_R^n} \langle u_n(s), G(s, u_n(s)) dW_s \rangle_{\mathrm{H}},$$

we see that from (6.5.7), condition (2.7.3) is satisfied for $\alpha = 1$, $Z = |u_0|_{\mathrm{H}}^2$ and $\phi(r) = C_{f,N}$. Since $\mathbb{E}(I(t)) = 0$, condition (2.7.4) is satisfied and hence all inequalities for the parameters (see (2.7.2)) are trivially satisfied. Thus if $|u_0|_{\mathrm{H}} \leq \delta$, we have

(6.5.8)
$$\sup_{n \in \mathbb{N}} \mathbb{E} \left[\left| u_n(t \wedge \tau_R^n) \right|_{\mathrm{H}}^2 + \frac{7}{4} \int_0^{t \wedge \tau_R^n} \left| \nabla u_n(s) \right|_{L^2}^2 ds + 2 \int_0^{t \wedge \tau_R^n} \left\| u_n(s) \right\|_{L^4}^4 ds \right] \le C_T(\delta).$$

In particular

(6.5.9)
$$\sup_{n \in \mathbb{N}} \left(\sup_{t \in [0,T]} \mathbb{E} \left| u_n(t \wedge \tau_R^n) \right|_{\mathrm{H}}^2 \right) \le C_T(\delta)$$

Hence, using (6.5.8) and (6.5.9) we infer that

(6.5.10)
$$\sup_{n \in \mathbb{N}} \mathbb{E} \int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{L^{4}}^{4} \leq \tilde{C}_{T}(|u_{0}|_{H}^{2}) =: C_{3}(\delta).$$

Since we are interested in the estimates involving V norm of u. We apply the Itô formula to $\phi(x) = |\nabla x|_{L^2}^2$ and the process $u_n(t)$, obtaining

$$\begin{aligned} |\nabla u_{n}(t \wedge \tau_{R}^{n})|_{L^{2}}^{2} &= |\nabla (P_{n}u_{0})|_{L^{2}}^{2} + 2\int_{0}^{t \wedge \tau_{R}^{n}} ((u_{n}(s), -Au_{n}(s) - B_{n}(u_{n}(s)) - g_{n}(u_{n}(s)))) \, ds \\ &+ 2\int_{0}^{t \wedge \tau_{R}^{n}} ((u_{n}(s), f_{n}(u_{n}(s)))) \, ds + 2\int_{0}^{t \wedge \tau_{R}^{n}} ((u_{n}(s), G_{n}(s, u_{n}(s)) \, dW_{s})) \\ &+ \int_{0}^{t \wedge \tau_{R}^{n}} \|\nabla (G_{n}(s, u_{n}(s)))\|_{\mathcal{J}_{2}(\ell^{2}; \mathrm{H})}^{2} \, ds, \end{aligned}$$
(6.5.11)

where $((\cdot, \cdot))$ is as defined in (6.2.1). Using Lemma 6.2.1, assumptions (A1) and (A2), boundedness of P_n in H, estimates (6.4.8), (6.4.13), the Cauchy-Schwarz and the Young inequality, we get

$$\begin{split} |\nabla u_{n}(t \wedge \tau_{R}^{n})|_{L^{2}}^{2} &\leq |\nabla u_{0}|_{L^{2}}^{2} - 2\int_{0}^{t \wedge \tau_{R}^{n}} |A u_{n}(s)|_{L^{2}}^{2} ds + \int_{0}^{t \wedge \tau_{R}^{n}} |A u_{n}(s)|_{L^{2}}^{2} ds \\ &+ \int_{0}^{t \wedge \tau_{R}^{n}} \left| |u_{n}(s)| \cdot |\nabla u_{n}(s)| \right|_{L^{2}}^{2} ds + 2C_{N} \int_{0}^{t \wedge \tau_{R}^{n}} |\nabla u_{n}(s)|_{L^{2}}^{2} ds \\ &- 4\int_{0}^{t \wedge \tau_{R}^{n}} \left| |u_{n}(s)| \cdot |\nabla u_{n}(s)| \right|_{L^{2}}^{2} ds + \int_{0}^{t \wedge \tau_{R}^{n}} |A u_{n}(s)|_{L^{2}}^{2} |f(u_{n}(s))|_{L^{2}} ds \\ &+ \frac{1}{2} \int_{0}^{t \wedge \tau_{R}^{n}} |A u_{n}(s)|_{L^{2}}^{2} ds + C_{T,\sigma} \int_{0}^{t \wedge \tau_{R}^{n}} |\nabla u_{n}(s)|_{L^{2}}^{2} ds \\ &+ 2 \int_{0}^{t \wedge \tau_{R}^{n}} ((u_{n}(s), G(s, u_{n}(s))) dW_{s})) \\ &\leq |\nabla u_{0}|_{L^{2}}^{2} - \frac{1}{2} \int_{0}^{t \wedge \tau_{R}^{n}} |A u_{n}(s)|_{L^{2}}^{2} ds - 3 \int_{0}^{t \wedge \tau_{R}^{n}} ||u_{n}(s)| \cdot |\nabla u_{n}(s)||_{L^{2}}^{2} ds \\ &+ C_{T,\sigma,N} \int_{0}^{t \wedge \tau_{R}^{n}} |\nabla u_{n}(s)|_{L^{2}}^{2} ds + \frac{1}{4} \int_{0}^{t \wedge \tau_{R}^{n}} |A u_{n}(s)|_{L^{2}}^{2} ds + C_{f} \int_{0}^{t \wedge \tau_{R}^{n}} |u_{n}(s)|_{H}^{2} ds \\ &+ 2 \int_{0}^{t \wedge \tau_{R}^{n}} ((u_{n}(s), G(s, u_{n}(s)) dW_{s})). \end{split}$$

On rearranging we have

$$\begin{aligned} |\nabla u_{n}(t \wedge \tau_{R}^{n})|_{L^{2}}^{2} + \frac{1}{4} \int_{0}^{t \wedge \tau_{R}^{n}} |A u_{n}(s)|_{L^{2}}^{2} ds + 3 \int_{0}^{t \wedge \tau_{R}^{n}} \left| |u_{n}(s)| \cdot |\nabla u_{n}(s)| \right|_{L^{2}}^{2} ds \\ &\leq |\nabla u_{0}|_{L^{2}}^{2} + C_{T,\sigma,N} \int_{0}^{t \wedge \tau_{R}^{n}} |\nabla u_{n}(s)|_{L^{2}}^{2} ds + C_{f} \int_{0}^{t \wedge \tau_{R}^{n}} |u_{n}(s)|_{H}^{2} ds \\ &+ 2 \int_{0}^{t \wedge \tau_{R}^{n}} ((u_{n}(s), G(s, u_{n}(s)) dW_{s})). \end{aligned}$$

$$(6.5.12)$$

Now since the process $\mu_n(t \wedge t \wedge \tau_R^n), t \in [0, T]$

$$\mu_n(t \wedge \tau_R^n) = \int_0^{t \wedge \tau_R^n} ((u_n(s), G(s, u_n(s)) \, dW_s)), \quad t \in [0, T]$$

is a \mathbb{F} -martingale, as by Lemma 6.2.1 and (6.5.1) we have the following inequalities

$$\begin{split} \mathbb{E} \int_{0}^{t \wedge \tau_{R}^{n}} \left| \left((u_{n}(s), G(s, u_{n}(s))) \right) \right|^{2} ds &\leq \mathbb{E} \int_{0}^{t \wedge \tau_{R}^{n}} |\mathbf{A}u_{n}(s)|_{L^{2}}^{2} \|G(s, u_{n}(s))\|_{\mathcal{F}_{2}(\ell^{2}; \mathbf{H})}^{2} ds \\ &\leq \frac{1}{4} \mathbb{E} \int_{0}^{t \wedge \tau_{R}^{n}} |\mathbf{A}u_{n}(s)|_{L^{2}}^{2} |\nabla u_{n}(s)|_{L^{2}}^{2} ds < \infty, \end{split}$$

where to establish the last inequality we have used the equivalences of norm from Proposition 6.4.9. Thus $\mathbb{E}[\mu_n(t)] = 0$.

Again as before, by applying Lemma 2.7.20 for

$$\begin{aligned} X(t) &= |\nabla u_n(t \wedge \tau_R^n)|_{L^2}^2, \qquad Y(t) = \frac{1}{4} \int_0^{t \wedge \tau_R^n} |A u_n(s)|_{L^2}^2 ds + 3 \int_0^{t \wedge \tau_R^n} \left| |u_n(s)| \cdot |\nabla u_n(s)| \right|_{L^2}^2 ds, \\ I(t) &= 2\mu_n(t) = 2 \int_0^{t \wedge \tau_R^n} ((u_n(s), G(s, u_n(s)) dW_s)), \end{aligned}$$

the inequalities (2.7.3) and (2.7.4) are satisfied. Thus from (6.5.9) and (6.5.11), if $||u_0||_V \le \rho$, then

$$(6.5.13) \sup_{n \in \mathbb{N}} \mathbb{E}\left[\left| \nabla u_n(t \wedge \tau_R^n) \right|_{L^2}^2 + \frac{1}{4} \int_0^{t \wedge \tau_R^n} \left| A u_n(s) \right|_{L^2}^2 ds + 3 \int_0^{t \wedge \tau_R^n} \left| \left| u_n(s) \right| \cdot \left| \nabla u_n(s) \right| \right|_{L^2}^2 ds \right] \le C_T(\rho).$$

In particular

(6.5.14)
$$\sup_{n \in \mathbb{N}} \left(\sup_{t \in [0,T]} \mathbb{E} |\nabla u_n(t \wedge \tau_R^n)|_{L^2}^2 \right) \le C_T(\rho)$$

From (6.5.14) and (6.5.13), we have the following estimate

(6.5.15)
$$\sup_{n\in\mathbb{N}}\mathbb{E}\int_{0}^{T\wedge\tau_{R}^{n}}\left|\operatorname{A}u_{n}(t)\right|_{L^{2}}^{2}dt\leq C_{T}(\rho).$$

Note that $|u|_{D(A)}^2 := |u|_{L^2}^2 + |Au|_{L^2}^2$. Thus from (6.5.9) and (6.5.15) we can infer (6.5.3). On combining (6.5.9) and (6.5.14), we get

(6.5.16)
$$\sup_{n \in \mathbb{N}} \left(\sup_{t \in [0,T]} \mathbb{E} \left\| u_n(t \wedge \tau_R^n) \right\|_{\mathcal{V}}^2 \right) \le C_T(\rho)$$

Using the Burkholer-Davis-Gundy inequality, the definition of $\langle \cdot, \cdot \rangle_V$ and the Young's inequality, for every $\varepsilon > 0$ we obtain

$$\begin{split} \mathbb{E} \sup_{t \in [0,T]} \int_{0}^{t \wedge \tau_{R}^{n}} \langle u_{n}(s), G(s, u_{n}(s)) dW_{s} \rangle_{\mathrm{V}} \\ &= \mathbb{E} \left[\int_{0}^{T \wedge \tau_{R}^{n}} \left| \langle u_{n}(s), G(s, u_{n}(s)) \rangle_{\mathrm{H}} + \left(\left(u_{n}(s), G(s, u_{n}(s)) \right) \right) \right|^{2} ds \right]^{1/2} \\ &\leq \mathbb{E} \left[\int_{0}^{T \wedge \tau_{R}^{n}} \|G(s, u_{n}(s))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds \right]^{1/2} \\ &\leq \frac{1}{2} \mathbb{E} \left[\sup_{s \in [0,T]} |\nabla u_{n}(s \wedge \tau_{R}^{n})|_{L^{2}}^{2} \int_{0}^{T \wedge \tau_{R}^{n}} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds \right]^{1/2} \\ &\leq \varepsilon \mathbb{E} \sup_{s \in [0,T]} |\nabla u_{n}(s \wedge \tau_{R}^{n})|_{L^{2}}^{2} + C_{\varepsilon} \mathbb{E} \int_{0}^{T \wedge \tau_{R}^{n}} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds. \end{split}$$

On combining (6.5.7) and (6.5.12), then using (6.5.3), (6.5.9), (6.5.14), (6.5.17) and Lemma 2.7.20, we can infer (6.5.2). Thus the proof of the lemma is complete.

In the next lemma we will use the estimates from Lemma 6.5.1 to establish higher order estimates.

Lemma 6.5.2. Let τ_R^n be as defined in (6.5.1). For all $\rho > 0$ and $p \in [1,3]$ there exist positive constants $C_1(p,\rho)$, $C_2(p,\rho)$ such that if $||u_0||_V \le \rho$, then

(6.5.18)
$$\sup_{n \in \mathbb{N}} \mathbb{E} \left(\sup_{t \in [0,T]} \left\| u_n(t \wedge \tau_R^n) \right\|_{\mathrm{V}}^{2p} \right) \le C_1(p,\rho),$$

(6.5.19)
$$\sup_{n \in \mathbb{N}} \mathbb{E} \int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s)|_{L^{2}}^{2} ds \leq C_{2}(p,\rho).$$

Proof. Let $p \in [1,3]$. Then by using the Itô formula for $\xi(t) = ||u_n(t)||_V^2$, $\phi(x) = x^p$, equations (6.5.5), (6.5.11) and the definition of $|| \cdot ||_V$, we obtain

$$\begin{split} \|u_{n}(t\wedge\tau_{R}^{n})\|_{\mathcal{V}}^{2p} &= \|u_{n}(0)\|_{\mathcal{V}}^{2p} - 2p\int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \left(|\nabla u_{n}(s)|_{L^{2}}^{2} + |Au_{n}(s)|_{L^{2}}^{2}\right) ds \\ &- 2p\int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \langle u_{n}(s), B_{n}(u_{n}(s)) \rangle_{\mathcal{V}} ds - 2p\int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \langle u_{n}(s), g_{n}(u_{n}(s)) \rangle_{\mathcal{V}} ds \\ &- 2p\int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \langle u_{n}(s), f_{n}(u_{n}(s)) \rangle_{\mathcal{V}} ds + p\int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \|G_{n}(s, u_{n}(s))\|_{\mathcal{F}_{2}(\ell^{2}; \mathcal{V})}^{2} ds \\ &+ 2p(p-1)\int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-2)} \langle u_{n}(s), G_{n}(s, u_{n}(s)) \rangle_{\mathcal{V}}^{2} ds \end{split}$$

(6.5.20)

$$+2p\int_0^{t\wedge\tau_R^n}\|u_n(s)\|_{\mathrm{V}}^{2(p-1)}\langle u_n(s),G_n(s,u_n(s))dW_s\rangle_{\mathrm{V}}.$$

Using Lemma 6.2.1, the definition of g (6.2.13), boundedness of P_n in V and assumption (A1), we can simplify (6.5.20)

$$\begin{split} \|u_{n}(t \wedge \tau_{R}^{n})\|_{\mathcal{V}}^{2p} &\leq \|u_{n}(0)\|_{\mathcal{V}}^{2p} - 2p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \left(|\nabla u_{n}(s)|_{L^{2}}^{2} + |Au_{n}(s)|_{L^{2}}^{2} \right) ds \\ &+ 2p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \left[\frac{1}{2} |Au_{n}(s)|_{L^{2}}^{2} + \frac{1}{2} ||u_{n}(s)| \cdot |\nabla u_{n}(s)||_{L^{2}}^{2} \right] ds \\ &- 2p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \langle u_{n}(s), |u_{n}(s)|^{2}u_{n}(s) - Nu_{n}(s)\rangle_{\mathcal{V}} ds \\ &+ 2p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \left[C_{f} |u_{n}(s)|_{H}^{2} + \frac{1}{4} |Au_{n}(s)|_{L^{2}}^{2} \right] ds \\ &+ p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \left[\frac{1}{2} |Au_{n}(s)|_{L^{2}}^{2} + C_{T,\sigma} |\nabla u_{n}(s)|_{L^{2}}^{2} \right] ds \\ &+ 2p (p-1) \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \langle u_{n}(s), G(s, u_{n}(s))\|_{\mathcal{F}_{2}(\ell^{2};\mathcal{V})}^{2} \|u_{n}(s)\|_{\mathcal{V}}^{2} ds \\ &+ 2p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathcal{V}}^{2(p-1)} \langle u_{n}(s), G(s, u_{n}(s)) dW_{s} \rangle_{\mathcal{V}}. \end{split}$$

On rearranging we get

$$\begin{split} \|u_{n}(t\wedge\tau_{R}^{n})\|_{V}^{2p} &\leq \|u_{n}(0)\|_{V}^{2p} - \frac{p}{2} \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s)|_{L^{2}}^{2} ds \\ &- 2p \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|^{2(p-1)} |\nabla u_{n}(s)|_{L^{2}}^{2} ds + p \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} ||u_{n}(s)| \cdot |\nabla u_{n}(s)||_{L^{2}}^{2} ds \\ &- 2p \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} ||\sqrt{g(|u(s)|^{2})}| \cdot |u(s)||_{L^{2}}^{2} ds + C_{N} \cdot 2p \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |\nabla u_{n}(s)|_{L^{2}}^{2} ds \\ &- 4p \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} ||u_{n}(s)| \cdot |\nabla u_{n}(s)||_{L^{2}}^{2} ds + 2p C_{f} \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |u_{n}(s)|_{H}^{2} ds \\ &+ p C_{T,\sigma} \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |\nabla u_{n}(s)|_{L^{2}}^{2} ds + 2p \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} \langle u_{n}(s), G(s, u_{n}(s)) dW_{s} \rangle_{V} \\ &+ 2p(p-1) \int_{0}^{t\wedge\tau_{R}^{n}} \|u_{n}(s)\|^{2(p-1)} \left[\frac{1}{4} |Au_{n}(s)|_{L^{2}}^{2} + C_{T,\sigma} |\nabla u_{n}(s)|_{L^{2}}^{2} ds \right] ds \end{split}$$

which on further simplification yields

$$\begin{split} \|u_{n}(t \wedge \tau_{R}^{n})\|_{V}^{2p} &+ \frac{p(3-p)}{2} \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s)|_{L^{2}}^{2} ds \\ &+ 3p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} ||u_{n}(s)| \cdot |\nabla u_{n}(s)||_{L^{2}}^{2} ds + 2p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} ||\sqrt{g(|u(s)|^{2})}| \cdot |u(s)||_{L^{2}}^{2} ds \\ &\leq \|u_{n}(0)\|_{V}^{2p} + C_{T,\sigma,N,p} \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |\nabla u_{n}(s)|_{L^{2}}^{2} ds + C_{f} \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |u_{n}(s)|_{H}^{2} ds \end{split}$$

(6.5.21)

$$+2p\int_0^{t\wedge\tau_R^n}\|u_n(s)\|_{\mathbf{V}}^{2(p-1)}\langle u_n(s),G(s,u_n(s))dW_s\rangle_{\mathbf{V}}.$$

As before we will show that the process $\mu_n(t \wedge t \wedge \tau_R^n), t \in [0,T]$

$$\mu_n(t \wedge \tau_R^n) = \int_0^{t \wedge \tau_R^n} \|u_n(s)\|_{\mathbf{V}}^{2(p-1)} \langle u_n(s), G(s, u_n(s)) \, dW_s \rangle_{\mathbf{V}}, \quad t \in [0, T]$$

is a $\mathbb F\text{-martingale}.$ By Lemma 6.2.1 and (6.5.1) we have the following inequalities

$$\begin{split} \mathbb{E} &\int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathbf{V}}^{4(p-1)} |\langle u_{n}(s), G(s, u_{n}(s)) \rangle_{\mathbf{V}}|^{2} ds \\ &\leq \mathbb{E} \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathbf{V}}^{4(p-1)} |u_{n}(s)|_{\mathrm{D}(\mathbf{A})}^{2} \|G(s, u_{n}(s))\|_{\mathcal{F}_{2}(\ell^{2};\mathbf{H})}^{2} ds \\ &\leq \frac{1}{4} \mathbb{E} \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{\mathbf{V}}^{4(p-1)} |u_{n}(s)|_{\mathrm{D}(\mathbf{A})}^{2} |\nabla u_{n}(s)|_{L^{2}}^{2} ds < \infty, \end{split}$$

where the finiteness of the integral follows from Proposition 6.4.9. Hence $\mathbb{E}[\mu_n(t)] = 0$.

Since $|u_n(s)|_{H} \le ||u_n(s)||_V$ and $|\nabla u_n(s)|_{L^2} \le ||u_n(s)||_V$ on applying the generalised version of the Gronwall Lemma (Lemma 2.7.20) for

$$X(t) = \|u_n(t \wedge \tau_R^n)\|_{V}^{2p}, \qquad I(t) = 2p \int_0^{t \wedge \tau_R^n} \|u_n(s)\|_{V}^{2(p-1)} \langle u_n(s), G(s, u_n(s)) dW_s \rangle_{V},$$

and

$$\begin{split} Y(t) &= \frac{p(3-p)}{2} \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s)|_{L^{2}}^{2} ds + 3p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} \big| |u_{n}(s)| \cdot |\nabla u_{n}(s)| \big|_{L^{2}}^{2} ds \\ &+ 2p \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} \big| |\sqrt{g(|u(s)|^{2})}| \cdot |u(s)| \big|_{L^{2}}^{2} ds, \end{split}$$

we have

(6.5.22)
$$\sup_{n \in \mathbb{N}} \left(\sup_{t \in [0,T]} \mathbb{E} \| u_n(t \wedge \tau_R^n) \|_{\mathbf{V}}^{2p} \right) \le C_{T,p} \| u_0 \|_{\mathbf{V}}^{2p}, \quad p \in [1,3].$$

Using (6.5.22) in (6.5.20), we also obtain

(6.5.23)
$$\sup_{n \in \mathbb{N}} \mathbb{E} \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |Au_{n}(s)|_{L^{2}}^{2} ds \leq C_{T,p} \|u_{0}\|_{V}^{2p} := C_{2}(p,\rho), \quad p \in [1,3].$$

Now we are left to show the estimate (6.5.18). Using the Burkholder- Davis- Gundy inequality, Lemma 6.2.1 and the Young inequality for every $\varepsilon > 0$, we get

$$\begin{split} \mathbb{E} \sup_{t \in [0,T]} \int_{0}^{t \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} \langle u_{n}(s), G(s, u_{n}(s)) dW_{s} \rangle_{V} \\ &\leq \mathbb{E} \left[\int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{4(p-1)} |\langle u_{n}(s), G(s, u_{n}(s)) \rangle_{V}|^{2} ds \right]^{1/2} \\ &\leq \mathbb{E} \left[\int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{4(p-1)} \|G(s, u_{n}(s))\|_{\mathcal{F}_{2}(\ell^{2}; \mathrm{H})}^{2} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds \right]^{1/2} \\ &\leq \frac{1}{4} \mathbb{E} \left[\int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{4(p-1)} |\nabla u_{n}(s)|_{L^{2}}^{2} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds \right]^{1/2} \\ &\leq \frac{1}{4} \mathbb{E} \left[\int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2p} \|u_{n}(s)\|_{V}^{2(p-1)} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds \right]^{1/2} \\ &\leq \frac{1}{4} \mathbb{E} \left[\sup_{t \in [0,T]} \|u_{n}(t \wedge \tau_{R}^{n})\|_{V}^{2p} \int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|_{V}^{2(p-1)} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds \right]^{1/2} \\ &\leq \varepsilon \mathbb{E} \sup_{t \in [0,T]} \|u_{n}(t \wedge \tau_{R}^{n})\|_{V}^{2p} + C_{\varepsilon} \mathbb{E} \int_{0}^{T \wedge \tau_{R}^{n}} \|u_{n}(s)\|^{2(p-1)} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds. \end{split}$$

$$(6.5.24)$$

Thus from (6.5.21) and using (6.5.23), (6.5.24) and Lemma 2.7.20, we have

$$\mathbb{E}\left(\sup_{t\in[0,T]} \|u_n(t\wedge\tau_R^n)\|_{\mathbf{V}}^{2p}\right) \le C_{T,p}\left(\mathbb{E}\|u_0\|_{\mathbf{V}}^{2p}\right) + \varepsilon \mathbb{E}\left(\sup_{t\in[0,T]} \|u_n(t\wedge\tau_R^n)\|_{\mathbf{V}}^{2p}\right) + C_{T,p,\varepsilon}$$

Choosing ε small enough we get

n

(6.5.25)
$$\sup_{n \in \mathbb{N}} \mathbb{E} \left(\sup_{t \in [0,T]} \|u_n(t \wedge \tau_R^n)\|_{\mathbf{V}}^{2p} \right) \le C_{T,p} \left(\|u_0\|_{\mathbf{V}}^{2p} \right) := C_1(p,\rho), \quad p \in [1,3].$$

6.5.2 Tightness of measures

For each $n \in \mathbb{N}$, the solution u_n of the truncated equation (6.4.26) defines a measure $\mathcal{L}(u_n)$ on $(\mathcal{Z}_T, \mathcal{T})$, defined in (6.3.5). In this subsection we will prove that this sequence of measures defined on \mathcal{Z}_T is tight.

Lemma 6.5.3. The set of measures $\{\mathscr{L}(u_n), n \in \mathbb{N}\}$ is tight on $(\mathscr{Z}_T, \mathscr{T})$.

Proof. We recall the definition of stopping time, τ_R^n

$$\tau_R^n := \inf\{t \in [0,T] : \|u_n(t)\|_{\mathcal{V}} \ge R\}.$$

We will use Corollary 6.3.4 to prove the tightness of measures. According to estimates (6.5.2) and (6.5.3), conditions (a) and (b) are satisfied. Thus it is sufficient to prove that the sequence $(u_n)_{n \in \mathbb{N}}$

satisfies the Aldous condition [A] in H. By (6.4.26), for $t \in [0, T \wedge \tau_R^n]$ we have

$$\begin{split} u_n(t) &= u_n(0) - \int_0^t \mathbf{A}_n u_n(s) \, ds - \int_0^t B_n(u_n(s)) \, ds - \int_0^t g_n(u_n(s)) \, ds + \int_0^t f_n(u_n(s)) \, ds \\ &+ \int_0^t G_n(s, u_n(s)) \, dW(s) \\ &:= J_1^n + J_2^n(t) + J_3^n(t) + J_4^n(t) + J_5^n(t) + J_6^n(t), \quad t \in [0, T \wedge \tau_R^n]. \end{split}$$

Let $s, t \in [0, T]$, s < t and $\theta := t - s$. First we will establish estimates for each term of the above equality.

Ad. J_2^n . Since $A: D(A) \to H$ is a bounded linear map, then by the Hölder inequality and estimate (6.5.3), we have the following inequalities

$$\mathbb{E}\left[|J_{2}^{n}(t\wedge\tau_{R}^{n})-J_{2}^{n}(s\wedge\tau_{R}^{n})|_{\mathrm{H}}\right] = \mathbb{E}\left|\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}\mathsf{A}_{n}u_{n}(s)ds\right|_{\mathrm{H}} \leq \mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}|\mathsf{A}u_{n}(s)|_{\mathrm{H}}ds$$

$$(6.5.26) \leq c\theta^{\frac{1}{2}}\left(\mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}|u_{n}(s)|_{\mathrm{D}(\mathrm{A})}^{2}ds\right)^{\frac{1}{2}} \leq c(C_{2}(R))^{\frac{1}{2}}\cdot\theta := c_{2}\cdot\theta$$

Ad. J_3^n . $B : D(A) \times V \to H$ is bilinear and continuous and $P_n : H \to H$ is bounded then by Lemma 6.4.5, the Cauchy-Schwarz inequality and the estimates (6.5.2), (6.5.3) we have

$$\mathbb{E}\left[|J_{3}^{n}(t \wedge \tau_{R}^{n}) - J_{3}^{n}(s \wedge \tau_{R}^{n})|_{\mathrm{H}}\right] = \mathbb{E}\left|\int_{s \wedge \tau_{R}^{n}}^{t \wedge \tau_{R}^{n}} P_{n}B(u_{n}(s))ds\right|_{\mathrm{H}} \leq \mathbb{E}\int_{s \wedge \tau_{R}^{n}}^{t \wedge \tau_{R}^{n}} |P_{n}B(u_{n}(s), u_{n}(s))|_{\mathrm{H}}ds$$

$$\leq \mathbb{E}\int_{s \wedge \tau_{R}^{n}}^{t \wedge \tau_{R}^{n}} \|B\| \cdot |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}\|u_{n}(s)\|_{\mathrm{V}}ds$$

$$\leq \|B\| \cdot \mathbb{E}\left(\left[\sup_{t \in [0,T]}\|u_{n}(t \wedge \tau_{R}^{n})\|_{\mathrm{V}}^{2}\right]^{1/2} \cdot \theta^{1/2} \left[\int_{0}^{t \wedge \tau_{R}^{n}} |u_{n}(s)|_{\mathrm{D}(\mathrm{A})}ds\right]^{1/2}\right)$$

$$\leq (C_{1}(R))^{1/2}(C_{2}(R))^{1/2} \cdot \theta^{1/2} := c_{3} \cdot \theta^{1/2}.$$

Ad. J_4^n . Since $H^1 \hookrightarrow L^6$, then by the definition of g and estimate (6.5.18) (for p = 2), we have

$$\begin{split} & \mathbb{E}\left[|J_{4}^{n}(t\wedge\tau_{R}^{n}) - J_{4}^{n}(s\wedge\tau_{R}^{n})|_{\mathrm{H}}\right] = \mathbb{E}\left|\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}g_{n}(u_{n}(s))ds\right|_{\mathrm{H}} \leq \mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}|P_{n}(\Pi g(|u_{n}(s)|^{2})u_{n}(s))|_{\mathrm{H}}ds \\ & \leq \mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}|g(|u_{n}(s)|^{2})u_{n}(s)|_{L^{2}}ds \leq \mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}\left(\int_{\mathbb{R}^{3}}|u_{n}(s,x)|^{6}dx\right)^{1/2}ds \\ & = \mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}\|u_{n}(s)\|_{L^{6}}^{3}ds \leq C \mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}\|u_{n}(s)\|_{\mathrm{V}}^{3}ds \leq C \left[\mathbb{E}\left(\sup_{t\in[0,T]}\|u_{n}(t\wedge\tau_{R}^{n})\|_{\mathrm{V}}^{4}\right)\right]^{3/4}\theta \end{split}$$

(6.5.28)

$$\leq C \cdot (C_1(2,R))^{3/4} \cdot \theta := c_4 \cdot \theta.$$

Ad. J_5^n . Using the assumption **H1**, (6.5.9) and the Cauchy-Schwarz inequality, we obtain the following inequalities

$$\mathbb{E}\left[|J_{5}^{n}(t\wedge\tau_{R}^{n}) - J_{5}^{n}(s\wedge\tau_{R}^{n})|_{\mathrm{H}}\right] = \mathbb{E}\left|\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}} P_{n}(\Pi f(u_{n}(s))ds)\right|_{\mathrm{H}} \leq \mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}} |f(u_{n}(s))|_{\mathrm{H}}ds$$

$$\leq \mathbb{E}\left(\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}} |f(u_{n}(s))|_{\mathrm{H}}^{2}ds\right)^{1/2} \cdot \theta^{1/2} \leq C_{f}\left(\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}} |u_{n}(s)|_{\mathrm{H}}^{2}ds\right)^{1/2} \theta^{\frac{1}{2}}$$

$$\leq C_{f}\left(C(R)\right)^{1/2}\theta := c_{F} \cdot \theta$$

 $(6.5.29) \leq C_f \left(C(R) \right)^{1/2} \theta := c_5 \cdot \theta.$

Ad. J_6^n . Using the Itô isometry, Lemma 6.2.1 and (6.5.2), we obtain the following

$$\mathbb{E}\left[\left|J_{6}^{n}(t\wedge\tau_{R}^{n})-J_{6}^{n}(s\wedge\tau_{R}^{n})\right|_{\mathrm{H}}^{2}\right] = \mathbb{E}\left|\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}G_{n}(s,u_{n}(s))dW(s)\right|_{\mathrm{H}}^{2} \\ \leq c \mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}\left\|G(s,u_{n}(s))\right\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2}ds \leq \frac{c}{4}\mathbb{E}\int_{s\wedge\tau_{R}^{n}}^{t\wedge\tau_{R}^{n}}\left|\nabla u_{n}(s)\right|_{L^{2}}^{2}ds \\ \leq \frac{c}{4}\mathbb{E}\left(\sup_{t\in[0,T]}\left\|u_{n}(t\wedge\tau_{R}^{n})\right\|_{\mathrm{V}}^{2}\right)\theta \leq \frac{1}{4}C_{1}(R)\cdot\theta := c_{6}\cdot\theta.$$

$$(6.5.30)$$

Let us fix $\kappa > 0$ and $\varepsilon > 0$. By the Chebyshev's inequality and estimates (6.5.26) - (6.5.29), we obtain

$$\mathbb{P}(\{|J_i^n(t \wedge \tau_R^n) - J_i^n(s \wedge \tau_R^n)|_{\mathbf{H}} \ge \kappa\}) \le \frac{1}{\kappa} \mathbb{E}\left[|J_i^n(t \wedge \tau_R^n) - J_i^n(s \wedge \tau_R^n)|_{\mathbf{H}}\right] \le \frac{c_i \theta}{\kappa}; \quad n \in \mathbb{N},$$

where $i = 1, \dots, 5$. Let $\delta_i = \frac{\kappa}{c_i} \varepsilon$. Then
$$\sup_{n \in \mathbb{N}} \sup_{0 \le \theta \le \delta_i} \mathbb{P}(\{|J_i^n(t \wedge \tau_R^n) - J_i^n(s \wedge \tau_R^n)|_{\mathbf{H}} \ge \kappa\}) \le \varepsilon, \quad i = 1 \dots 5.$$

By the Chebyshev inequality and (6.5.30), we have

$$\mathbb{P}(\{|J_6^n(t \wedge \tau_R^n) - J_6^n(s \wedge \tau_R^n)|_{\mathcal{H}} \ge \kappa\}) \le \frac{1}{\kappa^2} \mathbb{E}\left[|J_6^n(t \wedge \tau_R^n) - J_6^n(s \wedge \tau_R^n)|_{\mathcal{H}}^2\right] \le \frac{c_6\theta}{\kappa^2}; \quad n \in \mathbb{N}$$

Let $\delta_6 = \frac{\kappa^2}{c_6} \varepsilon$. Then
$$\sup_{n \in \mathbb{N}} \sup_{0 \le \theta \le \delta_6} \mathbb{P}(\{|J_6^n(t \wedge \tau_R^n) - J_6^n(s \wedge \tau_R^n)|_{\mathcal{H}} \ge \kappa\}) \le \varepsilon.$$

Since [**A**] holds for each term J_i^n , i = 1, 2, ..., 6; we infer that it holds also for $(u_n)_{n \in \mathbb{N}}$. Thus the proof of lemma can be concluded by invoking Corollary 6.3.4.

Now we will state the main theorem of this section.

Theorem 6.5.4. Let assumptions (**H1**) and (**H2**) be satisfied. Then there exists a martingale solution $(\hat{\Omega}, \hat{\mathscr{F}}, \hat{\mathbb{F}}, \hat{\mathbb{P}}, u)$ of problem (6.2.16) such that

(6.5.31)
$$\hat{\mathbb{E}}\left[\sup_{t\in[0,T]}\|u(t)\|_{\mathrm{V}}^{2}+\int_{0}^{T}|u(t)|_{\mathrm{D}(\mathrm{A})}^{2}dt\right]<\infty.$$

In the following subsection we will prove Theorem 6.5.4 in several steps.

6.5.3 Proof of Theorem 6.5.4

By Lemma 6.5.3 the set of measures $\{\mathscr{L}(u_n), n \in \mathbb{N}\}$ is tight on the space $(\mathscr{Z}_T, \mathscr{T})$ defined by (6.3.5). Hence by Corollary 6.3.6 there exist a subsequence $(n_k)_{k \in \mathbb{N}}$, a probability space $(\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{P}})$ and, on this space, \mathscr{Z}_T -valued random variables $\tilde{u}, \tilde{u}_{n_k}, k \geq 1$ such that

(6.5.32) \tilde{u}_{n_k} has the same law as u_{n_k} and $\tilde{u}_{n_k} \to \tilde{u}$ in \mathcal{Z}_T , $\tilde{\mathbb{P}}$ -a.s.

 $\tilde{u}_{n_k} \to \tilde{u}$ in $\mathcal{Z}_T, \tilde{\mathbb{P}}$ – a.s. precisely means that

$$\begin{split} \tilde{u}_{n_k} &\to \tilde{u} \text{ in } \mathscr{C}([0,T];\mathbf{U}'), \\ \tilde{u}_{n_k} &\to \tilde{u} \text{ in } L^2(0,T;\mathbf{D}(\mathbf{A})), \\ \tilde{u}_{n_k} &\to \tilde{u} \text{ in } L^2(0,T;\mathbf{H}_{\text{loc}}), \\ \tilde{u}_{n_k} &\to \tilde{u} \text{ in } \mathscr{C}([0,T];\mathbf{V}_w). \end{split}$$

Let us denote the subsequence (\tilde{u}_{n_k}) again by $(\tilde{u}_n)_{n \in \mathbb{N}}$.

The following auxiliary result which is needed in the proof of Theorem 6.5.4, cannot be deduced directly from the Kuratowski Theorem (see Theorem 2.5.17).

Lemma 6.5.5. Let T > 0 and \mathcal{Z}_T be as defined in (6.3.5). Then the following three sets $\mathscr{C}([0,T];V) \cap \mathcal{Z}_T$, $\mathscr{C}([0,T];H_n) \cap \mathcal{Z}_T$ and $L^2(0,T;D(A)) \cap \mathcal{Z}_T$ are Borel subsets of \mathcal{Z}_T .

In order to prove Lemma 6.5.5 we will need the following space:

$$L^2_{\operatorname{loc}}([0,T]\times\mathbb{R}^3) = \left\{ u: [0,T]\times\mathbb{R}^3 \to \mathbb{R}^3: \int_0^T \int_{|x|\leq R} |u(x,t)|^2 \, dx \, dt < \infty, \forall R > 0 \right\}.$$

 $L^2_{\text{loc}}([0,T] \times \mathbb{R}^3)$ is complete under the family of semi-norms

$$\rho_R := \left[\int_0^T \int_{|x| \le R} |u(x,t)|^2 \, dx \, dt \right]^{1/2}.$$

In particular it's a Frechét space with the metric

$$d(u,v) = \sum_{n\geq 1} \frac{1}{2^n} \frac{\rho_n(u-v)}{1+\rho_n(u-v)}.$$

Remark 6.5.6. $L^2(0,T;H_{\text{loc}}) \subset L^2_{\text{loc}}([0,T] \times \mathbb{R}^3)$ and we can define the open sets in $L^2(0,T;H_{\text{loc}})$ by restricting the metric d to $L^2(0,T;H_{\text{loc}})$. Hence $L^2(0,T;H_{\text{loc}})$ is a topological space with the trace topology from $L^2_{\text{loc}}([0,T] \times \mathbb{R}^3)$.

Let us define a new topological space:

$$\tilde{\mathcal{Z}}_T := \mathscr{C}([0,T]; \mathbf{U}') \cap L^2_{\operatorname{loc}}([0,T] \times \mathbb{R}^3) \cap L^2_w(0,T; \mathbf{D}(\mathbf{A})) \cap \mathscr{C}([0,T]; \mathbf{V}_w).$$

Note that $\tilde{\mathcal{Z}}_T$ and \mathcal{Z}_T are same as a set. Because $L^2_{\text{loc}}([0,T] \times \mathbb{R}^3) \cap L^2_w(0,T;\mathbb{D}(A))$ and $L^2(0,T;\mathbb{H}_{\text{loc}}) \cap L^2_w(0,T;\mathbb{D}(A))$ are same as a set. $L^2(0,T;\mathbb{H}_{\text{loc}}) \subset L^2_{\text{loc}}([0,T] \times \mathbb{R}^3)$ and the only extra elements in $L^2_{\text{loc}}([0,T] \times \mathbb{R}^3)$ are the ones which are locally square integrable but have non-zero divergence. But the intersection of $L^2_{\text{loc}}([0,T] \times \mathbb{R}^3)$ with $L^2_w(0,T;\mathbb{D}(A))$ eliminates those elements as the divergence free condition is imposed by the second set.

By Remark 6.5.6 and Lemma C.2, $\tilde{\mathcal{Z}}_T$ and \mathcal{Z}_T have the same topologies. Thus we will prove Lemma 6.5.5 for $\tilde{\mathcal{Z}}_T$ instead of \mathcal{Z}_T .

Proof of Lemma 6.5.5 First of all $\mathscr{C}([0,T]; V) \subset \mathscr{C}([0,T]; U') \cap L^2_{loc}([0,T] \times \mathbb{R}^3)$. Secondly, $\mathscr{C}([0,T]; V)$ and $\mathscr{C}([0,T]; U') \cap L^2_{loc}([0,T] \times \mathbb{R}^3)$ are Polish spaces. And finally, since V is continuously embedded in U', the map

$$i: \mathscr{C}([0,T]; \mathbb{V}) \to \mathscr{C}([0,T]; \mathbb{U}') \cap L^2_{\text{loc}}([0,T] \times \mathbb{R}^3),$$

is continuous and hence Borel. Thus by application of the Kuratowski Theorem, $\mathscr{C}([0,T];V)$ is a Borel subset of $\mathscr{C}([0,T];U') \cap L^2_{loc}([0,T] \times \mathbb{R}^3)$. Therefore by Lemma C.1, $\mathscr{C}([0,T];V) \cap \tilde{\mathcal{Z}}_T$ is a Borel subset of $\mathscr{C}([0,T];U') \cap L^2_{loc}([0,T] \times \mathbb{R}^3) \cap \tilde{\mathcal{Z}}_T$ which is equal to $\tilde{\mathcal{Z}}_T$. We can show in the same way in the case of $\mathscr{C}([0,T];H_n) \cap \mathcal{Z}_T$.

Similarly we can show that $L^2(0,T; D(A)) \cap \tilde{\mathcal{Z}}_T$ is a Borel subset of $\tilde{\mathcal{Z}}_T$. $L^2(0,T; D(A)) \hookrightarrow L^2_{loc}([0,T] \times \mathbb{R}^3)$ and both are Polish spaces thus by application of the Kuratowski Theorem, $L^2(0,T; D(A))$ is a Borel subset of $L^2_{loc}([0,T] \times \mathbb{R}^3)$. Finally, we can conclude the proof of theorem by Lemma C.1.

By Lemma 6.5.5, $\mathscr{C}([0,T]; \mathbf{H}_n)$ is a Borel subset of $\mathscr{C}([0,T]; \mathbf{U}') \cap L^2(0,T; \mathbf{H}_{loc})$. Since $u_n \in \mathscr{C}([0,T]; \mathbf{H}_n), \mathbb{P}$ -a.s., and \tilde{u}_n, u_n have the same laws on \mathscr{Z}_T , thus

(6.5.33)
$$\mathscr{L}(\tilde{u}_n)(\mathscr{C}([0,T];\mathbf{H}_n) = 1, \quad n \in \mathbb{N}.$$

Since $\mathscr{C}([0,T];V) \cap \mathscr{Z}_T$ and $L^2(0,T;D(A)) \cap \mathscr{Z}_T$ are Borel subsets of \mathscr{Z}_T (Lemma 6.5.5) and \tilde{u}_n and u_n have the same laws on \mathscr{Z}_T ; from (6.5.18) and (6.5.3), we have for $p \in [1,3]$

(6.5.34)
$$\sup_{n \in \mathbb{N}} \tilde{\mathbb{E}} \left(\sup_{0 \le s \le T} \| \tilde{u}_n(s) \|_{\mathbf{V}}^{2p} \right) \le C_1(p),$$

(6.5.35)
$$\sup_{n \in \mathbb{N}} \tilde{\mathbb{E}} \left[\int_0^T | \tilde{u}_n(s) |_{\mathbf{D}(\mathbf{A})}^2 ds \right] \le C_2(\| u_0 \|_{\mathbf{V}}^2).$$

Since $\mathscr{C}([0,T]; H_n)$ is continuously embedded in $L^4(0,T; L^4)$ and \tilde{u}_n , u_n have same law μ on $\mathscr{C}([0,T]; H_n)$, we have

$$\begin{split} \tilde{\mathbb{E}} \int_{0}^{T} \|\tilde{u}_{n}(s)\|_{L^{4}}^{4} ds &= \int_{\tilde{\Omega}} \left[\int_{0}^{T} \|\tilde{u}_{n}(s,\omega)\|_{L^{4}}^{4} ds \right] d\tilde{\mathbb{P}}(\omega) = \int_{L^{4}(0,T;L^{4})} \left[\int_{0}^{T} \|y\|_{L^{4}}^{4} ds \right] d\mu(y) \\ &= \int_{\mathscr{C}([0,T];\mathcal{H}_{n})} \left[\int_{0}^{T} \|y\|_{L^{4}}^{4} ds \right] d\mu(y) = \int_{L^{4}(0,T;L^{4})} \left[\int_{0}^{T} \|y\|_{L^{4}}^{4} ds \right] d\mu(y) \\ &= \int_{\Omega} \left[\int_{0}^{T} \|u_{n}(s,\omega)\|_{L^{4}}^{4} ds \right] d\mathbb{P}(\omega) = \mathbb{E} \int_{0}^{T} \|u_{n}(s)\|_{L^{4}}^{4} ds \,. \end{split}$$

Thus, by estimate (6.5.4) we infer

(6.5.36)
$$\sup_{n \in N} \tilde{\mathbb{E}} \int_0^T \|\tilde{u}_n(s)\|_{L^4}^4 dt \le C_3(|u_0|_{\mathrm{H}}^2).$$

By inequality (6.5.34) we infer that the sequence (\tilde{u}_n) contains a subsequence, still denoted by (\tilde{u}_n) convergent weakly in the space $L^2([0,T] \times \tilde{\Omega}; D(A))$. Since by (6.5.32) $\tilde{\mathbb{P}}$ -a.s $\tilde{u}_n \to \tilde{u}$ in \mathcal{Z}_T , we conclude that $\tilde{u} \in L^2([0,T] \times \tilde{\Omega}; D(A))$, i.e.

(6.5.37)
$$\widetilde{\mathbb{E}}\left[\int_0^T |\tilde{u}(s)|^2_{\mathrm{D}(\mathrm{A})} ds\right] < \infty.$$

Similarly by inequality (6.5.34) for p = 1 we can choose a subsequence of (\tilde{u}_n) convergent weak star in the space $L^2(\tilde{\Omega}; L^{\infty}(0, T; V))$ and using convergences (6.5.32), we infer that

(6.5.38)
$$\widetilde{\mathbb{E}}\left(\sup_{0\leq s\leq T}\|\widetilde{u}(s)\|_{\mathbf{V}}^{2}\right) < \infty.$$

For each $n \ge 1$, let us consider a process \tilde{M}_n with trajectories in $\mathscr{C}([0,T];H_n)$, in particular in $\mathscr{C}([0,T];H)$, defined by

(6.5.39)
$$\tilde{M}_{n}(t) = \tilde{u}_{n}(t) - P_{n}\tilde{u}(0) + \int_{0}^{t} A\tilde{u}_{n}(s)ds + \int_{0}^{t} B_{n}(\tilde{u}_{n}(s))ds + \int_{0}^{t} g_{n}(\tilde{u}_{n}(s))ds - \int_{0}^{t} f_{n}(\tilde{u}_{n}(s))ds, \qquad t \in [0,T].$$

Lemma 6.5.7. \tilde{M}_n is a square integrable martingale with respect to the filtration $\tilde{\mathbb{F}}_n = (\tilde{\mathscr{F}}_{n,t})$, where $\tilde{\mathscr{F}}_{n,t} = \sigma\{\tilde{u}_n(s), s \leq t\}$, with the quadratic variation

(6.5.40)
$$\langle \langle \tilde{M}_n \rangle \rangle_t = \int_0^t \|G_n(s, \tilde{u}_n(s))\|_{\mathcal{T}_2(\ell^2; \mathrm{H})}^2 ds$$

Proof. Indeed since \tilde{u}_n and u_n have the same laws, for all $s, t \in [0, T], s \leq t$, then for all bounded continuous functions h on $\mathscr{C}([0,s]; V_w)$, and all $\psi, \zeta \in V_{\gamma}$ (for $\gamma > \frac{d}{2}$), we have

(6.5.41)
$$\tilde{\mathbb{E}}\left[\langle \tilde{M}_n(t) - \tilde{M}_n(s), \psi \rangle h(\tilde{u}_{n|[0,s]})\right] = 0$$

and

(6.5.42)
$$\widetilde{\mathbb{E}}\Big[\Big(\langle \tilde{M}_n(t), \psi \rangle \langle \tilde{M}_n(t), \zeta \rangle - \langle \tilde{M}_n(s), \psi \rangle \langle \tilde{M}_n(s), \zeta \rangle \\ - \int_s^t \langle (G(\sigma, \tilde{u}_n(\sigma)))^* P_n \psi, (G(\sigma, \tilde{u}_n(\sigma)))^* P_n \zeta \rangle_{\ell^2} \, d\sigma \Big) \cdot h(\tilde{u}_{n|[0,s]}) \Big] = 0.$$

Lemma 6.5.8. Let us define a process \tilde{M} for $t \in [0, T]$ by

(6.5.43)

$$\tilde{M}(t) = \tilde{u}(t) - \tilde{u}(0) + \int_0^t A\tilde{u}(s) ds + \int_0^t B(\tilde{u}(s)) ds + \int_0^t \Pi(g(|\tilde{u}(s)|^2)\tilde{u}(s)) ds$$

$$-\int_0^t \Pi f(\tilde{u}(s)) ds.$$

Then \tilde{M} is an H-valued continuous process.

Proof. Since $\tilde{u} \in \mathcal{C}([0, T]; V)$ we just need to show that each of the remaining terms on the RHS of (6.5.43) are H-valued a.s. and well-defined.

Using the Cauchy-Schwarz inequality repeatedly and (6.5.37) we have the following inequalities

$$\tilde{\mathbb{E}} \int_0^T |\mathrm{A}\tilde{u}(s)|_{\mathrm{H}} \, ds \leq T^{1/2} \left(\tilde{\mathbb{E}} \int_0^T |\tilde{u}(s)|_{\mathrm{D}(\mathrm{A})}^2 \, ds \right)^{1/2} < \infty.$$

Since $H^{k,p} \hookrightarrow L^{\infty}$ for every k > d/p, hence there exists a C > 0 such that $||u||_{L^{\infty}} \le C ||u||_{H^{2,2}}$ for every $u \in H^{2,2}$. Thus by the Cauchy-Schwarz inequality, (6.5.37) and (6.5.38) we obtain the following estimate

$$\begin{split} &\tilde{\mathbb{E}} \int_{0}^{T} |B(\tilde{u}(s))|_{\mathrm{H}} \, ds \leq T^{1/2} \tilde{\mathbb{E}} \left(\int_{0}^{T} |\tilde{u}(s) \cdot \nabla \tilde{u}(s)|_{L^{2}}^{2} \, ds \right)^{1/2} \\ &\leq T^{1/2} \tilde{\mathbb{E}} \left(\int_{0}^{T} \|\tilde{u}(s)\|_{L^{\infty}}^{2} |\nabla \tilde{u}(s)|_{L^{2}}^{2} \, ds \right)^{1/2} \leq T^{1/2} \, C \, \tilde{\mathbb{E}} \left(\int_{0}^{T} |\tilde{u}(s)|_{\mathrm{D}(\mathrm{A})}^{2} \|\tilde{u}(s)\|_{\mathrm{V}}^{2} \, ds \right)^{1/2} \\ &\leq T^{1/2} \, C \, \left[\tilde{\mathbb{E}} \sup_{s \in [0,T]} \|\tilde{u}(s)\|_{\mathrm{V}}^{2} \right]^{1/2} \left[\tilde{\mathbb{E}} \int_{0}^{T} |\tilde{u}(s)|_{\mathrm{D}(\mathrm{A})}^{2} \, ds \right]^{1/2} < \infty. \end{split}$$

We know that for d = 3, $H^{1,2} \hookrightarrow L^6$, thus using (6.2.14), convergences (6.5.32) and (6.5.34), we get

$$\begin{split} \tilde{\mathbb{E}} &\int_0^T \left| \Pi g(|\tilde{u}(s)|^2) \tilde{u}(s) \right|_{\mathcal{H}} ds \leq \tilde{\mathbb{E}} \int_0^T |g(|\tilde{u}(s)|^2) \tilde{u}(s)|_{L^2} ds \leq \tilde{\mathbb{E}} \int_0^T \|\tilde{u}(s)\|_{L^6}^3 ds \\ &\leq C \tilde{\mathbb{E}} \int_0^T \|\tilde{u}(s)\|_{\mathcal{V}}^3 ds \leq C \left(\tilde{\mathbb{E}} \sup_{s \in [0,T]} \|\tilde{u}(s)\|_{\mathcal{V}}^4 \right)^{3/4} T < \infty. \end{split}$$

Using the assumptions (A1) and (6.5.38) we can show that

$$\begin{split} \tilde{\mathbb{E}} &\int_{0}^{T} |\Pi f(\tilde{u}(s))|_{\mathrm{H}} \, ds \leq \tilde{\mathbb{E}} \int_{0}^{T} |f(\tilde{u}(s))|_{L^{2}} \, ds \leq T^{1/2} \, \tilde{\mathbb{E}} \left(\int_{0}^{T} |f(\tilde{u}(s))|_{L^{2}}^{2} \, ds \right)^{1/2} \\ &\leq T^{1/2} \, C_{f} \left(\tilde{\mathbb{E}} \int_{0}^{T} |\tilde{u}(s)|_{\mathrm{H}}^{2} \, ds \right)^{1/2} < \infty \,. \end{split}$$

This concludes the proof of the lemma.

Lemma 6.5.9. Let us fix $\gamma > \frac{d}{2}$. If $u \in L^2(0,T;H) \cap L^4(0,T;L^4)$ and $(u_n)_{n \in \mathbb{N}}$ be a bounded sequence in $L^2(0,T;H) \cap L^4(0,T;L^4)$ such that $u_n \to u$ in $L^2(0,T;H_{loc})$, then for all $r,t \in [0,T]$ and all $\psi \in V_{\gamma}$:

(6.5.44)
$$\lim_{n \to \infty} \int_r^t \langle g(|u_n(s)|^2) u_n(s), \psi \rangle \, ds = \int_r^t \langle g(|u(s)|^2) u(s), \psi \rangle \, ds.$$

Here $\langle \cdot, \cdot \rangle$ denotes the duality pairing between V_{γ} and V'_{γ} .

Proof. We will prove the lemma in two steps.

Step I

Let us fix $\gamma > \frac{d}{2}$ and $r, t \in [0, T]$. Assume first that $\psi \in \mathcal{V}$. Then there exists a R > 0 such that $\operatorname{supp}(\psi)$ is a compact subset of \mathcal{O}_R . There exists a constant $C \ge 0$ such that

$$\begin{aligned} |\langle g(|u|^{2})u,\psi\rangle| &= \left| \int_{\mathcal{O}_{R}} g(|u(x)|^{2})u(x)\psi(x)dx \right| \le |g(|u|^{2})|_{L^{2}(\mathcal{O}_{R})} ||u||_{L^{2}(\mathcal{O}_{R})}^{2} ||\psi||_{L^{\infty}(\mathcal{O}_{R})} \\ (6.5.45) &\le \left| |u|^{2} \right|_{L^{2}(\mathcal{O}_{R})} ||u||_{L^{2}(\mathcal{O}_{R})} ||\psi||_{L^{\infty}} \le C ||u||_{L^{4}}^{2} ||u||_{L^{2}(\mathcal{O}_{R})} ||\psi||_{V_{\gamma}}, \qquad u \in \mathcal{H} \cap L^{4}, \end{aligned}$$

where we used (6.2.7) to establish the last inequality. We have

$$g(|u_n|^2)u_n - g(|u|^2)u = g(|u_n|^2)(u_n - u) + \left[g(|u_n|^2) - g(|u|^2)\right]u.$$

Thus using the estimate (6.5.45), the Hölder inequality, (6.2.15) and the Cauchy-Schwarz inequality, we obtain

$$\begin{split} \left| \int_{r}^{t} \langle g(|u_{n}(s)|^{2})u_{n}(s),\psi\rangle \, ds - \int_{r}^{t} \langle g_{n}(|u(s)|^{2})u(s),\psi\rangle \, ds \right| \\ &\leq \left| \int_{r}^{t} \langle g(|u_{n}(s)|^{2})(u_{n}(s) - u(s)),\psi\rangle \, ds \right| + \left| \int_{r}^{t} \langle \left(g(|u_{n}(s)|^{2}) - g(|u(s)|^{2}) \right)u(s),\psi\rangle \, ds \\ &\leq C \int_{r}^{t} \|u_{n}(s)\|_{L^{4}}^{2} |u_{n}(s) - u(s)|_{L^{2}(\mathcal{O}_{R})} \|\psi\|_{V_{Y}} \, ds \\ &+ 2 \int_{r}^{t} \left| \langle |u_{n}(s) - u(s)|(|u_{n}(s)| + |u(s)|)u(s),\psi\rangle \right| \, ds \\ &\leq C \|\psi\|_{V_{Y}} \int_{r}^{t} \|u_{n}(s)\|_{L^{4}}^{2} |u_{n}(s) - u(s)|_{L^{2}(\mathcal{O}_{R})} \, ds \\ &+ 2C \|\psi\|_{V_{Y}} \int_{s}^{t} |u_{n}(s) - u(s)|_{L^{2}(\mathcal{O}_{R})} \Big| u(s)[|u_{n}(s)| + |u(s)|] \Big|_{L^{2}(\mathcal{O}_{R})} \, ds. \end{split}$$

Thus by the Hölder inequality, we get

$$\begin{split} \left| \int_{r}^{t} \langle g(|u_{n}(s)|^{2})u_{n}(s),\psi\rangle \, ds - \int_{r}^{t} \langle g_{n}(|u(s)|^{2})u(s),\psi\rangle \, ds \right| \\ &\leq C \|\psi\|_{\mathcal{V}_{\gamma}} \bigg[|u_{n}|_{L^{4}(0,T;L^{4})}^{2} |u_{n} - u|_{L^{2}(0,T;L^{2}(\mathcal{O}_{R}))} \\ &\quad + 2|u_{n} - u|_{L^{2}(0,T;L^{2}(\mathcal{O}_{R}))} \left[\int_{r}^{t} |u(s)[|u_{n}(s)| + |u(s)|] \big|_{L^{2}(\mathcal{O}_{R})}^{2} \, ds \bigg]^{1/2} \bigg] \\ &\leq C \bigg[|u_{n}|_{L^{4}(0,T;L^{4})}^{2} + 2|u|_{L^{4}(0,T;L^{4})}^{2} \bigg(|u_{n}|_{L^{4}(0,T;L^{4})}^{2} + |u|_{L^{4}(0,T;L^{4})}^{2} \bigg)^{1/2} \bigg] |u_{n} - u|_{L^{2}(0,T;L^{2}(\mathcal{O}_{R}))} \|\psi\|_{\mathcal{V}_{\gamma}}. \end{split}$$

Since $u_n \to u$ in $L^2(0,T;H_{\text{loc}})$ we infer that (6.5.44) holds for every $\psi \in \mathcal{V}$. Step II

Let $\psi \in V_{\gamma}$ and $\varepsilon > 0$. Then there exists a $\psi_{\varepsilon} \in \mathcal{V}$ such that $\|\psi_{\varepsilon} - \psi\|_{V_{\gamma}} < \varepsilon$. Hence, we get

$$\begin{aligned} \left| \langle g(|u_{n}|^{2})u_{n} - g(|u|^{2})u, \psi \rangle \right| \\ &\leq \left| \langle g(|u_{n}|^{2})u_{n} - g(|u|^{2})u, \psi_{\varepsilon} \rangle \right| + \left| \langle g(|u_{n}|^{2})u_{n} - g(|u|^{2})u, \psi - \psi_{\varepsilon} \rangle \right| \\ &\leq \left| \langle g(|u_{n}|^{2})u_{n} - g(|u|^{2})u, \psi_{\varepsilon} \rangle \right| + \left[\|g(|u_{n}|^{2})u_{n}\|_{V_{\gamma}'} + \|g(|u|^{2})u\|_{V_{\gamma}'} \right] \|\psi - \psi_{\varepsilon}\|_{V_{\gamma}}. \end{aligned}$$

$$(6.5.46)$$

Since \mathcal{V} is dense in V_{γ} , (6.5.45) holds for all $\psi \in V_{\gamma}$. In particular, there exists a constant C > 0 such that

(6.5.47)
$$\|g(|u|^2)u\|_{V_{\gamma}} \le C \|u\|_{L^4}^2 \|u\|_{H}, \qquad u \in H \cap L^4.$$

Using (6.5.46), (6.5.47) and the Cauchy-Schwarz inequality we have following inequalities

$$\begin{split} \left| \int_{r}^{t} \langle g(|u_{n}(s)|^{2})u_{n}(s) - g(|u(s)|^{2})u(s),\psi \rangle ds \right| \\ &\leq \varepsilon C \int_{r}^{t} \left(\|u_{n}(s)\|_{L^{4}}^{2} |u_{n}(s)|_{\mathrm{H}} + \|u(s)\|_{L^{4}}^{2} |u(s)|_{\mathrm{H}} \right) ds \\ &+ \left| \int_{r}^{t} \langle g(|u_{n}(s)|^{2})u_{n}(s) - g(|u(s)|^{2})u(s),\psi_{\varepsilon} \rangle ds \right| \\ &\leq \varepsilon C \left[\|u_{n}\|_{L^{4}(0,T;L^{4})}^{2} \|u_{n}\|_{L^{2}(0,T;\mathrm{H})} + \|u\|_{L^{4}(0,T;L^{4})}^{2} \|u\|_{L^{2}(0,T;\mathrm{H})} \right] \\ &+ \left| \int_{r}^{t} \langle g(|u_{n}(s)|^{2})u_{n}(s) - g(|u(s)|^{2})u(s),\psi_{\varepsilon} \rangle ds \right|. \end{split}$$

Hence by **Step I** and the assumptions on u, u_n there exists a M > 0 such that

$$\lim_{n\to\infty} \sup_{n\to\infty} \left| \int_r^t \langle g(|u_n(s)|^2) u_n(s) - g(|u(s)|^2) u(s), \psi \rangle \, ds \right| \leq M\varepsilon.$$

Since $\varepsilon > 0$ is arbitrary we conclude the proof.

Corollary 6.5.10. Let us fix $\gamma > \frac{d}{2}$. If $u \in L^2(0,T;H) \cap L^4(0,T;L^4)$ and $(u_n)_{n \in \mathbb{N}}$ be a bounded sequence in $L^2(0,T;H) \cap L^4(0,T;L^4)$ such that $u_n \to u$ in $L^2(0,T;H_{loc})$, then for all $r, t \in [0,T]$ and all $\psi \in V_{\gamma}$:

(6.5.48)
$$\lim_{n \to \infty} \int_r^t \langle g(|u_n(s)|^2) u_n(s), P_n \psi \rangle \, ds = \int_r^t \langle g(|u(s)|^2) u(s), \psi \rangle \, ds.$$

Here $\langle \cdot, \cdot \rangle$ denotes the duality pairing between V_{γ} and V'_{γ} .

Proof. Let us fix $\gamma > \frac{d}{2}$ and take $r, t \in [0, T]$ and $\psi \in V_{\gamma}$. We have

$$\begin{split} \int_r^t \langle g(|u_n(s)|^2)u_n(s), P_n\psi \rangle \, ds &= \int_r^t \langle g(|u_n(s)|^2)u_n(s), P_n\psi - \psi \rangle \, ds + \int_r^t \langle g(|u_n(s)|^2)u_n(s), \psi \rangle \, ds \\ &:= I_1(n) + I_2(n). \end{split}$$

We will consider each of these integrals individually. Using the estimate from (6.5.47), we have

$$|I_{1}(n)| \leq \int_{r}^{t} \|g(|u_{n}(s)|^{2})u_{n}(s)\|_{V_{\gamma}^{\prime}} \|P_{n}\psi - \psi\|_{V_{\gamma}} ds$$

$$\leq \|P_{n}\psi - \psi\|_{V_{\gamma}} \int_{r}^{t} \|u_{n}(s)\|_{L^{4}}^{2} |u_{n}(s)|_{H} ds.$$

Since the sequence $(u_n)_{n \in \mathbb{N}}$ is bounded in $L^2(0,T;H) \cap L^4(0,T;L^4)$ and $P_n \psi \to \psi$ in V_{γ} , we infer

$$\lim_{n \to \infty} I_1(n) = 0$$

By Lemma 6.5.9, we conclude

$$\lim_{n\to\infty}I_2(n)=\int_r^t\langle g(|u(s)|^2)u(s),\psi\rangle\,ds.$$

Lemma 6.5.11. For all $s, t \in [0, T]$ such that $s \le t$ and $\gamma > \frac{d}{2}$:

- (a) $\lim_{n\to\infty} \langle \tilde{u}_n(t), P_n \psi \rangle = \langle \tilde{u}(t), \psi \rangle, \quad \tilde{\mathbb{P}}\text{-}a.s., \quad \psi \in \mathbf{V},$
- (b) $\lim_{n\to\infty} \int_s^t \langle A\tilde{u}_n(\sigma), P_n\psi \rangle_H ds = \int_s^t \langle A\tilde{u}(\sigma), \psi \rangle_H d\sigma$, $\tilde{\mathbb{P}}$ -a.s., $\psi \in H$,
- (c) $\lim_{n\to\infty}\int_s^t \langle B(\tilde{u}_n(\sigma)), P_n\psi\rangle d\sigma = \int_s^t \langle B(\tilde{u}(\sigma)),\psi\rangle d\sigma, \quad \tilde{\mathbb{P}}\text{-}a.s., \quad \psi \in \mathcal{V}_{\gamma},$
- (d) $\lim_{n\to\infty}\int_s^t \langle g(|\tilde{u}_n(\sigma)|^2)\tilde{u}_n(\sigma), P_n\psi\rangle \, d\sigma = \int_s^t \langle g(|\tilde{u}(\sigma)|^2)\tilde{u}(\sigma),\psi\rangle \, d\sigma, \ \tilde{\mathbb{P}}\text{-}a.s., \quad \psi \in \mathcal{V}_\gamma,$
- (e) $\lim_{n\to\infty}\int_s^t \langle f(\tilde{u}_n(\sigma)), P_n\psi\rangle d\sigma = \int_s^t \langle f(\tilde{u}(\sigma)), \psi\rangle d\sigma, \quad \tilde{\mathbb{P}}\text{-}a.s., \quad \psi \in \mathcal{V}_{\gamma},$

where $\langle\cdot,\cdot\rangle$ denotes the duality pairing between appropriate spaces.

Proof. Let us fix $s, t \in [0, T]$, $s \le t$ and $\gamma > \frac{d}{2}$. By (6.5.32) we know that

(6.5.49)
$$\tilde{u}_n \to \tilde{u} \text{ in } \mathscr{C}([0,T]; \mathbf{U}') \cap L^2_w(0,T; \mathbf{D}(\mathbf{A})) \cap L^2(0,T; \mathbf{H}_{\mathsf{loc}}) \cap \mathscr{C}([0,T]; V_w), \quad \mathbb{P}\text{-a.s.}$$

Let $\psi \in V$. Since $\tilde{u}_n \to \tilde{u}$ in $\mathscr{C}([0, T]; V_w) \tilde{\mathbb{P}}$ -a.s., from (6.5.34) \tilde{u}_n is uniformly bounded in $\mathscr{C}([0, T]; V_w)$ and $P_n \psi \to \psi$ in V, thus

$$\lim_{n \to \infty} \langle \tilde{u}_n(t), P_n \psi \rangle - \langle \tilde{u}(t), \psi \rangle = \lim_{n \to \infty} \langle \tilde{u}_n(t) - \tilde{u}(t), \psi \rangle + \lim_{n \to \infty} \langle \tilde{u}_n(t), P_n \psi - \psi \rangle = 0 \quad \tilde{\mathbb{P}}\text{-a.s.}$$

Hence, we infer that assertion (a) holds.

Let $\psi \in H$. Since by (6.5.49) $\tilde{u}_n \to \tilde{u}$ in $L^2_w(0,T; D(A)) \tilde{\mathbb{P}}$ -a.s., from (6.5.35) \tilde{u}_n is uniformly bounded in $L^2_w(0,T; D(A))$ and $P_n \psi \to \psi$ in H. Thus, we have $\tilde{\mathbb{P}}$ -a.s.,

$$\lim_{n \to \infty} \int_{s}^{t} \langle A\tilde{u}_{n}(\sigma), P_{n}\psi \rangle_{H} d\sigma - \int_{s}^{t} \langle A\tilde{u}(\sigma), \psi \rangle_{H} d\sigma$$
$$= \lim_{n \to \infty} \int_{s}^{t} \langle A\tilde{u}_{n}(\sigma) - A\tilde{u}(\sigma), \psi \rangle_{H} d\sigma + \lim_{n \to \infty} \int_{s}^{t} \langle A\tilde{u}_{n}(\sigma), P_{n}\psi - \psi \rangle_{H} d\sigma \to 0.$$

Hence, we have shown that assertion (b) is true.

For every $\psi \in V_{\gamma}$ assertion (*c*) follows directly from [23, Lemma B.1] and a modification of Corollary 6.5.10.

By (6.5.49) $\tilde{u}_n \to \tilde{u}$ in $L^2(0,T;H_{\text{loc}})$. From Lemma 6.5.1, convergences (6.5.49) and estimate (6.5.36) the sequence (\tilde{u}_n) is bounded in $L^2(0,T;H) \cap L^4(0,T;L^4)$ and $\tilde{u} \in L^2(0,T;H) \cap L^4(0,T;L^4)$. Thus, using Corollary 6.5.10 we infer that (*d*) holds for every $\psi \in V_{\gamma}$.

Now we are left to deal with (*e*). Let $\psi \in V_{\gamma}$,

$$\begin{split} &\int_{s}^{t} \langle f(\tilde{u}_{n}(\sigma)), P_{n}\psi \rangle \, ds - \int_{0}^{t} \langle f(\tilde{u}(\sigma), \psi \rangle \, d\sigma \\ &= \int_{0}^{t} \langle f(\tilde{u}_{n}(\sigma)) - f(\tilde{u}(\sigma)), \psi \rangle \, d\sigma + \int_{0}^{t} \langle f(\tilde{u}_{n}(\sigma)), P_{n}\psi - \psi \rangle \, d\sigma \end{split}$$

Since $V_{\gamma} \hookrightarrow H$, by the Cauchy-Schwarz inequality we have

$$\begin{split} &\int_{s}^{t} \langle f(\tilde{u}_{n}(\sigma)), P_{n}\psi\rangle ds - \int_{0}^{t} \langle f(\tilde{u}(\sigma),\psi\rangle d\sigma \\ &\leq \int_{0}^{t} \langle f(\tilde{u}_{n}(\sigma)) - f(\tilde{u}(\sigma)),\psi\rangle d\sigma + \int_{s}^{t} \|f(u_{n}(s))\|_{\mathbf{V}_{\gamma}^{\prime}} \|P_{n}\psi - \psi\|_{\mathbf{V}_{\gamma}} d\sigma \\ &\leq \int_{0}^{t} \langle f(\tilde{u}_{n}(\sigma)) - f(\tilde{u}(\sigma)),\psi\rangle d\sigma + C_{f} \|P_{n}\psi - \psi\|_{\mathbf{V}_{\gamma}} \int_{s}^{t} \|\tilde{u}_{n}(\sigma)\|_{\mathbf{H}} d\sigma \\ &:= I_{1}(n) + I_{2}(n). \end{split}$$

Since $\tilde{u}_n \to \tilde{u}$ in $L^2(0, T; H_{\text{loc}})$ and \tilde{u}_n is a bounded sequence in $L^2(0, T; H)$, $I_1(n)$ can be shown to converge to zero as $n \to \infty$ following the methodology of Lemma 6.5.9 and Corollary 6.5.10. Since $P_n \psi \to \psi$ in V_{γ} , $I_2(n) \to 0$ as $n \to \infty$. This completes the proof of Lemma 6.5.11.

The proofs of Lemmas 6.5.12, 6.5.15 and 6.5.17 follow the similar methodology as that of Lemmas 5.6 - 5.8 [23] and Lemmas 5.5.9 - 5.5.12.

Let *h* be the bounded continuous function on $\mathscr{C}([0,T]; \mathbf{U}')$ and $\tilde{\mathbb{F}} = (\tilde{\mathscr{F}}_t) = \sigma\{\tilde{u}(s), s \leq t\}$ be the filtration of sigma fields generated by the process \tilde{u} .

Lemma 6.5.12. For all $s, t \in [0, T]$, such that $s \le t$ and all $\psi \in V_{\gamma}$:

(6.5.50)
$$\lim_{n \to \infty} \tilde{\mathbb{E}}\left[\langle \tilde{M}_n(t) - \tilde{M}_n(s), \psi \rangle h(\tilde{u}_{n|[0,s]}) \right] = \tilde{\mathbb{E}}\left[\langle \tilde{M}(t) - \tilde{M}(s), \psi \rangle h(\tilde{u}_{|[0,s]}) \right].$$

Proof. Let us fix $s, t \in [0, T], s \le t$ and $\psi \in V_{\gamma}$. By identity (6.5.39) we have

$$\begin{split} \langle \tilde{M}_{n}(t) - \tilde{M}_{n}(s), \psi \rangle &= \langle \tilde{u}_{n}(t) - \tilde{u}_{n}(s), P_{n}\psi \rangle + \int_{s}^{t} \langle A\tilde{u}_{n}(\sigma), P_{n}\psi \rangle \, d\sigma \\ &+ \int_{s}^{t} \langle B(\tilde{u}_{n}(\sigma)), P_{n}\psi \rangle \, d\sigma + \int_{s}^{t} \langle g(|\tilde{u}_{n}(\sigma)|^{2})\tilde{u}_{n}(\sigma), P_{n}\psi \rangle \, d\sigma - \int_{s}^{t} \langle f(\tilde{u}_{n}(\sigma)), P_{n}\psi \rangle \, d\sigma. \end{split}$$

By Lemma 6.5.11, we infer that

(6.5.51)
$$\lim_{n \to \infty} \langle \tilde{M}_n(t) - \tilde{M}_n(s), \psi \rangle = \langle \tilde{M}(t) - \tilde{M}(s), \psi \rangle, \quad \tilde{\mathbb{P}}\text{-a.s.}$$

In order to prove (6.5.50) we first observe that since $\tilde{u}_n \to \tilde{u}$ in \mathcal{Z}_T , in particular in $\mathscr{C}([0,T]; U')$ and h is a bounded continuous function on $\mathscr{C}([0,T]; U')$, we get

(6.5.52)
$$\lim_{n \to \infty} h(\tilde{u}_{n|[0,s]}) = h(\tilde{u}_{|[0,s]}),$$

and

(6.5.53)
$$\sup_{n \in \mathbb{N}} \|h(\tilde{u}_{n|[0,s]})\|_{L^{\infty}} < \infty.$$

Let us define a sequence of $\mathbb{R}-valued$ random variables :

$$f_n(\omega) := \left[\langle \tilde{M}_n(t,\omega), \psi \rangle - \langle \tilde{M}_n(s,\omega), \psi \rangle \right] h(\tilde{u}_{n|[0,s]}), \quad \omega \in \tilde{\Omega}.$$

We will prove that the functions $\{f_n\}_{n\in\mathbb{N}}$ are uniformly integrable in order to apply the Vitali theorem. We claim that

(6.5.54)
$$\sup_{n \in \mathbb{N}} \tilde{\mathbb{E}}[|f_n|^2] < \infty.$$

Since $H \hookrightarrow V'_{\gamma}$ then by the Cauchy-Schwarz inequality, for each $n \in \mathbb{N}$ we have

(6.5.55)
$$\tilde{\mathbb{E}}[|f_n|^2] \le 2c \|h \circ \tilde{u}_n\|_{L^{\infty}}^2 |\psi|_{V_{\gamma}}^2 \tilde{\mathbb{E}}\left[|\tilde{M}_n(t)|_H^2 + |\tilde{M}_n(s)|_H^2\right]$$

Since \tilde{M}_n is a continuous martingale with quadratic variation defined in (6.5.40), by the Burkholder-Davis-Gundy inequality we obtain

(6.5.56)
$$\tilde{\mathbb{E}}\left[\sup_{t\in[0,T]}|\tilde{M}_{n}(t)|_{\mathrm{H}}^{2}\right] \leq c\tilde{\mathbb{E}}\left[\int_{0}^{T}\|G_{n}(\sigma,\tilde{u}_{n}(\sigma))\|_{\mathcal{T}_{2}(\ell^{2};\mathrm{H})}^{2}d\sigma\right]$$

Since $P_n: H \to H$ is a contraction and by Lemma 6.2.1, (6.5.18) for p = 1, we have

$$\begin{split} & \tilde{\mathbb{E}}\left[\int_{0}^{T}\|G_{n}(\sigma,\tilde{u}_{n}(\sigma))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2}d\sigma\right] \leq \tilde{\mathbb{E}}\left[\int_{0}^{T}\|G(\sigma,\tilde{u}_{n}(\sigma))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2}d\sigma\right] \\ & \leq \tilde{\mathbb{E}}\left[\int_{0}^{T}\frac{1}{4}|\nabla\tilde{u}_{n}(\sigma)|_{L^{2}}^{2}d\sigma\right] \leq \tilde{\mathbb{E}}\left[\sup_{\sigma\in[0,T]}\|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2}\right]T < \infty. \end{split}$$

Then by (6.5.55) and (6.5.57) we see that (6.5.54) holds. Since the sequence $\{f_n\}_{n \in \mathbb{N}}$ is uniformly integrable and by (6.5.51) it is $\tilde{\mathbb{P}}$ -a.s. point-wise convergent, then application of the Vitali Theorem completes the proof of the lemma.

Remark 6.5.13. Using the Burkholder-Davis-Gundy inequality we have proved a stronger claim (6.5.56) than what we needed.

From Lemma 6.5.7 and Lemma 6.5.12 we have the following corollary.

Corollary 6.5.14. For all $s, t \in [0, T]$ such that $s \le t$:

$$\mathbb{E}\left(\tilde{M}(t)-\tilde{M}(s)\big|\tilde{\mathscr{F}}_t\right)=0.$$

Lemma 6.5.15. For all $s, t \in [0, T]$ such that $s \le t$ and all $\psi, \zeta \in V_{\gamma}$:

$$\begin{split} &\lim_{n\to\infty} \tilde{\mathbb{E}}\Big[\Big(\langle \tilde{M}_n(t),\psi\rangle\langle \tilde{M}_n(t),\zeta\rangle-\langle \tilde{M}_n(s),\psi\rangle\langle \tilde{M}_n(s),\zeta\rangle\Big)h(\tilde{u}_{n\mid[0,s]})\Big]\\ &=\tilde{\mathbb{E}}\Big[\Big(\langle \tilde{M}(t),\psi\rangle\langle \tilde{M}(t),\zeta\rangle-\langle \tilde{M}(s),\psi\rangle\langle \tilde{M}(s),\zeta\rangle\Big)h(\tilde{u}_{\mid[0,s]})\Big],\end{split}$$

where $\langle \cdot, \cdot \rangle$ denotes the appropriate dual pairing.

Proof. Let us fix $s, t \in [0, T]$ such that $s \le t$ and $\psi, \zeta \in V_{\gamma}$ and define \mathbb{R} -valued random variables f_n and f by

$$\begin{split} f_{n}(\omega) &:= \Big(\langle \tilde{M}_{n}(t,\omega),\psi \rangle \langle \tilde{M}_{n}(t,\omega),\zeta \rangle - \langle \tilde{M}_{n}(s,\omega),\psi \rangle \langle \tilde{M}_{n}(s,\omega),\zeta \rangle \Big) h(\tilde{u}_{n\mid [0,s]}(\omega)), \\ f(\omega) &:= \Big(\langle \tilde{M}(t,\omega),\psi \rangle \langle \tilde{M}(t,\omega),\zeta \rangle - \langle \tilde{M}(s,\omega),\psi \rangle \langle \tilde{M}(s,\omega),\zeta \rangle \Big) h(\tilde{u}_{\mid [0,s]}(\omega)), \quad \omega \in \tilde{\Omega}. \end{split}$$

By Lemma 6.5.11 or more precisely by (6.5.51) and (6.5.52) we infer that $\lim_{n\to\infty} f_n(\omega) = f(\omega)$, for $\tilde{\mathbb{P}}$ almost all $\omega \in \tilde{\Omega}$. We will prove that the functions $\{f_n\}_{n\in\mathbb{N}}$ are uniformly integrable. We claim that for some r > 1,

$$(6.5.58) \qquad \qquad \sup_{n \in \mathbb{N}} \tilde{\mathbb{E}}\left[|f_n|^r \right] < \infty$$

For each $n \in \mathbb{N}$ as before we have

(6.5.59)
$$\tilde{\mathbb{E}}\left[\left|f_{n}\right|^{r}\right] \leq C \|h \circ \tilde{u}_{n}\|_{L^{\infty}}^{r} \|\psi\|_{V_{Y}}^{r} \|\zeta\|_{V_{Y}}^{r} \tilde{\mathbb{E}}\left[\left|\tilde{M}_{n}(t)\right|_{H}^{2r} + |\tilde{M}_{n}(s)|_{H}^{2r}\right].$$

Since \tilde{M}_n is a continuous martingale with quadratic variation defined in (6.5.40), by the Burkholder-Davis-Gundy inequality we obtain

(6.5.60)
$$\tilde{\mathbb{E}}\left[\sup_{t\in[0,T]}|\tilde{M}_{n}(t)|_{\mathrm{H}}^{2r}\right] \leq c\tilde{\mathbb{E}}\left[\int_{0}^{T}\|G_{n}(\sigma,\tilde{u}_{n}(\sigma))\|_{\mathcal{T}_{2}(\ell^{2};\mathrm{H})}^{2}d\sigma\right]^{r}$$

Since P_n : H \rightarrow H is a contraction and by Lemma 6.2.1 we have

$$\begin{split} \tilde{\mathbb{E}}\left[\int_{0}^{T}\|G_{n}(\sigma,\tilde{u}_{n}(\sigma))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2}d\sigma\right]^{r} &\leq \tilde{\mathbb{E}}\left[\int_{0}^{T}\|G(\sigma,\tilde{u}_{n}(\sigma))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2}d\sigma\right]^{r} \\ &\leq \tilde{\mathbb{E}}\left[\int_{0}^{T}\frac{1}{4}|\nabla\tilde{u}_{n}(\sigma)|_{L^{2}}^{2}d\sigma\right]^{r} \leq C_{r}T^{r-1}\tilde{\mathbb{E}}\left[\int_{0}^{T}\|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2r}d\sigma\right] \\ (6.5.61) &\leq C_{r}\tilde{\mathbb{E}}\left[\sup_{\sigma\in[0,T]}\|\tilde{u}_{n}(\sigma)\|_{\mathrm{V}}^{2r}\right]T^{r}. \end{split}$$

Thus, if $r \in [1,3]$ then, by (6.5.18), (6.5.53) and (6.5.59) - (6.5.61) we infer that (6.5.58) holds. Hence by application of the Vitali theorem

(6.5.62)
$$\lim_{n \to \infty} \tilde{\mathbb{E}}[f_n] = \tilde{\mathbb{E}}[f].$$

We will be using the following notations in the following lemmas. $V'(\mathcal{O}_R)$ is the dual space to $V(\mathcal{O}_R)$, where

$$V(\mathcal{O}_R)$$
: = the closure of $\mathcal{V}(\mathcal{O}_R)$ in $H^1(\mathcal{O}_R, \mathbb{R}^3)$

where $\mathcal{V}(\mathcal{O}_R)$ denotes the space of all divergence free vector fields of class \mathscr{C}^{∞} with compact supports contained in \mathcal{O}_R . We recall that $H_{\mathcal{O}_R}$ is the space of restrictions to the subset \mathcal{O}_R of elements of the space H i.e.,

$$\mathbf{H}_{\mathcal{O}_R} := \left\{ u_{\mid \mathcal{O}_R} \colon u \in \mathbf{H} \right\},\,$$

with the scalar product defined by

$$\langle u, \mathbf{v} \rangle_{\mathbf{H}_{\mathcal{O}_R}} := \int_{\mathcal{O}_R} u(x) \mathbf{v}(x) dx, \quad u, \mathbf{v} \in \mathbf{H}_{\mathcal{O}_R}.$$

Lemma 6.5.16. The map $G: \mathbb{H}_{\mathcal{O}_R} \to \mathcal{T}_2(\ell^2; \mathbb{V}'(\mathcal{O}_R))$ given by (6.2.12) is well defined and there exists some constant $C_R > 0$ such that

(6.5.63)
$$\|G(u)\|_{\mathcal{F}_{2}(\ell^{2}; V'(\mathcal{O}_{R}))} \leq C_{R} \|u\|_{H_{\mathcal{O}_{P}}}, \quad u \in \mathbf{H}.$$

Moreover, for every $\psi \in \mathcal{V}$ the mapping $H \ni u \mapsto \langle G(u), \psi \rangle \in \ell^2$ is continuous, if in the space H we consider the Fréchet topology inherited from the space $L^2_{loc}(\mathbb{R}^3, \mathbb{R}^3)$.

Proof. Let $\sigma = (\sigma^1, ..., \sigma^d) : \overline{\mathcal{O}} \to \mathbb{R}^d$ and fix R > 0. Let $u \in \mathcal{V}(\mathcal{O}_R)$. Then

(6.5.64)
$$\sum_{j=1}^{d} \frac{\partial}{\partial x_j} (\sigma^j u) = \sum_{j=1}^{d} \left(\frac{\partial \sigma^j}{\partial x_j} u + \sigma^j \frac{\partial u}{\partial x_j} \right) = (\operatorname{div} \sigma) u + \sum_{j=1}^{d} \sigma^j \frac{\partial u}{\partial x_j}.$$

Let $v \in \mathcal{V}(\mathcal{O}_R)$. Since v on the boundary $\partial \mathcal{O}_R$ is equal to zero, thus using the integration by parts formula, we obtain for $v \in \mathcal{V}(\mathcal{O}_R)$

$$\int_{\mathcal{O}} \left(\sum_{j=1}^{d} \sigma^{j} \frac{\partial u}{\partial x_{j}} \right) \mathbf{v} \, dx = \sum_{j=1}^{d} \int_{\mathcal{O}} \frac{\partial}{\partial x_{j}} (\sigma^{j} u) \mathbf{v} \, dx - \int_{\mathcal{O}} (\operatorname{div} \sigma) u \, \mathbf{v} \, dx$$
$$= -\sum_{j=1}^{d} \int_{\mathcal{O}} (\sigma^{j} u) \frac{\partial \mathbf{v}}{\partial x_{j}} \, dx - \int_{\mathcal{O}} (\operatorname{div} \sigma) u \, \mathbf{v} \, dx.$$

Using the Hölder inequality, we obtain

$$(6.5.65) \qquad \qquad \left| \int_{\mathcal{O}_R} \left(\sum_{j=1}^d \sigma^j \frac{\partial u}{\partial x_j} \right) \mathbf{v} \, dx \right| \le \|\sigma\|_{L^{\infty}} \|u\|_{\mathbf{H}_{\mathcal{O}_R}} \|\mathbf{v}\|_{\mathbf{V}(\mathcal{O}_R)} + \|\operatorname{div}\sigma\|_{L^{\infty}} \|u\|_{\mathbf{H}_{\mathcal{O}_R}} \|\mathbf{v}\|_{\mathbf{V}(\mathcal{O}_R)}$$

Therefore, if we define a linear functional \hat{B}_R by

$$\hat{B}_R \mathbf{v} := \int_{\mathcal{O}_R} \Big(\sum_{j=1}^d \sigma^j \frac{\partial u}{\partial x_j} \Big) \mathbf{v} \, dx, \qquad \mathbf{v} \in \mathcal{V}(\mathcal{O}_R),$$

we infer that it is bounded in the norm of the space $V(\mathcal{O}_R)$. Thus it can be uniquely extended to a linear bounded functional (denoted also by \hat{B}_R) on $V(\mathcal{O}_R)$. Moreover, by estimate (6.5.65) we have the following inequality

$$\|\hat{B}_R\|_{\mathcal{V}(\mathcal{O}_R)} \le \left(\|\sigma\|_{L^{\infty}} + \|\operatorname{div}\sigma\|_{L^{\infty}}\right)|u|_{\mathcal{H}_{\mathcal{O}_R}}$$

or equivalently

$$(6.5.66) \qquad \qquad \|(\sigma \cdot \nabla)u\|_{\mathcal{V}(\mathcal{O}_R)} \le \left(\|\sigma\|_{L^{\infty}} + \|\operatorname{div}\sigma\|_{L^{\infty}}\right) \cdot |u|_{\mathcal{H}_{\mathcal{O}_R}}$$

Since by equality (6.2.12), $G(u)(e_j) = \prod [(\sigma_j \cdot \nabla)u]$, for some orthonormal basis $\{e_j\}_{j=1}^{\infty}$ of ℓ^2 , we get by estimate (6.5.66)

$$\|G(u)\|_{\mathcal{T}_{2}(\ell^{2}; \mathcal{V}'(\mathcal{O}_{R}))} = \left[\sum_{j=1}^{\infty} \|G(u)(e_{j})\|_{\mathcal{V}'(\mathcal{O}_{R})}^{2}\right]^{1/2} \leq \left(\|\sigma\|_{\ell^{2}} + \|\operatorname{div}\sigma\|_{\ell^{2}}\right) \cdot |u|_{\operatorname{H}_{\mathcal{O}_{R}}}.$$

Therefore, using the assumption (A2), $G(u) \in \mathcal{T}_2(\ell^2, V'(\mathcal{O}_R))$ and

$$\|G(u)\|_{\mathcal{T}_2(\ell^2, \mathcal{V}'(\mathcal{O}_R))} \leq C_R \cdot |u|_{\mathcal{H}_{\mathcal{O}_P}}.$$

By estimate (6.5.63) and the continuity of the embedding $\mathcal{T}_2(\ell^2, \mathsf{V}'(\mathcal{O}_R)) \hookrightarrow \mathcal{L}(\ell^2, \mathsf{V}'(\mathcal{O}_R))$, we obtain

$$\|G(u)y\|_{\mathcal{V}(\mathcal{O}_R)} \leq C(R) |u|_{\mathcal{H}_{\mathcal{O}_R}} \|y\|_{\ell^2}, \qquad u \in \mathcal{H}, \quad y \in \ell^2$$

for some constant C(R) > 0. Thus, for any $\psi \in V(\mathcal{O}_R)$

(6.5.67)
$$|(G(u)y)\psi| \le C(R)|u|_{\mathcal{H}_{\mathcal{O}_R}} ||y||_{\ell^2} ||\psi||_{\mathcal{V}(\mathcal{O}_R)}, \qquad u \in \mathcal{H}, \quad y \in \ell^2.$$

Now we identify $_{V'}\langle G(\cdot),\psi\rangle_V$ with the mapping $\psi^{**}G\colon H\to (\ell^2)'$ defined by

$$(\psi^{**}G(u))y := (G(u)y)\psi \in \mathbb{R}, \quad u \in \mathcal{H}, \quad y \in \ell^2.$$

Thus, from the inequality (6.5.67), we infer that

(6.5.68)
$$\|\psi^{**}G(u)\|_{\ell^2} \le C(R) \|\psi\|_{V} |u|_{H_{\mathcal{O}_R}}.$$

Therefore, if we fix $\psi \in \mathcal{V}$ then, there exists $R_0 > 0$ such that $\operatorname{supp} \psi$ is a compact subset of \mathcal{O}_{R_0} . Since *G* is linear, estimate (6.5.68) with $R := R_0$ yields that the mapping

$$L^2_{\operatorname{loc}}(\mathbb{R}^3,\mathbb{R}^3) \supset \operatorname{H} \ni u \mapsto \psi^{**}G(u) \in \ell^2$$

is continuous in the Fréchet topology inherited on the space H from the space $L^2_{loc}(\mathbb{R}^3, \mathbb{R}^3)$, concluding the proof of the lemma.

Lemma 6.5.17 (Convergence of quadratic variations). For any $s, t \in [0,T]$ and $\psi, \zeta \in V_{\gamma}$, we have

$$\lim_{n \to \infty} \tilde{\mathbb{E}} \left[\left(\int_{s}^{t} \langle (G(\sigma, \tilde{u}_{n}(\sigma)))^{*} P_{n} \psi, (G(\sigma, \tilde{u}_{n}(\sigma)))^{*} P_{n} \zeta \rangle_{\ell^{2}} d\sigma \right) \cdot h(\tilde{u}_{n \mid [0, s]}) \right]$$
$$= \tilde{\mathbb{E}} \left[\left(\int_{s}^{t} \langle (G(\sigma, \tilde{u}(\sigma)))^{*} \psi, (G(\sigma, \tilde{u}(\sigma))^{*} \zeta \rangle_{\ell^{2}} d\sigma \right) \cdot h(\tilde{u}_{\mid [0, s]}) \right],$$

where $\langle \cdot, \cdot \rangle_{\ell^2}$ is the inner product in ℓ^2 .

Proof. Let us fix $\psi, \zeta \in V_{\gamma}$ and define a sequence of random variables by

$$f_n(\omega) \colon = \left(\int_s^t \langle (G(\sigma, \tilde{u}_n(\sigma, \omega)))^* P_n \psi, (G(\sigma, \tilde{u}_n(\sigma, \omega)))^* P_n \zeta \rangle_{\ell^2} d\sigma \right) \cdot h(\tilde{u}_{n|[0,s]}), \quad \omega \in \tilde{\Omega}.$$

We will prove that the functions are uniformly integrable and convergent $\tilde{\mathbb{P}}$ -a.s. We will prove that for some r > 1,

(6.5.69)
$$\sup_{n \in \mathbb{N}} \tilde{\mathbb{E}}[|f_n|^r] < \infty.$$

For some c > 0 we have the following inequalities

$$|(G(\sigma,\tilde{u}_n(\sigma,\omega)))^*P_n\psi|_{\ell^2} \le ||G(\sigma,\tilde{u}_n(\sigma,\omega))||_{\mathscr{L}(\ell^2;\mathrm{H})}|P_n\psi|_{\mathrm{H}} \le \frac{1}{2}|\nabla\tilde{u}_n(\sigma,\omega)|_{L^2}|\psi|_{\mathrm{H}},$$

and thus

$$\begin{split} \mathbb{E}|f_{n}|^{r} &= \mathbb{E}\left|\left(\int_{s}^{t} \langle (G(\sigma,\tilde{u}_{n}(\sigma)))^{*}P_{n}\psi, (G(\sigma,\tilde{u}_{n}(\sigma)))^{*}P_{n}\zeta\rangle_{\ell^{2}}d\sigma\right) \cdot h(\tilde{u}_{n\mid[0,s]})\right|^{r} \\ &\leq \|h\circ\tilde{u}_{n}\|_{L^{\infty}}^{r} \mathbb{E}\left(\int_{s}^{t} |(G(\sigma,\tilde{u}_{n}(\sigma,\omega)))^{*}P_{n}\psi|_{\ell^{2}}|(G(\sigma,\tilde{u}_{n}(\sigma,\omega)))^{*}P_{n}\zeta|_{\ell^{2}}d\sigma\right)^{r} \\ &\leq c^{4}\|h\circ\tilde{u}_{n}\|_{L^{\infty}}^{r}|\psi|_{\mathrm{H}}^{r}|\zeta|_{\mathrm{H}}^{r} \mathbb{E}\left(\int_{s}^{t} \|\tilde{u}_{n}(\sigma,\omega)\|_{\mathrm{V}}^{2}d\sigma\right)^{r}. \end{split}$$

Using the Hölder inequality, we get

$$\mathbb{E}\left(\int_{s}^{t}\|\tilde{u}_{n}(\sigma,\omega)\|_{\mathbf{V}}^{2}d\sigma\right)^{r} \leq (t-s)^{r-1}\mathbb{E}\int_{s}^{t}\|\tilde{u}_{n}(\sigma,\omega)\|_{\mathbf{V}}^{2r}d\sigma \leq C\mathbb{E}\left(\sup_{\sigma\in[0,T]}\|\tilde{u}_{n}(\sigma,\omega)\|_{\mathbf{V}}^{2r}\right)$$

for some C > 0. Thus

$$\mathbb{E}|f_n|^r \le \tilde{C}\mathbb{E}\left(\sup_{\sigma \in [0,T]} \|\tilde{u}_n(\sigma,\omega)\|_{\mathbf{V}}^{2r}\right)$$

for some $\tilde{C} > 0$. Hence by (6.5.18) for $r \in (1,3]$

$$\sup_{n\geq 1} \tilde{\mathbb{E}} |f_n|^r \leq \tilde{C} \sup_{n\geq 1} \tilde{\mathbb{E}} \left[\sup_{\sigma\in[0,T]} \|\tilde{u}_n(\sigma,\omega)\|_{V}^{2r} \right] \leq \tilde{C}C_1(r) < \infty,$$

inferring (6.5.69).

Pointwise convergence : Next, we have to prove the following pointwise convergence for a fix $\omega \in \tilde{\Omega}$, i.e. we will show that for a fix $\omega \in \tilde{\Omega}$

(6.5.70)
$$\lim_{n \to \infty} \int_{s}^{t} \left\langle (G(\sigma, \tilde{u}_{n}(\sigma, \omega)))^{*} P_{n} \psi, (G(\sigma, \tilde{u}_{n}(\sigma, \omega)))^{*} P_{n} \zeta \right\rangle_{\ell^{2}} d\sigma$$
$$= \int_{s}^{t} \left\langle (G(\sigma, \tilde{u}(\sigma, \omega)))^{*} \psi, (G(\sigma, \tilde{u}(\sigma, \omega))^{*} \zeta \right\rangle_{\ell^{2}} d\sigma.$$

Let us fix $\omega \in \tilde{\Omega}$ such that

- (i) $\tilde{u}_n(\cdot,\omega) \to \tilde{u}(\cdot,\omega)$ in $L^2(0,T,H_{loc})$,
- (ii) $\tilde{u}(\cdot,\omega) \in L^2(0,T;H)$ and the sequence $(\tilde{u}_n(\cdot,\omega))_{n\geq 1}$ is bounded in $\mathscr{C}([0,T];V)$.

Notice that, in order to prove (6.5.70) it is sufficient to prove that

(6.5.71)
$$G(\cdot, \tilde{u}_n(\cdot, \omega))^* P_n \psi \to G(\cdot, \tilde{u}(\cdot, \omega))^* \psi \quad \text{in} \quad L^2(s, t; \ell^2).$$

We have

$$\begin{split} &\int_{s}^{t} \left\| G(\sigma, \tilde{u}_{n}(\sigma, \omega))^{*} P_{n} \psi - G(\sigma, \tilde{u}(\sigma, \omega))^{*} \psi \right\|_{\ell^{2}}^{2} d\sigma \\ &\leq \int_{s}^{t} \Big(\left\| G(\sigma, \tilde{u}_{n}(\sigma, \omega))^{*} (P_{n} \psi - \psi) \right\|_{\ell^{2}} + \left\| G(\sigma, \tilde{u}_{n}(\sigma, \omega))^{*} \psi - G(\sigma, \tilde{u}(\sigma, \omega))^{*} \psi \right\|_{\ell^{2}} \Big)^{2} d\sigma \\ &\leq 2 \int_{s}^{t} \left\| G(\sigma, \tilde{u}_{n}(\sigma, \omega))^{*} \right\|_{\mathscr{L}(\mathrm{H}, \ell^{2})}^{2} \cdot \left| P_{n} \psi - \psi \right|_{\mathrm{H}}^{2} d\sigma + 2 \int_{s}^{t} \left\| G(\sigma, \tilde{u}_{n}(\sigma, \omega))^{*} \psi - G(\sigma, \tilde{u}(\sigma, \omega))^{*} \psi \right\|_{\ell^{2}}^{2} d\sigma \\ &(6.5.72) \\ &=: 2 \{ I_{1}(n) + I_{2}(n) \}. \end{split}$$

Let us consider the term $I_1(n)$. Since $\psi \in V_{\gamma}$, we have

$$\lim_{n\to\infty} |P_n\psi - \psi|_{\rm H} = 0.$$

By Lemma 6.2.1, the continuity of the embedding $\mathcal{T}_2(\ell^2, H) \hookrightarrow \mathscr{L}(\ell^2, H)$ and (ii), we infer that

$$\int_{s}^{t} \left\| G(\sigma, \tilde{u}_{n}(\sigma, \omega))^{*} \right\|_{\mathcal{L}(\mathbf{H}, \ell^{2})}^{2} d\sigma \leq C \int_{s}^{t} \left| \nabla \tilde{u}_{n}(\sigma, \omega) \right|_{L^{2}}^{2} d\sigma \leq \tilde{C}T \sup_{n \geq 1} \left\| \tilde{u}_{n}(\omega) \right\|_{\mathscr{C}([0,T]; \mathbf{V})} \leq K$$

for some constant K > 0. Thus

$$\lim_{n\to\infty} I_1(n) = \lim_{n\to\infty} \int_s^t \left\| G(\sigma, \tilde{u}_n(\sigma, \omega))^* \right\|_{\mathscr{L}(\mathrm{H}, \ell^2)}^2 \cdot |P_n \psi - \psi|_{\mathrm{H}}^2 d\sigma = 0.$$

Let us move to the term $I_2(n)$ in (6.5.72). We will prove that for every $\psi \in V_{\gamma}$ the term $I_2(n)$ tends to zero as $n \to \infty$. Assume first that $\psi \in \mathcal{V}$. Then there exists R > 0 such that $\operatorname{supp} \psi$ is a compact subset of \mathcal{O}_R . Since $\tilde{u}_n(\cdot, \omega) \to \tilde{u}(\cdot, \omega)$ in $L^2(0, T; H_{\text{loc}})$, then in particular

$$\lim_{n\to\infty} q_{T,R} \big(\tilde{u}_n(\cdot,\omega) - \tilde{u}(\cdot,\omega) \big) = 0,$$

where $q_{T,R}$ is the seminorm defined by (6.3.1). In other words, $\tilde{u}_n(\cdot, \omega) \to \tilde{u}(\cdot, \omega)$ in $L^2(0,T; H_{\mathcal{O}_R})$. Therefore there exists a subsequence $(\tilde{u}_{n_k}(\cdot, \omega))_k$ such that

$$\tilde{u}_{n_k}(\sigma,\omega) \to \tilde{u}(\sigma,\omega) \quad \text{ in } H_{\mathcal{O}_R} \text{ for almost all } \sigma \in [0,T] \text{ as } k \to \infty.$$

Hence by Lemma 6.5.16

$$G\big(\sigma, \tilde{u}_{n_k}(\sigma, \omega)\big)^* \psi \to G\big(\sigma, \tilde{u}(\sigma, \omega)\big)^* \psi \quad \text{ in } \ell^2 \text{ for almost all } \sigma \in [0, T] \text{ as } k \to \infty.$$

In conclusion, by the Vitali Theorem

$$\lim_{k\to\infty}\int_s^t \|G\big(\tilde{u}_{n_k}(\sigma,\omega)\big)^*\psi - G\big(\tilde{u}(\sigma,\omega)\big)^*\psi\|_{\ell^2}^2\,d\sigma = 0 \qquad \text{for } \psi\in\mathcal{V}\,.$$

Repeating the above reasoning for all subsequences, we infer that from every subsequence of the sequence $(G(\sigma, \tilde{u}_n(\sigma, \omega))^* \psi)_n$ we can choose the subsequence convergent in $L^2(s, t; \ell^2)$ to the same

limit. Thus the whole sequence $(G(\sigma, \tilde{u}_n(\sigma, \omega))^* \psi)_n$ is convergent to $G(\sigma, \tilde{u}(\sigma, \omega))^* \psi$ in $L^2(s, t; \ell^2)$. At the same time

$$\lim_{n \to \infty} I_2(n) = 0 \quad \text{for every } \psi \in \mathcal{V}.$$

If $\psi \in V_{\gamma}$ then for every $\varepsilon > 0$ we can find $\psi_{\varepsilon} \in \mathcal{V}$ such that $\|\psi - \psi_{\varepsilon}\|_{V_{\gamma}} < \varepsilon$. By the continuity of embeddings $\mathcal{T}_{2}(\ell^{2}, \mathrm{H}) \hookrightarrow \mathcal{L}(\ell^{2}, \mathrm{V}_{\gamma}')$, Lemma 6.2.1 and (ii), we obtain

$$\begin{split} &\int_{s}^{t} \left\| G(\sigma,\tilde{u}_{n}(\sigma,\omega))^{*}\psi - G(\sigma,\tilde{u}(\sigma,\omega))^{*}\psi \right\|_{\ell^{2}}^{2}d\sigma \\ &\leq 2\int_{s}^{t} \left\| [G(\sigma,\tilde{u}_{n}(\sigma,\omega))^{*} - G(\sigma,\tilde{u}(\sigma,\omega))^{*}](\psi - \psi_{\varepsilon}) \right\|_{\ell^{2}}^{2}d\sigma \\ &\quad + 2\int_{s}^{t} \left\| [G(\sigma,\tilde{u}_{n}(\sigma,\omega))^{*} - G(\tilde{u}(\sigma,\omega))^{*}]\psi_{\varepsilon} \right\|_{\ell^{2}}^{2}d\sigma \\ &\leq 4\int_{s}^{t} \left[\left\| G(\sigma,\tilde{u}_{n}(\sigma,\omega)) \right\|_{\mathscr{L}(\ell^{2},V_{\gamma}^{\prime})}^{2} + \left\| G(\sigma,\tilde{u}(\sigma,\omega)) \right\|_{\mathscr{L}(\ell^{2},V_{\gamma}^{\prime})}^{2} \right] \|\psi - \psi_{\varepsilon}\|_{V_{\gamma}}^{2}d\sigma \\ &\quad + 2\int_{s}^{t} \left\| [G(\sigma,\tilde{u}_{n}(\sigma,\omega))^{*} - G(\sigma,\tilde{u}(\sigma,\omega))^{*}]\psi_{\varepsilon} \right\|_{\ell^{2}}^{2}d\sigma \\ &\leq c T \left(\left\| \tilde{u}_{n}(\cdot,\omega) \right\|_{\mathscr{L}(0,T;V)}^{2} + \left\| \tilde{u}(\cdot,\omega) \right\|_{\mathscr{L}(0,T;V)}^{2} \right) \cdot \varepsilon^{2} + 2\int_{s}^{t} \left\| [G(\sigma,\tilde{u}_{n}(\sigma,\omega))^{*}]\psi_{\varepsilon} \right\|_{\ell^{2}}^{2}d\sigma \\ &\leq C\varepsilon^{2} + 2\int_{s}^{t} \left\| [G(\sigma,\tilde{u}_{n}(\sigma,\omega))^{*} - G(\sigma,\tilde{u}(\sigma,\omega))^{*}]\psi_{\varepsilon} \right\|_{\ell^{2}}^{2}d\sigma, \end{split}$$

for some positive constants *c* and *C*. Passing to the upper limit as $n \to \infty$, we infer that

$$\limsup_{n\to\infty}\int_s^t \left\|G(\sigma,\tilde{u}_n(\sigma,\omega))^*\psi-G(\sigma,\tilde{u}(\sigma,\omega))^*\psi\right\|_{\ell^2}^2 d\sigma \leq C\varepsilon^2.$$

In conclusion, we proved that

$$\lim_{n \to \infty} \int_{s}^{t} \left\| G(\sigma, \tilde{u}_{n}(\sigma, \omega))^{*} \psi - G(\sigma, \tilde{u}(\sigma, \omega))^{*} \psi \right\|_{\ell^{2}}^{2} d\sigma = 0$$

which completes the proof of (6.5.71). Thus, by (6.5.69), convergence (6.5.70) and Vitali Theorem, we conclude the proof of Lemma 6.5.17.

By Lemma 6.5.12 we can pass to the limit in (6.5.41). By Lemmas 6.5.15 and 6.5.17 we can pass to the limit in (6.5.42) as well. After passing to the limits we infer that for all $\psi, \zeta \in V_{\gamma}$ and all bounded continuous functions h on $\mathscr{C}([0,T]; U')$:

(6.5.73)
$$\widetilde{\mathbb{E}}\left[\langle \tilde{M}(t) - \tilde{M}(s), \psi \rangle h(\tilde{u}_{|[0,s]})\right] = 0,$$

and

(6.5.74)
$$\begin{split} \tilde{\mathbb{E}}\Big[\Big(\langle \tilde{M}(t),\psi\rangle\langle \tilde{M}(t),\zeta\rangle-\langle \tilde{M}(s),\psi\rangle\langle \tilde{M}(s),\zeta\rangle \\ &-\int_{s}^{t}\langle (G(r,\tilde{u}(r)))^{*}\psi,(G(r,\tilde{u}(r)))^{*}\zeta\rangle_{\ell^{2}}\,dr\Big)\cdot h(\tilde{u}_{|[0,s]})\Big]=0, \end{split}$$

where $\langle \cdot, \cdot \rangle$ is the dual pairing between V'_{γ} and V_{γ} .

From Lemma 6.5.7, Lemma 6.5.15 and Lemma 6.5.17, we infer the following corollary.

Corollary 6.5.18. *For* $t \in [0, T]$

$$\langle \langle \tilde{M} \rangle \rangle_t = \int_0^t \| G(s, \tilde{u}(s)) \|_{\mathcal{T}_2(\ell^2; \mathrm{H})} ds, \qquad t \in [0, T].$$

Continuation of the proof of Theorem 6.5.4. Now we apply the idea analogous to Chapter 5, see also [38, Section 8.3]. By Lemma 6.5.8 and Corollary 6.5.14, we infer that $\tilde{M}(t), t \in [0, T]$ is an H-valued continuous square integrable martingale with respect to the filtration $\tilde{\mathbb{F}} = (\tilde{\mathscr{F}}_t)$. Moreover, by Corollary 6.5.18 the quadratic variation of \tilde{M} is given by

(6.5.75)
$$\langle \langle \tilde{M} \rangle \rangle_t = \int_0^t \|G(s, \tilde{u}(s))\|_{\mathcal{F}_2(\ell^2; \mathbf{H})} ds \qquad t \in [0, T].$$

Therefore by the martingale representation theorem, there exist

- a stochastic basis $(\tilde{\tilde{\Omega}}, \tilde{\tilde{\mathscr{F}}}, \tilde{\tilde{\mathbb{P}}}),$
- a cylindrical Wiener process $\tilde{\tilde{W}}(t)$ on ℓ^2 ,
- and a progressively measurable process $\tilde{\tilde{u}}(t)$ such that for all $t \in [0, T]$ and all $v \in \mathcal{V}$:

$$\begin{split} \langle \tilde{\tilde{u}}(t), \mathbf{v} \rangle &- \langle \tilde{\tilde{u}}(0), \mathbf{v} \rangle + \int_0^t \langle A \tilde{\tilde{u}}(s), \mathbf{v} \rangle \, ds + \int_0^t \langle B(\tilde{\tilde{u}}(s), \tilde{\tilde{u}}(s)), \mathbf{v} \rangle \, ds + \int_0^t \langle g(|\tilde{\tilde{u}}(s)|^2) \tilde{\tilde{u}}(s), \mathbf{v} \rangle \, ds \\ &= \int_0^t \langle f(\tilde{\tilde{u}}(s)), \mathbf{v} \rangle \, ds + \left\langle \int_0^t G(s, \tilde{\tilde{u}}(s)) \, d\tilde{\tilde{W}}(s), \mathbf{v} \right\rangle. \end{split}$$

Thus the conditions from Definition 6.3.7 hold with $(\hat{\Omega}, \hat{\mathscr{F}}, \hat{\mathbb{F}}, \hat{\mathbb{P}}) = (\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{F}}, \tilde{\mathbb{P}}), \hat{W} = \tilde{W}$ and $\hat{u} = \tilde{u}$. The proof of Theorem 6.5.4 is thus complete.

6.5.4 Uniqueness and strong solutions

In this subsection we will show that the solutions of (6.2.16) are pathwise unique and that the martingale solution of (6.2.16) is the strong solution. Let us recall the definition of pathwise unique solutions.

Definition 6.5.19. Let $(\Omega, \mathscr{F}, \mathbb{F}, \mathbb{P}, W, u^i)$, i = 1, 2 be the martingale solutions of (6.2.16) with $u^i(0) = u_0$, i = 1, 2. Then we say that the solutions are pathwise unique if \mathbb{P} -a.s. for all $t \in [0, T]$, $u^1(t) = u^2(t)$.

Theorem 6.5.20. Assume that the assumptions (A1) and (A2) are satisfied. If u_1, u_2 are two solutions of (6.2.16) defined on the same filtered probability space $(\hat{\Omega}, \hat{\mathscr{F}}, \hat{\mathbb{F}}, \hat{\mathbb{P}})$ then $\hat{\mathbb{P}}$ -a.s. for all $t \in [0, T], u_1(t) = u_2(t)$.

The theorem has been proved in [76, Theorem 3.7].

Theorem 6.5.21. Assume that assumptions (A1) and (A2) are satisfied. Then there exists a path-wise unique strong solution $u \in \mathcal{C}([0,T];V) \cap L^2(0,T;D(A))$ of (6.2.16) such that

$$\sup_{t\in[0,T]} \|u(t)\|_{\mathbf{V}}^2 + \int_0^T |u(t)|_{\mathbf{D}(\mathbf{A})}^2 dt < \infty.$$

Proof. Since by Theorem 6.5.4 there exists a martingale solution and by Theorem 6.5.20 it is pathwise unique, the existence of strong solution follows from [68, Theorem 2] or by the Yamada-Watanabe Theorem [95].

6.6 Invariant measures

In the following, we consider time homogeneous damped tamed NSEs, i.e. the coefficients f, σ are independent of t and furthermore $f \in H$ is not dependent on u. The time homogeneous damped tamed NSEs in abstract form are given by

(6.6.1)
$$\begin{cases} du(t) = \left[-A_{\alpha}u(t) - B(u(t)) - \Pi[g(|u(t)|^2)u(t)] + \Pi f \right] dt + \sum_{j=1}^{\infty} G_j(u(t)) dW_j(t), \\ u(0) = u_0 \in \mathbf{V}, \end{cases}$$

where $A_{\alpha} = \alpha I - v\Delta$ for some $\alpha \in \mathbb{R}$ and v > 0 is the viscosity. The operator *B* and the cylindrical Wiener process $W = (W_j)_{j=1}^{\infty}$ on ℓ^2 , is same as defined in Section 6.2 and G_j are as defined in (6.2.11).

Let $\mathscr{B}_b(V)$ denote the set of all bounded and Borel measurable functions on V. For any $\varphi \in \mathscr{B}_b(V)$, $t \ge 0$, we define a function $T_t \varphi \colon V \to \mathbb{R}$ by

(6.6.2)
$$T_t \varphi(\mathbf{v}) := \mathbb{E} \left(\varphi(u(t; \mathbf{v})) \right), \quad \mathbf{v} \in \mathbf{V}.$$

It follows from Theorem 6.6.2 and Ondrejat [69] (see also [17]) that $T_t \varphi \in \mathscr{B}_b(V)$ and $\{T_t\}_{t\geq 0}$ is a semigroup on $\mathscr{B}_b(V)$. Also since this unique solution to (6.6.1) has *a.e.* path in $\mathscr{C}([0,T];V)$, it is also a Markov semigroup (see [69, Theorem 27]). Moreover, $\{T_t\}_{t\geq 0}$ is a Feller semigroup, i.e. T_t maps $C_b(V)$ into itself.

Next we state the main result of this section, regarding invariant measures:

Theorem 6.6.1. Let for every $\alpha > 0$, the assumptions $(\mathbf{A1})' - (\mathbf{A3})'$ be satisfied. Then there exists an invariant measure $\mu \in \mathscr{P}(V)$ of the semigroup $(T_t)_{t\geq 0}$ defined by (6.6.2), such that for any $t \geq 0$ and $\varphi \in \mathrm{SC}_b(V_w)$

$$\int_{\mathcal{V}} T_t \varphi(u) \mu(du) = \int_{\mathcal{V}} \varphi(u) \mu(du).$$

If T_t is sequentially weakly Feller Markov semigroup then for every $\varphi \in SC_b(V_w)$, $T_t\varphi \in SC_b(V_w) \subset B_b(V)$ (see [17, 62] for the definitions and inclusions of the spaces); therefore the integral on LHS in Theorem 6.6.1 makes sense.

Now we list the assumptions that we make on the coefficients f and σ along with a coercivity type assumption, see [72].

Assumptions. (A1)' The function $f : \mathbb{R}^3 \to \mathbb{R}^3$, is time independent and H-valued.

 $(\mathbf{A2})'$ A measurable function $\sigma : \mathbb{R}^3 \to \mathbb{R}^3$ of C^1 class with respect to the x-variable and for all $x \in \mathbb{R}^3$ there exists a constant $C_{\sigma} > 0$ such that

$$\|\partial_{x^j}\sigma(x)\|_{\ell^2} \le C_\sigma, \quad j=1,2,3$$

and, for all $x \in \mathbb{R}^3$,

$$\|\sigma(x)\|_{\ell^2}^2 \le \frac{1}{4}.$$

(A3)' there exists a $\delta > 0$ such that

$$2\nu |\nabla u|_{L^2}^2 - \|G(u)\|_{\mathcal{F}_2(\ell^2;\mathbf{H})}^2 \ge 2\delta |\nabla u|_{L^2}^2.$$

The following theorem regarding the existence of a pathwise unique strong solution to the time homogeneous damped tamed NSEs (6.6.1) can be proved by modifying the proofs of Theorem 6.5.4 and Theorem 6.5.20 to incorporate the extra linear term αu .

Theorem 6.6.2. Assume that assumptions (A1)' and (A2)' are satisfied. Then for every $u_0 \in V$, there exists a path-wise unique strong solution u of (6.6.1) for every T > 0 such that $u \in \mathscr{C}([0,T];V) \cap L^2(0,T;D(A))$, \mathbb{P} -a.s.

For the fixed initial data $u_0 = v \in V$ we denote the (pathwise) unique solution of (6.6.1), whose existence is proved in Theorem 6.6.2 by u(t;v). Then $\{u(t;v) : v \in V, t \ge 0\}$ forms a strong Markov process with state space V. We have the following result:

Lemma 6.6.3. For $v, v' \in V$ and R > 0, define

$$\tau_R^{\mathbf{v}} := \inf\{t \in [0, T] : \|u(t; \mathbf{v})\|_{\mathbf{V}} > R\},\$$

and

$$\tau_R^{\mathbf{v},\mathbf{v}'} := \tau_R^{\mathbf{v}} \wedge \tau_R^{\mathbf{v}'}.$$

Suppose that assumptions (A1)' and (A2)' hold, then

$$\mathbb{E} \| u(t \wedge \tau_R^{\mathbf{v}, \mathbf{v}'}; \mathbf{v}) - u(t \wedge \tau_R^{\mathbf{v}, \mathbf{v}'}; \mathbf{v}') \|_{\mathbf{V}}^2 \le C_{t, R} \cdot \| \mathbf{v} - \mathbf{v}' \|_{\mathbf{V}}^2.$$

Proof. Let u(t) := u(t; v), $\tilde{u}(t) := u(t; v')$, and

$$w(t) := u(t) - \tilde{u}(t).$$

Set $t_R \coloneqq \tau_R^{\mathbf{v},\mathbf{v}'} \wedge t.$ By Itô Lemma, we have

$$\begin{split} \|w(t)\|_{\mathcal{V}}^{2} &= \|w(0)\|_{\mathcal{V}}^{2} - 2\int_{0}^{t_{R}} \langle \mathbf{A}_{\alpha}w(s), w(s) \rangle_{\mathcal{V}} ds - 2\int_{0}^{t_{R}} \langle B(u(s)) - B(\tilde{u}(s)), w(s) \rangle_{\mathcal{V}} ds \\ &- 2\int_{0}^{t_{R}} \langle g(|u(s)|^{2})u(s) - g(|\tilde{u}(s)|^{2})\tilde{u}(s), w(s) \rangle_{\mathcal{V}} ds + \int_{0}^{t_{R}} \|G(u(s)) - G(\tilde{u}(s))\|_{\mathcal{F}_{2}(\ell^{2};\mathcal{V})}^{2} ds \\ &+ 2\int_{0}^{t_{R}} \langle \sum_{j=1}^{\infty} \left(G_{j}(u(s)) - G_{j}(\tilde{u}(s))\right) dW_{j}(s), w(s) \rangle_{\mathcal{V}} \\ &:= \|w(0)\|_{\mathcal{V}}^{2} + I_{1}(t_{R}) + I_{2}(t_{R}) + I_{3}(t_{R}) + I_{4}(t_{R}) + I_{5}(t_{R}). \end{split}$$

Now we deal with each term individually. From the definition of A_{α} , we have

$$I_{1}(t_{R}) = -2\int_{0}^{t_{R}} \langle \mathbf{A}_{\alpha}w(s), w(s) \rangle_{\mathbf{V}} ds = -2\alpha \int_{0}^{t_{R}} \|w(s)\|_{\mathbf{V}}^{2} - 2\nu \int_{0}^{t_{R}} |w(s)|_{H^{2}}^{2} + 2\nu \int_{0}^{t_{R}} |w(s)|_{\mathbf{H}}^{2} ds = -2\alpha \int_{0}^{t_{R}} \|w(s)\|_{\mathbf{V}}^{2} - 2\nu \int_{0}^{t_{R}} |w(s)|_{H^{2}}^{2} + 2\nu \int_{0}^{t_{R}} |w(s)|_{\mathbf{H}}^{2} ds = -2\alpha \int_{0}^{t_{R}} \|w(s)\|_{\mathbf{V}}^{2} - 2\nu \int_{0}^{t_{R}} |w(s)|_{H^{2}}^{2} + 2\nu \int_{0}^{t_{R}} |w(s)|_{\mathbf{H}}^{2} ds = -2\alpha \int_{0}^{t_{R}} \|w(s)\|_{\mathbf{V}}^{2} - 2\nu \int_{0}^{t_{R}} |w(s)|_{H^{2}}^{2} + 2\nu \int_{0}^{t_{R}} |w(s)|_{\mathbf{H}}^{2} ds = -2\alpha \int_{0}^{t_{R}} \|w(s)\|_{\mathbf{V}}^{2} - 2\nu \int_{0}^{t_{R}} |w(s)|_{\mathbf{V}}^{2} + 2\nu \int_{0}^{t_{R}} |w(s)|_{\mathbf{V}}^{2} ds = -2\alpha \int_{0}^{t_{R}} \|w(s)\|_{\mathbf{V}}^{2} - 2\nu \int_{0}^{t_{R}} |w(s)|_{\mathbf{V}}^{2} ds = -2\alpha \int_{0}^{t_{R}} \|w(s)\|_{\mathbf{V}}^{2} + 2\nu \int_{0}^{t_{R}} |w(s)|_{\mathbf{V}}^{2} ds = -2\alpha \int_{0}^{t_{R}} \|w(s)\|_{\mathbf{V}}^{2} ds = -2\alpha \int_{0}^{t_$$

By definition of the operator B and the Hölder inequality, we get

$$\begin{split} |I_{2}(t_{R})| &= 2 \left| \int_{0}^{t_{R}} \langle B(u(s)) - B(\tilde{u}(s)), w(s) \rangle_{V} ds \right| \\ &= 2 \left| \int_{0}^{t_{R}} \langle B(w(s), u(s)), w(s) \rangle_{V} ds + \int_{0}^{t} \langle B(\tilde{u}(s), w(s)), w(s) \rangle_{V} ds \right| \\ &\leq 2 \int_{0}^{t_{R}} |w(s) \cdot \nabla u(s)|_{L^{2}} \left(|w(s)|_{H} + |Aw(s)|_{L^{2}} \right) ds \\ &+ 2 \int_{0}^{t_{R}} |\tilde{u}(s) \cdot \nabla w(s)|_{L^{2}} \left(|w(s)|_{H} + |Aw(s)|_{L^{2}} \right) ds. \end{split}$$

Using Lemma 6.2.1, the Gagliardo-Nirenberg inequality (6.2.9), the Hölder inequality, the Sobolev embedding theorem $(H^1 \hookrightarrow L^6)$ and the Young inequality, we obtain the following estimate on the second term,

$$\begin{split} |I_{2}(t_{R})| &\leq C \int_{0}^{t_{R}} |w(s) \cdot \nabla u(s)|_{L^{2}}^{2} ds + C_{v} \int_{0}^{t_{R}} |\tilde{u}(s) \cdot \nabla w(s)|_{L^{2}}^{2} ds + \frac{v}{2} \int_{0}^{t_{R}} |w(s)|_{\mathrm{D(A)}}^{2} ds \\ &\leq C \int_{0}^{t_{R}} ||w(s)||_{L^{\infty}}^{2} |\nabla u(s)|_{L^{2}}^{2} + C_{v} \int_{0}^{t_{R}} ||u(s)||_{L^{6}}^{2} ||\nabla w(s)||_{L^{3}}^{2} ds + \frac{v}{2} \int_{0}^{t_{R}} ||w(s)||_{\mathrm{D(A)}}^{2} ds \\ &\leq C_{R} \int_{0}^{t_{R}} ||w(s)||_{H^{2}}^{3/2} |w(s)|_{\mathrm{H}}^{1/2} ds + C_{R,v} \int_{0}^{t_{R}} ||\nabla w(s)||_{H^{1}} ||\nabla w(s)||_{L^{2}} ds \\ &\quad + \frac{v}{2} \int_{0}^{t_{R}} ||w(s)||_{\mathrm{D(A)}}^{2} ds \\ &\leq C_{R,v} \int_{0}^{t_{R}} ||w(s)||_{\mathrm{V}}^{2} ds + \frac{v}{2} \int_{0}^{t_{R}} ||w(s)||_{H^{2}}^{2} ds + \frac{v}{2} \int_{0}^{t_{R}} |w(s)|_{\mathrm{D(A)}}^{2} ds. \end{split}$$

Since $|g(r) - g(r')| \le |r - r'|$, the Hölder inequality and the Young inequality gives

$$\begin{split} |I_{3}(t_{R})| &= 2 \left| \int_{0}^{t_{R}} \langle g(|u(s)|^{2})u(s) - g(|\tilde{u}(s)|^{2})\tilde{u}(s), w(s) \rangle_{\mathrm{V}} ds \right| \\ &\leq 4 \int_{0}^{t_{R}} \langle |w(s)|(|u(s)| + |\tilde{u}(s)|)\tilde{u}(s) + |u(s)|^{2}w(s), w(s) \rangle_{\mathrm{V}} ds \\ &\leq C_{v} \int_{0}^{t_{R}} |w(s) \cdot (|u(s)| + |\tilde{u}(s)|)^{2} |_{\mathrm{H}}^{2} + \frac{v}{4} \int_{0}^{t_{R}} |w(s)|_{\mathrm{D}(\mathrm{A})}^{2} ds. \end{split}$$

Using the Hölder inequality, the Sobolev embedding theorem and the Gagliardo-Nirenberg inequality (6.2.9) we get the following estimates,

$$\begin{split} |I_{3}(t_{R})| &\leq C_{\nu} \int_{0}^{t_{R}} \|w(s)\|_{L^{6}}^{2} \||u(s)| + |\tilde{u}(s)|\|_{L^{6}}^{4} ds + \frac{\nu}{4} \int_{0}^{t_{R}} |w(s)|_{\mathrm{D(A)}}^{2} ds \\ &\leq C_{R,\nu} \int_{0}^{t_{R}} \|w(s)\|_{H^{2}}^{2} |w(s)|_{\mathrm{H}} ds + \frac{\nu}{4} \int_{0}^{t_{R}} |w(s)|_{\mathrm{D(A)}}^{2} ds \\ &\leq C_{R,\nu} \int_{0}^{t_{R}} \|w(s)\|_{\mathrm{V}}^{2} ds + \frac{\nu}{4} \int_{0}^{t_{R}} \|w(s)\|_{H^{2}}^{2} ds + \frac{\nu}{4} \int_{0}^{t_{R}} |w(s)|_{\mathrm{D(A)}}^{2} ds. \end{split}$$

Using assumption (A2)' and Lemma 6.2.1 we get the following estimate on I_4

$$|I_4(t_R)| = \left| \int_0^{t_R} \|G(u(s)) - G(\tilde{u}(s))\|_{\mathcal{J}_2(\ell^2; \mathbb{V})}^2 ds \right| \le \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{L^2}^2 ds = \frac{1}{2} \int_0^{t_R} |Aw(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_0^{t_R} |\nabla w(s)|_{\mathrm{H}}^2 ds + C_\sigma \int_$$

Since $\mathbb{E}(I_5(t_R)) = 0$, by combining all the above estimates and using $|u|_{D(A)} \le ||u||_{H^2}$, we get

$$\begin{split} \mathbb{E} \|w(t \wedge \tau_{R}^{\mathbf{v}, \mathbf{v}'})\|_{\mathbf{V}}^{2} &\leq \|w(0)\|_{\mathbf{V}}^{2} + C_{R, \alpha, \nu} \int_{0}^{t_{R}} \mathbb{E} \|w(s)\|_{\mathbf{V}}^{2} ds \\ &\leq \|\mathbf{v} - \mathbf{v}'\|_{\mathbf{V}}^{2} + C_{R, \alpha, \nu} \int_{0}^{t} \mathbb{E} \|w(s \wedge \tau_{R}^{\mathbf{v}, \mathbf{v}'})\|_{\mathbf{V}}^{2} ds. \end{split}$$

The desired estimate follows from the application of generalised Gronwall Lemma.

For a metric space \mathbb{U} , we use $\mathscr{P}(\mathbb{U})$ to denote the total of all probability measures on \mathbb{U} . We will use the following theorem from Maslowski-Seidler [62] to prove the existence of invariant measures.

Theorem 6.6.4. Assume that

- (i) the semigroup $\{T_t\}_{t\geq 0}$, defined by (6.6.2) is sequentially weakly Feller in V,
- (ii) for any $\varepsilon > 0$ there exists R > 0 such that

$$\sup_{T \ge 1} \frac{1}{T} \int_0^T \mathbb{P}(\{\|u(t;u_0)\|_{V} > R\}) dt < \varepsilon.$$

Then there exists at least one invariant measure for (6.6.1).

6.6.1 Boundedness in probability

Lemma 6.6.5. Let $u_0 \in V$. Then, under the assumptions of Theorem 6.6.1, for every $\varepsilon > 0$, there exists R > 0 such that

(6.6.3)
$$\sup_{T \ge 1} \frac{1}{T} \int_0^T \mathbb{P}(\{\|u(t;u_0)\|_{\mathcal{V}} > R\}) dt < \varepsilon$$

Proof. Using the Itô lemma for the function $|x|_{\rm H}^2$ and the process u(t), we have

(6.6.4)
$$\begin{aligned} \frac{1}{2}|u(t)|_{\mathrm{H}}^{2} &= \frac{1}{2}|u_{0}|_{\mathrm{H}}^{2} + \int_{0}^{t} \langle -\mathrm{A}_{\alpha}u(s) - B(u(s)) - \Pi(g(|u(s)|^{2})u(s)) + \Pi f, u(s) \rangle_{\mathrm{H}} \, ds \\ &+ \int_{0}^{t} \sum_{j=1}^{\infty} \langle G_{j}(u(s))dW_{j}(s), u(s) \rangle_{\mathrm{H}} + \frac{1}{2} \int_{0}^{t} \|G(u(s))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2} \, ds. \end{aligned}$$

Now we deal with each term individually.

(6.6.5)
$$\langle \mathbf{A}_{\alpha} u, u \rangle_{\mathbf{H}} = \alpha \| u \|_{\mathbf{H}}^{2} + \nu \| \nabla u \|_{L^{2}}^{2}.$$

$$(6.6.6) \qquad \langle B(u), u \rangle_{\rm H} = 0.$$

(6.6.7)
$$\langle \Pi g(|u|^2)u, u \rangle_{\mathrm{H}} = \left| |\sqrt{g(|u|^2)}| \cdot |u| \right|_{L^2}^2.$$

Using the assumptions on *f*, for some $\beta > 0$ we obtain the following estimate:

(6.6.8)
$$\langle \Pi f, u \rangle_{\mathrm{H}} \le \|f\|_{\mathrm{V}'} \|u\|_{\mathrm{V}} \le \frac{1}{4\beta} \|f\|_{\mathrm{V}'}^2 + \beta \|u\|_{\mathrm{V}}^2$$

Since u is the solution of (6.6.1) and satisfies the estimates (6.4.25) (Theorem 6.5.4 and Theorem 6.5.21 hold for the tamed NSEs, but we can also prove similar theorems for the damped tamed NSEs too), we can show that the process

$$M(t) = \int_0^t \langle u(s), \sum_{j=1}^\infty G_j(u(s)) dW_j(s) \rangle_{\mathrm{H}},$$

is a \mathbb{F} -martingale. Thus taking expectation in (6.6.4) and using the estimates (6.6.5) - (6.6.8), we infer

$$\begin{split} \frac{1}{2} \mathbb{E} |u(t)|_{\mathrm{H}}^2 &\leq \frac{1}{2} |u_0|_{\mathrm{H}}^2 - \alpha \mathbb{E} \int_0^t |u(s)|_{\mathrm{H}}^2 ds - \nu \mathbb{E} \int_0^t |\nabla u(s)|_{L^2}^2 ds \\ &- \mathbb{E} \int_0^t \left| |\sqrt{g(|u(s)|^2)}| \cdot |u(s)| \Big|_{L^2}^2 ds + \frac{1}{4\beta} \mathbb{E} \int_0^t \|f\|_{\mathrm{V}'}^2 ds + \beta \mathbb{E} \int_0^t \|u(s)\|_{\mathrm{V}}^2 ds \\ &+ \frac{1}{2} \int_0^t \|G(u(s))\|_{\mathcal{T}_2(\ell^2;\mathrm{H})}^2 ds. \end{split}$$

On rearranging, we get

$$\begin{aligned} \frac{1}{2} \mathbb{E} |u(t)|_{\mathrm{H}}^{2} + \frac{1}{2} \mathbb{E} \int_{0}^{t} \left(2\nu |\nabla u(s)|_{L^{2}}^{2} - \|G(u(s))\|_{\mathcal{F}_{2}(\ell^{2};\mathrm{H})}^{2} \right) ds + \mathbb{E} \int_{0}^{t} \left| |\sqrt{g(|u(s)|^{2})} | \cdot |u(s)| \right|_{L^{2}}^{2} ds \\ (6.6.9) &\leq \frac{1}{2} |u_{0}|_{\mathrm{H}}^{2} + \frac{1}{4\beta} T \|f\|_{\mathrm{V}}^{2} + \beta \mathbb{E} \int_{0}^{t} |\nabla u(s)|_{L^{2}}^{2} ds + (\beta - \alpha) \mathbb{E} \int_{0}^{t} |u(s)|_{\mathrm{H}}^{2} ds. \end{aligned}$$

Now using the assumption (A3)' in (6.6.9), we obtain

$$\begin{split} \frac{1}{2} \mathbb{E} \left| u(t) \right|_{\mathrm{H}}^{2} + (\delta - \beta) \mathbb{E} \int_{0}^{t} \left| \nabla u(s) \right|_{L^{2}}^{2} ds + (\alpha - \beta) \mathbb{E} \int_{0}^{t} \left| u(s) \right|_{\mathrm{H}}^{2} ds \\ + \mathbb{E} \int_{0}^{t} \left| \left| \sqrt{g(|u(s)|^{2})} \right| \cdot |u(s)| \right|_{L^{2}}^{2} ds &\leq \frac{1}{2} |u_{0}|_{\mathrm{H}}^{2} + \frac{1}{4\beta} T \| f \|_{\mathrm{V}'}^{2} \end{split}$$

Choosing $\beta \leq \frac{1}{2} \min{\{\delta, \alpha\}}$ yields

$$\begin{split} \frac{1}{2} \mathbb{E} |u(t)|_{\mathrm{H}}^2 + \frac{\delta}{2} \mathbb{E} \int_0^t |\nabla u(s)|_{L^2}^2 ds + \frac{\alpha}{2} \mathbb{E} \int_0^t |u(s)|_{\mathrm{H}}^2 ds + \mathbb{E} \int_0^t \left| |\sqrt{g(|u(s)|^2)} | \cdot |u(s)| \right|_{L^2}^2 ds \\ & \leq \frac{1}{2} |u_0|_{\mathrm{H}}^2 + \frac{1}{4\beta} T \|f\|_{\mathrm{V}'}^2. \end{split}$$

Therefore for $\gamma = \frac{1}{2} \min{\{\alpha, \delta\}}$,

$$\frac{1}{2}\mathbb{E}|u(t)|_{\mathrm{H}}^{2} + \gamma \mathbb{E}\int_{0}^{t} \|u(s)\|_{\mathrm{V}}^{2} ds + \mathbb{E}\int_{0}^{t} \left| |\sqrt{g(|u(s)|^{2})}| \cdot |u(s)| \right|_{L^{2}}^{2} ds \leq \frac{1}{2}|u_{0}|_{\mathrm{H}}^{2} + \frac{1}{4\beta}T \|f\|_{\mathrm{V}}^{2} + \frac{1$$

Thus for any T > 0, we infer that

(6.6.10)
$$\frac{1}{T} \int_0^T \mathbb{E} \|u(s)\|_V^2 ds \le \frac{1}{2\gamma T} \|u_0\|_H^2 + \frac{1}{4\gamma\beta} \|f\|_{V'}^2.$$

Using the Chebyshev inequality and inequality (6.6.10), we infer that for every $T \ge 0$

$$\begin{split} \frac{1}{T} \int_0^T \mathbb{P}(\{\|u(t, u_0)\|_{\mathcal{V}} > R\}) dt &\leq \frac{1}{TR^2} \int_0^T \mathbb{E} \|u(t)\|_{\mathcal{V}}^2 dt \\ &\leq \frac{1}{R^2} \left[\frac{1}{2\gamma T} |u_0|_{\mathcal{H}}^2 + \frac{1}{4\gamma\beta} \|f\|_{\mathcal{V}}^2 \right]. \end{split}$$

Now for sufficiently large R > 0 depending on ε , $|u_0|_H$ and $||f||_{V'}$ the assertion follows.

6.6.2 Sequentially weak Feller property

We are left to verify the assumption (*i*) of Theorem 6.6.4, i.e. the Markov semigroup $\{T_t\}_{t\geq 0}$ is sequentially weakly Feller in V. In other words we want to show that for any t > 0 and any bounded and weakly continuous function $\varphi: V \to \mathbb{R}$, if $\xi_n \to \xi$ weakly in V, then

$$(6.6.11) T_t \varphi(\xi_n) \to T_t \varphi(\xi)$$

In Theorem 5.7.7, we proved for stochastic constrained Navier-Stokes equations that the martingale solution of SCNSE continuously depends on the initial data. We have a similar result for time homogeneous damped tamed NSEs, which can be proved analogously, see also [25, Theorem 4.11].

Theorem 6.6.6. Assume that $(u_{0,n})_{n=1}^{\infty}$ is a V-valued sequence that is convergent weakly to $u_0 \in V$. Let

$$(\Omega_n, \mathscr{F}_n, \mathbb{F}_n, \mathbb{P}_n, W_n, u_n)$$

be a martingale solution of problem (6.6.1) on $[0,\infty)$ with the initial data $u_{0,n}$. Then for every T > 0 there exist

- a subsequence $(n_k)_k$,
- a stochastic basis (Ω̃, ℱ, ℙ̃, ℙ̃),
- a cylindrical Wiener process $\tilde{W}(t) = \left(\tilde{W}^{j}(t)\right)_{j=1}^{\infty}$ on ℓ^{2} ,
- and $\tilde{\mathbb{F}}$ -progressively measurable processes \tilde{u} , $(\tilde{u}_{n_k})_{k\geq 1}$ (defined on this basis) with laws supported in \mathcal{Z}_T (see (6.3.5)) such that

(6.6.12) \tilde{u}_{n_k} has the same law as u_{n_k} on \mathcal{Z}_T and $\tilde{u}_{n_k} \to \tilde{u}$ in \mathcal{Z}_T , $\tilde{\mathbb{P}}$ - a.s.

and the system

$$\left(\tilde{\Omega},\tilde{\mathscr{F}},\tilde{\mathbb{F}},\tilde{\mathbb{P}},\tilde{W},\tilde{u}\right)$$

is a martingale solution to problem (6.6.1) on the interval [0,T] with the initial data u_0 . In particular, for all $t \in [0,T]$ and all $v \in V$

$$\begin{split} &\langle \tilde{u}(t), \mathbf{v} \rangle + \int_0^t \langle \mathbf{A}_{\alpha} \tilde{u}(s), \mathbf{v} \rangle \, ds + \int_0^t \langle B(\tilde{u}(s)), \mathbf{v} \rangle \, ds + \int_0^t \langle g(|\tilde{u}(s)|^2) \tilde{u}(s), \mathbf{v} \rangle \, ds \\ &= \langle \tilde{u}(0), \mathbf{v} \rangle_{\mathbf{V}} + \int_0^t \langle f, \mathbf{v} \rangle \, ds + \langle \int_0^t \sum_{j=1}^\infty G_j(s, \tilde{u}(s)) \, dW_j(s), \mathbf{v} \rangle, \quad \tilde{\mathbb{P}}\text{-}a.s. \end{split}$$

Moreover, the process \tilde{u} satisfies the following inequality

(6.6.13)
$$\tilde{\mathbb{E}}\left[\sup_{s\in[0,T]}\|\tilde{u}(s)\|_{V}^{2} + \int_{0}^{T}|\tilde{u}(s)|_{D(A)}^{2}ds\right] < \infty.$$

We will need the uniqueness in law of solutions of (6.6.1). We define the uniqueness in law here:

Definition 6.6.7. Let $(\Omega^i, \mathscr{F}^i, \mathbb{P}^i, \mathbb{W}^i, u^i), i = 1, 2$ be the martingale solutions of (6.6.1) with $u^i(0) = u_0, i = 1, 2$. Then we say that the solutions are unique in law if

$$\operatorname{Law}_{\mathbb{P}^1}(u^1) = \operatorname{Law}_{\mathbb{P}^2}(u^2) \text{ on } \mathscr{C}([0,\infty); V_w) \cap L^2([0,\infty); D(A)),$$

where $\text{Law}_{\mathbb{P}^{i}}(u^{i}), i = 1, 2$ are by definition probability measures on $\mathscr{C}([0,\infty); V_{w}) \cap L^{2}([0,\infty); D(A))$.

Lemma 6.6.8. Assume that assumptions (A1)' - (A2)' are satisfied. Then the martingale solution of (6.6.1) are unique in law.

The proof of the above lemma is the direct application of Theorems 2 and 11 of [68] once we have proved the existence of a pathwise unique martingale solution of (6.6.1), which follows from Theorem 6.6.2.

Lemma 6.6.9. The semigroup $\{T_t\}_{t\geq 0}$ is sequentially weakly Feller in V.

Proof. Let us choose and fix $0 < t \le T, \xi \in V$ and $\varphi : V \to \mathbb{R}$ be a bounded weakly continuous function. Need to show that $T_t \varphi$ is sequentially weakly Feller in V. For this aim let us choose an V-valued sequence (ξ_n) weakly convergent in V to ξ . Since the function $T_t \varphi : V \to \mathbb{R}$ is bounded, we only need to prove (6.6.11).

Let $u_n(\cdot) = u(\cdot,\xi_n)$ be a strong solution of (6.6.1) on [0,T] with the initial data ξ_n and let $u(\cdot) = u(\cdot,\xi)$ be a strong solution of (6.6.1) with the initial data ξ . We assume these processes are defined on the stochastic basis $(\Omega, \mathcal{F}, \mathbb{F}, \mathbb{P}, W)$. By Theorem 6.6.6 there exist

- a subsequence $(n_k)_k$,
- a stochastic basis $(\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{F}}, \tilde{\mathbb{P}})$, where $\tilde{\mathbb{F}} = \{\tilde{\mathscr{F}}_s\}_{s \in [0,T]}$,
- a cylindrical Wiener process $\tilde{W}(t) = (\tilde{W}^j(t))_{i=1}^{\infty}$ on ℓ^2 ,

• and progressively measurable processes $\tilde{u}(s), (\tilde{u}_{n_k}(s))_{k\geq 1}, s \in [0, T]$ (defined on this basis) with laws supported in \mathcal{Z}_T such that

(6.6.14) \tilde{u}_{n_k} has the same law as u_{n_k} on \mathcal{Z}_T and $\tilde{u}_{n_k} \to \tilde{u}$ in \mathcal{Z}_T , $\tilde{\mathbb{P}} - a.s.$

and the system

(6.6.15)
$$(\tilde{\Omega}, \tilde{\mathscr{F}}, \tilde{\mathbb{F}}, \tilde{\mathbb{P}}, \tilde{W}, \tilde{u})$$

is a martingale solution to (6.6.1) on the interval [0, T] with the initial data ξ .

In particular, by (6.6.14), P-almost surely

$$\tilde{u}_{n_k}(t) \rightarrow \tilde{u}(t)$$
 weakly in V.

Since the function $\varphi: V \to \mathbb{R}$ is sequentially weakly continuous, we infer that $\tilde{\mathbb{P}}$ -a.s.,

$$\varphi(\tilde{u}_{n_k}(t)) \to \varphi(\tilde{u}(t))$$
 in \mathbb{R} .

Since the function φ is also bounded, by the Lebesgue dominated convergence theorem we infer that

(6.6.16)
$$\lim_{k \to \infty} \tilde{\mathbb{E}} \left[\varphi(\tilde{u}_{n_k}(t)) \right] = \tilde{\mathbb{E}} \left[\varphi(\tilde{u}(t)) \right].$$

From the equality of laws of \tilde{u}_{n_k} and u_{n_k} , $k \in \mathbb{N}$, on the space \mathcal{Z}_T we infer that \tilde{u}_{n_k} and u_{n_k} have the same laws on V_w and so

(6.6.17)
$$\tilde{\mathbb{E}}\left[\varphi(\tilde{u}_{n_k}(t))\right] = \mathbb{E}\left[\varphi(u_{n_k}(t))\right].$$

On the other hand, R.H.S. of (6.6.17) is equal by (6.6.2), to $T_t \varphi(\xi_{n_k})$.

Since *u*, by assumption, is a martingale solution of (6.6.1) with the initial data ξ and by the above \tilde{u} is also a solution of (6.6.1) with the initial data ξ . Thus, by Lemma 6.6.8, we infer that

the processes u and \tilde{u} have the same law on the space \mathcal{Z}_T .

Hence

(6.6.18)
$$\tilde{\mathbb{E}}\left[\varphi(\tilde{u}(t))\right] = \mathbb{E}\left[\varphi(u(t))\right].$$

As before, the R.H.S. of (6.6.18) is equal by (6.6.2), to $T_t \varphi(\xi)$.

Thus by equations (6.6.16), (6.6.17) and (6.6.18), we infer

$$\lim_{k\to\infty} T_t \varphi(\xi_{n_k}) = T_t \varphi(\xi)$$

Using the subsequence argument, we can deduce that the whole sequence $(T_t \varphi(\xi_n))_{n \in \mathbb{N}}$ is convergent and

$$\lim_{n \to \infty} T_t \varphi(\xi_n) = T_t \varphi(\xi).$$

Thus the existence of an invariant measure is established by using Theorem 6.6.4, Lemmas 6.6.5 and 6.6.9; completing the proof of Theorem 6.6.1.


OPEN PROBLEMS AND FUTURE DIRECTIONS

n the final chapter of this thesis, we formulate some of the open problems that arise from the work carried out here and will also list down other related problems that I plan to explore in the future.

7.1 Open problems

7.1.1 Constrained Navier-Stokes equations on a general bounded domain

In Chapter 4, we proved the existence of the global solution to the CNSE

(7.1.1)
$$\begin{cases} \frac{du}{dt} + \mathrm{A}u - |\nabla u|_{L^2}^2 u + B(u, u) = 0, \\ u(0) = u_0 \in \mathrm{V} \cap \mathcal{M}, \end{cases}$$

on a two dimensional bounded domain with periodic boundary conditions. We were also able to establish the existence of local solutions to (7.1.1) as well as invariance of the manifold \mathcal{M} , with Dirichlet boundary conditions. Our approach of proving boundedness of the enstrophy using the gradient type structure of the equation obviously fails, since the orthogonal property (3.2.5) is not satisfied in this case. Thus, it will be interesting to see if one can prove the existence of a global solution to (7.1.1) on a general bounded domain, possibly by obtaining suitable bounds on the gradient norm of the solution.

7.1.2 Lower bound on the regularity of the initial data

At the end of Chapter 4 we provided a formal analysis to show that for the CNSE

(7.1.2)
$$\begin{cases} \frac{du}{dt} + Au - |\nabla u|^2_{L^2} u + B(u, u) = 0, \\ u(0) = u_0 \in \hat{\nabla} \cap \mathcal{M}, \end{cases}$$

where $\hat{V} = D(A^{\frac{\alpha}{2}})$ is as in (4.5.3), to be well-posed we need α to be at least $\frac{1}{2}$. It's important to note that we consider these equations on a bounded domain with the periodic boundary conditions. Since, the formal analysis indicates that the initial data u_0 at least needs to be in $D(A^{1/4})$ for (7.1.2) to be well posed, one should prove the same rigorously. Another reason to explore this problem is to compare it with the two dimensional Navier-Stokes equations, where the well-posedness results for $u_0 \in H$ and V (see [88]), for more regular initial data with a "compatibility condition" on the boundary of the domain (see [89]) and for $u_0 \in \hat{V} = D(A^{\frac{\alpha}{2}})$, $\alpha \in (1, \frac{3}{2})$ (see [12, Proposition 3.3]) are known.

7.1.3 Stochastic constrained Navier-Stokes equations on \mathbb{R}^2

In Chapter 5 we proved the existence of a strong pathwise unique solution to the SCNSE

(7.1.3)
$$\begin{cases} du(t) + [Au(t) + B(u(t))]dt = |\nabla u(t)|^2_{L^2}u(t)dt + Cu(t) \circ dW(t) \\ u(0) = u_0, \end{cases}$$

on a two dimensional bounded domain with periodic boundary conditions, i.e. on a torus. We believe that an analogous result holds also in the Euclidean space \mathbb{R}^2 , but so far we have not addressed this problem (even the existence of a martingale solution). Here we briefly describe a possible approach to the problem (inspired from Chapter 6 and [23]).

Step I

In the spirit of Chapter 6, one considers truncated equations on infinite dimensional space rather than Faedo-Galerkin approximations on finite dimensional space. The non-commutativity of the classical Faedo-Galerkin projection operator with the gradient operator (∇) on the whole Euclidean space, as mentioned in Chapter 6, is the motivation towards studying truncated SPDEs instead of approximated finite dimensional SDEs.

Step II

The second major difference between our method and the approach taken in the case of a torus is in the choice of the space \mathcal{Z}_T . We will have to prove the tightness of the laws of the solutions of the truncated SPDEs on a different space instead of the one used in the case of a torus, see Eq. 5.4.3. Due to the lack of compactness of embedding $H^1 \hookrightarrow L^2$ on \mathbb{R}^3 , we will use L^2_{loc} space to deduce strong convergence of approximating solutions. We will also require an auxiliary space U, as in Chapter 6. In contrast to [23] we will not use this space to prove convergence, but just to establish the Aldous condition, see Definition 2.9.10. Once we have the tightness of the laws, the rest of the proof of the existence of a martingale solution should be analogous to the existence proof for stochastic tamed Navier-Stokes equations, see Theorem 6.5.4.

7.1.4 Existence of invariant measures for stochastic constrained Navier-Stokes equations

In Theorem 5.8.1, we proved that the family of semigroups generated by the solution of (7.1.3) are sequentially weakly Feller, which along with the other necessary condition corresponding to boundedness in probability implies the existence of invariant measures. The proof will essentially use Theorem 6.6.4 (see [62]), but we haven't been able to verify the second condition and this remains for now an open problem.

7.1.5 Stochastic tamed Navier-Stokes equations: Invariant measures

Röckner and Zhang proved the existence of a unique invariant measure for the time homogeneous stochastic tamed Navier-Stokes equations on \mathbb{T}^3 [76]. We established the existence of invariant measures for the time homogeneous damped tamed Navier-Stokes equations

(7.1.4)
$$\begin{cases} du(t) = \left[-A_{\alpha}u(t) - B(u(t)) - \Pi[g(|u(t)|^2)u(t)] + \Pi f \right] dt + \sum_{j=1}^{\infty} G_j(u(t)) dW_t^j, \\ u(0) = u_0 \in \mathbb{V}, \end{cases}$$

where $A_{\alpha} = \alpha I - \nu \Delta$ for some $\alpha \in \mathbb{R}$ and $\nu > 0$, on the whole Euclidean space. The question of uniqueness of invariant measure remains open.

Another interesting problem in this direction could be to prove the existence of invariant measures for the time homogeneous tamed Navier-Stokes equations on a general bounded domain after establishing the existence of solutions.

7.2 Possible future research directions

7.2.1 Stochastic 2D viscous shallow water equations

Bresch and Desjardins [8] studied a two dimensional viscous shallow water model with friction term in a bounded domain with periodic boundary conditions. They proved the existence of global weak solutions. To the best of my knowledge, this problem has not been studied in a stochastic setting. Thus, I would like to study the well-posedness of the following SDE which will be a good starting point to understand the dynamics of such a model under random forcing.

(7.2.1)
$$\begin{cases} \partial_t h + \operatorname{div}(h \, u) = 0, \\ d(h \, u) + \left[\operatorname{div}(h \, u \otimes u) + \frac{(h \, u)^{\perp}}{R_0} + r_0 u + r_1 h | u | u \\ -\kappa h \nabla \Delta h + \frac{h \nabla h}{\mathrm{Fr}^2} - v \operatorname{div}(h \nabla u) \right] dt = h f \, dW, \end{cases}$$

supplemented with initial conditions

(7.2.2)
$$h|_{t=0} = h_0, \qquad (h u)|_{t=0} = m_0.$$

In the equation (7.2.1) u denotes the horizontal mean velocity field, h the depth variation, Fr > 0 the Froude number, $R_0 > 0$ the Rossby number and $\kappa \ge 0$ the capillary coefficient. The terms $r_0 u$ and $r_1 h |u| u$ correspond to the drag terms and $W = (W^1, W^2)$ is a two dimensional Wiener process where W^1 and W^2 are real-valued independent Brownian motions defined on a suitable probability space $(\Omega, \mathcal{F}, \mathbb{F}, \mathbb{P})$.

Even though the motivations behind the following two problems are completely different, they are related in the sense that both provide an approximation scheme for CNSE in deterministic and stochastic setting respectively. The motivation for studying such a constrained problem (7.2.5) is that these equations should be a better approximation of the Euler equations (for small viscosity) since, for the Euler equations, the energy of (sufficiently smooth) solutions is constant (see [31]).

7.2.2 Slightly compressible approximation of constrained Navier-Stokes equations

Rubinstein *et.al.* [78] constructed an asymptotic solution for small ε to the following reactiondiffusion problem on $\Omega \subset \mathbb{R}^m$:

(7.2.3)
$$\begin{cases} \frac{\partial u}{\partial t} = \varepsilon \Delta u - \varepsilon^{-1} \nabla V(u) \\ u(x,0,\varepsilon) = g(x), \qquad \partial_n u = 0 \text{ on } \partial \Omega. \end{cases}$$

They showed that at each $x \in \Omega$, u tends quickly to a minimum of V(u). Motivated by their work, I would like to show that asymptotically (as $\varepsilon \to 0$) the solution to

(7.2.4)
$$\frac{\partial u^{\varepsilon}}{\partial t} = v \Delta u^{\varepsilon} + u^{\varepsilon} \cdot \nabla u^{\varepsilon} - \varepsilon^{-1} \nabla_{L^2} \varphi(u^{\varepsilon})$$

on \mathbb{T}^2 , where

$$\varphi \colon \mathbb{R}^2 \ni u \to \frac{1}{4} (|u|_{\mathrm{H}}^2 - 1)^2 \in \mathbb{R},$$

converges to the solution of Navier-Stokes equations whose L^2 -norm is conserved and equal to 1:

(7.2.5)
$$\frac{\partial u}{\partial t} = v\Delta u + u \cdot \nabla u + v |\nabla u|_{L^2}^2 u.$$

We have already established the existence of the global solution to (7.2.5) on \mathbb{T}^2 in Chapter 4.

7.2.3 Stochastic hyperbolic constrained Navier-Stokes equations

The motion of a particle of mass μ under the impact of a random forcing $b(q) + \sigma(q)W$ with the damping coefficient proportional to the speed is described according to the Newton law

(7.2.6)
$$\mu \ddot{q}_{t}^{\mu} = b(q_{t}^{\mu}) + \sigma(q_{t}^{\mu}) \dot{W}_{t} - \alpha \dot{q}_{t}^{\mu}, \quad q_{0}^{\mu} = q \in \mathbb{R}^{n}, \quad \dot{q}_{0}^{\mu} = p \in \mathbb{R}^{n}.$$

In practice, the dynamics of the position q_t is of interest and thus in principal we are working with a system involving twice as many variables (q_t, \dot{q}_t) . Valid approximations that reduce the state space are crucial for both theoretical and computational applications.

Smoluchowski showed that for Brownian particle under an external field, there exists a Markov process which under certain circumstances is a good approximation to the position of the Ornstein-Uhlenbeck process (governs the dynamics of a Brownian particle). Developing on the same, it has been shown [67] that for very small μ , q_t^{μ} can be approximated by the solution of the first order equation

(7.2.7)
$$\dot{q}_t = b(q_t) + \sigma(q_t) \dot{W}_t, \quad q_0 = q \in \mathbb{R}^N.$$

In short, this small mass limit or the Smoluchowski-Kramers approximation of the system (7.2.6) reduces the state space from (q_t, \dot{q}_t) to q_t cutting the dimension of state space to half.

Apart from it's application in reducing the computational cost significantly Smoluchowski-Kramers approximation is also used by several mathematicians and physicists to study fundamental mathematical notions, like invariant measures and large time behaviour. In particular, Cerrai and Freidlin [34, 35, 43] have shown that the solution of damped wave equations perturbed by stochastic forcing converges to the solution of corresponding stochastic heat equation. They also established relations between stationary distributions and large deviations of a general class of SPDEs and their limiting equations. Existence of such convergences make it possible to analyse the simpler equation (first order) in order to understand the large time behaviour and other asymptotics of second order equation.

Brenier *et.al.* [7] showed that the solution of a damped wave equation converges to the solution of Navier-Stokes equations on a two dimensional torus in small mass limit, which could be seen as a deterministic version of Smoluchowski-Kramers approximation. I would like to consider a similar problem in the context of stochastic hyperbolic CNSEs. To be precise, I would like to show that as $\mu \rightarrow 0$ the solution of

(7.2.8)
$$\mu \frac{\partial^2 u^{\mu}}{\partial t^2} + \frac{\partial u^{\mu}}{\partial t} = v \Delta u^{\mu} + u^{\mu} \cdot \nabla u^{\mu} + v |\nabla u^{\mu}|^2_{L^2} u^{\mu} + g(\cdot, u^{\mu}) \frac{\partial W}{\partial t},$$

converges to the solution of SCNSE with the same external forcing. This would also enable us to address the questions of stationary distributions and large deviations for the stochastic hyperbolic CNSE.



ORTHOGONALITY OF BILINEAR MAP TO THE STOKES OPERATOR

In this appendix we show that the bilinear map $B: V \times V \to V'$, defined in Chapter 3 is orthogonal to the Stokes operator A in H on \mathbb{R}^2 . This in fact holds true for any bounded domain with periodic boundary conditions and the proof for that can be found in [90]. The proof of the following lemma is motivated from the same.

Lemma A.1. Let $x \in \mathbb{R}^2$ and $u \in D(A)$, then

(A.0.1)
$$\langle B(u,u), Au \rangle_{\mathrm{H}} = 0, \ \forall u \in \mathrm{D}(\mathrm{A}).$$

Proof. Let $u \in D(A)$ then, by the definition of B(u, v) and Au,

$$\langle B(u,u), Au \rangle_{\mathrm{H}} = \int_{\mathcal{O}} (u(x) \cdot \nabla) u(x) \cdot Au(x) dx$$

$$= \sum_{i,j,k=1}^{2} \int_{\mathcal{O}} (u_{i}D_{i}u_{j})(-\Delta u_{j}) dx$$

$$= -\sum_{i,j,k=1}^{2} \int_{\mathcal{O}} u_{i}D_{i}u_{j}D_{k}^{2}u_{j} dx.$$

Now by integration by parts and the Stokes formula

$$\langle B(u,u), Au \rangle_{\mathrm{H}} = -\left(\sum_{i,j,k=1}^{2} u_i D_i u_j D_k u_j\right) \Big|_{\partial \mathcal{O}} + \sum_{i,j,k=1}^{2} \int_{\mathcal{O}} D_k (u_i D_i u_j) D_k u_j dx = \sum_{i,j,k=1}^{2} \int_{\mathcal{O}} D_k u_i D_i u_j D_k u_j dx + \sum_{i,j,k=1}^{2} \int_{\mathcal{O}} u_i D_{ki} u_j D_k u_j dx.$$

Now we will show that each of the terms in RHS will vanish. We will consider the first term and show that it vanishes.

$$\begin{split} \sum_{i,j,k=1}^{2} D_{k} u_{i} D_{i} u_{j} D_{k} u_{j} &= (D_{1} u_{1})^{3} + D_{1} u_{2} D_{2} u_{1} D_{1} u_{1} + D_{1} u_{1} (D_{1} u_{2})^{2} + (D_{1} u_{2})^{2} D_{2} u_{2} \\ &+ (D_{2} u_{1})^{2} D_{1} u_{1} + D_{2} u_{2} (D_{2} u_{1})^{2} + D_{2} u_{1} D_{1} u_{2} D_{2} u_{2} + (D_{2} u_{2})^{3} \\ &= (D_{1} u_{1} + D_{2} u_{2}) \left[(D_{1} u_{1})^{2} + (D_{2} u_{2})^{2} - D_{1} u_{1} D_{2} u_{2} \right] \\ &+ D_{1} u_{2} D_{2} u_{1} (D_{1} u_{1} + D_{2} u_{2}) + (D_{1} u_{2})^{2} (D_{1} u_{1} + D_{2} u_{2}) \\ &+ (D_{2} u_{1})^{2} (D_{1} u_{1} + D_{2} u_{2}). \end{split}$$

Now since $\nabla \cdot u = D_1 u_1 + D_2 u_2 = 0$, the first term vanishes identically.

The second term vanishes because

$$2\sum_{i,j,k=1}^{2} \int_{\mathcal{O}} u_{i} D_{ki} u_{j} D_{k} u_{j} dx = \sum_{i,j,k=1}^{2} \int_{\mathcal{O}} u_{i} D_{i} (D_{k} u_{j})^{2} dx$$
$$= \left(\sum_{i,j,k=1}^{2} u_{i} (D_{k} u_{j})^{2}\right) \Big|_{\partial \mathcal{O}} - \sum_{i,j,k=1}^{2} \int_{\mathcal{O}} D_{i} u_{i} (D_{k} u_{j})^{2} dx$$
$$= -\sum_{j,k=1}^{2} \int_{\mathcal{O}} (\nabla \cdot u) (D_{k} u_{j})^{2} dx = 0.$$

Thus we have shown that for every $u \in D(A)$, $\langle B(u,u), Au \rangle_{H} = 0$.



Some results in the support of Section 4.4

This appendix is dedicated to the proof of Remark 4.4.1, which plays a crucial role in proving the convergence of solution of Constrained Navier-Stokes to the solution of Euler equations in the inviscid limit, Theorem 4.1.2.

Remark B.1. If $\nabla \cdot u = 0$ and $\operatorname{Curl}(u) = 0$, then u is constant by Hodge decomposition. In particular, if $u \in V$ and $\operatorname{Curl}(u) = 0$, then u = 0.

Proof of Remark 4.4.1. We want to show that Curl is a linear isomorphism between V and $L^2_0(\mathbb{T}^2)$. It is clear that the map

$$\operatorname{Curl}: \mathbf{V} \ni u \mapsto \omega = \operatorname{Curl}(u) \in L^2_0(\mathbb{T}^2),$$

is linear and continuous. Hence in order to prove the Remark 4.4.1 it is sufficient to find a continuous linear map

$$(B.0.1) \qquad \qquad \Lambda: L^2_0(\mathbb{T}^2) \to V$$

such that,

(B.0.2)
$$\operatorname{Curl} \circ \Lambda = \operatorname{id} \operatorname{on} L_0^2(\mathbb{T}^2),$$

(B.0.3)
$$\Lambda \circ \operatorname{Curl} = \operatorname{id} \operatorname{on} V.$$

Let $\omega \in L_0^2(\mathbb{T}^2)$ then by elliptic regularity [46] (applies also for $p \neq 2$) there exists a unique $\psi \in L_0^2(\mathbb{T}^2) \cap H^2(\mathbb{T}^2)$ such that

 $(B.0.4) \qquad \qquad \Delta \psi = \omega,$

and the map

$$L_0^2 \ni \omega \mapsto \psi \in L_0^2 \cap H^2,$$

is bounded. Let us put $u = -\nabla^{\perp} \psi$, i.e.

(B.0.5)
$$u = (-D_2 \psi, D_1 \psi).$$

Then $u \in H^1(\mathbb{T}^2)$ and $\nabla \cdot u = 0$ in the weak sense. Thus $u \in V$. Using all of this we define the bounded linear map $\Lambda : L^2_0(\mathbb{T}^2) \ni \omega \mapsto u \in V$. Now we are left to check that (B.0.2) and (B.0.3) holds for this Λ .

Let us take $\omega \in L^2_0(\mathbb{T}^2)$ and put $u := \Lambda(\omega) \in V$. Now considering LHS of (B.0.2),

$$(\operatorname{Curl} \circ \Lambda)(\omega) = \operatorname{Curl}(u) = D_1 u_2 - D_2 u_1$$
$$= D_1 D_1 \psi - (-D_2 D_2 \psi) = \Delta \psi = \omega$$

where we have used the definitions of ψ and u from (B.0.4) and (B.0.5). Hence we have established (B.0.2).

Now we take $v \in V$ and put $\omega = \operatorname{Curl}(v) \in L^2_0(\mathbb{T}^2)$. Define $\psi \in L^2_0(\mathbb{T}^2) \cap H^2(\mathbb{T}^2)$ by

$$(B.0.6) \qquad \qquad \Delta \psi = \omega.$$

Observe that in view of (4.4.2) we have

$$\Delta \varphi = \operatorname{Curl}(-D_2 \varphi, D_1 \varphi), \quad \varphi \in H^2(\mathbb{T}^2).$$

Thus by (B.0.6) and the definition of u from (B.0.5) we obtain

$$\operatorname{Curl}(u) = \operatorname{Curl}(v),$$

where $u = -\nabla^{\perp} \psi \in V$. Therefore using Remark B.1 u = v, thus proving that Curl is a linear isomorphism between V and $L_0^2(\mathbb{T}^2)$. It is straightforward to show (4.4.3). Thus we are left to prove (4.4.4).

Let us fix $p \in (1,\infty)$ and take $u \in H^{1,p}(\mathbb{T}^2)$. Denote $\omega = \operatorname{Curl}(u) \in L_0^p(\mathbb{T}^2)$. From the first part of the proof there exists a bounded linear map $\Lambda : L_0^p(\mathbb{T}^2) \to H^{1,p}(\mathbb{T}^2)$

$$\Lambda: L^p_0 \ni \omega \mapsto u \in H^{1,p},$$

such that

Curl
$$\circ \Lambda$$
 = id on $L_0^p(\mathbb{T}^2)$.

In particular, there exists a $C'_p > 0$,

$$\|\Lambda\omega\|_{H^{1,p}(\mathbb{T}^2)} \le C'_p \|\omega\|_{L^p(\mathbb{T}^2)}, \quad \omega \in L^p_0(\mathbb{T}^2).$$

Hence

(B.0.7)
$$\|\nabla \Lambda \omega\|_{L^p(\mathbb{T}^2)} \le C'_p \|\omega\|_{L^p(\mathbb{T}^2)}, \quad \omega \in L^p_0(\mathbb{T}^2)$$

Taking now $u \in H^{1,p}(\mathbb{T}^2)$. Putting $\omega = \operatorname{Curl}(u)$ so that $\Lambda \omega = u$ from (B.0.7) we infer (B.0.8),

(B.0.8) $\|\nabla u\|_{L^{p}(\mathbb{T}^{2})} \leq C_{p} \|\omega\|_{L^{p}(\mathbb{T}^{2})}.$

Now since $\|\omega\|_{L^p(\mathbb{T}^2)} \le \|\omega\|_{L^{\infty}(\mathbb{T}^2)}$ for every p, we can establish (4.4.4).



KURATOWSKI THEOREM

The main objective of this appendix is to establish the preliminaries that are required to prove Lemmas 5.7.1 and 6.5.5. The proof of lemmas heavily rely on the Kuratowski Theorem [70, Theorem 3.9], which we recall below for the sake of completeness.

Theorem C.1. Assume that X_1, X_2 are the Polish spaces with their Borel σ -fields denoted respectively by $\mathscr{B}(X_1), \mathscr{B}(X_2)$. If $\varphi: X_1 \to X_2$ is an injective Borel measurable map then for any $E_1 \in \mathscr{B}(X_1), E_2 := \varphi(E_1) \in \mathscr{B}(X_2)$.

Next two lemmas are the main results of this appendix.

Lemma C.1. Let X_1, X_2 and Z be topological spaces such that X_1 is a Borel subset of X_2 . Then $X_1 \cap Z$ is a Borel subset of $X_2 \cap Z$, where $X_2 \cap Z$ is a topological space too, with the topology given by

(C.0.1)
$$\tau(X_2 \cap Z) = \{A \cap B : A \in \tau(X_2), B \in \tau(Z)\}.$$

Proof. Since the Borel σ -filed on $X_2 \cap Z$ is the smallest σ -field generated by $\tau(X_2 \cap Z)$, i.e. $\mathscr{B}(X_2 \cap Z) = \sigma(\tau(X_2 \cap Z))$, in order to prove the lemma it is enough to show that $\forall Y \in \mathscr{B}(X_1)$

(C.0.2)
$$Y \cap Z \in \mathscr{B}(X_2 \cap Z).$$

Firstly, we show that (C.0.2) holds for all $Y \in \tau(X_1)$. Since $X_1 \in \mathscr{B}(X_2)$, $X_1 \subset X_2$ and has trace topology from X_2 , i.e $\forall Y \in \tau(X_1)$ there exists a $C \in \tau(X_2)$ such that

$$Y = C \cap X_1.$$

As $X_1 \in \mathscr{B}(X_2)$ there exists a countable collection $\{K_i\}_{i \in \mathbb{N}}$ of open subsets of X_2 such that

$$X_1 = \bigcup_{i \in \mathbb{N}} K_i \; .$$

Therefore,

$$Y \cap Z = C \cap X_1 \cap Z = C \cap \left(\bigcup_{i \in \mathbb{N}} K_i\right) \cap Z = \bigcup_{i \in \mathbb{N}} (C \cap K_i) \cap Z$$

Since $C \in \tau(X_2)$, for every $i \in \mathbb{N}$, $C \cap K_i$ is open in X_2 and there exists a collection $\{B_j\}_{j \in \mathbb{N}} \in \tau(X_2)$ such that

$$\bigcup_{i\in\mathbb{N}} (C\cap K_i) = \bigcup_{j\in\mathbb{N}} B_j$$

Thus

$$Y \cap Z = \bigcup_{j \in \mathbb{N}} (B_j \cap Z)$$
,

and for every $j \in \mathbb{N}$, $B_j \cap Z \in \mathscr{B}(X_2 \cap Z)$. Since $\mathscr{B}(X_2 \cap Z)$ is a σ -field, the countable union also belongs to $\mathscr{B}(X_2 \cap Z)$, proving (C.0.2) for every $Y \in \tau(X_1)$.

Secondly, we implement the method of good sets to prove (C.0.2) for a larger class of subsets of X_1 . Let

$$\mathscr{G} = \{ A \subset X_1 \colon A \cap Z \in \mathscr{B}(X_2 \cap Z) \}.$$

Claim: \mathscr{G} is a σ -field.

- i) $X_1 \in \mathcal{G}$ since $X_1 \subset X_1$ and $X_1 \in \tau(X_1)$ by the definition of topology.
- ii) Let $A \in \mathcal{G}$. We want to show that $A^c := X_1 \setminus A \in \mathcal{G}$, i.e. $A^c \subset X_1$ and $A^c \cap Z \in \mathcal{B}(X_2 \cap Z)$. Since $A \in \mathcal{G}$, $A \subset X_1$ and $A \cap Z \in \mathcal{B}(X_2 \cap Z)$. Clearly $A^c = X_1 \setminus A \subset X_1$. Since $A \cap Z \in \mathcal{B}(X_2 \cap Z)$, then by the definition of σ -field

$$^{c}(A \cap Z) := (X_{2} \cap Z) \setminus (A \cap Z) \in \mathscr{B}(X_{2} \cap Z).$$

We have the following set relations

$$\begin{split} {}^{c}(A \cap Z) &= {}^{c}A \cup {}^{c}Z = \left[(X_{2} \cap Z) \setminus A \right] \cup \left[(X_{2} \cap Z) \setminus Z \right] \\ &= \left[(X_{2} \setminus A) \cap Z \right] \cup \emptyset = (X_{2} \setminus A) \cap Z \\ &= \left[A^{c} \cup (X_{2} \setminus X_{1}) \right] \cap Z \\ &= \left(A^{c} \cap Z \right) \cup \left[(X_{2} \setminus X_{1}) \cap Z \right] \\ &= \left(A^{c} \cap Z \right) \cup \left[(X_{2} \setminus X_{1}) \cap Z \right] \end{split}$$

Now in the above identity ${}^{c}(A \cap Z)$, ${}^{c}X_{1}$ belongs to $\mathscr{B}(X_{2} \cap Z)$ and hence $A^{c} \cap Z \in \mathscr{B}(X_{2} \cap Z)$, inferring $A^{c} \in \mathscr{G}$.

iii) Let $\{A_i\}_{i \in \mathbb{N}} \in \mathcal{G}$. Then $A_i \subset X_1$ for every $i \in \mathbb{N}$ hence

$$\bigcup_{i\in\mathbb{N}}A_i\subset X_1.$$

Also, the following holds

$$\left(\bigcup_{i\in\mathbb{N}}A_i\right)\cap Z=\bigcup_{i\in\mathbb{N}}(A_i\cap Z)$$

Since $A_i \in \mathcal{G}, A_i \cap Z \in \mathcal{B}(X_2 \cap Z)$ and as $\mathcal{B}(X_2 \cap Z)$ is a σ -field

$$\bigcup_{i\in\mathbb{N}} (A_i\cap Z)\in\mathscr{B}(X_2\cap Z).$$

From *i*) – *iii*) we can infer that \mathscr{G} is a σ -field. We have already shown that $\tau(X_1) \subset \mathscr{G}$ thus

$$\mathscr{B}(X_1) = \sigma(\tau(X_1)) \subset \mathscr{G}.$$

Therefore, we have shown that for every $Y \in \mathscr{B}(X_1), Y \cap Z \in \mathscr{B}(X_2 \cap Z)$.

Lemma C.2. Let X_1, X_2, Y be topological spaces such that $X_1 \subset X_2$, X_1 has trace topology from X_2 and $X_1 \cap Y = X_2 \cap Y$ then

$$\tau(X_1 \cap Y) = \tau(X_2 \cap Y).$$

Proof. The topologies of $X_1 \cap Y$ and $X_2 \cap Y$ denoted by $\tau(X_1 \cap Y)$ and $\tau(X_2 \cap Y)$ respectively are given by

$$\tau(X_1 \cap Y) = \text{generated by } \{A \cap B : A \in \tau(X_1), B \in \tau(Y)\},\$$

$$\tau(X_2 \cap Y) = \text{generated by } \{C \cap B : C \in \tau(X_2), B \in \tau(Y)\}.$$

Since X_1 has a trace topology from X_2 , for every $A \in \tau(X_1)$ there exists a $C \in \tau(X_2)$ such that $A = C \cap X_1$. Thus

$$\tau(X_1 \cap Y) = \text{generated by } \{C \cap X_1 \cap B \colon C \in \tau(X_2), B \in \tau(Y)\}.$$

Thus all we are left to show is $C \cap X_1 \cap B = C \cap B$ for every $C \in \tau(X_2)$ and $B \in \tau(Y)$. Since $X_1 \cap Y = X_2 \cap Y$, we have the following set relations

$$C \cap X_1 \cap B = (C \cap X_1) \cap (Y \cap B) = (C \cap X_1 \cap Y) \cap B$$
$$= (C \cap X_2 \cap Y) \cap B = (C \cap X_2) \cap (Y \cap B) = C \cap B.$$



CONVERGENCE OF P_n

Here we present various convergence results for the orthogonal projection P_n as given by (6.4.1) (for more details see Section 6.4) as $n \to \infty$.

Lemma D.1. Let $\gamma > \frac{d}{2}$ and $P_n : H \to H_n$ be the orthogonal projection. Then as $n \to \infty$

- (i) $P_n \psi \to \psi$ in H for $\psi \in \mathbf{H}$,
- (*ii*) $P_n \psi \to \psi$ in V for $\psi \in V$,
- (*iii*) $P_n \psi \rightarrow \psi$ in V_{γ} for $\psi \in V_{\gamma}$.

Proof. Let $\psi \in H$, then by (6.4.1) and Plancherel Theorem, we have

$$|P_n\psi - \psi|_{\mathrm{H}}^2 = \int_{\mathbb{R}^3} |\mathscr{F}(P_n\psi)(\xi) - \hat{\psi}(\xi)|^2 d\xi = \int_{\mathbb{R}^3} |\mathbb{I}_{B_n}(\xi)\hat{\psi}(\xi) - \hat{\psi}(\xi)|^2 d\xi = \int_{|\xi| > n} |\hat{\psi}(\xi)|^2 d\xi.$$

Now since $\psi \in H$ using Lebesgue dominated convergence theorem it can be shown that

$$\lim_{n\to\infty}\int_{|\xi|>n}|\hat{\psi}(\xi)|^2\,d\xi=0,$$

which infers (i).

Let $\psi \in V$, then by (6.4.1) and the definition of V–norm we get

$$\begin{split} \|P_n\psi - \psi\|_{\mathbf{V}}^2 &= \int_{\mathbb{R}^3} (1 + |\xi|^2) \left| \mathscr{F}(P_n\psi)(\xi) - \hat{\psi}(\xi) \right|^2 d\xi = \int_{\mathbb{R}^3} (1 + |\xi|^2) \left| \mathbbm{1}_{B_n}(\xi) \hat{\psi}(\xi) - \hat{\psi}(\xi) \right|^2 d\xi \\ &= \int_{|\xi| > n} (1 + |\xi|^2) |\hat{\psi}(\xi)|^2 d\xi. \end{split}$$

Again using the Lebesgue dominated convergence theorem and the fact that $\psi \in V$, we can show that

$$\lim_{n \to \infty} \int_{|\xi| > n} (1 + |\xi|^2) |\hat{\psi}(\xi)|^2 d\xi = 0,$$

thus proving (ii).

Let $\psi \in V_{\gamma},$ then by (6.4.1) and the definition of $V_{\gamma}-norm$ we get

$$\begin{split} \|P_n\psi - \psi\|_{\mathcal{V}_{\gamma}}^2 &= \int_{\mathbb{R}^3} (1 + |\xi|^2)^{\gamma} \left| \mathscr{F}(P_n\psi)(\xi) - \hat{\psi}(\xi) \right|^2 d\xi = \int_{\mathbb{R}^3} (1 + |\xi|^2)^{\gamma} \left| \mathbb{1}_{B_n}(\xi) \hat{\psi}(\xi) - \hat{\psi}(\xi) \right|^2 d\xi \\ &= \int_{|\xi| > n} (1 + |\xi|^2)^{\gamma} \left| \hat{\psi}(\xi) \right|^2 d\xi. \end{split}$$

Similarly as before it can be shown that

$$\lim_{n \to \infty} \int_{|\xi| > n} (1 + |\xi|^2)^{\gamma} |\hat{\psi}(\xi)|^2 d\xi = 0,$$

as $\psi \in V_{\gamma}$, which concludes the proof.

BIBLIOGRAPHY

- S. Albeverio, Z. Brzeźniak and J. Wu Existence of global solutions and invariant measures for stochastic differential equations driven by Poisson type noise with non-Lipschitz coefficients, Journal of Mathematical Analysis and Applications. 371, 309 – 322 (2010).
- [2] F. Alouges and A. Soyeur, On global weak solutions for Landau-Lifshitz equations: existence and nonuniqueness, Nonlinear Analysis 18, 1071 – 1084 (1992).
- [3] A. Badrikian, Séminaire sur les fonctions aléatoires linéaires et les mesures cylindriques, Lecture Notes in Mathematics, 139, Springer Verlag (1970).
- [4] L. Banas, Z. Brzeźniak, M. Neklyudov, and A. Prohl, Stochastic ferromagnetism: analysis and numerics, Studies in Mathematics, Berlin, Boston: De Gruyter (2013).
- [5] C. Bardos, Existence et unicité de la solution de l'équation d'Euler en dimension deux, Journal of Mathematical Analysis and Applications, 40, 769 – 790 (1972).
- [6] V. Barbu, G. Da Prato and L. Tubaro, A reflection type problem of the stochastic 2-D Navier-Stokes equations with periodic conditions, Electronic Communications in Probability 16, 304 – 313 (2011).
- [7] Y. Brenier, R. Natalini and M. Ouel, On a relaxation approximation of the incompressible Navier-Stokes equations, Proceedings of the American Mathematical Society, 132(4), 1021 – 1028 (2004).
- [8] D. Bresch and B. Desjardins, Existence of global weak solutions for a 2D viscous shallow water equations and convergence to the quasi-geostrophic model, Communications in Mathematical Physics, 238, 211 – 223 (2003).
- [9] H. Brezis, Analyse fonctionnelle, Masson (1983).
- [10] Z. Brzeźniak, and A. Carroll, Approximations of the Wong ñ Zakai type for stochastic differential equations in M-type 2 Banach spaces with applications to loop spaces, Séminare de Probabilités XXXVII, Springer Lecture Notes in Mathematics 1832, 251 – 289 (2003).
- [11] Z. Brzeźniak and A. Carroll, The stochastic geometric heat equation, In preparation.
- [12] Z. Brzeźniak, S. Cerrai and M. Freidlin Quasipotential and exit time for 2D Stochastic Navier-Stokes equations driven by space time white noise, Probability Theory and Related Fields, 162(3), 739 – 793 (2015).
- [13] Z. Brzeźniak and G. Dhariwal, Stochastic Constrained Navier-Stokes Equations on \mathbb{T}^2 ,

Preprint arxiv:1701.01385v2 (2017).

- [14] Z. Brzeźniak, G. Dhariwal and M. Mariani, 2D Constrained Navier-Stokes equations, Preprint arXiv:1606.08360v2 (Submitted, 2016).
- [15] Z. Brzeźniak and K. D. Elworthy, Stochastic differential equations on Banach manifolds, Methods of Functional Analysis and Topology, 6, 43 – 84 (2000).
- [16] Z. Brzeźniak and B. Ferrario, 2D Navier-Stokes equation in Besov spaces of negative order, Nonlinear Analysis, 70, 3902 – 3916 (2009).
- [17] Z. Brzeźniak, B. Ferrario, Stationary solutions for stochastic damped Navier-Stokes equations in \mathbb{R}^d , (In preparation, 2016).
- [18] Z. Brzeźniak, F. Flandoli and M. Maurelli, Existence and uniqueness for stochastic 2D Euler flows with bounded vorticity, Archive for Rational Mechanics and Analysis, 221(1), 107 – 142 (2016).
- [19] Z. Brzeźniak, B. Goldys, and T. Jegaraj, Weak solutions of a stochastic Landau-Lifshitz-Gilbert Equation, Applied Mathematics Research Express 2013, 1, 1 – 33 (2013).
- [20] Z. Brzeźniak and E. Hausenblas, Martingale solutions for stochastic equations of reaction diffusion type driven by Lévy noise or Poisson random measure, Preprint arXiv:1010.5933v1 (2010).
- [21] Z. Brzeźniak and J. Hussain, *Global solutions of non-linear heat equation with solutions in a Hilbert manifold*. In preparation.
- [22] Z. Brzeźniak and L. Li, Weak solutions of the Stochastic LandauñLifshitzñGilbert Equations with nonzero anisotrophy energy, Applied Mathematical Research Express 2016, 2, 334 – 375 (2016).
- [23] Z. Brzeźniak and E. Motyl, Existence of a martingale solution of the stochastic Navier-Stokes equations in unbounded 2D and 3D-domains, Journal of Differential Equations, 254(4), 1627 – 1685 (2013).
- [24] Z. Brzeźniak and E. Motyl, The existence of martingale solutions to the stochastic Boussinesq equations, Global and Stochastic Analysis, 1(2), 175 – 216 (2014).
- [25] Z. Brzeźniak, E. Motyl and M. Ondreját, *Invariant measure for the stochastic Navier-Stokes* equations in unbounded 2D domains, Submitted to the Annals of Probability (2016).
- [26] Z. Brzeźniak and M. Ondreját, Weak solutions to stochastic wave equations with values in Riemannian manifolds, Communications in Partial Differential Equations, 36, 1624 – 1653 (2011).
- [27] Z. Brzeźniak and M. Ondreját, Stochastic geometric wave equations with values in compact Riemannian homogeneous spaces, Annals of Probability, 41, 1938 – 1977 (2013).
- [28] Z. Brzeźniak and Y. Zastawniak, Basic stochastic processes, Springer (1999).
- [29] N. Burq, P. Gérard, N. Tzvetkov, The Cauchy problem for the nonlinear Schrödinger equation

on a compact manifold, Journal of Nonlinear Mathematical Physics, 10(1), 12 – 27 (2003).

- [30] L. Caffarelli and F. Lin, Nonlocal Heat Flows preserving the L^2 Energy, Discrete and Continuous Dynamical Systems, 23, 49 64 (2009).
- [31] E. Caglioti, M. Pulvirenti and F. Rousset, On a constrained 2D Navier-Stokes Equation, Communications in Mathematical Physics, 290, 651 – 677 (2009).
- [32] A. Carroll, The stochastic nonlinear heat equation, PhD Thesis, University of Hull (1999).
- [33] L. Cattabriga, Su un problema al contorno relativo al sistema di equazioni di Stokes, Rendiconti del Seminario Matematico della Università di Padova, 31, 308 – 340 (1961).
- [34] S. Cerrai and M. Freidlin, Smoluchowski-Kramers approximation for a general class of SPDEs, Journal of Evolution Equations, 6, 657 – 689 (2006).
- [35] S. Cerrai and M. Freidlin, On the Smoluchowski-Kramers approximation for a system with an infinite number of degrees of freedom, Probability Theory and Related Fields, 135, 363 – 394 (2006).
- [36] J. Chemin, Perfect Incompressible Fluids, Oxford University Press (1998).
- [37] P. Constantin and J. Wu, Inviscid limit for vortex patches, Nonlinearity, 8, 735 742 (1995).
- [38] G. Da Prato and J. Zabczyk, Stochastic Equations in Infinite dimensions, Second Edition, Cambridge University Press (2014).
- [39] L. de Simon, Uníapplicazione della teoria degli integrali singolari allo studio delle equazioni differenziali lineari astratte del primo ordine, Rendiconti del Seminario Matematico della Università di Padova, 34, 205 – 223 (1964).
- [40] J. Duan and A. Millet, Large deviations for the Boussinesq equations under random influences, Stochastic Processes and their Applications, 119(6), 2052 – 2081 (2009).
- [41] J. Eells and J. H. Sampson, *Harmonic mappings of Riemannian manifolds*, American Journal of Mathematics, 86, 109 – 160 (1964).
- [42] C. L. Fefferman, D. S. McCormick, J. C. Robinson and J. L. Rodrigo, Higher order commutator estimates and local existence for the non-resistive MHD equations and related models, Journal of Functional Analysis, 267, 1035 – 1056 (2014).
- [43] M. Freidlin, Some remarks on the Smoluchowski-Kramers approximation, Journal of Statistical Physics, 117(3/4), 617 – 634 (2004).
- [44] T. Funaki, A stochastic partial differential equation with values in manifold, Journal of Functional Analysis, 109, 257 – 258 (1992).
- [45] I. Gallagher, and F. Planchon, On global infinite energy solutions to the Navier-Stokes equations in two dimensions, Archive for Rational Mechanics and Analysis, 161, 307 – 337 (2002).
- [46] D. Gilbarg, and N. Trudinger, Elliptic Partial Differential Equations of Second Order, Springer-Verlag Berlin Heidelberg (2001).

- [47] E. Hopf, *Uber die Anfangswertaufgabe für die hydrodynamischen Grundgleichungen*, Mathematische Nachrichten, **4**, 213 231 (1951).
- [48] J. Hussain Analysis of some deterministic and stochastic evolution equations with solutions taking values in an infinite dimensional Hilbert manifold, PhD Thesis, University of York (2015).
- [49] A. Jakubowski, The almost sure Skorokhod representation for subsequences in nonmetric spaces, Teoriya Veroyatnostei i ee Primeneniya, 42(1), 209 216 (1997); translation in Theory of Probability and its Applications, 42(1), 167 174 (1998).
- [50] J. Jost, Postmodern Analysis, Third Edition, Springer-Verlag (2005).
- [51] T. Kato, A remark on a theorem of C. Bardos on the 2D-Euler equation.
- [52] T. Kato, and G. Ponce, Well-posedness of the Euler and Navier-Stokes equations in the Lebesgue spaces $L_s^p(\mathbb{R}^2)$, Revista Matemática Iberoamericana, **2**, 73 88 (1986).
- [53] K. Kuratowski, Topologie, Vol. I (French) 3'eme ed. Monografie Matematyczne, Tom XX, Polskie Towarzystwo Matematyczne, Warszawa, (1952).
- [54] O. A. Ladyzhenskaia, Solution "in the large" of the nonstationary boundary value problems for the Navier-Stokes system with two space variables, Communications on Pure and Applied Mathematics, 12(3), 427 – 433 (1959).
- [55] J. Leray, Sur le mouvement d'un liquide visquex emplissant l'espace, Acta Mathematica, 63, 193 – 248 (1934).
- [56] J. L. Lions, Quelques méthodes de résolution des problèmes aux limites non linéaires, Dunod, Paris (1969).
- [57] G. Łukaszewicz and P. Kalita, Navier-Stokes Equations An Introduction with Application, Springer International Publishing (2016).
- [58] A. Lunardi, Interpolation Theory, Second Edition, Edizioni della Normale (2009).
- [59] T. J. Lyons, Differential equations driven by rough signals, Revista Matemática Iberoamericana 14(2), 215 – 310 (1998).
- [60] D. S. Malik, J. M. Moderson and M. K. Sen, Fundamentals of abstract algebra, McGraw-Hill (1997).
- [61] U. Manna and A. A. Panda *Higher order regularity and blow-up criterion for semi-dissipative and ideal Boussinesq equations*, Submitted (2016).
- [62] B. Maslowski and J. Seidler, On sequentially weakly Feller solutions to SPDE's, Rendiconti Lincei Matematica e Applicazioni. 10, 69 – 78 (1999).
- [63] M. Métivier, Semimartingales, Gruyter (1982).
- [64] M. Métivier, Stochastic Partial Differential Equations in Infinite Dimensional Spaces, Scuola Normale Superiore, Pisa (1988).
- [65] R. Mikulevicius and B.L. Rozovskii, Global L₂-solutions of stochastic Navier-Stokes equa-

tions, Annals of Probability, 33(1), 137 – 176 (2005).

- [66] E. Motyl, Stochastic hydrodynamic-type evolution equations driven by Lévy noise in 3D unbounded domains - abstract framework and applications, Stochastic Processes and their Applications 124, 2052 – 2097 (2014).
- [67] E. Nelson, *Dynamical theories of Brownian motion*, Second Edition, Princeton University Press (2001).
- [68] M. Ondreját, Uniqueness for stochastic evolution equations in Banach spaces, Dissertationes Mathematicae, **426**, 1 63 (2004).
- [69] M. Ondreját, Brownian representations of cylindrical local martingales, martingale problems and strong Markov property of weak solutions of SPDEs in Banach spaces, Czechoslovak Mathematical Journal, 55(130), 1003 – 1039 (2005).
- [70] K. R. Parthasarathy, Probability measures on metric spaces, Academic Press, New York (1967).
- [71] E. Pardoux, Équations aux dérivées partielles stochastiques de type monotone, Séminaire Jean Leray, 3, 1 – 10 (1975).
- [72] E. Pardoux, Stochastic partial differential equations and filtering of diffusion processes, Stochastics, 3, 127 – 167 (1979).
- [73] A. Pazy, Semigroups of Linear Operators and Applications to Partial Differential Equations, Springer-Verlag New York (1983).
- [74] H. Pham, Continuous-time stochastic control and optimization with financial applications, Springer (2009).
- [75] M. Röckner and X. Zhang, Tamed 3D Navier-Stokes Equation: Existence, Uniqueness and Regularity, Infinite Dimensional Analysis, Quantum Probability and Related Topics, 12(4), 525 – 549 (2009).
- [76] M. Röckner and X. Zhang, Stochastic Tamed 3D Navier-Stokes Equation: Existence, Uniqueness and Ergodicity, Probability Theory and Related Fields, 145, 211 – 267 (2009).
- [77] T. Runst and W. Sickel, Sobolev Spaces of fractional order, Nemytsky operators and Nonlinear Partial differential equations, W. de Gruyter and Co., Berllin (1996).
- [78] J. Rubinstein, P. Sternberg and J. B. Keller, Fast reaction, slow diffusion and curve shortening, SIAM Journal on Applied Mathematics, 49(1), 116 – 133 (1989).
- [79] W. Rudin, Real and complex analysis, Third Edition, McGraw-Hill (1987).
- [80] W. Rudin, Functional Analysis, Second Edition, McGraw-Hill (1991).
- [81] P. Rybka, Convergence of a heat flow on a Hilbert manifold, Proceedings of the Royal Society of Edinburgh, 136A, 851 – 862 (2006).
- [82] B. Schmalfuss, Qualitative properties of the stochastic Navier-Stokes equations, Nonlinear Analysis, 28(9), 1545 – 1563 (1997).

- [83] J. Shatah and M. Struwe, Regularity Results for Nonlinear Wave Equations, Annals of Mathematics, Second Series 138, 503 – 518 (1993).
- [84] J. Shatah and M. Struwe, *Geometric wave equations. Courant Lecture Notes in Mathematics*2, New York University, Courant Institute of Mathematical Sciences, New York (1998).
- [85] J. Shatah and M. Struwe, *The Cauchy problem for wave maps*, International Mathematics Research Notices, **11**, 555 – 571 (2002).
- [86] Z. Songmu, Non-linear evolution equations, Champan & Hall/CRC Press (2004).
- [87] D.W. Stroock and S.R.S. Varadhan, Multidimensional Diffusion Processes, Springer, Berlin (1979).
- [88] R. Temam, Navier-Stokes Equations: Theory and Numerical Analysis, North-Holland Publishing Company (1979).
- [89] R. Temam, Behaviour at time t = 0 of the solutions of semi-linear evolution equations, Journal of Differential Equations, 43(1), 73 92 (1981).
- [90] R. Temam, Navier-Stokes Equations and Nonlinear Functional Analysis: Second Edition, SIAM (1995).
- [91] H. Triebel, Fourier analysis and function spaces, Teubner (1977).
- [92] H. Triebel, Interpolation Theory, Function Spaces, Differential Operators, North-Holland Publishing Company (1978).
- [93] I. I. Vrabie, C_0 -Semigroups and applications, First Edition, Elsevier (2003).
- [94] M. J. Vishik and A. V. Fursikov, *Mathematical problems of statistical hydrodynamics*, Kluwer Academic Publishers, Dordrecht (1988).
- [95] T. Yamada and S. Watanabe On the uniqueness of solutions of stochastic differential equations, Journal of Mathematics Kyoto University, 11(1), 155 – 167 (1971).
- [96] N. Young, An introduction to Hilbert space, Second Edition, Cambridge University Press (1995).